Thesis Title

Investigating the Strategic Approach to Cyberspace Culture and its Alignment with Vision 2030: The Case of Qatar Higher Education

Doctor Researcher: Amna Al Ansari Advised by: Dr Abrahim Althonayan Brunel University

Abstract

The increased reliance on technology and the Internet represents a predominant feature of the modern business environment, whereby more enterprises are shifting their operations to online platforms in order to reduce operational costs, reach wider audiences, and leverage cyberspace technologies to achieve their strategic objectives. Likewise, in the higher education sector, universities and other higher education institutions are also adopting cyberspace technologies so as to achieve targeted strategic objectives. However, with the adoption of cyberspace, other unprecedented consequences emerge, such as cybercrime, cyber threats, cyber bullying, and cyber terrorism. Additionally, where the actions of actors in cyberspace are not well guided, this further results in the inefficient use of cyberspace technologies and their consequent misalignment with targeted strategic objectives. Subsequently, cyberspace culture emerges as a pivotal component for enabling the efficient use of cyberspace to achieve strategic objectives. The goal of the current research is to investigate the strategic approach to cyberspace culture and its alignment with Qatar's higher education vision for 2030. A key motivation for undertaking this research concerned the implementation of the Qatar National Vision, 2030 (QNV 2030), which seeks to transform the country's economy from hydrocarbon-based to knowledge-based by investing in science and technology and enhancing the higher education sector. As such, increased uptake and usage of technology is anticipated in the near future as the QNV 2030 is achieved. However, without proper guidance regarding the use of cyberspace technologies, the physical infrastructure is likely to deter achieving the higher education vision for 2030.

In the research, data were collected by conducting semi-structured interviews with 18 senior executives and by moderating a focus group comprising six managerial executives. The participants were drawn from Qatari regulatory organisations, higher education institutions, and the Ministry of Transport and Communication (MOTC). The findings indicated that, in order to align cyberspace culture with strategic objectives, effective frameworks are needed. However, current cyberspace culture frameworks were ineffective in aligning cyberspace culture to strategic objectives due to their failure to incorporate the impact of human factors. As a result, this research proposed a cyberspace culture framework to align strategic objectives in higher education. The framework comprised external elements (government's strategic directive on Vision 2030), internal elements (higher education management and leadership), and culture (human factors). The

proposed framework was validated by the focus group discussion and findings from previous literature, thereby indicating that it could be employed to align cyberspace culture with strategic objectives in different economic sectors aside from higher education. Nevertheless, the research further acknowledges that, in future scholarly work, there is a need for researchers to employ the model in a real-world setup so as to evaluate its effectiveness in aligning cyberspace culture with strategic objectives.

Keywords: Cyberspace Culture; Technology; Qatar National Vision, 2030; Higher Education

Abstract	2
List of Figures	9
List of Tables	11
Terminologies	12
Abbreviations	15
Chapter 1: Introduction	16
1.1 Background	16
1.2 Problem Statement	20
1.3 Rationale of the Research	21
1.4 Research Aim	21
1.5 Research Objectives	22
1.6 Research Questions	22
1.7 Research Justification	22
1.8 Research Contribution	24
1.9 Research Methodology	24
1.10 Research Limitations	26
1.11 Thesis Outline	27
Chapter 2: Literature Review	28
2.1 Introduction	28
2.2 The Evolution of Cyberspace, Culture, and Core Advancements	28
2.2.1 Cyberspace History	29
2.2.2 Evolution of Cybersecurity	31
2.2.5 Definition of Cyberspace	54
2.2.5 Cyberspace Culture	39
2.2.6 Recent Advancement in Cyberspace Culture	44
2.3 Challenges in the Evolution of Cyberspace Culture	45
2.4 Qatar Vision, 2030	47
2.4.1 Overview of the ONV 2030	48
2.4.2 Contributions of the Cyberspace Culture to the QNV 2030	50
2.5 Cyberspace Culture in Service Industries and Higher Education	53
2.5.1 Cyberspace Culture and Service Industries in Qatar	54
2.5.2 Cyberspace Culture and Higher Education in Qatar	55
2.5.3 Culture Management as a Controversial Idea	58
2.6 Guiding Principles for Rulemaking in the Cyberspace Culture	59

Table of Contents

2.7 Recognition of Cyberspace Culture Openness	61
 2.8 Gaps in Cyberspace and the QNV 2030 2.8.1 Gaps in Cyberspace 2.8.2 Gaps in the QNV 2030 2.8.3 Trust among Distributed Parties	
2.9 Conclusion	67
Chapter 3: Literature Evaluation	68
3.1 Introduction	68
 3.2 Literature Evaluation	
3.3 Existent Literature Gaps in Cyberspace Culture and Higher Education	81
3.4 Conclusion	84
Chapter 4: Framework Development	86
4.1 Introduction	86
4.2 Organisation Culture	
 4.4 Cyberspace Culture Frameworks	92 93 93 94 94 96 98 100
4.5 Gaps in Existing Frameworks	
4.6 Cyberspace Culture Framework 4.6.1 Derivation of the Cyberspace Culture Framework	104 104
 4.7 Elements of the Cyberspace Culture Framework	
4.8 Conclusion	124
Chapter 5: Research Methodology	125

5.1 Introduction	125
5.2 Interpretivist Research Philosophy	127
5.3 Research Approach	128
5.3.1 Inductive Research	128
5.4 Research Strategy	129
5.4.1 Semi-Structured Interviews	129
5.4.2 Focus Group Discussions	131
5.5 Research Design	132
5.5.1 Research Process	134
	155
5.6 Data Collection Strategy	137 137
5.7 Ovalitativa Data Analysia	142
5.7 Qualitative Data Analysis	143
5.8 Research Quality	144
5.8.1 Renability	144 145
5.9 Research Ethics	1/6
5.10 Secure and	140
5.10 Summary	14/
Chapter 6: Primary Data Collection and Analysis	148
6.1 Introduction	148
6.2 Data Collection and Analysis Strategies	148
6.2.1 Data-Collection Strategy	148
6.2.2 Data Analysis Strategy	149
6.3 Interview Particulars	152
6.3.1 Interview Demographic Data	155
6.4 Focus Group Particulars	160
6.4.1 Respondents' Professional Responsibilities	162
6.4.2 Organisational Sector	164
6.4.3 Professional Roles	164 165
6.4.5 Gender	165
6.4.6 Nationality	166
6.4.7 Job Category	167
6.5 Thematic Analysis	168
6.5.1 Theme 1: External Elements and Attainment of the Higher Education 2030 Visio	n
(Q1–Q5, Appendix A)	178
6.5.3 Theme 3: Culture and the Higher Education ONV 2030 (O9 Appendix A)	200 211
and a second of a second of the second of th	

6.5.4 Theme 4: Proposed Cyberspace Culture Framework (Focus Group Q1 Appendix	A)
6.6 Conclusion	217
	220
Chapter 7: Discussion	228
7.1 Introduction	228
7.2 Thematic Analysis of Research Findings	228
7.2.1 Theme 1: External Elements and the Higher Education Vision, 2030 Achievement	1t 231
7.2.3 Theme 3: Culture and the H.E. Vision, 2030 Achievement	256
7.2.4 Theme 4: Proposed Cyberspace Culture Framework	259
7.3 Validated Proposed Cyberspace Culture Framework	266
7.3.1 Internal Elements	266
7.3.2 External Elements	267
7.3.4 Refinement of the Proposed Cyberspace Culture Framework	267
7.4 Implementation Guidance	272
7.4.1 Role of the Government (Vision 2030 Directive)	272
7.4.2 Role of Higher Education Leadership	273
7.4.3 Role of the Culture in Organisations	274
7.5 Conclusion	274
Chapter 8: Conclusion and Recommendations	276
8.1 Introduction	276
8.2 Results Determined by Research Aim (Chapter 1, Section 1.4)	276
8.3 Results Determined by Research Objectives (Chapter 1, Section 1.5)	278
8.4 Results Determined by Research Questions (Chapter 1, Section 1.6)	281
8.5 Research Limitations	283
8.5.1 Limitations in Literature Search	284
8.5.2 Limitations in Research Design	284
8.6 Research Contribution	285
8.6.1 Theoretical Contribution	285
8.6.2 Practical Contribution	286
8.7 Recommendations	287
8.8 Recommendation for Future Research	288
8.9 Research Conclusion	289
References	291
Appendix A: Interview Questions	316
Appendix B: Interview Particulars	326

Appendix C: Focus Group Particulars	329
Appendix D: Themes and Coding References	330

Figure 4-6: Information Security framework -STOPE101 Figure 5-6 Thematic analysis phases......144 Figure 6-13 Treemap illustrating distribution of research themes......174 Figure 7-2: Influence of the government's strategic economic directives on the H.E. vision 2030. Figure 7-3: Influence of the government's external pressure directives on H.E. vision 2030. .. 236 Figure 7-5 : Influence of compliance and advanced security on H.E. vision 2030 attainment. . 239 Figure 7-6: Influence of environmental uncertainty and political factors on H.E. vision 2030. 241 Figure 7-7: Influence of advanced learning and teaching tech on H.E. vision 2030 attainment. 243

List of Figures

Figure 7-8: Influence of management technologies on H.E. vision 2030 attainment	246
Figure 7-9: Influence of vision 2030 implementation framework on H.E. vision 2030 attait	inment.
	247
Figure 7-10: Influence of leadership and management strategies on H.E. vision 2030	249
Figure 7-11: Security awareness and continuous learning influencing H.E. vision 2030	
attainment	251
Figure 7-12: Perspectives on attitudes and behaviour influencing H.E. vision 2030 attainn	nent.
	253
Figure 7-13: Perspectives on security technical best practices influencing H.E. vision 203	0
attainment	254
Figure 7-14: Perspectives on the influence of culture on H.E. vision 2030 attainment	256
Figure 7-15: Challenges of fostering cyberspace culture in the Qatar context	258
Figure 7-16: Solutions to foster cyberspace culture in the Qatar context	259
Figure 7-17: Alignment of vision 2030 directives to H.E. vision 2030	260
Figure 7-18: Alignment of technological advancement in H.E. vision 2030	261
Figure 7-19: Alignment of vision 2030 implementation framework to H.E. vision 2030	262
Figure 7-20: Alignment of leadership and management strategies to H.E. vision 2030	263
Figure 7-21: Alignment of security policy to H.E. vision 2030.	264
Figure 7-22: Alignment of security policy to H.E. vision 2030.	265
Figure 7-23: Alignment of culture to H.E. vision 2030.	266
Figure 7-24: Refined proposed cyberspace culture framework	267

Table 2-1 Definition of cybersecurity	
Table 2-2 Definition of cyberspace	
Table 3-1 Literature evaluation framework	69
Table 3-2 Research literature evaluation.	
Table 4-1 Literature evaluation framework matrix	94
Table 4-2 Classification of cyberspace culture frameworks	
Table 4-3 Identification of the research gap	
Table 6-1 Particulars of the interviews	
Table 6-2 External elements codes, sub-themes and codes	178
Table 6-3 Internal elements codes and themes	
Table 6-4 Culture codes and themes	
Table 6-5 Cyberspace culture framework codes, sub-themes and codes	217
Table 7-1: Thematic map of research findings.	

List of Tables

Terminologies

Asynchronous communication:	Refers to a type of communication whereby users	
	engaging in the interaction are not required to be present at	
	the same time.	
Chat/chatrooms:	Describes a place whereby any number of computer users	
	can engage in live and online conversation and, as a result,	
	share messages with each other and communicate in real	
	time.	
Cyber classrooms:	A place for teaching and learning over a network, wherein a	
	teacher and students can speak to each other using	
	microphone as well as by typing questions and answers in a	
	message box. Many cyber classrooms have an electronic	
	whiteboard for teachers to use for visual explanation.	
Discussion board (Bulletin Board):	Describes a forum accessed over a network in which	
	different messages can be placed.	
Download:	Refers to the acquisition of information across a given	
	computer network.	
Emoticons:	Icons or symbols that represent people's emotional status,	
	such as happy, humorous, sad, or surprised.	
Elluminate:	A copyrighted web-conferencing software system allowing	
	people to write and speak to each other in real time	
	using a microphone and a headset. Elluminate also	
	offers White Board for visual explanations.	
Homepage:	Refers to the first page loaded whenever a user	
	launches their web browser. As such, the webpage	
	delineates an individual's personal page and the initial	
	point for exploring the website.	
Instant Messenger:	A computer application allowing instant text	
	communication between more than two people through a	
	shared network, such as the Internet.	

Internet:	Refers to a system of interlinked computer networks in an	
	international scope that facilitates data-communication	
	services such as remote login, transfer of files, and	
	electronic mail.	
Invisibility:	A condition in which people are present to each other online,	
	but are unable to see or be seen by each other	
Listserv:	Refers to a programme that handles a mailing list allowing	
	people to communicate with others who have subscribed to	
	the same list.	
Logging off, logging out:	An action whereby a password-protected session on a	
	computer system is ended.	
Logging in:	An act of provisioning both a username and password in	
	order to gain access to a computer system. The action	
	enables users to identify themselves in a given system. As a	
	result, diverse users can store files separately and securely	
	in one computer based on their distinct identities.	
Newsgroups:	Describes a component of the Internet in which users are	
	able to post and reply to messages sent by other users. An	
	asynchronous text-based forum for communication.	
Online:	Being Internet-connected. Online is employed as an	
	adjective describing the variety of activities that can be	
	undertaken on the Internet, such as shopping, chatting and	
	playing games.	
'rl':	A term derived from 'real life' during the period when	
	'virtual' and 'real' were termed as opposite. People	
	today consider the virtual and cyberspace to be 'real', but	
	the term 'rl' continues to be used to indicate space and	
	events not in cyberspace.	

Screen name:	A name chosen by a user when engaging in synchronous
	communication on Instant Messenger. The name can either
	be the user's real name or a varied form of the person's real
	name, the person's nickname, a totally made-up pseudonym,
	or any series of characters.
Smiley:	A sequential order of typed characters that produce the
	sideways image of an individual smiling, such as ':-)'. A
	smiley is used to represent emotion.
Synchronous communication:	Any type of communication whereby parties who interact
	are required to be present simultaneously.

ADDICVIACIONS
Advanced Research Projects Agency Network
Conseil Européen pour laRecherché Nucléaire
Computer Emergency Response Team
International Business Machines
Information and Communication Technology
Information Security Framework
Information Security Culture Framework
Internet Utilisation Statistics
Management Information Systems
Massachusetts Institute of Technology
Ministry of Transport and Communication
National Cyber Security Strategy
Qatar Foundation
Qatar's National Vision
Radio Frequency Identification
United Nations Educational, Scientific and Cultural Organisation
Small and Medium Enterprises
Stanford Research Institute
University of California, Los Angeles
Vice President
World Health Organisation
World Wide Web

Abbreviations

Chapter 1: Introduction

1.1 Background

The Internet has witnessed substantial growth over the past three decades, leading to the conceptualisation of a new communication dimension as a representation of cyberspace, considered the most significant achievement of modern man (Amant et al., 2017). Gibson (1984) first coined the phrase 'cyberspace' in the novel *Neuromancer*, where the author viewed it as a graphical representation of data obtained from the memory banks of all computers present in the human system. As such, cyberspace can be explained as a new communication medium emerging from globally interconnected computers. Cyberspace can further be delineated as the virtual environment developed by computing systems and the users who interact within it (Schatz et al., 2017). The Internet Utilisation Statistics (IUS) reported that the number of users has grown by 444.8% since 2000. Asia possesses the highest number of users (42%), followed by Europe (24.2%) and North America (13.54%). The rest of the users are distributed between Latin America/the Caribbean (10.4%), Africa (5.6%), and Oceania/Australia (1%; Gomez-Diago, 2012). Cyberspace culture comprises a pivotal aspect defining modern life as it empowers communities and individuals to organise and connect themselves in and through it (Boyle, 2017). The period between 2000 and 2010 witnessed an expansion in global Internet usage from 360 million to over 2 billion people. As this expansion of Internet usage continues, cyberspace culture is anticipated to continually grow and become interwoven in the daily life of individuals across the globe. Furthermore, given that a significant proportion of the population does not play any role in cyberspace, the advancement of wireless communication technologies is facilitating the emergence of interactive communication within horizontal networks. As a result, this enables different people to share synchronous and asynchronous messages through e-mails, chats, SMS, blogs, vlogs, podcasts, and wikis (Hunter, 2017).

For many years, women have occupied a significant portion of cyberspace at the behest of different motivations. For instance, in 2000, a study by the Pew Internet Institute revealed that women used cyberspace to maintain and establish relationships. In a separate study, Solis (2009) reported that platforms such as Facebook, Flickr, FriendFeed, myspace.com, and Twitter were utilised more by women than by men. As people enjoin themselves in networks, they access a wide range of available content on the Internet and can share both information and ideas concerning different

situations (Dohr *et al.*, 2010). Additionally, online access empowers individuals to complete diverse activities, such as shopping, completing online studies, or searching for jobs. These facilities further enable the human population to enjoy greater autonomy, in turn influencing their performance and empowering them to handle diverse issues. Women living in isolated regions can now also access opportunities and information they would be unable to access otherwise.

Cyberspace was recently defined as a world-changing domain characterised by the interlinked utilisation of the electromagnetic spectrum and electrons, the target of which is to construct, modify, exchange, use, share, save, and dispose of information and disrupt physical resources (Mayer *et al.*, 2014). Today, communication between humans is becoming increasingly computer-mediated and, over time, ubiquitous. Computers have become connected through high-speed networks both local and wide using wireless technologies. Interaction over high bandwidth is also boosting the speed of communication and offers the ability to transmit images, voices, and sound as well as data in a text format (Gungor *et al.*, 2011). Computer-based technologies are thus empowering the creation of entirely new interaction interfaces between humans and machines, along with a completely new virtual space for human–human interaction (Beck, 2014).

Collectively, these new and different communication spaces are referred to as cyberspace. Cyberspace comprises several elements: a) Telecommunications devices and physical infrastructures empowering the interconnection of networks of communication and technological systems; b) computerised systems and the associated software ensuring that domains operate and connect as required; c) networks linking computer systems together; d) networks that link the computer networks linking computer systems (the difference between (c) and (d) stems from the nature of organisation); e) access nodes for users and intermediaries that route nodes; and f) data residing in the systems (constituent data; Bryant, 2016; Smirnov, Levashova and Kashevnik, 2019).

A common school of thought posits that culture is associated with the past as well as the conservation and relaying of traditions in a particular community (Morin, 2016). However, a different cultural face is directed towards the future—namely, the forward-facing outlook of culture, comprising creation rather than a conservation of focus on memory and imagination. The

ability to discern new and unexpected scenarios to induct people into computer-mediated environments is required in the new environments they inhabit (Rouse and Dionisio, 2018). Nonetheless, despite the wide spread of computer systems, Rouse and Dionisio's understanding of functions as mediation tools remains relatively poor. Individuals can often feel disconnected in a technology-based world that appears strong and powerful in its physical presence while remaining hidden and difficult in another sense—namely, the understanding of the human experience they portray (Burda and Harding, 2013). Subsequently, understanding the present situation requires cultural imagination that involves speculating and affirming the likelihood of other forms of life that differ from the commonplace experiences to which users are accustomed in a natural manner as it shapes the mind through cultural transmission processes (Shweder *et al.*, 1998).

According to Gomez-Diago (2015), cyberspace culture embraces the set of attitudes, thought processes, values, and practices that grow together with cyberspace. By referencing the interaction between the different concepts, associations between culture and technology can be clearly identified. The lack of boundaries, whether nationally or locally in the environment, also encourages and enhances conversations between individuals drawn from diverse cultures (Gomez-Diago, 2012). Higher education cyberspace culture is delineated as an intellectual space culture to which users gain access via computers and which empowers users in the context of higher educational to collaborate with virtual reality, avatars, and text (Saunders *et al.*, 2009). In this cyberspace culture, educators and students are provided with a unique opportunity to undertake a wide range of computer-accessed learning, both planned and unplanned.

Avatars describe computer-based characters influenced by people in real life. In the past, several science-fiction writers highlighted that avatars enabled individuals to explore the imaginary world by using characters endowed with diversified features, ethics, and genders (Reamer, 2013). Furthermore, virtual reality also includes any forms of reality based on computers. Through cyberspace culture, students are empowered to engage in encounters that can be based on either fantasy or reality. Subsequently, examples of virtual reality include instantiations whereby the computer-mediated experience presumes a core defining feature of the environment, such as adventure games and computer-based flight simulators (Mahmoud and Auter, 2009).

Becker's (1994) study of teachers who deployed computers in an exemplary manner revealed that the school support for teachers, as well as the commensurate allocation of resources for the development of staff and smaller classes, heavily influenced the class learning environments where teachers were involved. Students' immersion in virtual reality was also anticipated to generate further diversified changes attributed to changes in culture. Internet believers from academia and industry today are deploying new technologies possessing all the features of culture, such as status, language, rituals, and symbols. In response, Bowers (1988) argued that computers behave as cultural devices, as they drive the philosophical viewpoints of the programmers and software designers who make them.

Lovink (2017) affirmed that not only is the Internet spread occurring rapidly, but there is also proof of an unplanned shift in culture. While some individuals inhabit a cultural space with features such as traditional text and conventional broadcast avenues, such as television, others instead engross themselves in cyberspace culture (Hanson, 2013). An argument by Postman (1985) postulated a diversion removed from conventional text and digitally produced typography images. In some instances, a variation of perspectives may occur between individuals who believe in the two types of culture (Snow, 1959). Snow postulated that scientists and literary intellectuals could not understand each other. Applying this analogy to higher education subsequently implies that it is likely that classrooms will emerge whereby teachers and students fail to differentiate the core principles of their counterpart's daily lives (Valimaa, 1998).

Cyberspace security and threats represent a concern for all stakeholders, not just the government, public authorities, commercial enterprises, or individuals (Caton, 2012). This research focusses on Qatar due to the country's overall objective of modernising its economy from being hydrocarbonbased to knowledge-based. In this context, investigating how to foster cyberspace culture in higher education will prove beneficial in the long term, as this will facilitate attaining the Qatar National Vision, 2030 (QNV 2030). A second justification for selecting Qatar is that the Researcher is able to easily collect data from senior executives in higher education and government regulatory bodies. Therefore, in order to transform Qatar into a highly advanced country by 2030, it will be necessary to provide information and communication technology (ICT) systems to empower government, private businesses, learning institutions, and individuals with access to the required knowledge and information. To this end, in 2014, a new cybercrime law was enacted in Qatar to ensure protection against diverse cybercrimes and preserve intellectual property (Foody *et al.*, 2017). However, some arguments have been advanced against these new laws based on the possibility that this might prohibit freedom of expression. Likewise, numerous countries across the globe are facing similar challenges in enacting cyber laws that incriminate illegal activities in social media and cyberspace.

Nonetheless, Qatar has made substantial progress in developing an advanced ICT sector that will offer advantages for all sectors of society. The accessibility and affordability of ICT is also expected to enhance access to different societal aspects and empower collaboration across different levels. Qatar's establishment of the National Cyber Security Committee in 2013 further promised to furnish governance structures to enhance the collaborative addressing of cybersecurity at the highest government levels. As a result, Qatar formulated a National Cyber Security Strategy (NCSS) to serve as a blueprint for progressing and enhancing cybersecurity across the country. The NCSS coalesces diverse initiatives, measures, and awareness programmes on cybersecurity with effective governance, producing an efficient long-term protective strategy (Al-Jaber, 2014). Subsequently, this empowers Qatar in its efforts to achieve its vision by 2030.

1.2 Problem Statement

In order to achieve the QNV 2030, Qatar has been significantly investing in the acquisition of advanced technology and enhancing the use of ICT in all sectors of society, including higher education. This strategy aligns with the country's primary objective of transforming its economy from a hydrocarbon-based alternative to one that is knowledge-based. However, closer inspection of these strategic objectives reveals eminent problems that are yet to be addressed. To begin, as Qatar invests in the uptake of ICT technology in the higher education sector, an unprecedented consequence that is likely to arise concerns the lack of appropriate cyberspace culture to guide learners and teachers in using the technology to achieve established objectives. In this regard, higher education providers who advocate for students to adopt cyberspace culture possess a duty of care towards the students. In the event that students face risks due to the dangerous cyberspace environments, then serious questions regarding education quality must be asked. It thereby becomes possible that future use of educational cyberspace culture is likely to disadvantage

students. For this reason, it is necessary for effective cyberspace culture to guide learners in higher education so as to avoid misusing the technology and facing other threats. To this end, Vestich (1997) investigated the notion that educational technologies and the Internet are causing providers to stray from sound education practices. An argument has also been advanced that the Internet world can be viewed as a technology-constructed fabrication that serves as a substitute for reality.

Furthermore, while educational cyberspace culture is expected to provide diverse experiences, students and teachers should not expect the same level of satisfaction in online experiences and real life (Hope, 2016). Nonetheless, as revealed by Pugalee and Robinson (1998), the increased pace of equipping teachers with skills in computer-mediated technology, as well as for providing students with access to the Internet and educational cyberspace, will need to be utilised cautiously to prevent long-term results from disappointing such efforts. This also emphasises the second role of cyberspace culture—namely, guiding teachers to harness computer-mediated technology to achieve different strategic objectives. Taking this context into consideration, the research questions raised in this thesis concern how attaining Qatar's 2030 higher educational vision affects the cultivation of cyberspace culture. Therefore, the primary issue tackled in this research concerns the need for an effective cyberspace culture that can guide the actions of both learners and teachers in cyberspace in an effort to ensure they achieve targeted strategic objectives.

1.3 Rationale of the Research

Investigating the current strategic approach to cyberspace culture is crucial for combating or reducing the various threats facing Qatar's higher education sector. The evolution of modern cyberspace culture, particularly over the past three decades, has also become one of the most heavily researched topics today. In particular, considerable research has been conducted concerning higher education cyberspace in numerous countries, aided by technology and the rising interest in the cyberspace industry. However, there remains a lack of specific research on cyberspace culture in higher education in the context of Qatar in particular. To counter this, the scope of this current research goes beyond the use of academic research publications by including industry journals and case study surveys on the topic, whether private or public.

1.4 Research Aim

The primary goal of this research is to investigate the strategic approach to cyberspace culture and its alignment with Qatar's strategic direction, as well as to develop an effective framework for aligning cyberspace culture with the vision of Qatar's strategic direction, supported by implementation guidance focussed primarily on the higher education sector.

1.5 Research Objectives

- 1. To evaluate current/existing cyberspace cultures' alignment with Qatar's strategic vision.
- 2. To identify the strengths and weaknesses of the various current frameworks regarding cyberspace culture.
- 3. To analyse the current strategic approach to cyberspace culture and its alignment with the country's strategic directions.
- 4. To analyse the QNV 2030 cyberspace strategy and how it is influenced by the attainment of the 2030 higher education vision.

1.6 Research Questions

- 1. How does cyberspace culture strategically align with the country's strategic directions?
- 2. How are current cyberspace culture frameworks aligned with the country's strategic directions?
- 3. Why is cyberspace culture important, and what literature gaps exist regarding cyberspace culture?
- 4. What kind of cyberspace strategy needs to be deployed by the government to align with the country's strategic directions?

1.7 Research Justification

Over the past three decades, significant research has been undertaken concerning the evolution of modern cyberspace culture given its undue influence on actors who inhabit the digital landscape. Research from organisations such as IUS has accordingly revealed that, over time, the number of cyberspace users has gradually increased in different parts of the world as technology has become more affordable and readily accessible. The higher education sector thus stands to benefit from the adoption of cyberspace and computer-mediated technology, as it facilitates not only learning, teaching, and content delivery, but also collaboration among researchers across different parts of the world. However, as higher education providers advocate for adopting cyberspace culture, increased risks remain for the various actors in the cyberspace environments, including cybercrime, cyberbullying, and cyberterrorism, among others. This necessitates serious debate regarding cyberspace culture's role in guiding the actions of students, teachers, and other actors in

cyberspace so as to minimise threats to the higher education sector and guarantee the success of computer-mediated technology in enhancing learning, teaching, and scientific research.

In Qatar's context, the goal of the human development pillar in the QNV 2030 is to enhance the human population's technical capacity through aspects such as investment in world-class scientific research and first-rate education to unlock said capacity. Furthermore, in enhancing the population's capacity, there has also been an increased focus on the delivery of e-education, whereby more investments are being made to deliver learning and teaching via technology. While such objectives will enable Qatari higher education institutions to achieve world-class education delivery, a further risk that is observed concerns the increased threat landscape that emerges from technology adoption. Therefore, to guarantee the success of investments in technological infrastructure for achieving the overall QNV 2030, a further need for cyberspace culture is emphasised to guide the actions of different actors, such as learners and educators, in the digital space.

The objectives of this research were to investigate the strategic approach to cyberspace culture and its alignment with Qatar's strategic direction in order to develop an effective framework for aligning cyberspace culture with the vision of Qatar's strategic direction, supported by implementation guidance focussed primarily on the higher education sector. This research thus responds to the need to foster positive cyberspace culture as technology-mediated learning becomes mainstream in the modern education environment. Furthermore, as the Qatar government depends on upskilling its population to drive a knowledge-based economy, there is an increased need to ensure that the adopted technology facilitates attaining the strategic goals and visions set by the government in the QNV 2030.

The Researcher is driven to pursue this research based on the importance of cyberspace culture for attaining the QNV 2030. In the current decade, Qatar has been investing in the delivery of higher education via technology (e-education). However, as such investments are being made across the various higher education institutions, an appropriate cyberspace culture to guide the actions of educators and learners in this digital space remains lacking. Therefore, the primary motivation for

undertaking this research stems from the disconnect between the attainment of the QNV 2030 and the existent cyberspace culture.

1.8 Research Contribution

- This research seeks to investigate how cyberspace culture generally aligns with the strategic directions of a country. As such, it will enhance the implementation of new procedures and policies focussed on improving cyberspace-culture strategies. Moreover, this research will be crucial for easing the conceptual problem associated with cyberspace by helping to improve the present strategies for investigating information security.
- 2. The primary contribution of this research comprises a newly developed framework to facilitate the achievement of cyberspace culture (proposed cyberspace culture framework, Figure 4-7). This developed framework plays an important role as a reference point and a guide for other researchers in the same field. Additionally, it provides cyberspace scholars and professionals with practical recommendations concerning cyberspace's role and importance in higher education institutions. Additionally, this research also serves as a scholarly source for different fields in the services sector.
- 3. Through an in-depth analysis of various strategic frameworks, this research will prove valuable in contributing to the current theoretical knowledge by means of a thorough review of various academic sources, as well as industrial literature and sources.
- 4. This overall research will represent a substantial contribution to the advancement of knowledge in the field of cyberspace culture.

1.9 Research Methodology

The chosen research methodology provides a suitable structure for conducting research effectively. The guideline provided by the research methodology helps the researcher to follow the proper direction (Adams, Khan, and Raeside, 2014). According to Williams (2007), all tools and techniques necessary for data collection and analysis could be defined in the research methodology, representing an exploration of knowledge through objective and systematic methods for identifying solutions to research problems.

To satisfy the objectives of this research, qualitative research has been adopted as guided by an interpretive philosophy. As explained by Creswell (2011), qualitative research methods are most suitable for assessing and developing a comprehensive understanding of a phenomenon within its

natural setting. Qualitative research is based on an inductive process employed to formulate a hypothesis (Punch, 2006). This research adopts the inductive approach to theory development whereby empirical data are collected and analysed first in order to help develop patterns of theory. This is unlike deductive studies, which begin by hypothesising predictive statements, which are then tested by collecting empirical data. According to Walliman (2017), inductive reasoning starts from specific observations or sensory experiences, from which a general conclusion is developed.

Regarding the research strategy, a case study approach is adopted here, as this research focusses on one single geographical context—Qatar's higher education. Mabry defined a case study as 'the empirical investigation of a specified or bounded phenomenon' (Mabry, 2008, p., 214). Adopting a case study is thus justified by the fact that its use aids in discovering and understanding participants' experiences and perspectives within their specific operation environment. This, in turn, allows the Researcher to understand participant voices regarding the cultivation of cyberspace culture.

Sampling also represents an important aspect of the research methodology, as this outlines the approach guiding participant selection. Participants for this research were identified using a non-random, non-probability sampling technique whereby participants were chosen based on a pre-determined criterion. In particular, for the final research sample, the Researcher only targeted cyberspace experts and professionals as well as policy experts who understand the 2030 flagship project due to their research area knowledge (Silverman, 2016). With the non-probability sampling technique, a random selection method is not necessary to select a sample from the population. Rather, subjective methods are employed to decide which component will be part of the sample (Etikan *et al.*, 2016). Probability sampling is not adopted here because it assigns an equal probability of selecting participants in a given population (Fink, 1995).

With the participants selected, the Researcher proceeded to collect the qualitative data by administering interviews and moderating focus groups, as they facilitate investigating participants' views at greater depth (Dornyei, 2007; Kvale and Brinkmann, 2009). Cohen *et al.* (2007) also argued that interviewing represents a beneficial method for exploring the construction and negotiation of meaning in a natural setting. The Researcher conducted the different interviews on

a one-on-one personal basis in Qatar based on the convenience of the respondents and the Researcher. Due to data-saturation concerns, the Researcher limited the interviews to 15 participants and focus groups to 10 members, wherein each focus group would be assigned only five participants. Similar themes were maintained in the questions presented to interviewees and focus groups so as to ensure data reliability. Furthermore, the Researcher ensured that questions were unambiguous and free of any bias by using clear and easy-to-understand language. Likewise, bias was eliminated by structuring the interview questions based on the conceptual framework developed in the research and presenting them to a test group to ensure they understood everything clearly. Following the data collection, data analysis was performed via a thematic analysis of the content. Content analysis was employed to analyse the data whereby the collected data were categorised into themes and sub-themes, and comparisons were made.

1.10 Research Limitations

This research nonetheless remains limited by several issues: First, since the qualitative data were collected by conducting interviews with diverse players involved in formulating the QNV 2030, there remains a likelihood that they might not be available to partake in the interviews. Second, since the data collected from interviews were analysed using thematic content analysis procedures, there is a likelihood that the analytical approach might not generate the targeted data, particularly where the data remain too complex to analyse. Third, a challenge is also anticipated in obtaining information on cyberspace culture from a strategy rather than a technical perspective. The Researcher sought to interview senior executives drawn from higher education institutions and the QNV 2030 committee, as well as to engage in discussions with the various groups. As such, due to the respondents' highly technical roles, a challenge is likely to be encountered in that the respondents might only provide data from a technical rather than strategic perspective.

1.11 Thesis Outline



Figure 1-1: Research outline.

Source: The Researcher

Chapter 2: Literature Review

2.1 Introduction

The cyberspace culture concept can be considered relatively new based on the fact that most academic research on cyberspace culture and related aspects has been published within the last three decades (Berman, 1999; Macfadyen, 2006; Bell, 2009; Murray and Murray, 2017). As such, the present chapter focusses on work published from the mid-1990s to the present. Furthermore, interest in cyberspace culture has progressively surpassed academic research in recent years, leading to the increased availability of data from other non-academic sources, such as the United Nations Educational, Scientific, and Cultural Organisation (UNESCO) (Diki-Kidiri, 2007). For this reason, the scope of this research considers not only academic resources, but also online articles, surveys, journals, and case studies describing this topic. The central purpose pursued in this chapter is to evaluate key contributions from cyberspace culture literature made by both scholarly and industrial researchers and higher education institutions in Qatar. To this end, the following areas are discussed:

- The evolution of the cyberspace environment and cyberspace culture over the last three decades;
- The most vital literature on cyberspace culture, basing the review on the higher education sector; and
- The value added, challenges experienced, advancements made, and oversight of the cyberspace culture.

The analysis of previous academic and industrial works on the cyberspace industry and the higher education sector revealed challenges, limitations, and advancements made in both areas, thereby enhancing the research. The analysis also furthers the research by opening up opportunities for development of the same nature or other related fields.

2.2 The Evolution of Cyberspace, Culture, and Core Advancements

This section seeks to introduce the underlying concepts employed throughout the chapter. To this end, this section has been further structured into six sub-sections so as to logically analyse the different concepts in a coherent manner. The first sub-section expounds on cyberspace history, ranging from its limited application during World War II to its wide-scale usage in the late 1980s and 1990s as the Internet and personal computers became more affordable and assumed a central role in both business and personal lifestyles. The second sub-section discusses the concept of cybersecurity and how it emerged following the need to protect legitimate users from threats and destructive attacks from other malicious users. In particular, it emphasises the lack of international laws governing the Internet as a contributing factor to the observed rise in cybercrime.

The third sub-section undertakes a critical review of diverse definitions of cyberspace from different authors. By examining these definitions, the goal is to highlight similarities and differences among them and to further illustrate how the concept has evolved over time. For instance, certain research questions ask whether any changes have occurred in the underlying concept as definitions have progressed from works of fiction, such as cyberpunk novels, to more factual and non-fiction scientific applications. The fourth sub-section defines culture and its constituent elements, whereas the fifth sub-section delves further into the concept of cyberspace culture, its impact, and underlying concepts. The final sub-section discusses the concept of modern-day cyberspace culture and how it has been morphed by diversity in technological innovations.

2.2.1 Cyberspace History

The phrase 'cyberspace' was coined in 1982 by a Canadian science fiction author, William Gibson, in an article that appeared in the Omni magazine, and thereafter in his novel *Neuromancer* (Jahshan, 2007). In his definition, Gibson conceptualised cyberspace as a virtual world created by computer networks and inhabited by intelligent beings. An excerpt from Gibson's novel describing cyberspace delineated the concept as 'a graphic representation of data abstracted from the banks of every computer in the human system' (Gibson, 1984 p.51). As such, the author advanced the idea of a space in the digital realm that could be inhabited and explored.

Examining the root definition of the term 'cyberspace' further reveals a similarity in concept to the connotation made by Gibson. According to Berdayes and Murphy (2000), cyberspace is constructed of two root words: 'cyber' and 'space'. Cyber derives from the Greek word for 'pilot', thereby advancing the idea of steering, controlling, or piloting. Space, meanwhile, appears in the English language with diverse meanings, such as a time lapse or duration or a physical expanse (Van Manen and Adams, 2009). A philosophical view of space by Hobbes connoted the idea of an

unbounded continuum void of matter (Berdayes and Murphy, 2000). Therefore, cyberspace communicates the idea of control and unboundedness, or concretion and abstraction. Gibson shared the same view by conceptualising cyberspace as an imaginary space in the digital landscape where intelligent beings are free to explore and engage creatively.

Based on this understanding, it can be argued that the conceptualisation of cyberspace can be traced back to the early WWII period, when allies were investigating how electronic technology could be employed to improve radar signals and developing machines that could compute complicated radar messages (Kaisler, 2016). While computers were not yet fully developed during this era, a different school of thought argues that there existed computing devices that created virtual platforms where allies could engage with one another. Axelord (2014) further highlighted that technological disruptions have been catalysing cyberspace since their inception during the WWII period. A particular disruption concerned the creation of the Advanced Research Projects Agency network (ARPAnet) by the United States Department of Defence in the 1960s (Liu and Albitz, 2009).

The ARPAnet connected research organisations and government agencies in the U.S., allowing them to share scarce computing resources. By the mid-1970s, educational institutions and the government developed an interconnection of computers via networks to share information and messaging. For instance, the UCLA and the Stanford Research Institute (SRI) were interconnected using the ARPAnet through packet-switching technology (Liu and Albitz, 2009). By the mid-1980s, the observed increase in network interconnectivity led to a technological gold rush and effectively brought about the creation of new fortunes and the unprecedented demise of older regimes; a new economy thus began to be developed by enthusiastic engineers and venture capitalists, leading to the creation of a new venture in the Information Super Highway (Brown, 2014).

In 1990, a new innovation was developed by Tim Berners-Lee, a computer programmer working at the Conseil Européen pour laRecherché Nucléaire (CERN) in Switzerland. Lee created a system that could empower individuals in a network to access research materials (Brown, 2014). Similarly, they could also share documents across different computer platforms without necessarily having to reformat them. By 1991, Lee had created the first browser, later named the

World Wide Web (WWW; Brown, 2014). As such, the Internet boom experienced during the 1990s led to the development of a new system (cyberspace) and a culture associated with it (cyberspace culture).

2.2.2 Evolution of Cybersecurity

This sub-section provides a brief overview of terms employed to describe the evolution of cybersecurity. The history of cybersecurity is reviewed first, after which definitions of cybersecurity, cyberspace, and culture are provided. It is worth nothing that the Researcher first evaluates cybersecurity and its history. Thereafter, definitions for cybersecurity, cyberspace, and culture are provided and the official definition of cyberspace culture adopted in this research finally outlined.

2.2.2.1 History of Cybersecurity

The period between the 1970s and 1980s witnessed a substantial increase in computer usage across universities, colleges, and households (Forest and Kinser, 2003). Part of the reason for this came from the advent of the personal desktop computer during this time—namely, the IBM personal computer in 1981 and the Apple Macintosh in 1984. Second, computer networking developed rapidly with the conception of distributed computing by the Massachusetts Institute of Technology (MIT) and Carnegie Mellon University (Forest and Kinser, 2003). Such networks were integral for enabling students, staff, and faculty to share computing resources.

The direct implication of these two innovations was that they increased the number of users who could engage in cyberspace. Individuals who could access personal desktop computers could now easily share computing resources with others through digital networks. Nevertheless, this increased access to cyberspace also led to several disadvantages as well. In particular, given the lack of laws governing cyberspace, some individuals could use it for malicious intentions, such as by accessing unauthorised documents. Rost and Glass (2011) highlighted the case of Markus Hess, a German hacker who compromised over 400 military computers with the intention of selling secrets to the Russian KGB. However, his efforts were eventually stopped after being detected by Clifford Stall, a system administrator from the Lawrence Berkeley Laboratory (Stoll, 2005).

By the mid-1980s, cybercrime had grown from the basic access to unauthorised files to serious crimes comprising the spread of viruses and worms, producing significant destruction across the

Internet. Wang and Kissel (2015) noted the Morris worm developed by a hacker (Robert Morris) in 1988 as an experiment to assess the Internet's size. The worm was designed to traverse different computer networks, use a known bug to intercept Unix terminals, and duplicate itself. In essence, Morris intended the worm to infect as many computers on the Internet as possible without leaving a trace or causing harm to computers. However, the worm led to unprecedented effects, such as denial of services due to its rapid rate of spreading.

Following these actions, significant concerns were raised for ensuring the Internet's security. For instance, the Computer Emergency Response Team (CERT) was founded to research systemic issues that might generate adverse effects on the Internet (Lucas and Moeller, 2004). Similarly, legislations were also passed to convict individuals who misused computing resources, such as the 1986 Computer Fraud and Abuse Act. From this, cybersecurity emerged as a highly important concept for developing positive cyberspace culture.

2.2.2.2 Cybersecurity Definition

Cybersecurity can be described as consisting of controls, technologies, and processes designed to protect data, networks, and systems from cyber-attacks (Dorsey *et al.*, 2017). Lindsay (2015) highlighted that when cybersecurity is effective, it lowers the likelihood of cyber-attacks and further ensures that an organisation and its individuals remain protected from the threat of unauthorised exploitation of their technologies, systems, or networks (Lindsay, 2015). An effective cybersecurity approach comprises several protection layers spread across programmes, computers, networks, or data that a user intends to secure. In organisations, the technology, people, and processes ought to complement one another so as to develop effective defences against cyber-attacks (Altner and Servi, 2016).

Numerous definitions of the term 'cybersecurity' have emerged over the years, as summarised in Table 2-1 below:

Year	Researcher	Cybersecurity definition	Significant features
2004	Moteff and	'Cyber-security typically means protecting both	Protection
	Parfomak	physical and cyber assets from operational	Importance of data
		failure or manipulation due to unauthorised	
		access to operating software or data'	

Table 2-1: Definition of cybersecurity.

2009	Hansen and	'protection against unwanted disclosure,	Importance of data
	Nissenbaum	modification, or destruction of data in a system	protection
		and also [to] the safeguarding of systems	Securing systems
		themselves'	
2009	Whitman and	'the protection of information and its critical	Protection of information
	Mattord	elements, including the systems and hardware	Process of protection
		that use, store, and transmit that information'	
2011	Rowe, Lunt and	'preservation of confidentiality, integrity and	Preservation of
	Ekstrom	availability of information in the cyberspace'	information
			Cyberspace
2013	Bambauer	Cyberspace is a place where systems, networks,	Cyberspace is a major
		and data are placed, the safety of which is	platform.
		termed cyber security. Various technological	
		mechanisms intervene in the requests for access	
		or control of the data held within a system	
2013	von Solms and	'Cyber security deals with both information-	ICT major medium
	van Niekerk,	based assets stored or transmitted using ICT and	
		non-information-based assets that are vulnerable	
		to threats via ICT'	

Source: The Researcher

A brief review of the definitions outlined in Table 2-2 reveals a confluence among the different researchers. First, all authors acknowledged that cybersecurity emphasises protecting data and information. Second, they also highlighted that cyberspace represents an important platform that provides meaning to cybersecurity. Alternative definitions of cybersecurity are delineated in Table 2-2 below:

Table 2-2: Alternative definition of cybersecurity.

Terminology	Definition
Cyber world	 Benarous, Kadri, and Bouridane (2016) considered the cyberworld and virtual world to be synonymous with the Internet. As such, they described the term as 'a set of machines linked to huge servers that provide different services used by machines' (p.372). Lehto and Neittaanmäki (2018) considered the cyberworld to be closely related to the cyber environment, describing it as comprising 'users, devices, networks.

	software, processes, information in transit or storage, and services or systems that can be connected to networks either directly or indirectly' (p.34).
Cyberspace	Bryant (2016) described cyberspace as 'a global domain within the information environment that consists of interdependent network of information technology infrastructures and resident data including the Internet, telecommunication networks, computer systems, embedded processors and controllers' (p.3).
Cyber ecosystem	Mazurczyk, Drobniak, and Moore (2016) described the cyber ecosystem as 'a community of cyber-organisms, i.e. non-human actors such as processes, programs, defensive and offensive systems that interact between themselves and with the environment' (p.23).

Source: Adapted from Andronache (2019)

2.2.3 Definition of Cyberspace

Gibson's previously discussed definition of cyberspace connoted the idea of a virtual landscape created by computer networks and inhabited by intelligent beings (1984). However, in order to clearly comprehend the term, it is expedient to examine it from the views of other scholars and industry practitioners. It is also important to highlight that the term has since gained significant interest from the non-academic community, including technological strategists, security and military professionals, and medical leaders, together with users of the technology domain (Dawson and Thomson, 2018).

In a literal sense, cyberspace can be described as the interconnection of different forms of technology. This assertion stems from the fact that the Internet backbone comprises three main types of technologies: hardware, software, and protocols (Baldauf and Stair, 2011). While hardware refers to the physical cables and devices that carry data, software describes the different programmes allowing users to interact with the Internet in order to access services and information. Protocols, meanwhile, describe the rules governing the established connections in networks (Bangia, 2005). Different arguments further consider cyberspace to refer to the utilisation of technology in computing environments. This view derives from the understanding that computers have represented an integral part of human culture since the early 1990s. As such, this represents a domain used to refer to the integration of different forms of technology in human beings' everyday life (Li *et al.*, 2017).

The combination of emerging trends in cyberspace, including computer technology and the Internet, has been defined as cyberspace culture (Khin *et al.*, 2016). Lévy (2001) further described cyberspace culture as the set of intellectual and material technologies, attitudes, values, and modes of thought that have developed simultaneously with the growth of cyberspace. In contrast, the author defined cyberspace itself as the material infrastructure of digital communications, the diverse information contained within it, and the intelligent beings that navigate and nourish it.

With this insight, it becomes easier to comprehend cyberspace culture as an ongoing trend in the current world that involves including computer and modern Internet technologies in various aspects of a country, such as health, education, infrastructure, and research. The contribution of this Internet and computer technology cannot go unnoticed, and considerable research has accordingly been conducted on this subject (Safa *et al.*, 2018). The majority of such publications examine this contribution in general, and they have proven vital for understanding cyberspace culture.

Cyberspace, as a concept of the Internet, implies that cyberspace usage is not limited solely to global sharing of information in large numbers, but also includes other relevant aspects, such as distributed creation of information and materials, social networking and network marketing, real-time streaming of information from the Internet, mass collaboration of people in the online space to achieve common goals and objectives, collaborative assessment of information shared and distributed online, social bookmarking, and cloud computing, which involves sharing and storing information on the Internet through cloud-storage services for real-time access and retrieval (Chase *et al.*, 2002).

All these services encourage commitment and participation in the online space, thus further encouraging the participation of more people. As a result, this contributes to shaping individual lives, including how people talk, their different communities, and even their identities. Bell (2001) expressed a similar view by postulating that the association between technology and society is two-way; as such, computers not only give form to ideas, experiences, and metaphors, but are also shaped by these aspects as well.

In order to better understand the cyberspace concept, several definitions are summarised in Table 2-3 below:
Table 2-3: Definition of cyberspace.

Year	author	Cybersecurity definition	Significant features
1984	Gibson, W.	'A consensual hallucination experienced	Teaching platform,
		daily by billions of legitimate operators, in	data, users.
		every nation, by children being taught	
		mathematical concepts A graphical	
		representation of data abstracted from the	
		banks of every computer in the human	
		system. Unthinkable complexity, lines of	
		light ranged in the non-space of the mind,	
		clusters and constellations of data. Like	
		city lights, receding'.	
1996	Strate, Lance, Ron	'Individuals could directly connect their	Virtual place,
	Jacobson, Stephanie	nervous systems to a global computer	connectivity.
	Gibson, and Gary	network referred as the matrix and	Nervous system.
	Gumpert	experience of virtual reality'.	
2000	Benedikt	'A common mental geography, built, in	Data, invisible place,
		turn, by consensus and revolution, canon,	sharing
		experiment; a territory swarming with data	
		and lies, with mind stuff and memories of	
		nature, with a million voices and two	
		million eyes in a silent, invisible concert to	
		enquiry, deal-making, dream sharing, and	
		simple beholding'.	
2001	Bell	Cyberspace as a cultural hub is created	The conglomeration
		from machines, people, and stories in	of machine, stories,
		everyday life.	and people in
			everyday life.
2006	Yar	'The interactional space or environment	Interactional space,
		created by linking computers together into	computers.
		a communication network'.	
2014	Rowland, Rice and	'The ever-expanding manifestation of the	Communications
	Shenoi	pervasive information and	infrastructures,
		communications infrastructure'.	information.

Source: The Researcher

The evaluation of different definitions highlighted in Table 2-2 leads to the authors' definition of cyberspace as a virtual platform, wherein interactions occur between technology and individuals

residing in the given space. This insight is important for advancing the discussion of the interrelation between cyberspace and culture.

2.2.4 Culture

A widespread definition of culture describes it as a set of assumptions, beliefs, values, policies, life orientations, and behavioural and procedural conventions shared by and influencing a group of people without necessarily determining the behaviour of each member or their interpretations of the meaning of behaviours by other individuals (Spencer-Oatey, 2008). Eliot (2014) further argued that culture assumes different meanings depending on its scope of application. Eliot (2014) viewed an individual's culture as depending on that of the group to which they belong, whereas the group's culture is further influenced by the greater society.

Subsequently, Eliot (2014) and Spencer-Oatey (2008) advanced the idea of culture influencing an individual, group, or entire society. Strauss and Quinn (2001) resonated with this view by revealing that culture can be anthropologically equated with socially learned behaviours and ideas. At a more detailed level, culture is found to arise from learned organised experiences created by members of a given population, including images and interpretations formed through other individuals, contemporaries, or even transmitted from past generations (Howley, 2016; Lovinck, 2017). Schein (1984) accordingly classified culture into several levels, as depicted in Figure 2-1 below:



Figure 2-1: The levels of culture and their interaction. Source: Schein (1984)

Figure 2-1 illustrates the three levels of culture and their anticipated interactions—namely, basic assumptions at the lower level with values and artefacts at the upper levels. Figure 2-1 illustrates culture's influence on observable behaviour through its impact on the value systems and mental models formed in individuals. This theory is elaborated further in Chapter 4.

2.2.5 Cyberspace Culture

A simplistic definition of cyberspace culture describes it as culture that emerges in cyberspace (St. Amant, 2007). St. Amant, however, posited that a universally accepted definition of the term remains lacking due to its underlying complexity. Le'vy (2001) argued that part of the reason for this difficulty stems from the fact that the culture of the future is associated with the 'universal-without-totality' concept. This implies that, though cyberspace culture is virtually present among

all humanity, it lacks totality due to the numerous differences present within it. Despite this finding, however, two definitions of cyberspace culture are considered in this research.

The first comes from Mihalache (2002), who described cyberspace culture as a collection of cultural products and cultures that are made possible or exist in the Internet, along with stories about them. This definition indicates that the Internet comprises an environment inhabited by and associated with its own culture. A more refined definition by Kiliç (2018) identified cyberspace culture as a concept that describes various cultural products and practices that have emerged in the modern world due to the interaction between computers and Internet technology, as well as human behavioural aspects such as hobbies, art, and language.

A confluence can be observed between the two definitions in that they both indicate that cyberspace culture emerges from the interaction of Internet technology and inhabitants residing within the virtual space. While human behaviour is highlighted in the Kiliç (2018) definition, however, this aspect is also assumed in the second definition (Mihalache, 2002) based on the fact that cultures encapsulate interaction. Other authors have developed similar arguments as well. For instance, Bell (2001) claimed that interaction between technology and society is two-way, with computers giving form to ideas, experiences, and metaphors while they themselves are shaped by these aspects. Based on this argument, it emerges that cyberspace culture does not exist independently; rather, it is formed by both the cyberspace virtual landscape and the inhabitants residing within it.

In order to develop an independent definition of cyberspace culture, the foundations of culture must first be examined. Four important underlying concepts of culture facilitate this. The first comprises the notion that culture consists of socially learned behaviours (Strauss and Quinn, 2001). The second concerns the notion that knowledge represents the underlying foundation of any culture at any point in human society's evolution, and as a result, knowledge-construction quality influences the very fabric of a given culture (Haarmann, 2007). Third, culture is founded on human–human interactions (Macfadyen, 2006), and fourth, language represents the carrier of cultural meanings invoked whenever interpersonal communication takes place (Schaller and Crandall, 2004).

From the analysis of the different foundational concepts on culture, it emerges that culture is founded on interpersonal interactions (human–human) made possible through the use of language and knowledge, and further learned through social behaviour. Therefore, to develop the definition of cyberspace culture, these aspects, together with the interaction of technology, need to be incorporated. Based on the different findings, the official definition of cyberspace culture in this research is delineated as follows:

Cyberspace culture describes a set of underlying, socially learned beliefs that emerge as humans interact with one another through computer-mediated Internet communication, and which influence how they construct knowledge to facilitate this interaction.

This definition comprehensively covers both positive and negative human interactions, where positive behaviour such as empowering discussions are reinforced through positive interaction while nefarious behaviour such as cybercrime emanates from interacting with other individuals who foster such negativity. Therefore, culture is relative and dependant on the nature of human–human interactions occurring through computer-mediated communication over the Internet.

The view that cyberspace culture serves as an avenue of human–human interaction further leads to the understanding that cyberspace culture comprises contributions from a wide scope of academic experts, digital enthusiasts, and computer scientists devoted to ensuring a progressive and advanced use of technology for the community's benefit, as well as the use of technology in cyber-social systems. Cyber-social systems refer to the application of technology in social systems, such as integrating technology in social interactions and utilising social media platforms in normal day-to-day activities (Zeng *et al.*, 2016). The aspect of socially learned behaviours is observed by examining cyber-social systems.

Cyberspace culture represents a concept integrating different reliable techniques for obtaining information on diversified topics rather than being limited to the fixity of information (Goi, 2009). The concept of the online cyberspace culture has thus allowed people to share ideas and information freely on the Internet via blogs, connecting people from different geographical regions

42

through the ideas to which they align and relate (Chase *et al.*, 2002). This can generate continual integrated meanings, and people can generate a common idea through collective intelligence and share their areas of expertise, and the expertise itself, to generate common goals and objectives.

Cyberspace culture connotes the idea of an avenue to develop different cultures through the use of new technology (Hakken, 1999). Through technology, more people can interact via cyberspace compared to physical interactions (Linhares de Souza *et al.*, 2012). People thus visit the Internet not only in search of entertainment alone, but also as a means of obtaining information they would otherwise not be able to receive, whether due to eminent restrictions in their country or because of geographical or social limitations (Oh *et al.*, 2018; Ding *et al.*, 2018). One can generate one's own resources to face situations that are difficult or forbidden to talk about, and thus actively contribute to society, such as on issues of women's rights and female genital mutilation. Through this ability to access forbidden content on the Internet, the connection between virtual and physical worlds of different cultures is therefore made possible, affecting society (Chase *et al.*, 2002). Accordingly, more and more people are becoming integrated into the core of cyber-culture, influencing society's ability to invent, produce, use, and interpret such vital forms of technology (Billone, 2016).

Cyberspace culture has thus made it possible for men and women who love communicating to create groups related to what interests them and concerning issues that directly affect their lives (Garcia, 2016). For instance, Gunitsky (2015) reported an increased utilisation of social media as a tool for attaining autocratic stability that enables non-democratic regimes to shape discussions held by the public and further undermine their opposition. A different study by Howard *et al.* (2011) further identified social media's significant influence on the Arab Spring through shaping political debates and spreading democratic ideas across international borders. Such studies highlight the increased use of cyberspace as an avenue to lead change on issues that directly affect people in the real world.

People with varied interests can also benefit from the ability to obtain different information regarding different types of content from various sources on the Internet and other forms of media and communication (Chang and Grabosky, 2017). Integrating these forms of technology, communication, and information retrieval and sharing thus encompasses the entire concept of

cyber technology. New cultures tend to arise whenever issues regarding cyberspace development are brought to the fore. In turn, this is shaped by networked computer technologies and related processes (Choo, 2011). The cyberspace culture issue is thus heavily influenced and consequently shaped by the interaction between computers and human beings. Furthermore, the Internet age that has emerged over the years involves making more individuals dependent on cyberspace, thus further enhancing the culture (Chang and Grabosky, 2017). Researchers have accordingly made significant efforts to develop and focus on cyberspace culture; as such, the influence of cyberspace must put into account the interaction between computers and human beings.

An increased number of questions have also been raised regarding whether a mixture of physical and local culture within the virtual cyberspace will lead to the creation of a new culture and orientation or if this will lead to integrating the two cultures without developing a new one. However, this is not dependent on any of the cultures, depending instead on the hybridisation and robustness of the cultures (Goodenow, 1996). In this way, the Internet culture has brought about the development of cultural shocks, leading to changes in the cultures, lifestyles, and cultural productions in one way or another. Therefore, serious thought should be given to the diversity and instability of the different cultures before a new culture is integrated into the lifestyle.

Cyberspace culture is chaotic by nature, as it changes constantly in unpredictable ways. Gunter (1995) argued that its chaotic nature can be deployed in education based on the bounded instability present in educational systems. The events unveil in ways that are unpredictable despite not being random. In higher education cyberspace culture, the aspect of turbulence and ambiguity identified by Myers (1990) leads to the view that chaos theory will enable educators to find some approaches to be of more assistance than others. In essence, it remains difficult for educators to predict which interest groups or cultural communities' students will engage with in cyberspace. The nature of computer-based applications changes, leading to changes to the features of the World Wide Web and other aspects of cyberspace culture. While email is largely based on words, graphics and sound are nonetheless increasing in popularity over time, for instance. Still more changes have also occurred on the World Wide Web as pages have become more likely to contain diverse multimedia, such as video clips and animations. The availability of bandwidth further implies that there will be

more capabilities for users to access heavier content such as videos, as well as to undertake live video conferencing (Fetterman, 1996).

Some authors have argued that, in this way, Internet-based applications heavily influence education (Kitching *et al.*, 2015). Cyberspace is expected to grow radically over the next few decades as advanced video conferencing technologies become cheaper and more reliable, leading to words in conventional email being substituted with more engaging virtual reality sounds and images (Mucic *et al.*, 2016), allowing email's anonymous word-based segments to likely be replaced by exchanges that occur in real-time, with individuals seeing one another as they interact. Lord *et al.* (2016) further predicted that networks with terabit speeds will overcome existing Internet infrastructures, allowing interactive broadband services to become common by 2025. Subsequently, diverse arguments of the 1990s identifying cyberspace culture as being largely anonymous and rich in text will become substantially difficult to sustain in the future.

2.2.6 Recent Advancement in Cyberspace Culture

As cyberspace culture evolves, it brings about a wide variety of complex security issues at a global and national scale. As a result, new strategies that comprehensively cover these security issues must be formulated in order to secure cyberspace operations. Consistent evolution of cyberspace culture eventually leads to numerous innovations and technological advancements to societies, governments, national defence systems, and economies (Gartzke, 2015). However, it is important to highlight that cyberspace culture has contributed to both positive and negative technological advancements. A side-by-side comparison of these positive and negative contributions reveals beneficial innovations such as secure money transfers, online distance learning, social media, and advancements in collaborative scientific research across the globe (Tran, Yonatany and Mahnke, 2016). In contrast, however, negative cyberspace culture has also brought about a rise in viruses, malware, and other forms of cybercrime, including terrorism (Goodman *et al.*, 2007).

As cyberspace culture is continuously advancing and developing in almost all sectors, it has been observed that further progress is being made in enabling connectivity of multiple devices across a network and, most importantly, across wireless platforms and the Internet. These advancements have been witnessed across various sectors, including higher education and healthcare. Additionally, the trends in cyberspace evolution have also led to the development of other

45

underlying factors, such as social, economic, and other related changes (Bendovschi, 2015). All these aspects of change and evolution in cyberspace bring us to a world in which the networking environment and cyberspace are supposed to be directly involved in the daily lives of the entire population. This subsequently involves changes in the technological, social, and economic aspects of cyberspace (Bower, 2001).

The two outstanding trends in these cyberspace changes comprise lower costs of producing processors, including a reduction in size and increase in processing speed, and the connectivity of networks and their flexibility. With these increasing advancements and technology, cyberspace is slowly evolving to a point where intelligence will be embedded within the machines and wireless technology will be developed to impressive levels. In turn, this means that cyberspace is slowly transforming from what it is today into a global overlay of technical infrastructure. Furthermore, cyberspace is also more involved in most aspects of our daily lives, such as communication, health, education, transport, and defence, among others (Neaimi and Ranginya, 2014). The trends in computer and network capabilities are highlighted below, as they lead to the overall evolvement of the network space.

2.3 Challenges in the Evolution of Cyberspace Culture

Inasmuch as the evolution of cyberspace culture contributes to diverse fields, such as innovation and education, health systems, commerce, and general societal advancements, several challenges must also be addressed in order to facilitate technology's implementation in the new era of cyberspace and to benefit the society and the world at large (Galinec *et al.*, 2017). According to Galinec *et al.* (2017), four key challenges and threats influence cyberspace culture: cyberwar, economic espionage, cybercrime, and cyberterrorism.

Cyberwar describes an action or actions initiated by an international organisation or a local nationgovernment in an attempt to cause damage to computers or information networks belonging to another nation, such as through the spread of computer viruses and the launch of denial-of-service attacks. Robertson and Arnold (2018) highlighted two real-life examples of cyberwarfare: The accusation that Russians manipulated the 2016 U.S. presidential elections through social media and the development of the Stuxnet worm by U.S. and Israeli intelligence in an effort to sabotage Iranian nuclear power projects. Baezner and Robin (2017) further reported cyberattacks by the U.S. and U.K. governments against Syria due to the nation's development and use of chemical weapons. Zetter (2016) additionally discussed the shutdown of Ukraine's power grid by Russian hackers in 2015.

Despite using different techniques, these different instances featured a confluence in that one or more nations launched an attack with the intention of causing harm to another. The Cyber Survey Report by Deloitte also reported that, in 2017, unauthorised access to systems represented the major threat faced by digital companies (Deloitte, 2017). Other threats mentioned included hacking (13%), malware and viruses (13%), denial-of-service attacks (5%), fraud (4%), and terrorism (3%).

According to Janczewski and Colarik (2005), cyberterrorism describes any form of premeditated or politically motivated attack by an individual, clandestine agent, or group against the information systems and computer data of another nation that results in violence against the attackers. The authors differentiated cyberterrorism from cyberwar by highlighting that cyberwar leads to enemy losses, whereas cyberterrorism leads to violence against the attackers. Bidgoli (2004) added that cyberterrorism also seeks to instil fear by producing uncertainty and confusion within a population with the goal of coercing its government to conform to a given political or social agenda. The threats stem from cyber insecurity and address the various security threats created by the Internet, the vulnerabilities in preserving the privacy and confidentiality of data, and the threats brought about by including cyberspace culture in the production of modern weapons.

The evolution and use of modern weapons in cyberspace could also threaten the relations between countries, especially when the weapons are misused, potentially leading to the development of inter-government wars. On the Internet, global criminal organisations are also misusing cyberspace to recruit new members and coordinate criminal activities, potentially by spreading false messages and encouraging violent actions online. Jacobson (2010) reported that diverse global terrorist groups from different parts of the world rely on the Internet to spread their nefarious agendas, recruit new agents, and raise funds to support their activities. Bieda and Halawi (2015) added that cyberspace offers a decentralised global platform that facilitates communication among individuals from diverse backgrounds. Through the Internet, terrorists can spread their agendas to

a large number of people by adopting specific platforms that guarantee anonymity, thereby ensuring they remain undetected (Bieda and Halawi, 2015). Furthermore, the Internet enables terrorists to interact with their sympathisers and targets in real time, making it easier to spread fear and uncertainty on a global scale.

However, cyberspace culture also facilitates access to positive information infrastructure. Cyberspace coexists with real space without necessarily being limited by national boundaries, thus enabling better and faster communication among individuals. Scientists and researchers can accordingly collaborate in diverse projects regardless of their locations across the world. Today, the evolution of cyberspace culture is occurring more quickly than the current technologies in the field of cybersecurity. Cyber warfare—a form of latent aggression initiated by one government or organised crime group—represents one of the many challenges faced by cyberspace culture today.

The issues that arise from using technology in areas such as education, health, and other government sectors, such as infrastructural development, must be addressed. For instance, this may involve issues such as the errors that may arise from using new methods and machines that have not been adequately tested. The attacks on government information may also lead to issues such as impersonation and financial attacks intended to weaken the government. In education and financial sectors, corruption can be easily developed, since utilising the Internet in corrupt activities may be untraceable, especially if performed in remote locations via the application of cloud technology (Galinec *et al.*, 2017). This makes aspects such as online education and elearning appear less promising due to this opening of opportunities for fraudulent activities.

In the following section, the Qatar National Vision, 2030 (QNV 2030) is reviewed in greater detail. The Researcher begins by providing an overview of the QNV 2030, after which each of the pillars in the QNV 2030 is examined.

2.4 Qatar Vision, 2030

In October, 2008, the QNV 2030 was launched as a plan to develop and transform Qatar into an advanced community capable of making certain achievements regarding sustainable development.

This vision plan divides its development goals into four pillars: social, economic, human, and environmental.

2.4.1 Overview of the QNV 2030

The QNV 2030 was created in 2008 in response to an identified need to build a sustainable economy based on knowledge instead of carbon fuels, which were challenged by their susceptibility to depletion (Qatar National Vision, 2030, 2008). As such, the QNV 2030 seeks to transform Qatar into a modern country capable of ensuring sustainable development and high living standards by 2030. This vision serves two main functions: Defining country outcomes on a long-term basis and outlining a framework to facilitate the creation and implementation of national policies and plans.

The QNV 2030 seeks to address several challenges ranging from modernisation and tradition preservation to managing growth and uncontrolled expansion. Similarly, it also seeks to bridge the present and future by providing a roadmap for successful and sustainable development. To that end, the QNV 2030 is founded on the previously mentioned four key pillars of development: human, environmental, social, and economic (Qatar National Vision, 2030, 2008). In regards to human development, Qatar aims to develop its people in an effort to ensure sustainability. The vision plan outlines several strategies to achieve this goal, such as prioritising and investing in world-class scientific research and first-rate education so as to unlock its people's potential; increasing the participation of Qatari people in its labour force in contrast to expatriates; attracting qualified expatriates; promoting access to better healthcare services; and boosting moral, traditional, and cultural heritage (Qatar National Vision, 2030, 2008).

With the social development pillar, the QNV 2030 strives to develop a caring and just society based on the highest moral standards and capable of playing significant roles in global and local partnerships. Locally, the government seeks to promote equality, tolerance, humanitarianism, and empowered women in order to contribute to different economic spheres. Globally, the vision plan seeks to improve partnerships and international cooperation between Qatar and its allies so as to enhance cultural exchange and coexistence levels (Qatar National Vision, 2030, 2008).

The economic pillar, meanwhile, seeks to promote sound management of the economy by fostering attractive business climates. Additionally, it aims to ensure responsible exploitation of hydrocarbon resources to cater to future generations and pushes a transition towards knowledge-based economies that are more globally competitive (Qatar National Vision, 2030, 2008).

Finally, the QNV 2030's environmental pillar seeks to ensure that the country balances meeting developmental goals and protecting the economy. To this end, the government aims to build legal systems and institutions that can vouch for the protection and preservation of the environment. Similarly, it also intends to create sustainable policies that will guide future urban and infrastructural development while minimising its impact on the environment. Figure 2-2 below summarises the four pillars of the QNV 2030:



Institutional development and modernisation

Figure 2-2: Qatar's National Vision, 2030 pillars. Source: Qatar National Vision, 2030 (2008)

An in-depth analysis of the four pillars reveals a need to adopt technology in order to successfully implement the outlined strategies. This stems from the fact that the plan specifies a goal of modernising Qatar by focussing on a knowledge-based economy and moving away from carbon fuels. However, by adopting technology, an additional concern arises concerning cyberspace

culture and its undue influence on attaining this vision. The following sub-section advances this discussion by examining contributions of cyberspace culture to the QNV 2030.

2.4.2 Contributions of the Cyberspace Culture to the QNV 2030

The current sub-section discusses cyberspace culture's different contributions to the QNV 2030. The first sub-section details how the culture is facilitating economic development in aspects such as securing financial services, boosting competitiveness, and enhancing healthcare. In the second sub-section, cyberspace culture's influence on the education sector is discussed further, and contributions to e-education are outlined. The final sub-section delves further into cyberspace culture's influence on the improvement of social services and environmental conservation.

2.4.2.1 Economic Development

Economic development represents a fundamental issue influencing a country's attainment of economic freedom and its provision of better services, opportunities, and ways of life for its citizens. In the hope of achieving this goal of economic freedom, Qatar seeks to create a working balance between the base of oil and knowledge in terms of driving the country's economy, which would subsequently lead to a sustainable environment in terms of business, encouraging equality in competition, attracting investors, and stimulating the economy's growth. The powerful technologies resulting from cyberspace culture can help Qatar attain this economic independence. For instance, technology will enable an easy transformation of functions previously performed on paper, such as creating documents and sending letters, which can be performed electronically via email (Goi, 2009). A second application concerns utilising the Internet to conduct trade. In this case, this refers to electronic forms of payment included to encourage international trade and remove trade barriers produced by different currencies and countries.

Furthermore, Qatar can also leverage advanced telecommunications technology in order to boost its economic competitiveness. For instance, by implementing transparent communication processes and liberalising telecommunications, Qatar can foster sustainable business environments to attract more investors to the country (Dvornik and Sabolić, 2007). It has also been argued by Dvornik and Sabolić (2007) that investments in telecommunication systems act as stimuli for economic growth. Furthermore, the society also benefits from the provision of more efficient telecommunication services. Additionally, significant research has been conducted on the current market trends that can be leveraged in order to obtain intelligence from the Internet prior to making decisions regarding business investments (Chang and Grabosky, 2017). Integrating powerful cognitive tools and machines also enhances trading activities. This includes the machine's ability to analyse large amounts of data produced in finance management, as well as how using such machines in trade increases efficiency and transaction speed (Pope, 2011). All these elements have increased the efficiency in conducting trading activities, thus building the country's economy.

Smart city solutions can also be leveraged in an effort to improve management and ensure better delivery of transport and logistics services, waste management, and control of urban centres (Kumar *et al.*, 2018). Kumar *et al.* (2018) further noted that smart technologies are improving the performance of urban services through intelligent and smarter digitalisation efforts. Similarly, urban authorities and policy-makers are being empowered to deliver services more responsively and efficiently. Caragliu and Del Bo (2019) further highlighted that smart city policies boost urban innovation efforts while Haarstad and Wathne (2019) revealed that adopting smartness increases the agenda to enhance sustainability in urban areas. Qatar thus possesses the potential to boost its economic performance by leveraging positive cyberspace culture.

Big data and artificial intelligence represent important emergent technologies that can benefit Qatar in enhancing its governance initiatives. For instance, these technologies can be employed to help minimise incidences of fraud and corruption in public institutions and enhance regulatory quality, as well as adherence to the rule of law (Curti and MihoV 2018). Big data can also be employed to enhance strategic management by facilitating the generation of important insights from existent voluminous data, thereby facilitating decision making and improving the planning of different tactical and operational activities (Silahtaroğlu and Alayoglu, 2016; Merendino *et al.*, 2018). Such insights can boost economic development, particularly when employed in different businesses in Qatar.

2.4.2.2 Education Development

In the education sector, the QNV 2030 strives to develop a strategic plan ensuring credibility in the education process and empowering Qatar's population through world-class education. To achieve these goals, the QNV 2030 seeks to involve the use of scientific and new technologies of research and the Internet (Gasser and Drolshammer, 2015). The e-education objective seeks to support a flexible learning environment allowing students, teachers, and the general population to

52

access information in a timely manner. Therefore, this implies that education will not be limited solely to school-going students, as adults will also be empowered to further their careers.

Qatar learning institutions can benefit significantly from implementing modern tech-based learning infrastructure that guarantees optimal learning for students and improves content delivery for the teachers. Talebian *et al.* (2014) revealed that ICT technologies alleviate constraints of place and time, thereby boosting learning among students. Adults keen on advancing their careers also benefit from the flexibility of learning at their own pace whilst undertaking their work objectives. Technology-based learning further boosts current educational models, shifting from the traditional teacher-centred approaches to more self-directed content-centred approaches allowing students to direct their own learning (Dunn and Kennedy, 2019). Bognar, Sablić, and Škugor (2018) further reported that flipped learning was also facilitating online discussions, thereby promoting the expression of thoughts and feelings in a friendly manner. Beyond this, Alamri, Watson, and Watson (2020) observed that blended learning environments facilitated personalised learning models in higher education, thereby supporting independent research.

Scientific research in Qatar will also be significantly boosted following the adoption of advanced technologies. However, this will involve acquiring advanced world-class equipment as well as implementing advanced scientific processes. As a result, innovation processes will be improved by improving knowledge-searching processes. Zhou *et al.* (2019) further highlighted how emerging technologies influence scientific knowledge convergence. Additionally, technology boosts collaborative efforts between Qatar and its allies, particularly in advancing science.

2.4.2.3 Improvement of Social Services

The improvement of cyberspace culture has helped Qatar in social areas, specifically in the improved provision of social services and social security (Cottier and Drolshammer, 2018). This also includes improved privacy and better enforcement of privacy concerns, such as confidentiality. Furthermore, the process of data mining, personal information identification, and fair information practices represent some of the areas that will be improved on the path to achieving the QNV 2030.

Data mining, which involves obtaining and using information efficiently, will be improved through services such as Googling of the information and by providing data centres where users can obtain information on certain aspects of their life and matters that concern and interest them (Fiddner, 2015). The identification of users in the online space will also help attain the goals of better data management and efficient planning regarding how the obtained data are employed. Addressing privacy concerns further helps achieve Qatar's goals of better data management and ICT-related issues regarding data use and management. Likewise, social services such as the provision of assistance for solving issues in the social space will also be improved.

The QNV 2030 has been deliberately created in order to align with the International Vision, 2030 for Sustainable Development created by the United Nations. This agenda seeks to empower people and the planet and to ensure prosperity. The primary goal is to ensure peace at the universal level and the general prosperity of the nations and their people. It is also intended to eradicate poverty in all spheres and areas, including poverty at extreme levels—one of the main challenges facing the globe and indispensable for attaining sustainable development goals. Such goals and targets are anticipated to generate action over the next 15 years in several critically important areas for humanity and the planet.

Regarding people, the goal is to end poverty and hunger so as to be accomplished in all forms and dimensions. This should subsequently lead to all human beings fulfilling the potential for a healthy environment with equality and dignity. Regarding the planet, the primary determination is to ensure it remains protected from destruction by promoting sustainable production and consumption that ensures the planet is protected from eventual degradation. Furthermore, this also involves sustainable natural resource management and urgently acting on climate change in order to support any generational requirements in the present and future.

2.5 Cyberspace Culture in Service Industries and Higher Education

This section is further divided into two sub-sections, the first of which examines cyberspace culture in the general service sector. Cyberspace culture's influence on the provision of services such as finance, healthcare, and education in general is thus discussed. Thereafter, the second sub-section details how cyberspace culture affects the provision of higher education in Qatar.

2.5.1 Cyberspace Culture and Service Industries in Qatar

The QNV 2030 seeks to modernise Qatar by 2030 through its four core pillars (social, human, economic, and environment). In-depth analysis reveals that cyberspace culture will directly influence their attainment based on the fact that the QNV 2030 aims to transform Qatar into a knowledge-based economy. Cyberspace culture will thus influence different sectors of the economy ranging from manufacturing and production to service delivery. The current sub-section highlights some of cyberspace culture's expected influences on service industries in the country.

To begin, cyberspace culture is anticipated to transform the delivery of government services through the adoption of e-government, defined as the deployment of information and communication technologies to transform the government's function to be more accessible, accountable, and effective (Al-Hakim, 2007). Through e-government, the government utilises ICT technologies to serve citizens and external clients by meeting their informational and transactional needs (Kettani and Moulin, 2015).

Sun *et al.* (2015) postulated that effective e-government delivery transforms the interrelationship between citizens and the government, leading to diverse benefits such as increased convenience and reduced costs for citizens as well as higher transparency levels and revenue streams for the government. Qatar residents will benefit from the launch of one-stop government portals providing access to different e-services, ranging from the issuance of visas and birth certificates to making payments for utility bills and license renewals.

As cyberspace culture improves access to government services, it further generates a ripple effect regarding the growth of large enterprises and SMEs. First, this is accomplished by enabling them to access required services easily, such as through faster application processes for official business permits and payment for different rates and utilities. Second, cyberspace culture also positively influences SMEs' operations, such as by enabling them to conduct business online through e-commerce platforms. Savrul *et al.* (2014) further revealed that advanced technology enables enterprises to cooperate on a global scale, transforming their commercial activities from local to international spheres. Third, cyberspace culture will also influence decision-making processes in enterprises by generating actionable intelligence through analysis of available data. Ghasemaghaei

(2019) highlighted that data analytics improves the quality of decisions, and thus advocated their use in business. Cyberspace culture is expected to transform Qatar's business industry, especially given that the country is shifting towards a knowledge-based economy. With increased reliance on technology, IT-consulting organisations are expected to emerge over time in order to support the knowledge economy.

The health industry will also be influenced by cyberspace culture through the adoption of electronic health platforms to efficiently manage patient records and allow both medical practitioners and patients to access important medical health information (Cummings *et al.*, 2009). As a result, it is expected that health-service delivery will improve in the country as patients more easily gain access to practitioners. E-health innovations thus possess the potential to improve health-service delivery, making it safer, efficient, affordable, and more universally accessible.

Finally, cyberspace culture is also anticipated to influence the provision of financial services (Claessens *et al.*, 2002). Studies have highlighted that new forms of banking, such as online and mobile banking, have emerged with the Internet's increased distribution. As a result, banks can serve more customers and offer a wide variety of services compared to traditional brick-and-mortar banking institutions. Directly, e-finance implies convenience for customers and quicker access to their finances, while indirectly, it also intimates an improvement in retail business, as transactions can be completed flexibly and independent of geographical locations. Banking institutions will also benefit from e-finance as customers use their services across different economic sectors, effectively generating higher revenue for them.

2.5.2 Cyberspace Culture and Higher Education in Qatar

Cyberspace culture affects higher education in Qatar in two main forms: teaching and learning and the management of learning institutions. The current sub-section discusses cyberspace culture's influence on these two different aspects.

2.5.2.1 Improvement of Learning and Teaching

It has been argued that the Internet world can be considered a fabrication developed using technology and serving as a substitute for reality. In essence, educational cyberspace culture can empower teachers and students with diverse experiences, thereby implying that the experiences might not represent a satisfactory alternative to real life due to their virtual nature. While an

increased speed for training teachers in computer-mediated technology has been identified (Pugalee and Robinson, 1998), along with a commensurate need to connect students to the Internet, it nonetheless remains important to ensure cyberspace is used cautiously so as to avoid disappointing long-term results.

The use of IT in institutions of learning has changed the life and learning process of learning centres. The most basic purpose of education is to gain knowledge of reading, writing, and arithmetic, which is considered literacy. However, the advent of the modern world has led to including critical thinking skills and information and technological literacy on this list as well. Chris Dede, a technology expert, stated in 1966 that all human beings must obtain knowledge in a specific field that will drive them towards a sea of information and specific qualifications. This concept would later be implemented in the technology sphere, leading to the revolution of cyberspace and the education sector in particular.

A shift is being observed in the education sphere as learners are moving towards modern approaches and abandoning the reliance on hardcopy books and other more traditional forms of education (Ho *et al.*, 2010). Some common approaches include using technology in learning, which encompasses the use of technological material or utilising the Internet, also known as online education. This term describes the use of computer and mediated communication via the Internet to engage in learning and teaching in an attempt to attain an institution's learning objectives. The concept of online education includes aspects such as online, web-based, Internet-based, and e-learning. Johnson et al. (2018) further highlights that utilising video-based teaching platforms in online environments also enables teachers to better prepare in the online teaching playgrounds. Xu and Dai (2019) continued by reporting that adopting situational teaching improved online learning experiences and, as a result, advocated for its adoption.

Another fundamental change noted in the education sector in regards to the integration of cyberspace culture involves the transformation of classrooms from conventional brick-and-mortar spaces to more abstract online classrooms that transcend geographical regions and integrate individuals of different nationalities in one classroom (Ho *et al.*, 2010). A second improvement concerns the integration of analytics in education cyberspace culture. Conventionally, student

performance has been analysed through manual approaches such as evaluating examination papers via marking and ranking students after collating all marks.

Analytics, however, is revolutionising the education sector, as teachers can now identify weak areas upon analysing student history, and thus can assist their learners appropriately. Advanced computing tools such as analytic software facilitates such tasks, making it easier for teachers to aid their students. The Internet further enables students to learn on their own in a flexible manner as they access a wide range of content stored in different databases for different education levels.

The use of technology in educational research represents another important evolution in regards to cyberspace and education. By adopting advanced computer technologies such as drones and machines, the research process is further enhanced and becomes much easier based on the access to different testing, analysis, and evaluation tools. Consequently, new approaches to delivery and teaching emerge through technology, leading to the development of new scientific skills in education. The primary importance of including technology in education is that this enables achieving the target of making education open to all, making this a crucial step in the future development of education.

2.5.2.2 Improvement of Management

Cyberspace culture has also led to the introduction of novel management approaches in higher education institutions. One such application includes student support technology, whereby advanced applications are employed by management in order to track students' performance and encourage them to remain on course and finish their programmes (Natow *et al.*, 2017). Though the support technologies are centred on students, however, they also provide important insight for the management to reduce attrition rates. Support technology applications include aspects such as reminding students about upcoming classes, ongoing exhibitions on campus, or setting appointments with course instructors and administrators. As students utilise these technologies, management can track student engagement levels and introduce appropriate strategies to correct any noted issues.

Additionally, cyberspace culture has brought about the utilisation of analytics and big data to enhance decision-making processes in higher education institutions (Liebowitz, 2016). One study

revealed that big data enables management to gain more actionable insight from the analysis of operational and administrative data. Luo (2019) further reported that utilising big data to manage student affairs was becoming a common trend in higher education and was advocated for improving normative management and leading to the adoption of scientific decision-making processes. Institutional analytics represent a core application associated with cyberspace culture that covers diverse aspects ranging from structural to policy analytics. Administrators are empowered to make better data-driven decisions across all major institutional departments while management can also employ the different analytics to make better policies supported by data (Daniel, 2014).

Daniel (2014) further highlighted that analytics also facilitate student-performance management through data-mining applications, which allow management to predict students who are unlikely to complete different courses by analysing already available diverse student data. For instance, this includes data on previous academic performance, classroom behaviour, number of re-enrolments, financial aid, and courses taken. Likewise, Jha, Jha, and O'Brien (2019) reported that tracking of student activity data using big data was also important for identifying students who were struggling to complete their course work and, as a result, led to lower dropout rates. Cyberspace culture is revolutionising management by providing important insights that would otherwise be obtained from outsourced experts.

2.5.3 Culture Management as a Controversial Idea

As discussed in this section, technology is being adopted to facilitate managing various activities and functions in higher education institutions. However, it is worth noting that management of culture represents a controversial idea, as culture is often understood as an emergent, anthropological, and natural concept as opposed to something that can be managed. This assertion is supported by Holbeche (2015), who argued that culture is often difficult to manage and even more complicated to change, as it is broad and deep. Likewise, culture is also deeply imprinted into the employees and interconnected with the firm's politics as well as its structures, people, and priorities (Holbeche, 2015). Alvesson (2014) further stresses that culture is as significant and difficult to manage as it is difficult to understand and use in a thoughtful manner. In another study Buil, de Chernatony, and Martínez (2012) postulate that culture is inherently complex to define due to its multifaceted nature; for instance, it can be categorised into diverse dimensions such as collectivism, individualism, femininity, and masculinity, among others. As a result, researchers on culture often utilise demographic variables such as race, ethnicity, language, and socioeconomic status as a proxy for culture, thereby leading to an overreliance on group comparisons. The suggestion from different studies (Buil, de Chernatony and Martínez, 2012; Alvesson, 2014; Holbeche, 2015) is that culture can be considered an abstract concept that remains difficult to define, as it exists within individuals and can only be evaluated using proxies. Therefore, any attempts to manage culture produce unanticipated reactions or resistance in the form of counter-cultures. In the same vein, it is also difficult to develop a standard approach to manage culture, as it differs among different individual groups.

2.6 Guiding Principles for Rulemaking in the Cyberspace Culture

The legal policy-making instruments significantly influence policy formulation and implementation when conceptualising cyber rules. As a result, this plays an important role that must be carefully examined before a final decision can be made regarding the cyberspace culture rules and frameworks that are developed. One observation is that political forces have always focussed on striving to be part of the team that administers and organises cyberspace despite the fact that the scientific world and private actors were all active in major milestone developments.

Politics heavily influence the decision-making processes related to cyberspace culture. In most cases, two core powers influence decision-making, including how the government has maintained dominance and the way power has been actively or inactively distributed in the government (Baker, 2016). Government power involves the organisation of political power and how it influences the overall decision-making process of cyberspace- and Internet-governing rules (Barzilai, 2004). The rules developed to govern the cyberspace culture frameworks should thus be developed according to the national governments' competency. However, this includes not only the security issues, but also the usage of cyberspace.

The current and ongoing challenges concerning the regulation of cyberspace culture require a more collective decision-making process covering a broader concept than the traditional form (Weber, 2012). However, one aspect that must be considered involves the influence of international political law. Cyberspace represents an international concept that transcends national boundaries, thus requiring policy-making processes to assume an international approach (Barzilai, 2003). As

a result, global governance will be utilised in order to collectively consider the input of different governments and geographical and global entities, since cyberspace is not limited by geographical boundaries and exerts its influence internationally.

Therefore, it is imperative that global governance includes the collective efforts of cyberspace as an international concept whilst ensuring political leaders remain accountable and responsible in their roles. As such, this must acknowledge the fact that cyberspace culture problems are solved at an international level that transcends the particular governments dealing with them. Subsequently, creation of cyberspace culture frameworks should include collective efforts of entire domestic areas and theories of governments that are independent. The consideration of the type of principle that should be adopted should also be performed in consideration of the general guidelines of independent governments. Tackling the problems for ensuring safety should also be performed with the governments' independence in mind.

An in-depth analysis of the different regulation models that might have led to the development of a new legal order indicates that fresh approaches ought to be employed to build a legal framework suited to cyberspace culture. Subsequently, this will lead to the recognition of a need for the framework's development by concerned members of the civil society and business owners (Weber, 2012). This assessment should also reveal and conclude that, even if the new models are put into practice and established properly, the assessment remains insufficient for constructing an entire legal framework. As such, in general, this includes the process of identifying and obtaining the scope and limits that may lead to a proper establishment of the pillars of cyberspace culture that will be reliable in the future.

The development of the legal framework should further consider the optimal intended conditions, and the safety of society should further be taken into close consideration (Goi, 2009). Utopia, a concept developed by Sir Thomas Morus in 1516, is not a suitable consideration during a rule development plan, as it bears difficulties in identifying data constraints and is inappropriate for combining different elements of a plan in an effort to develop a framework that is globally accepted.

2.7 Recognition of Cyberspace Culture Openness

A side-by-side comparison of cyberspace culture and the information available on the Internet reveals a peculiar finding—namely, that the shared information represents an individual, and the individual thus represents information shared in cyberspace (Rahmed, 2018). This finding directly implies a need for a user-centred approach when devising laws, rules, and regulations to govern cyberspace culture and eventually benefit individuals who live in the digital landscape (Goi, 2009). A need for openness thus arises in developing rules in cyberspace culture. Openness further intimates that the rules are developed transparently without any bias or favouring for a given group, organisation or individual.

Furthermore, the creation of rules to govern cyberspace culture should adhere to rules of neutrality, transparency, and the switching or contract exit (Audibert and Murray, 2016). The common questions that must therefore be considered when creating the rules—and later in their implementation—is summarised as follows:

- 1. What should be considered in order to ensure transparency and openness between the stakeholders, consumer groups, and standard experts?
- 2. How can it best be ensured that the benefits to be gained from cyberspace culture are maintained even with increased scrutiny and multiple arrangements of the stakeholders?
- 3. How can the users of cyberspace culture be provided with the best and most efficient methods of understanding cyberspace's influence on their personal information?

The freedoms of information and communication are crucial for developing the overall cyberspace culture environment. The openness of cyberspace thus coherently aligns with the principle that the Internet represents a public sphere and must be seen as a universality concept (Goi, 2009). When the same is viewed in this perspective, the Internet's openness can be utilised to combat the fragmentation of the network structure (Fiddner, 2015). The European Commission accordingly outlined that the vision of the cyberspace governorship must feature one network that is fragmented.

Incorporating cultural diversity and freedom of expression together with addressing the privacy concerns of cyberspace aids in developing the proper framework for embracing cyberspace culture (Goi, 2009). Other relevant sections include the Internet's neutrality, openness of its standards, open process, multi-linguist, multiple accesses to information, the ethics concerned, and the quality of content (Silverman, 2016). These aspects represent vital elements that must be considered in developing the cyberspace culture and framework.

Fabre *et al.* (2016) emphasised that cyberspace law must adhere to and enshrine three general principles: openness, neutrality, and interoperability. However, this introduces diverse challenges, as some governments and markets are bound to mount resistance against the creation of universal laws. To that end, the authors identified several solutions that could be considered. On the one hand, a Confucian model of self-regulation was proposed wherein citizens generally respect established traditions without utilising any instruments of coercion. However, this approach has been criticised for assuming that all individuals possess similar knowledge regarding the use of cyberspace—a finding that is incorrect, as a few individuals, such as hackers, possess mastery over cyberspace.

On the other hand, it has also been recommended to adopt cyberspace rules that parallel international environmental laws (Fabre *et al.*, 2016). With this approach, any misdeed by citizens of a nation are borne by the government that failed to prevent their occurrence. In hindsight, however, this leads back to the concept of territoriality, as it refutes the universality concept of cyberspace.

A third approach has also been suggested whereby frameworks parallel to international public law are considered (Fabre *et al.*, 2016). With this approach, the laws assume a universality concept in the same way that international public law focusses on a common heritage of all humanity. The Internet would therefore become an asset shared universally by all members and regulated through international treaties. Despite highlighting the different options, however, the authors nonetheless acknowledged that cyberspace regulation has yet to be resolved.

2.8 Gaps in Cyberspace and the QNV 2030

In this section, the gaps identified in cyberspace and the QNV 2030 are delineated. Beyond this, other issues are discussed as well, such as trust among stakeholders of the QNV 2030 and enterprise cyberspace management.

2.8.1 Gaps in Cyberspace

Several research gaps exist regarding the use, development, and governance of cyberspace culture, and particularly the development of the information infrastructure (Silver, 2016). Furthermore, gaps have also been identified in several other areas regarding the application of cyberspace, including the use of computers, their related products, and accessories such as desktop operating systems, routers, and devices designed to facilitate the transfer of messages between two or more computers (Brunn, 2014). However, a different school of thought views this as more of a problem rather than a gap. The development of Internet infrastructure has evolved significantly over the last three decades, changing how cyberspace is viewed and correspondingly utilised to conduct business, run operations, and provide government activities, as well as to ensure people use cyberspace safely.

As these changes become a more permanent feature of life, this further implies an aspect of irreversibility whereby technology will become embedded in almost every kind of activity. Accordingly, this highlights a need for the general population to integrate advanced engineering and technological advancements in their day-to-day business activities. Beyond this, other cyberspace-affiliated aspects have also evolved and improved over time, such as the security considerations of cyberspace culture, use of cyberspace in academic learning, and private use of cyberspace developments in learning (Brunn, 2014). The finding that cyberspace culture encompasses a wide variety of research areas further indicates the existence of gaps that should be carefully examined and filled (Silver, 2016).

2.8.2 Gaps in the QNV 2030

An additional factor that also needs to be thoroughly researched concerns the wide-scale use of cyberspace and the different types of associated problems that arise concerning data security when shared through smaller computer networks and the more expansive Internet platform. Similarly, a concern should also be raised regarding the erosion of people's social, political, and cultural values

due to the Internet's influence. The Internet platform creates a habitable digital landscape wherein different individuals live and, as a result, form a culture for its usage. The United States, for instance, has experienced a vast investment in cyberspace, leading to the development of a cyberspace culture that, unfortunately, has not yet been sufficiently analysed. Further analysis reveals that cybersecurity is the only broad area regarding cyberspace that has been sufficiently analysed despite not being well related to the procedures involved in attaining the QNV 2030 (Brunn, 2014).

The QNV 2030 project thus needs to keenly examine and understand the implications of cyberspace culture that heavily influence the attainment of the QNV 2030, and this should also be related to the education sector. The pillar of economic development, for example, represents one of the major areas guiding the QNV 2030 project. The QNV 2030's strategy of economic development tends to focus on how natural resources could be harnessed together in order to develop an effective management strategy for the QNV 2030. The lack of proper research concerning the effects of cyberspace culture and the approaches employed to integrate it into the QNV 2030 will hinder its successful attainment. As such, the different key aspects of economic development must be included in the cyberspace review.

Integrating cyberspace culture in economic development also possesses an anticipated impact on the population. Specifically, this pillar focusses on integrating Islamic concepts and cultures throughout Qatar as well as rapid humanitarian development in certain areas (Repko, 2018). Subsequently, this calls for research regarding the impact of cyberspace and cyberspace culture and how it affects or may be used to attain the QNV 2030. The implementation of social programmes that call for keen and continuous monitoring should also be strictly observed in order to ensure that all the frameworks and development areas are considered (Fiddner, 2015). The forging of stronger terms regarding the relationship with other countries and regional development comprises another key area that has not yet been adequately researched, including how it will be affected by the integration of cyberspace culture and use.

Human development, which in this case encompasses the health system, education system, cultural awareness, and impact on citizens, is also vital in research. The focus on creating increased and

improved cultural awareness, improved healthcare, and increased use of the Internet and related material in the education system is also vital, but this has not yet been adequately researched (Repko, 2018). In order to address the low rate of Qataris joining the workforce, the use of cyberspace technology should be integrated to foster such developments and attain the QNV 2030 in the area of human development. Therefore, this represents a gap that should be filled and which requires the aeration of a framework to address it.

Environment development comprises another area that should be carefully analysed to ensure successful attainment of the QNV 2030. In this case, the QNV 2030 must be aligned with cyberspace culture in an effort to further generate Internet awareness of environmental conservation and to educate the population regarding its importance (Fiddner, 2015). Research should thus be conducted through the funding of the Qatar National Vision in a bid fill this gap. Consequently, three key research areas are identified and summarised in the following sub-section in an effort to fill said gaps.

2.8.3 Trust among Distributed Parties

Attaining the QNV 2030 involves cooperation between different entities and parties; as a result, it is necessary to establish a working and beneficial relationship without involving a third party or a central society or organisation. The existing research, particularly concerning how different entities and organisations need to establish trust in their data and the cyberspace technologies, has highlighted their needs based on the enterprise users (Silver, 2016). Current research has not adequately filled the gap in this area, however, and so more solutions should be developed by conducting further research.

The scale, complexity, autonomy, and dynamism required to address issues pertaining to the use of cyberspace technology and infrastructure should also be developed and highlighted thorough research. The current research needs to address the models' need for entities that are autonomous and organisations that are geographically distributed. Therefore, a research gap exists regarding the secure relationships on the network in peer-to-peer or other network settings (Fiddner, 2015). In order to achieve such goals as community, health, education, and economic development, further research is necessary.

2.8.4 Enterprise Cyberspace Management

As different entities collaborate in achieving the QNV 2030, a need further arises to link different structural elements of infrastructure, owned by different individuals and enterprises, to cyberspace. Enterprise cyberspace management represents a challenge that must be leveraged so as to integrate different frameworks into one model that can be used to develop cyberspace. The primary function is thus to combine different forms of security and computer mechanisms that will control the behaviour and usage of cyberspace in specific ways. While the market possesses a proliferation concerning the best ways research on cyberspace use could be maximised, a gap remains concerning enterprise cyberspace management (Scharfenort, 2012). For a technology that is still emerging, the research on the legal policy makings and different frameworks governing the same remains under investigation, and it should be carefully analysed and developed in order to lay out implications and their anticipated effects for attaining the QNV 2030, including how they can be mitigated and regulated appropriately (Fiddner, 2015).

The research to be undertaken should emphasise developing the structure of different dynamics, identifying technology implications for possible users, and developing and deploying the usage of cyberspace in general. Beyond this, research should also be undertaken to identify the implications of implementing the alternative and structures for allocating the different positions in diverse application areas in cyberspace. In particular, it is necessary to conduct thorough and comprehensive research on improving cyberspace and the role it plays in attaining the QNV 2030 (Silver, 2016). The collection and retaining or use of infrastructure, and the policies that should be implemented, should also be carefully investigated to determine how they affect attaining the QNV 2030.

2.8.5 Metrics and Models

Organisations, individuals, and critical infrastructure, as well as the associated sectors, must rely on information to implement the risks and challenges that come with adopting and implementing cyberspace and infrastructure (Fiddner, 2015). For the different organisations and individuals to effectively manage cyberspace, which includes accepting, transferring, and externalising the imposed risks, along with applying the necessary resources to reduce them, research must be conducted to manage the same. Research is thus needed to provide a foundation for obtaining and appropriately using data in cyberspace (Weber, 2012). The benefits, impacts, and costs for the

economy, the health system, the education system, and the QNV 2030 goal should also be considered during the research, as a gap exists in this area as well.

2.9 Conclusion

This chapter reviewed diverse literature sources in order to identify current knowledge regarding various thematic areas in the research. Definitions of different terms such as cyberspace, culture, and cybersecurity were detailed in the first section, followed by challenges for the evolution of cyberspace culture. The chapter further reviewed the QNV 2030, demonstrating its interaction with cyberspace culture. In the fourth section, cyberspace culture's impact on service industries and higher education was also explored with respect to Qatar, where different technologies were identified. Guiding principles for developing rules in cyberspace culture were also examined. Finally, gaps in the QNV 2030 were identified and discussed, including aspects such as trust among distributed parties and enterprise cyberspace management. In the following chapter, the reviewed literature is evaluated.

Chapter 3: Literature Evaluation

3.1 Introduction

The current chapter furthers the discussion from the previous chapter (Chapter 2) by critically evaluating the presented literature. By examining the different highlighted thematic areas, the Researcher establishes the existence of cyberspace culture, its alignment with the QNV 2030, and prevalent challenges prohibiting its further development. Beyond this, the Researcher also focusses on establishing the formality of cyberspace culture and its effectiveness in the higher education sector. To that end, the evaluation highlights gaps in reviewed literature that develop into baselines for developing theoretical frameworks guiding the development of cyberspace culture and aligning it to the QNV 2030 vision for higher education. The different thematic areas associated with cyberspace culture are discussed in the first section of the chapter. Particular focus is directed towards establishing the definition of cyberspace culture and its implication for the higher education sector. Thereafter, the rationale for the proposed frameworks (presented in Chapter 4) is detailed.

3.2 Literature Evaluation

In this section, literature detailed under different thematic areas in Chapter 2 is evaluated using the four-quadrant framework (Althonayan, 2003). The motivation of adopting this tool stems from the fact that it categorises reviewed research according to purpose and outcome. With the purpose category, the framework breaks the research further into visionary and implementational. Since the research focusses on cyberspace culture and attaining the QNV 2030, the visionary research focusses on the vision of cyberspace culture and how it affects attaining the QNV 2030 in higher education. Furthermore, the concern is to establish the nature of the current cyberspace culture and its implications as it develops over time.

Implementational research focusses on the practical aspects of the implementation process, such as detailing practical recommendations to facilitate adopting positive cyberspace cultures in higher education institutions, thereby implementing the QNV 2030. Both visionary and implementational research are further sub-divided into descriptive and prescriptive outcomes, leading to a fourquadrant research matrix facilitating the evaluation of the studied literature. The four quadrants include the following: i) visionary descriptive, ii) visionary prescriptive, iii) implementational descriptive, and iv) implementational descriptive (Althonayan, 2003). Table 3-1 below summarises the literature evaluation framework based on the four-quadrant tool:

Research Philosophy				
		Visionary	Implementational	
		Quadrant I	Quadrant III	
	Descriptive	 Describes cyberspace culture definitions and outlines their impact on culture formation Theoretical definitions of core cyberspace culture concepts and their impact on higher education. Research based on theoretical assumptions supported by literature. Likelihood to detail conceptual frameworks that model the interrelation between theoretical concepts. 	 Describes frameworks and processes of implementing cyberspace culture to facilitate the attainment of the QNV 2030 in higher education Establishes the existence of frameworks to foster cyberspace culture in higher education. Describes cyberspace culture implementation issues in higher education supported by literature. Implementation guidelines and discussions, as well as theoretical frameworks, are considered. 	
		Quadrant II	Quadrant IV	
Research Outcomes	Prescriptive	 Outlines prescriptive approaches of cyberspace culture and discusses its alignment to the QNV 2030's goal for higher education Prescriptive discussion of the impact of cyberspace culture on the QNV 2030's higher education goal. 	 Outlines prescriptive cyberspace culture alignment approaches and discusses processes of aligning it in higher education The alignment of cyberspace culture towards the QNV 	

Table 3-1: Literature evaluation framework.

Likely to introduce prescriptive	2030 goal for higher
theoretical frameworks illustrating	education may be discussed.
how cyberspace culture ought to be	Challenges and practical
aligned to the QNV 2030's higher	recommendations regarding
education vision.	the alignment of cyberspace
Literature-supported discussions	culture and the QNV 2030's
concerning the vision of aligning	goal for higher education
cyberspace culture to the QNV	highlighted.
2030 in higher education.	• Detail benefits of aligning
	cyberspace culture and the
	QNV 2030's goal for higher
	education

Source: Adopted from (Althonayan, 2003)

Table 3-1 above details the different concepts examined in this research concerning the deployment of the research quadrant framework. The Researcher assigned the different literature studies (discussed in Chapter 2) into respective quadrants so as to highlight literature gaps influencing the current research. A side-by-side comparison of the different quadrants further highlights the particular aspects comprising the literature gap.

3.2.1 Quadrant I Analysis

One of the core aspects of this research involved understanding the cyberspace concept and how it fostered culture, thereby influencing the actions and behaviours of actors who inhabit the virtual landscape. To that end, Section 2.2 defined several important concepts affiliated with cyberspace: cyberspace, cyberspace culture, and cybersecurity. In defining the term 'cyberspace', the reviewed literature demonstrated that the concept's definition has broadened and complexified since its inception in Gibson's dystopian novel *Neuromancer* in 1984 (Gibson, 1984). Additionally, over time, numerous definitions have emerged across different disciplines ranging from anthropology to philosophy. Therefore, this sub-section examines the different viewpoints for defining cyberspace as an abstract and imaginary space is discussed, after which the transition to a cultural space is detailed. It is worth noting that the studies examined in these two sub-sections are specific to quadrant I in Table 3-1 above.

3.2.1.1 Cyberspace as an Abstract Concept

Gibson (1984) described the cyberspace concept as a data representation that was graphical in nature and drawn from the memory banks of all computers in the human system. The significance of this definition is that it conceptualises cyberspace as an abstract and virtual non-physical concept. Second, it highlights that interaction with computers represents a fundamental aspect of cyberspace. However, Gibson's (1984) definition remains limited in that it fails to clearly outline the type of inhabitants and interactions that exist in this space. An additional philosophical view by Thomas Hobbes in the *Phantasm of Space* (Slowik, 2014) conveyed a similar idea by delineating cyberspace as an unbounded continuum void of matter (Berdayes and Murphy, 2000).

Benedikt (2000) offered a more detailed definition by relating cyberspace to a world possessing a global traffic of secrets, entertainments, measurements, knowledge, and other forms of alterhuman agency assuming a unique structure (sounds, presences, sights) of a vast, blossoming electronic light never before seen on the earth's surface. Benedikt further added that cyberspace represents a common mental geography built by revolution and consensus, and where diverse agents interact with one another to share data, make deals, or share dreams (Benedikt, 2000).

While the definitions by these three different authors shed light on the abstract and imaginary nature of cyberspace, they fail to address how cyberspace intertwines with everyday life in the modern age. For the most part, the definition by Benedikt (2000) intricately describes the nature of interaction within cyberspace through senses such as voice and eyesight, as well as the achievements made by these interactions, including enquiry, making deals, sharing dreams, and simply beholding. It is only upon review of Bell's (2001) work that the concept of cyberspace culture in the modern age comes to light. According to the author, cyberspace represents a hypothetical expanse between computers that allows people to develop new selves and worlds—an aspect described as a symbolic story (Bell, 2001).

Bell (2001) added that cyberspace facilitates interaction between remote actors (material stories) through hardware and software. In this way, the author emphasised the nature of cyberspace as being centred on people, their cognition of self, and the experiences they create in the space. The manner in which people interact and experience cyberspace further serves as a negotiation of

symbolic elements and materials, referred to as experiential stories (Bell, 2001). Consequently, the understanding of cyberspace shifts from an imaginary abstract concept to one centred on individuals and the identities they can assume in such spaces.

3.2.1.2 Cyberspace as a Unique Cultural Concept

Turkle (1995) visualised cyberspace as a simulation culture wherein, even though people fail to see one another, they are presented with an opportunity to create unique communities where they can engage with other individuals on a daily basis and establish deep and intricate relationships. The author thus argued that cyberspace experiences can only be discerned in a cultural context due to the fact that cyberspace creates a new environment for undertaking social and cultural engagement.

Escobar *et al.* (1994) assumed an anthropological view of cyberspace, emphasising the opinion that technology serves as a cultural innovation emanating from specific cultural conditions and facilitating the creation of new cultural and social conditions. The author argued that cyberspace is a culture based on the view that it originates from an established socio-cultural matrix. Therefore, a need arises to discern the two cyberspace phases, the first of which concerns its representation of an imaginary landscape created between computer networks while the second involves its conceptualisation as an expanse that enables interactions between humans, thereby creating new socio-cultural communities and human experiences positioned in the different contexts of their day-to-day lives.

As a cultural space, cyberspace influences the members who live within it (Spencer-Oatey, 2008; Eliot, 2014). This assertion is based on the fact that an individual's culture remains dependent on that of the group to which he or she belongs while the culture of said group is further influenced by the larger society. People interact on a human–human basis through computer-mediated networks, as well as with the software and hardware of Internet technology. As such, the construction of knowledge and the learning of different behaviours in cyberspace occurs in a unique fashion due to the inherent nature of the space.

Consequently, it is necessary to understand the features of cyberspace as a unique cultural space. One such feature involves the universal-without-totality concept, which implies that, though
cyberspace culture is virtually present among all humanity, it lacks totality due to the numerous differences present within it (Le'vy, 2001). Universality possesses several implications, indicating a lack of a central governing authority over the entire space. As a result, negative behaviour has emerged and threatened cyberspace culture over the years, including cyberwar, economic espionage, cybercrime, and cyberterrorism (Galinec *et al.*, 2017).

A second feature concerns cyberspace's invisibility, a concept whereby people are present with each other online, but remain unable to see one another. The significance of invisibility is that it can influence the formation of unique identities and personal characteristics via cultural and social interactions mediated by computer networks (Hine, 2000; Jones, 1995; Nakamura, 2002; Palloff and Pratt, 1999; Turkle, 1995). When people are invisible to one another, it becomes possible to invent multiple personas, leading to the formation of unique identities, both online and in real life (Turkle, 1995).

In the realm of higher education, the diversity of identity and personality formation under conditions of invisibility and universal-without-totality in cyberspace represents a highly important research concept despite the little attention it has received thus far. For this reason, the following sub-section examines the interaction of cyberspace culture and higher education, including current gaps regarding their alignment.

3.2.2 Quadrant II Analysis

This sub-section analyses quadrant II, whereby cyberspace as a tool of learning is discussed further.

3.2.2.1 Cyberspace as a Tool of Learning

In quadrant II, the Researcher outlined cyberspace culture's visionary prescriptive influence on higher education, thereby describing what is expected when the two align accordingly. The QNV 2030 focusses on boosting teaching initiatives, making a shift from traditional teacher-centred approaches to more self-directed and content-centred approaches wherein students are positioned to direct their own learning (Dunn and Kennedy, 2019). Additionally, the literature revealed that the QNV 2030 seeks to empower the scientific community by encouraging collaboration with other researchers in different parts of the world and facilitating the convergence of scientific knowledge in the country (Zhou *et al.*, 2019). However, in order to comprehend the different ways in which

cyberspace culture will influence the attainment of these goals, and to outline current research gaps, it is expedient to examine how cyberspace culture influences learners and teachers separately. As such, this leads to the population of literature in quadrant II.

The evaluation of diverse research studies indicates a divergence in the views held by researchers and educational technologists regarding cyberspace, with researchers adopting the view that cyberspace serves as a cultural space whereas educational technologists perceive cyberspace as an important tool facilitating meaningful learning (Crossman, 1997; Harasim, 1993; Jonassen *et al.*, 1999; Khan, 1997; Land and Bayne, 2012; Zoto *et al.*, 2018). Each of the two paradigms significantly influence the actions undertaken by educators and learners in cyberspace. The general argument advanced in this research is that educational technologists, in an effort to maximise the amount of learning that can be achieved in cyberspace, advocate its adoption as a teaching site.

As a result, cyberspace is viewed as a separate entity empowering human beings to complete different activities. For instance, based on this view, educators and learners utilise cyberspace to solve problems and construct meaningful learning (Duffy and Jonassen, 1992; Jonassen, 1993). When solving problems, both educators and learners search for information on cyberspace and compare different content in order to draw meaningful insights (Jonassen *et al.*, 2003). With the rapid advancement of technology leading to the Internet of things, mobile computing, cloud computing, and big data, cyberspace has grown rapidly to contain diverse types of information that enhance problem solving (Fang *et al.*, 2017). Furthermore, educators also rely on the technologies to create course content and instructional materials more efficiently by building on the research work of other scholars.

Cyberspace further augments the learning process by providing a virtual presence and enabling learners to engage in learning activities without limits of time or place (Hung and Chen, 2001; De Witt and Gloerfeld, 2017). Mobile devices and digital media encourage personalisation and diversified learning formats, as well as different forms of communication and cooperation. Subsequently, learners can experience authentic situations on cyberspace as they boost their learning efforts.

Third, cyberspace also boosts learners' participation and engagement levels, particularly in situations where some feel disadvantaged in taking part in classroom discussions due to factors such as language barriers (Szücs *et al.*, 2013). In other instances, only a small group of learners might dominate classroom discussions, thereby preventing the majority from actively engaging in class (Cope and Kalantzis, 2017). However, through tools such as discussion boards, which are readily available in cyberspace, less-active learners can engage with other students and instructors in ways that are more comfortable to them (Luyegu, 2016). Beyond this, instructors can also use the discussion boards to evaluate student understanding of different concepts taught in class.

Finally, cyberspace also provides numerous tools facilitating collaboration among learners. The social theory of constructivism and learning (Bednar *et al.*, 1992; Lave and Wenger, 1991) posits that collaboration activities help learners appreciate multiple perspectives and construct shared meanings. Furthermore, core cyberspace technologies such as email, instant messenger, and discussion groups extend the diversity of collaborative work outside the classroom, thus fostering further learning regardless of location (Fabos and Young, 1999; McLellan, 1998; Reynolds, 2002; Riel, 1995). Since these technologies are widely adopted, this indicates that they will influence how users learn and collaborate.

3.2.3 Quadrant III and IV Evaluation

This sub-section focusses on literature assigned to quadrants III and IV, which focus on implementational research aspects. Quadrant III literature has descriptively established the existence of frameworks aligning cyberspace culture and higher education, followed by discussions of anticipated challenges in their creation. With quadrant IV literature, the Researcher discussed the prescriptive benefits of implementing cyberspace culture within Qatar's higher education vision and highlighted current challenges in achieving this alignment. Furthermore, the literature also covers practical recommendations to facilitate the alignment.

The literature in quadrants III and IV focussed on aligning cyberspace culture to the QNV 2030 in higher education in terms of implementation. Descriptively, this implies identifying the different ways in which cyberspace culture can either prohibit or enhance the attainment of Qatar's vision for higher education in quadrant III while, prescriptively, this involves identifying implementational issues that will affect their interaction in quadrant IV. A brief analysis of cyberspace culture's contribution to the achievement of the QNV 2030, as detailed in Sub-Section 2.4.2.2, demonstrates that it emphasised empowering students (both adults and college students) to acquire learning more flexibly without limits of time and place (Talebian *et al.*, 2014). In effect, this highlights the significant role that cyberspace will play in the learning undertaken by users.

3.2.4 Quadrant III Analysis

Researchers have further conceptualised cyberspace as a cultural space (Bell, 2001; St. Amant, 2007). Interactions between technology and society have been argued to occur in a two-way process wherein computers give form to ideas, experiences, and metaphors while they themselves are shaped by these aspects (Bell, 2001). Therefore, inasmuch as cyberspace augments the abilities of learners and educators, it is also important to highlight that, as individuals interact with technology on a day-to-day basis, cyberspace culture emerges and influences different aspects of this interaction, such as identity formation and the social behaviours they adopt in the virtual landscape. These two interrelated aspects cannot be ignored, as using cyberspace to augment one's problem-solving capabilities, for instance, directly translates to spending significant amounts of time in the virtual landscape and, subsequently, creating identities within this space. The literature examined in this sub-section is specific to quadrant III, which focusses on descriptive approaches to aligning cyberspace culture to the QNV 2030's goal for higher education, as stipulated in the literature. Therefore, this sub-section examines how cyberspace, as a cultural space, influences actors (educators and learners) and identifies current gaps that can be filled through this research.

The first examined aspect concerns cyberspace culture's influence over how learning occurs in cyberspace. Jones (1998) argued that, since cyberspace represents a cultural space, a social theory of learning should be employed as a lens for examining learning in this digital space. The author further stated that cyberspace, as a cultural society, comprises an ethnographic site inhabited by people with complex power relationships. Markham's (1998) research on cyberspace also described it as an ethnographic site in which participants think, feel, and act differently in the digital space. In Sub-Section 3.2.2, De Witt and Gloerfeld (2017) postulated that cyberspace boosts learning by providing students with flexible platforms that offer authentic, personalised experiences not limited to time or place. This view resonates with the situated theory of learning, which posits that learning is unintentional and occurs within authentic activities, context, and

culture (Lave and Wenger, 1991). As such, situated learning can also be considered an appropriate lens for examining how learning occurs in cyberspace.

The second discussed aspect pertains to how learning is delivered in cyberspace. The author highlighted that modern cyberspace is structured in such a way that only a few empowered entities provide the larger un-empowered population with digital technology platforms to undertake different activities. For instance, an analysis of social media vendors revealed that few renowned vendors, such as Twitter and Facebook, control the greater number of users on the platform. In the educational technology space, Gregg *et al.* (2017) reported that, while numerous vendors of elearning platforms exist in the world, only a few of them dominate this space, such as Blackboard, Cisco systems, and Pearson Education. Most higher education institutions tend to rely on major providers as opposed to establishing their own learning platforms due to best practices, which advocate adopting more established systems due to their security and innovation capabilities. In effect, this leads to the notion of 'gatekeepers', or large entities that control how learning occurs in cyberspace. Directly, this implies that creative freedom is restricted, and so the Internet morphs into a place where users become consumers rather than creators (Lessig, 2002).

The third investigated aspect concerns the creation of content in cyberspace and its impact on creators. Despite the fears associated with cyberspace gatekeepers and their influence on limiting creativity within the digital landscape, cyberspace remains highly expansive with flows of global information full of pulses associated with personal experiences (Bell, 2009). As such, in spite of prevalent limitations posed by gatekeepers, digital cyberspace still avails opportunities for different individuals to author, remix, create, and share their freedom fuelled by the Internet to perform diverse activities (Lessig, 2002; Lankshear and Knobel, 2008). Therefore, individuals will continue to produce new content and identities and, subsequently, new cultures. Interests in control of institutions have thus not succeeded in limiting creative freedoms, and individuality remains in global forums. One argument that has been advanced is that personal experiences are still founded in customs, habits, and values, both locally and regionally (Palfrey and Gasser, 2008). The enablement of global culture by digital technologies intimates that individuals will be situated in individual contexts and express themselves in cyberspace.

3.2.5 Quadrant IV Analysis

As digital technologies and cyberspace continue to permeate different aspects of society, diverse authors (Landow, 1992); Apple and Jungck, 1996; Raschke, 2003; Aarseth, 1997) have advanced further debates regarding whether computers act as vehicles of oppression or empowerment. For the purposes of this research, this issue is not moralistic as to whether digital technologies can be considered good or bad in the hands of educators, nor does it imply liberation is oppressive or emancipatory. Rather, it is dynamic in nature, shifting as its users, technologies, and purposes do. In this case, it is not the form that matters, but the content in terms of consumption and production and its effect on users' subject positioning and perceived identity. Martin-Albo and Gregorio-Godeo (2017) also argued that authoring facilitates the creation of identities in figured worlds as social and cultural products. However, in order to understand the formation of identity, the authors highlighted a need to understand this as a dynamic rather than static process. In this research, cyberspace serves as the figured world wherein teachers and educators form new identities through authoring.

Spivey (1997) further discussed the identity-authoring phase that emerges by creating academic, literary, and disciplinary texts, arguing that authors align with the themes and forms they develop through their work. As the author generates a set of texts, a union occurs as the individual and texts become intertwined with each other. In the context of this research, the concept of text is expanded to consider digital and multimodal texts in cyberspace. Therefore, the argument is advanced that teachers who generate online texts are comparable to academic and literary authors. Valdez and Omerbašić (2015) further highlighted that authoring spaces influence identity negotiation and the creation of experiences for both students and educators. Another argument by Soreide (2006) posited that construction and negotiation of identities occurs among teachers as they make presentations related to their work.

The current research thus relates critical theory and digital literacies as they pertain to teachers' authorship of self in educational social structures. Digital literacies and technologies possess no inherent morality (Meyers *et al.*, 2013). Instead, they augment human capabilities and, through technology, enable transformations that alter social structures (McLuhan, 1964). The individual's positions on different subjects is changed as they use technology, which further transforms their

senses and reworks their social structures and interactions. It is thus anticipated that the individual creates new positions aligned with their personal goals rather than being unconsciously influenced by external effects (Palfrey and Gasser, 2008). In this way, each technological stage enhances communication and transforms social structures and consciousness among humans (Ong, 1989).

The phrase 'digital literacy' has been used to express how students, learners, and individuals in society interact with one another through computers (Helyer, 2015). However, with advancements in technology, new interfaces are being created via a wide array of devices, such as mp3 players, mobile devices, and smartphones. As such, the definition of digital literacy should be expanded to cover such aspects (Bell, 2009; Misty, 2015). Such devices enable individuals to author different types of content, including text, audio, and images, thus influencing their positions and identities in cyberspace.

Cyberspace transforms the creation and management of identities as interactions are made more visible through social networks in online spaces, leading to diverse positions. Furthermore, boundaries differentiating identities in online and offline spaces have continued to blur as users represent identities across multiple spaces, cyber or otherwise (Palfrey and Gasser, 2008). Identities that have been created from reinvention and experimentation are presented in a variety of ways. However, a major concern stems from the fact that social identities are more varied and richer and are not highly controlled by individuals as they express themselves in cyberspace. Each individual in cyberspace develops his or her own hypertext that serves as a representation of the manner in which they see the world and how they place it as they decode cultural texts.

The works of Freire (1970), Lessig (2004), and Palfrey and Gasser (2008) framed the current Researcher's approach. A substantive framework of these works influenced the Researcher's thinking and lead to a focus on certain issues. Concerned with control, Freire (1970) explored liberating and oppressive actions. The actions taken by an individual contribute to transforming or reinforcing the status quo. Situated within this research, teachers' visibility increases through actions in cyberspace, perhaps contributing to social inequalities.

Lessig (2004), Palfrey, and Gasser (2008) examined issues related to digital literacies. Lessig (2004) explored creativity and how the Internet enables producing content through remixing. Creating in cyberspace allows individuals to actively participate in the producing culture. Palfrey and Gasser (2008) discussed how participating in online social structures such as Facebook contributes to the authorship of self, both online and offline. The implications are relevant to this research, which involves teachers who operate in cyberspace in increasingly visible ways. Extrapolating from the works of Freire (1970, 1973), Lessig (2004, 2008), and Palfrey and Gasser (2008), teachers' actions in cyberspace shape identity and culture in addition to possibly transforming or reinforcing the status quo within school social structures.

By examining teachers' actions and the online content they create, this research emphasises describing authorship and exploration of the self, as well as the likelihood for change in various practices of teaching. In itself, self-authoring is dynamic and integrates a form of constantly shifting empowerment comprised of horizons, situated literacies, and borders. The term '*situated literacies*' was employed by Street (2003) to define literacy as a practice that was social in domains and events rooted in knowledge concepts, identity, and being. Literacy is both social and active, as people conceptualise it based on their culture and the different ways in which they employ it. However, an additional factor regarding the creation of boundaries to limit authorship activities also emerges.

Nonetheless, the creation of online content provides a technique for boundaries to be crossed and dialogue to be initiated with other individuals. The argument advanced is that liberation and empowerment form part of self-authorship and are not given or won externally (Freire, 1970; McLaren, 1995). Empowerment begins with a self-perception shift as the individual author internally morphs from a state of oppression to empowerment, which further manifests externally. The narrative provided by Douglass (1995) demonstrates this point, highlighting the power of this shift in self-perception as the researcher's identity changes from a slave to an oppressed human being. The current research explores self-perception processes when teachers interact with digital spaces; the social structures further bring about empowerment and liberation actions, or else cause further oppression. This research thus also attempts to comprehend the shift in the authorship of teachers' selves, revealing positionings as they generate more content in cyberspace. Teachers

described what transpires as they intermix their lives with practices that are digital and professional in cyberspace.

3.3 Existent Literature Gaps in Cyberspace Culture and Higher Education

The analysis conducted in quadrant I detailed findings regarding how cyberspace culture could enhance the attainment of the QNV 2030. Based on the view that cyberspace can be used as a learning tool that augments the learning experience for students and teaching experience for instructors, an identified literature gap concerns the lack of frameworks guiding both parties in aligning cyberspace culture to the established goals. Such frameworks would help learners understand cyberspace as an external identity that helps augment their learning capabilities by improving collaboration and virtual presence and which aids in solving different problems. Stepwise guidance is necessary to lead novice learners to attain expertise in employing cyberspace culture to boost their learning experiences. Similarly, instructors also need to be guided so as to appreciate cyberspace's value in enhancing their teaching activities. For instance, the frameworks would guide educators in selecting the best cyberspace teaching tools and improve the preparation of course instruction.

Based on literature evaluated in quadrants II and III, several research gaps were also identified. The first concerns learning in cyberspace and questions how the unique and situated nature of cyberspace can be leveraged to enhance the QNV 2030's attainment. For instance, a need was noted for higher education institutions to adopt technologies available across multiple platforms, such as mobile devices or desktop computers, in order to provide a seamless learning experience and leverage situatedness. It is further argued that, since most of the learning that occurs in cyberspace is situated, it is necessary to develop best practices to ensure that the community of students and educators will easily boost their learning experiences in the digital space.

The best practices will guide novice learners in acquiring expertise through participation that is peripheral in the cyberspace community practice. Additionally, it was noted that, though a dominant viewpoint of cyberspace identifies it as a learning tool in educational technology, few studies have established how novice learners become experts through a community of practice in cyberspace. A greater number of studies have emphasised the effectiveness of communication tools that learners employ in cyberspace or the design of learning environments for scaffolding learners' meaningful learning. If educational technologists wish to design effective learning environments for successful learning in cyberspace, they should know how learning occurs there. Concerning how learning is delivered in cyberspace, a second research gap also emerged regarding whether Qatar would rely on dominant entities to supply educational technology to facilitate the QNV 2030's attainment in higher education. A need has arisen to prepare contingency plans for both scenarios whereby, if the institutions choose to adopt popular digital technologies to facilitate attaining the QNV 2030, frameworks and best practices must be put in place to guide the procurement of digital technologies employed in learning. The Researcher further reported that frameworks have been developed to facilitate integrating blended learning in higher education (Garrison and Vaughan, 2013), blogging in higher education (Kerawalla *et al.*, 2009), reengineering of educational technology education (Kelley and Kellam, 2009), and more. However, these are only to be used as a basis for understanding the integration process, as they are not specific to the situations in Qatar. Meanwhile, where institutions opt to develop their own technologies to facilitate cyberspace learning, the Researcher argues that there will be a need to develop frameworks and best practices that guide the development processes.

In regards to the creation of content in cyberspace, a third research gap was also identified in guiding authorship among teachers in digital landscape. This gap concerns the identities of teachers who employ websites and social networks to add to their teaching practices, as well as what happens to them as they use digital technologies to go about their lives and their teaching. Drawing on the perspectives identified in the ongoing debate concerning authorship and creative freedom in cyberspace, it has been revealed that cyberspace provides both learners and educators with an avenue to create new identities through global interactions. When left unchecked, these identities and social interactions are likely to lead to nefarious activities, such as terrorism and cybercrime. Therefore, the Researcher identified a gap in creating frameworks and best practices that will foster positive cyberspace culture to boost the attainment of the QNV 2030. However, these frameworks must be created in lieu of the fact that cyberspace governance remains challenged by the Internet's openness and the seeming lack of jurisdiction. To the best of the Researcher's knowledge, no such frameworks exist in Qatar, specifically in aligning cyberspace culture to the QNV 2030.

In quadrant IV, a fourth research gap was also identified regarding how cyberspace influences identity formation. From the revelation that personalised devices aid in authoring different types of content, authorship is thus not limited to educators and teachers, but also comes from learners as they employ cyberspace to create authentic experiences. In this research, the form of the digital devices, tools, and programmes is arbitrary; the focus remains on the teachers' situated practices, as well as the representation and understanding of ideas and selves. The current research thus explores authorship and subject positioning through social structures, creation of spaces online, digital mixing, and content in cyberspace.

Cyberspace culture studies have been employed to refer to studies that consider interactions of culture, technology, reality, and identity in cyberspace. Cyberspace culture studies are transdisciplinary, representing diverse strands of work across numerous subject areas (Bell, 2009). Cyberspace culture describes the ways in which life is modified by representations and practices embedded in cyberspace. In effect, this describes a manner of thinking about interactions between people and technology based on narratives outlining the different ways in which digital technologies have changed lives and led to the construction of new identities (Bell, 2009).

Research Philosophy					
		Visionary	Implementational		
		Quadrant I	Quadrant III		
	Descriptive	(Gibson, 1984)	(Gregg, Holsing and Rocco, 2017)		
Research Outcomes		(Turkle, 1995)	(Lessig, 2002)		
		(Jones, 1995)	(Carr, 2009)		
		(Palloff and Pratt, 1999)	(Garrison and Vaughan, 2013)		
		(Berdayes and Murphy, 2000)	(Kerawalla et al., 2009)		
		(Escobar, 2000)	(Kelley and Kellam, 2009)		
		(Hine, 2000)	(Valdez and Omerbašić, 2015)		
		(Bell, 2001)			
		(Le'vy, 2001)			
		(Nakamura, 2002)			
		(Spencer-Oatey, 2008)			
		(Eliot, 2014)			

Table 3-2: Research literature evaluation.

	(Galinec <i>et al.</i> , 2017)	
	Quadrant II	Quadrant IV
	(Talebian, Mohammadi and Rezvanfar,	(Jones, 1998)
	2014)	(Markham, 1998)
	(Dunn and Kennedy, 2019)	(Bell, 2001)
	(Zhou <i>et al.</i> , 2019)	(St. Amant, 2007)
	(Crossman, 1997) (Harasim, 1993)	(Kiliç, 2018)
	(Jonassen, Peck, and Wilson, 1999)	(De Witt and Gloerfeld, 2017)
	(Khan, 1997)	(Lave and Wenger, 1991)
	(Land and Bayne, 2012) (Zoto et al.,	(Bell, 2009)
ive	2018)	(Palfrey and Gasser, 2008)
ript	(Duffy and Jonassen, 1992)	
esc	(Jonassen, 1993) (Jonassen et al., 2003)	
Pr	(Fang <i>et al.</i> , 2017)	
	(Hung and Chen, 2001)	
	(De Witt and Gloerfeld, 2017)	
	(Szücs <i>et al.</i> , 2013)	
	(Cope and Kalantzis, 2017)	
	(Luyegu, 2016) (Bednar et al., 1992)	
	(Lave and Wenger, 1991)	
	(Fabos and Young, 1999) (McLellan,	
	1998) (Reynolds, 2002)	
	(Riel, 1995)	

Source: Adopted from (Althonayan, 2003)

3.4 Conclusion

This chapter evaluated the reviewed literature in depth based on the four-quadrant framework analysis approach. Subsequently, several research gaps were identified in each of the quadrants, effectively providing a rationale for the framework development chapter. The view that cyberspace can be employed as a learning tool that augments students' learning experience and instructors' teaching experience indicates a gap concerning the lack of frameworks guiding both parties to adopt cyberspace culture appropriately. Furthermore, cyberspace's unique and situated nature further indicates a need for best practices to guide novice users in learning institutions to adopt technologies that are available across multiple platforms, such as mobile devices and desktop computers, in order to provide a seamless learning experience and leverage situatedness. Additionally, only a limited number of studies have outlined how novice learners become experts through a community of practice in cyberspace. The importance of frameworks to guide such development thus emerges in order to ensure meaningful learning in diverse learning environments. Similarly, such frameworks guide the creation of content in cyberspace identity formation without adversely affecting the actors in such spaces. In the next chapter, the procedure adopted for developing this research's conceptual framework is detailed. Likewise, existent related frameworks and theoretical models are also reviewed to facilitate the model's conceptualisation.

Chapter 4: Framework Development

4.1 Introduction

In the previous chapter, the Researcher presented a literature review of different themes in the research, after which current research gaps were identified. Cyberspace culture can be considered a relatively new concept based on the fact that the influence of technological culture, as a new research paradigm, has not been widely studied over the last three decades (Manuel, 2010). As the rate of change increases in the field of information and communication technology (ICT), the levels of risks posed to information-asset security similarly increase. As a result, information security culture is focussing on socio-cultural measures that produce an important influence on supporting security methods to improve employees' daily lives (Schlienger and Teufel, 2003). Therefore, processes for understanding and identifying beliefs, ideas, and values of different groups based on information security are essential for shaping and guiding business and security-affiliated strategies (Ramachandran *et al.*, 2008).

Malcolmson (2009) argued that the security culture adopted in an organisation could significantly influence the security of its human and physical resources. The author posited that security culture affects the nature of interactions occurring between employees in the organisation and established procedures outlining acceptable and unacceptable behaviour. Nonetheless, diverse theories and models have been developed to explain how culture is created and maintained. However, limited studies have examined how cyberspace culture is fostered and maintained, thus requiring further investigation. In this chapter, the key cultural theories of Schein (1984) and Hofstede (1983) are analysed, along with current frameworks of cyberspace culture.

Following this, the proposed framework is presented after careful analysis of the existent literature gaps. Four cyberspace culture frameworks are selected to guide the development of said proposed framework: the ISF (Alnatheer and Nelson, 2009), the ISCF (Da Veiga and Eloff, 2010), the Information Security Culture Framework (Shabbaan, 2014), and the Information Security STOPE framework (Al-Hogail, 2015). These frameworks are examined further in this chapter.

4.2 Organisation Culture

The focus of this research concerns gaining an understanding of how cyberspace culture facilitates the QNV 2030's implementation in higher education. Since this research conceptualises higher education institutions as organisations, theories of organisation culture are deemed appropriate for understanding how cyberspace culture influences their operation. Furthermore, drawing from other cyberspace culture studies on higher education (Okada and Ferreira, 2012; Junqueira, 2015; Teixeira *et al.*, 2017), the Researcher determined that organisations and communities are referred to as the basis of the research. Therefore, organisation culture is adopted as a lens for researching the different aspects associated with fostering cyberspace culture.

A confluence observed from the review of diverse studies concerns a lack of a single accepted definition of organisational culture in the literature, given that numerous scholars and authors have interpreted culture in different ways over the years. According to Louis (1980), organisational culture can be considered a group of shared meanings by people that, though largely tacit among members, are nonetheless clearly relevant and distinctive to the given group and are also passed to new members. Chatman and Eunyoung (2003), meanwhile, further defined this as an arrangement of shared assumptions, values, and beliefs that direct people's awareness of suitable and unsuitable behaviour.

A different perspective by Alvesson (2003) described culture as 'an umbrella concept for a way of thinking which takes a serious interest in cultural and symbolic phenomena'. The author further added assumptions of social reality and values to the definition. A more popular and widely referenced definition by Schein (1992) posited that organisation culture describes a set of essential assumptions shared by a group as they learn how to solve their problems of internal integration and external adaptation, and which have worked well enough that they are considered sufficiently valid to be taught to new members as the proper way to think, perceive, and feel in regards to the problems.

While diverse theories have broadly explained how culture is created, this research adopts two theories based on their appropriateness for this research—namely, Hofstede's cultural dimensions theory and Edger Schein's organisation culture theory. A major reason for choosing these two theories stems from their wide-scale acceptance in the scientific community and their use in diverse cultural studies in different contexts and geographical regions across the world

4.3.1 Hofstede's Cultural Dimension

As reiterated in the previous sub-section, organisation behaviour comprises an important element, and Hofstede (1983) strongly believed that the presence of local and national cultural groups can influence organisational behaviour. As such, this theory was developed to comprehend cultural

differences across countries and differentiate various methods of conducting business across different cultures. An argument advanced by the author was that organisational culture is constrained by national culture (Hofstede, 2001). Culture, as a largely human process, involves different approaches for implementation across different societies. Each society possesses a unique set of values, beliefs, and way of life, which present themselves as cultural differences. Hofstede outlined five dimensions of culture: individualism vs collectivism, power distance, uncertainty avoidance, and masculinity versus femininity. Figure 4-1 below summarises these dimensions:



Figure 4-1: Cultural dimensions. Source: Hofstede (2001)

Power distance focusses on the degree to which less-powerful individuals accept that power distribution in the society is actually unequal. Uncertainty avoidance concerns people in society that cannot accept ambiguities or any sort of uncertainty, while individualism and collectivism emphasise people's preference to either be alone or to hold strong ties with people in society and live in a knitted sort of close network. The masculinity dimension focusses on the society's preference for being heroic, assertive, and hard working for their achievements while femininity represents the cooperative, modest, and quality-of-life-based traits. Long-term orientation focusses on finding virtue in the long term.

The key contribution of Hofstede's theory in this research concerns the identification of different cultural dimensions that characterise different regions or countries. As each dimension demonstrates different behaviour in various societies, this helps contextualise Qatar as a masculine or patriarchal society characterised by high power distance and uncertainty avoidance. Directly, this indicates that Qatar possesses a conservative culture and is highly likely to resist changes in their cultural attitudes. The findings in the current research will therefore refute or confirm such claims.

With individualism, people tend to look after only themselves and their immediate blood relations. In contrast, collectivist cultures feature people being integrated into solid groups and looking after the entire group in exchange for intemperate loyalty. According to Hofstede (2010), the United States scores highly as an individualistic culture, unlike Indonesia, which records a lower score. Inequalities also occur in human societies in terms of power and wealth. As a result, power distances emerge that can be considered large or small. Countries such as Malaysia possess large power distances, thereby indicating that humans are perceived as less important. Such societies emphasise inequality, unlike the Netherlands, where its small power distance boosts equality among people.

The uncertainty of the future implies that people ought not to be upset, but should rather demonstrate tolerance for ambiguity. Hofstede asserted that this dimension was not familiar among many individuals, especially in the case of the IBM, as the author resorted to using available data from its archives (Hofstede, 2009). In countries featuring high uncertainty avoidance, such as Japan, Hofstede revealed that they are aggressive and respond better to navigation aspects, unlike Singapore, a country possessing low uncertainty avoidance and that tolerates such behaviour. Where countries display high masculinity, traditional roles are anticipated, with men's values being more pronounced than those of women. However, an overlap of gender roles can be observed in highly feminine countries, as men not only dominate, but also facilitate (Hofstede, 1984, 2010). In spite of the wide recognition of Hofstede's framework, it has also been criticised by numerous authors. For instance, McSweeney (2002) did not accept this framework as a universal approach, believing instead that surveys were not a reliable method and that nations should not be employed as a unit for examining cultural differences. The author further elaborated that this framework was not reliable, as employee results drawn from one company could not represent an entire score of the nation as a whole, and relying on only five dimensions was not sufficient to completely

represent cultural characteristics. McSweeney (2002) further argued that IBM data were old and could not apply in the modern century. Orr and Hauser (2008), meanwhile, expressed the same criticism when evaluating Hofstede's dimensions. The authors identified a need to thoroughly reexamine theoretical constructs within the 21st century context of behavioural patterns and crosscultural attitudes. Signorini *et al.* (2009) continued by criticising the 'over-simplification' of differences in culture and inconsistencies between categories in Hofstede's 5-D model in addition to identifying fault in it being static rather than dynamic. Other scholars (Kwon, 2012; Yeh, 1983; Fang, 2003) have similarly expressed reservations regarding Hofstede's model, but even still, it remains a widely employed framework around the world.

In the field of higher education, Hofstede's dimensions have been employed in numerous studies, making the theory more dominant in this research. For instance, Sriwindono and Yahya (2012) utilised these cultural dimensions to explore how they affected the acceptance of Management Information Systems (MISs) among Indonesian academics. Cheung and Chan (2009) further used the model to understand how cultural dimensions aligned with processes for meeting the demands of competitive economies through education. Additionally, Viberg and Grönlund (2013) employed it to investigate students' attitudes regarding the use of mobile technology for learning second and foreign languages. Popov *et al.* (2012), meanwhile, employed this model to identify challenges faced by multi-cultural students in undertaking group work, as well as their perceptions of the challenges.

4.3.2 Schein's Model of Culture (1984)

According to Schein (1984), organisational culture can be categorised into three major titles functioning as a guiding model for companies to understand the behaviour of individuals working within those companies. Despite being presented in the 1980s, Schein's model of organisational culture has been used globally by authors and scholars. This model was developed to evaluate the formation of culture within an organisation. Edgar Schein believed that culture could not be developed within one day, instead arguing that culture development represents a continuous process undertaken after numerous activities have been conducted by the individual, including problem-solving, customer and supplier relationship procedures, and stakeholder engagement (Schein, 1984).

The knowledge and experience gained from these activities helps form the shape of the workplace, described as its culture (Schein, 2013). Therefore, culture is broadly defined as a pattern of

essential shared assumptions as a group learns to solve its problems of adaptation externally and integration internally in ways that have worked well enough to be considered valid, and thus sufficient to be passed on to new employees as the proper way to think, feel, and perceive relative to those problems (Schein, 1992). Figure 4-2 below illustrates this model:

Figure 4-2: The levels of culture and their interaction. Source: Schein (1984)



According to Schein's model, culture possesses three levels: The first category—artefacts and symbols—includes any tangible or physical elements in an organisation that can be easily seen and experienced. The elements could include company or product logos, architecture, processes, and even corporate clothing. These elements are also observable to external parties, not just employees. The second category—espoused values—comprises strategies, principles, and rules of conduct for the organisation. Leaders' and managers' behaviour in relation to organisation goals and strategy depict espoused values practiced within an organisation. The third category—basic assumptions—comprises strongly embedded beliefs within the organisation that are considered a

core element of culture. These three categories can function as guiding principles for understanding organisational culture.

According to Schein (2013), leadership represents a key source of values and beliefs, and as such, the most important issue that leaders must consider involves understanding deeper culture levels and dealing with any anxiety that emanates from challenging assumptions. Although Schein's model is well recognised in the scientific community, however, several reservations have been advanced against it. De Jonge (2006), for instance, argued that, since the model is based solely on surveys and interviews with employees, these attributes cannot be considered a major element of culture. Furthermore, De Jonge (2006) claimed that this level represents the basic dynamic element of organisational culture that is often overlooked by organisational behaviourists.

One of the justifications supporting the adoption of Schein's organisation culture model in the research stems from the fact that it has been adopted by other researchers to examine how organisational culture influences strategy implementation. For instance, Hogan and Coote (2014) employed the framework to conceptualise innovation processes in law organisations while Dauber *et al.* (2012) employed it to develop a configuration model of culture, and thus to explore dynamic relations between organisational strategy and culture and external environmental factors. Ostroff *et al.* (2012), meanwhile, also used the model to understand the interaction between organisational cultures and external environmental factors. In the higher education sector, Taye, Sang, and Muthanna (2019) further utilised Schein's model of organisational culture to develop a conceptual framework to study organisational culture's influence on performance.

A second justification for adopting Schein's framework of organisation culture stems from the fact that Schein provides the most popular and accepted definition of culture adopted in the context of organisations (Lacatus, 2013). As such, Schein's theory emphasises the fact that leaders play a role in shaping and reinforcing an organisation's culture. This research subsequently employs this theory to understand how leaders shape and reinforce the culture of higher education organisations.

4.4 Cyberspace Culture Frameworks

Primarily, the focus of cyberspace security has largely concerned technical security measures, leaving human behaviour regarding information security unstudied in the past. In response, a need was identified for addressing this 'human factor'. In previous years, authors have accordingly explored how individuals perceive information security and why they are encouraged to adhere to

security practices. Moreover, a strong relation exists between organisational culture and behaviour, as culture influences behaviour. Similarly, it can be argued that cyberspace culture is formed on the basis of employees' cyber behaviour, with cyber behaviour referring to employees' interaction with information assets and security.

While diverse cyberspace culture frameworks are available in the literature, only a few are elaborated in this research based on their suitability and relevance to the research questions. The selected frameworks largely focus on the higher-level adoption of secure practices rather than adoption at only an organisation level. Beyond this, the frameworks are also employed to understand the formation and dissemination of information security process at a higher level, where work culture, as well as the culture of a geographic location and social constructs, are studied.

4.4.1 Four-Quadrant Framework Evaluation

The current research analyses and compares the prominent works in this field using a four-quadrant framework, wherein all the research and resolution aspects of these literature studies are observed. The adopted approach resembles the one utilised in Section 3.2, where the reviewed literature was analysed using this framework. Tables 3-1 and 3-2 discuss the evaluated literature. The comparison and analysis of these four selected frameworks can be developed using the four-quadrant framework as a literature evaluation tool. The framework is primarily employed to categorise the academic and industry-based contributions to the literature. The framework categorises research according to the research aim (visionary or implementation) and the final output, categorised as either descriptive or prescriptive (Althonayan, 2003).

Research classified as visionary focusses on an idea, vision, or proposal in literature as opposed to the dynamics and implementation process. Implementation research, meanwhile, focusses on practical recommendations for a particular procedure. An interesting fact to consider is that such studies can be either descriptive or prescriptive in their outcome (Althonayan, 2003). A better way to view the comparisons in the four-quadrant framework would be through a matrix. Such a matrix can sum up the four main categories as follows:

- I) Visionary and descriptive;
- II) Visionary and prescriptive;
- III) Implementation and descriptive; and
- IV) Implementation and prescriptive.

	Research Philosophy				
	Descriptive	Visionary	Implementational		
Research Outcomes		Quadrant I	Quadrant III		
		 Describes information security culture in workplaces. A sort of conceptual framework can be proposed. Implementation may not have been discussed. Almost all supporting research is based on description. 	 Issues related to information security culture will be discussed. A diagnostic framework may be discussed and detailed. Guidelines related to implementation are discussed. Supporting research and examples are empirical. 		
	Prescriptive	Quadrant II1) Possible integration within an organisation is discussed.2) Theoretical framework detailing the culture of information systems may be mentioned.3) The research will give at least the most basic version of the implementation.	Quadrant IV1) The integration of information security culture within all major areas of the organisation is discussed.2) Steps, challenges, and practical application of the integration will be discussed.3) The advantages of the execution as per the		
			empirical data will be discussed.		

Table 4-1: Literature evaluation framework matrix.

Source: Adopted from Althonayan (2003)

This framework is utilised to comprehensively classify literature by determining each study's contributions. The categorisation of academic and industry research is performed in order to explicate the present literature gap, which will significantly influence research.

4.4.2 ISF Framework

This framework is relevant to the current research, as it was developed in the context of Saudi Arabia, located in the same geographical region as Qatar. As such, similarities in landscape and can be interlinked, providing a detailed evolution of major differences in the organisational setting. According to Alnatheer and Nelson (2009), information security culture and practice can be influenced by numerous factors, such as those ranging from information security awareness and training programmes, policy, risk analysis and compliance with information security management standardisation, top support from management, and organisational culture.

The broad factors are grouped into three major themes: corporate citizenship, comprising awareness and training programmes on information security; legal and regulatory environment, made up of standardisation management and best practices; and information policies on information security and corporate governance, consisting of support for information security management from top management, compliance with information security, and risk analysis. Similarly, the framework adds a cultural theme that stipulates a culture that is both national and organisational and which strongly influences security-culture adoption. Nonetheless, Alnatheer and Nelson (2009) highlights that, despite the improved economic development observed over the years in Saudi Arabia, its information security culture in workplaces remains in its infancy. As such, it usually includes measures of culture and ethics introduced so as to enhance an organisation's security-related behaviour and usually said to be an organisation's subculture.

The framework also advocates aligning corporate governance, legal and regulatory environment, and corporate citizenship in any given country. Furthermore, it also recommends aligning these aspects to the national culture, forming the primary foundation of organisational culture. Additionally, further development of security culture is equally important for developing an information management culture in any country. Figure 4-3 below presents this in detail, where it can be observed that national culture represents an important aspect in this regard. However, the missing element of organisational behaviour still needs to be investigated, as it can differ from one organisation to another.



Figure 4-2: ISF framework. Source: Alnatheer and Nelson (2009)

However, the research by Alnatheer and Nelson (2009) stands out in that the authors focussed on the information security culture in Saudi Arabia. Moreover, Western standards remain in a stage of infancy where companies' information security protocols are concerned. The authors determined that Saudi Arabia's social, economic, and cultural environment deeply influences organisations' information security culture. Moreover, it was found that employees largely possessed an extremely lax attitude regarding information security. Therefore, this framework is visionary and descriptive, as it proposes ideal ways in which information security can be improved and describes the various conditions for practicing information security in Saudi culture. Subsequently, the framework falls under quadrant I due to its descriptive nature.

4.4.3 ISCF Framework

One noteworthy aspect is that organisational behaviour and information security components must be aligned together in order to develop an information security cultural framework. The framework comprises three interlinked levels offering a major source of cultural formation. The process begins with information security components conducted in the organisation. The components are conceptualised as factors that influence behaviours towards securing information in an organisation when different components are executed, such as regular interactions between employees depicting certain information security behaviour. The regular interaction establishes a certain culture, thereafter considered the information security culture (Da Veiga, 2008). Components could include a security policy (Robbins, 1998) serving as a guideline for completing assigned tasks, and the organisation can also develop a certain strategy to implement the policy that later develops certain assumptions and values. Therefore, it can be understood that information security cultures represent a combination of artefacts, values, and assumptions developed from information security behaviour. As elaborated by Da Veiga (2008), on the organisational tier, training sessions on information security policy can be identified as artefacts resulting from the policy component.

Moreover, Veiga and Eloff (2014) provided organisations with the rationale for the need to be aware of the process through which culture would eventually determine the level of risks posed to information assets. Similarly, with the understanding of the prescription of employee behaviour, this article offers guidelines for developing the most successful strategies for managing employee behaviour and the development of efficient organisations promoting the elimination or minimisation of possible internal and external sources of threats to the company's information security system. In the end, however, it is imperative to note that the framework instrument for information security culture, as developed by various organisations, should always consider several factors, such as organisational culture and employee behaviour.

Da Veiga and Eloff (2010) utilised empirical studies to validate the proposed security framework. The primary goal was to teach the general human resources of an organisation to actively and consciously engage in information security activities so as to minimise information breach. To this end, the authors conducted implementation and prescriptive research targeting the organisations' need for a complete framework that considered the lack of a security-conscious culture. Based on its prescriptive and implementation status, the framework falls under quadrant IV. Figure 4-4 below summarises the ISCF framework:



Figure 4-3: ISCF framework. Source: Da Veiga and Elof (2010)

4.4.4 Information Security Culture Framework

The research further explored information security governance and diverse cultural factors influencing information security management. Figure 4-5 summarises the information security culture framework:



Figure 4-4: Information security culture framework.

Source: Shabbaan (2014)

Shabbaan (2014) explored information security governance by incorporating ISO/IEC 27002 best practices in information security management. The author argued that the practices could help an organisation enhance its governance practices and provide a benchmark for current practices. Shabbaan (2014) also presented descriptive research exploring how culture influenced information security management. Similarly, an implementational framework for improving information security management in non-profit organisations was also developed.

A breakthrough was achieved due to the framework also using ISO/IEC 27002 best practices in information security management in an effort to obtain required technical guidance in this research. The research framework emphasises priorities and implementation steps of information security and enhancing governance practices in information security among developing countries. In identifying these steps, the research further helped facilitate adopting new technologies in organisations. While the framework focusses specifically on Zanzibar, the research effectively drew its findings and conclusions from evaluating various socio-cultural dimensions in significantly socio-technical contexts.

In this sense, there is a need to consider priorities such as cultural, legal, device, and ethical dimensions of the society before enhancing approaches for implementing strategies that would be crucial for supporting the integrations of information security and information security management within the society. At the same time, the author highlighted implementational practices by providing knowledge on possible steps, changes, and transformations in information security and information security management utilisation in developing nations such as Zanzibar. As a result, the framework falls under quadrant III due to its descriptive and implementation aspects. Additionally, the identification of cultural issues influencing the adoption of information security and information security management should be addressed as a means of enhancing the adoption of the most imperative frameworks that would also offer opportunities for future research and evaluations of the manner in which Zanzibar and other developing nations could develop enhanced methods of information security governance. Beyond this, such a move would also help ensure profitability and the achievement of sustainable competitive advantages among organisations operating in such regions, as attested by Bne (2013).

4.4.5 Information Security Framework – STOPE

This framework seeks to clarify the association between diverse factors linked to the information security culture in order to serve as a guideline for professionals and managers in the implementation process. Al-Hogail (2015) proposed such a framework and discussed various components of human factors, such as training, responsibility, society, management, and regulations. The framework proposes to reinforce information security culture by validating from experts the practices that can be employed to develop a culture that protects the information assets. The proposed framework also integrates principles of change management that direct the usage and installation of an information security culture. The framework is validated by allowing experts to offer their perceptions regarding the completeness and correctness of the framework and all activities. The framework can also help companies build an operational culture of information security to safeguard business assets. Furthermore, in this visionary and prescriptive framework, Al-Hogail sought to describe the implementation of strategies in IT. Specifically, these strategies concern the importance of design and the validation of information security culture. Within this cultural perspective, the most imperative issue relates to the implications of employees' perceptions, as well as security behaviour within the organisational culture.

However, contrary to the works of Alnatheer and Nelson (2009), this article emphasised the need for developing an organisational culture that effectively reduces information threats perpetrated by insiders within the company. While developing various implementation strategies, the article focussed on the need for a comprehensive information security culture framework structured by considering people, technology, environment, strategy, and organisation. Combining these elements brings forth the realisation of STOPE as a means of forming a successful information security framework that adequately addresses various information security issues while also maximising not only profitability, but also overall sustainability of competitive advantage. In focussing on implementation strategies for information security culture design and validation, the author conducted evidence-based research utilising expert responses and survey outcomes. This research approach produces the understanding of a prescriptive study, as evaluated by Blessing and Amaresh (2009). As a result, the framework falls under quadrant II. Figure 4-6 below summarises the STOPE framework:

Issues:	Scope:					Development	
Human Behavior Diamond	S	т	ο	Р	E	<u>Tool:</u> Change Management	
	-						
Preparedness	Prepare employees to behave securely through training, awareness, knowledge acquisition, and change in perception.					Focus groups	
		Change agents					
					Motivation		
Responsibility	Ensure employees are behaving securely through monitoring and control, reward and deterrence, and applicability					Milestones and measures	
	Ensure management support by showing management commitment,				Management support		
Management	effective communication and interaction, and facilitation of resources.					Resources	
						Communication	
Society and Regulations	Consider external factors such as national culture, ethical conduct, government initiatives, and legal and regulations system				Culture analysis		
Outcome: behavior (artifacts), values, assumptions and knowledge to enhance information security							
Older State	TIME New State						

Figure 4-5: Information security framework – STOPE.

Source: Al-Hogail (2015)

In summary, the four cyberspace culture frameworks are assigned to the four-quadrant matrix, as illustrated in Table 4-2 below:

	Research Philosophy						
Outcomes		Visionary	Implementational				
	Descriptive	Quadrant I	Quadrant III				
		ISF Framework	Information Security Culture				
		(Alnatheer and Nelson, 2009)	Framework (Shabbaan, 2014)				
Research	ve	Quadrant II	Quadrant IV				
	cripti	Information Security framework –	ISCF				
	Preso	STOPE (Al-Hogail, 2015)	(Da Veiga and Elof, 2010)				

Table 4-2: Classification of cyberspace culture frameworks.

Source: Adopted from Althonayan (2003)

In this section, current information security frameworks were reviewed so as to guide the development of this research's conceptual framework. The following section examines the various gaps present in the reviewed frameworks.

4.5 Gaps in Existing Frameworks

Across the different frameworks, one observed confluence was that they all focus on teaching and proposing ways in which the human factor of information security can be dealt with. Most authors dealt with the technical aspects of information security, but less work was conducted on cyberspace culture. The relationship between company employees and cyberspace culture practices were evaluated in all these studies, however, and it was ultimately determined that employee awareness and training were vital for minimising security threats (Verizon, 2009). Most of the organisations lacked a satisfactory information security culture, and as such, an average of 2–3% of potential business was lost with clients who valued information security, while leakage of ideas and confidential information also affected the organisation's profits, along with its integrity (Mcllwrath, 2016).

However, a glaring research gap was identified across these four frameworks in that, while instructions and ideas for implementing these frameworks were provided in the studies, the authors failed to address how a cyberspace culture framework can be maintained following implementation. Moreover, these studies did not emphasise leadership's role in the cyberspace culture. The reviewed frameworks in Section 4.4 should also focus on the leadership roles, as they possess considerable significance for the organisations involved. The leadership decisions can help organisations avoid any future potential loss of business, for instance. Beyond this, leadership also plays a critical role in supporting governance regarding the implication of cyberspace culture. Moreover, leadership should also support and focus on building the cybersecurity culture within the organisation. The development of such a cyberspace culture can help the leadership to develop a greater business environment in the modern era. Human mistakes will always arise, and malicious human sentiment will never be completely eliminated, so it is necessary to develop possible measures to keep cyberspace culture safe once it has been implemented.

Elements	Alnatheer and Nelson's Framework	Da Veiga and Elof's Framework	Shabbaan's Framework	Al-Hogail's Framework
Leadership	Х	Х	Х	
Policy		Х	Х	
Technical competency		Х	Х	
Government regulation	Х		Х	X
Organisation culture	Х	Х	Х	Х
Management / governance	Х	Х		Х
Security culture	X	X	X	
Change management		Х		X

Table 4-3: Identification of the framewor	rk gap.	
---	---------	--

Source: The Researcher

Table 4-3 above displays the research gap according to the selected criteria. Based on the evaluation, the first, second, and third frameworks mentioned the aspect of security culture (Alnatheer and Nelson, 2009; Shabbaan, 2014; Da Veiga and Elof, 2010). However, the frameworks failed to highlight the development of cyberspace culture, thereby motivating the development of the proposed framework. Regardless, the various authors identified some of the components necessary for fostering cyberspace culture, such as leadership and government regulation.

4.6 Cyberspace Culture Framework

In this section, a framework is proposed for fostering cyberspace culture in line with the research gaps identified in Sub-Section 3.2.2 in the previous chapter. The sub-section begins by briefly summarising said research gaps and providing a rationale for variables that should be considered in the framework. Thereafter, framework elements are discussed and summarised in the second sub-section.

4.6.1 Derivation of the Cyberspace Culture Framework

In Sub-Section 3.2.2, several research gaps were identified in regards to aligning cyberspace culture with the general vision for higher education. A review of the vision revealed that it sought to achieve three key objectives: boost teaching initiatives by making a shift from traditional teacher-centred approaches to more self-directed and content-centred approaches; enhancing access to learning in cyberspace; and promoting the scientific community by facilitating collaboration with other authors in different parts of the globe and converging scientific knowledge. Based on these goals, the Researcher identified current research gaps in aligning cyberspace culture to the country's established vision.

By considering cyberspace as a learning tool to augment problem solving, enhance engagement, and provide authentic learning experiences, the Researcher observed a gap in enabling cyberspace adoption by learners and educators to attain the set objectives. Since cyberspace is viewed as a tool to augment human capabilities, it is necessary to outline best practices to guide the adoption of technology-based tools. The goal of the vision is to advance higher education institutions such that learning becomes learner-centred and teaching practices are boosted to world-class levels. However, without guidelines and best practices detailing how to adopt such technologies, this

vision cannot be attained. Therefore, the role of leadership and regulation emerges as a pivotal aspect in the proposed cyberspace culture framework in Figure 4-7.

Furthermore, by considering cyberspace as a cultural space, a research gap was identified in guiding how learning occurs in cyberspace. Based on the theory of situated learning (Lave and Wenger, 1991), this thesis argues that, since learning is unintentional and occurs within authentic activities, it is necessary to leverage the situated and personalised nature of cyberspace. As such, this raises a concern regarding how learners would be guided in attaining expertise through a wide array of devices and platforms without veering from the goals established in the higher education vision. Therefore, the aspect of espoused values and assumptions emerges in the proposed framework. The Researcher argues that learners should not perceive learning as limited solely to certain devices, such as computers. Rather, smartphones and mobile devices that deliver audio or image-based content should also be considered. As such, this implies that assumptions such as 'I believe that learning through a mobile device will be beneficial' should be promoted through the creation of policies. Additionally, this also implies that technical training is further required to facilitate adopting modern tech-based tools.

A third research gap was also identified in outlining how learning was delivered in cyberspace. Though it was argued that reliance on corporate gatekeepers would stifle creativity and freedom of self-expression, it would nonetheless take time to develop custom learning platforms and to market them in higher education institutions. As such, best practices to facilitate the selection of educational technology vendors are recommended. Subsequently, this indicates a need for leadership and responsibility so as to guide procurement processes and ensure the tools deliver as required. Frameworks guiding change management and behaviour change are also identified as a pivotal point in enhancing cyberspace culture.

The fourth research gap concerns the creation of content through authorship in cyberspace. Cyberspace technologies are being accessed not only by students to learn different subjects, but also by teachers to develop instructional content. As such, teachers who produced sets of texts online created identities similar to authors of academic and literary texts. The identified gap thus pertains to the identities of teachers who utilise social networks and websites to complement their teaching practices and, subsequently, what happened to them as they employed digital technologies to go about their lives and their teaching. In order to ensure that teachers maintain their identities in cyberspace, a need for responsibility, leadership, and regulation arises.

In summary, the review of previous theoretical frameworks in Section 4.4 revealed that the human aspect has largely been ignored by most authors, who have focussed more on the technical aspects of culture. The current research seeks to fill this gap by highlighting important aspects that should be considered in managing the human aspect of culture development. To this end, the following section accordingly describes the components of the proposed cyberspace culture framework, a diagrammatic representation of which is presented in Figure 4-7.

4.7 Elements of the Cyberspace Culture Framework

The analysis of current research gaps in the previous section highlighted several pivotal aspects that directly influence the successful alignment of cyberspace culture and higher education. These aspects are discussed further in this section. Before this, though, it is instructive to highlight that elements describe the particular components that make up the framework, (e.g. internal elements such as leadership and management strategies). However, factors also include the individual issues associated with a particular element; for instance, under the internal element leadership and management strategies, accountability represents a factor.

To begin with, the internal elements of the cyberspace culture framework are elaborated further. In describing how leadership and management strategies are formulated, insights from the ISCF framework (Da Veiga and Eloff, 2010) are adopted. Likewise, inputs from the STOPE framework (Al-Hogail, 2015) are further utilised in developing the security policy aspect while the information security culture framework (Shabbaan, 2014) guides the conceptualisation of best security practices and technical competency. In the second sub-section, the external elements of the cyberspace culture framework are delineated, whereby the government's role in influencing policy, regulations, leadership, and technical competency is examined. Further analysis of the QNV 2030's influence on the strategy-formulation process is also undertaken. The third sub-section discusses insights from theoretical perspectives regarding culture formation (Schein, 1984; Hofstede, 1987). Examining the different theories highlights important dimensions evaluated in the research, such as the bi-directional relationship between levels and dimensions of culture. In the fourth sub-section, the contribution of the cyberspace culture framework is elaborated further, highlighting new insights regarding cyberspace culture creation.

4.7.1 Internal Elements of the Cyberspace Culture Framework

This sub-section discusses the various internal elements of the cyberspace culture framework. In essence, internal elements describe aspects over which higher learning organisations can exert control. Such aspects include strategies adopted in management and leadership, security policies, and technical competency in security, as well as best practices implemented.

4.7.1.1 Leadership and Management Strategies

Leadership and management strategies adopted in higher education institutions are also important for guiding behaviour change and the actions of learners and educators in cyberspace, as well as within organisations. Accordingly, the ISCF framework (Da Veiga and Eloff, 2010) identified a need for responsible leaders who guide the daily interactions of employees, thus leading to culture formation.

4.7.1.1.1 Improving Accountability

According to Leonard et al. (2016), various aspects of leadership, such as personal qualities, administrative actions, and behaviours influence culture formation and maintenance within a given organisation. By definition, accountability describes the quality or state of being accountable, leading to an obligation or willingness to accept responsibility for one's actions. Ghanem and Castelli (2019) argued that accountability in leadership is crucial for protecting business and organisation ethics, as accountable leaders facilitate the development of ethical organisational perspectives. Their study accordingly identified accountability as a significant predictor of ethical leadership. Likewise, moral competence was identified as an important moderating factor, as increased moral competence enhanced positive effects of ethical and accountable leadership. Leaders are tasked with the responsibility of selecting appropriate technologies to facilitate authorship for teachers and educators as well as learning technologies for the students. Without proper strategies in place, novice educators and learners will experience significant challenges in enhancing their technological competence. The analysis of these aspects highlights that leadership plays an important role in guiding the QNV 2030's implementation in higher education, and thus illustrates a need to ensure accountability. Accountable leadership is accordingly associated with the capacity of higher education leaders to implement effective technologies that facilitate attaining the 2030 higher education vision and the cultivation of a positive cyberspace culture. Leaders should be accountable to the outcome of the strategies they propose and implement within higher education institutions.

4.7.1.1.2 Risk Management

Strategy formulated by leadership and management plays an important role in boosting individual learning institutions' competitiveness. Leaders in particular play an important role in managing the risks that may be faced by higher education institutions by fostering an appropriate culture within their organisations (Fleming, 2004). Furthermore, leaders are also tasked with selecting and implementing strategies to facilitate attaining different objectives in higher education institutions while also ensuring conflicts are resolved and accountability standards maintained. In this regard, the nature of adopted strategies are associated with risks as they directly influence the progress of developing cyberspace culture in the organisations. In a different study, Tsai (2011) also reported that an organisation's culture directly influences the adopted leadership behaviour and, subsequently, employees' job satisfaction, attitudes, and behaviour. Therefore, the risk management strategies arise in two ways: The first arises because leaders determine how various strategic objectives are achieved through fostering appropriate culture. As a result, risks associated with accomplishing the strategies emerge, as they may fail to facilitate attaining strategic objectives. The argument advanced in this regard is that the formulation and implementation of ineffective risk management strategies by organisational leaders introduces risks to the organisation in the event that the strategies fail to address targeted problems. This assertion is supported by Fadun (2013), who argued that effective risk management helped prevent the failure of risk management strategies adopted in a business firm. Second, as reported by Haughey (2020), leaders play a role in developing cybersecurity resilience frameworks that enable their organisations to detect and contain different types of cyber threats that may be launched against them. To minimise cyber threats and attacks, the researcher adds that there remains a need for management and leadership to not only employ competent chief information security officers, but also to nurture a culture of cyber resilience and create formal policies for cyber security (Haughey, 2020). Therefore, this indicates that the nature of management strategies implemented to enforce a given culture will likewise directly influence the various factors.

4.7.1.2 Security Policy

In addition to responsible leadership, security policy also emerged as an important element guiding the actions of different employees in higher education institutions.
4.7.1.2.1 Security Awareness

The proposed cyberspace culture framework adopts the preparedness and responsibility aspects of the STOPE framework (Al-Hogail, 2015). With preparedness, STOPE highlights the importance of preparing employees to behave securely through awareness, training, and perception changes. Similarly, the responsibility aspect outlines the value of ensuring employees behave securely through monitoring and control, as well as through deterrence and reward. The proposed framework advocates creating policies to ensure users' information-security behaviour by outlining acceptable and unacceptable uses of cyberspace. As users are guided to use cyberspace in an acceptable manner, further emphasising how internal security policies influence user security knowledge and awareness.

According to Ng (2010), security awareness training represents an effective approach for making users aware of the need for securing their environments. In this way, the author revealed that increased security awareness and training can improve an organisation's information security performance. Directly, this implies that security training positively influences employee knowledge and skill in regards to security management. An additional view by Roper, Fischer, and Grau (2006) added that employee performance is affected by three types of influences: environmental, skill and knowledge, and motivations and attitudes. Therefore, assessing how security training influences employee knowledge and skills, as well as attitudes and motivations, is crucial. An argument by Adomi (2011) further posited that information security best practices in higher education institutions, such as mandatory training, should be taken seriously, as they influence all user levels in higher education institutions. As such, the implemented policies should be critically evaluated to ensure increased efficiency in providing core education and community services.

4.7.1.2.2 Continuous Learning

The aspect of continuous learning was also indirectly emphasised in that, as security policies outline the acceptable uses of cyberspace, users are guided to continually improve on this knowledge, thereby enhancing their knowledge and awareness. With policies, standards, and best practices in place, a greater emphasis is placed on ensuring responsible use of cyberspace so as to facilitate learning and authorship. Furthermore, higher levels of awareness ensure higher education institutions are protected from cybercrime activities.

A different argument by Kanwar, Balasubramanian, and Carr (2019) claimed that continuous or lifelong learning is characterised by continual upskilling and reskilling in order to keep up with unexpected changes in industry and advances in technology. As such, this was viewed as an important component of sustainable development. In the context of security policies developed in higher education institutions, lifelong or continuous learning can thus be understood as the continual upskilling and reskilling of employees with appropriate skills matching the requirements of advanced technology. Security policies establish important knowledge facets in which employees must be competent so as to ensure they secure the resources and data within the institutions. In this regard, employee effectiveness in continual learning influences the achievement of the 2030 higher education vision and the overall cultivation of cyberspace culture. However, the inverse is also true in that, where employees have not fully developed continuous learning practices, this will likely to lead to vulnerabilities within the higher education institutions that can be exploited by malicious actors, and thus adversely influence attaining the QNV 2030 in higher education. For instance, where security policies require employees to constantly change their passwords, some employees may fail to continually learn new approaches to enhance their passwords, thus making it easier for attackers to steal data and resources in the institutions. Directly, this leads to the failure to achieve the higher education 2030 vision or the cultivation of a positive cyberspace culture.

4.7.1.2.3 Attitudes and Behaviour

According to Adomi (2011), information security policies determine the efficiency in providing core education and community services in higher education institutions. A further finding by Lim *et al.* (2009) identified a relationship between organisation culture and the information security culture developed in an organisation. As such, it is necessary to provide time to positively influence employee beliefs, values, and behaviour. The implication is that the adopted security policies influence employee knowledge and security awareness as well as the attitudes and behaviour they develop towards fostering cyberspace culture. Continuing, the argument by Adomi (2011), which cited that policies influence employee efficiency in providing core services, further implies that they influence the level of security awareness and security competency among employees. As such, a need arises to evaluate the two aspects in the context of cyberspace culture. Furthermore, since policies affect employees' knowledge and skill level, this additionally influences their attitudes and behaviour towards cyberspace culture.

The argument advanced in this context is that the formulated and adopted security policies influence the organisational culture of higher education institutions. For instance, policies ensuring resource protection may require employees to restrict their use of computers solely to designated work roles or to avoid bringing their own devices into the workplace. Such policies influence the employees' attitudes and behaviour, which may be either positive or negative. In the event that negative attitudes are developed towards security policies, this adversely influences the attainment of the higher education QNV 2030, as the employees fail to support them. This further leads to the inability to foster a positive cyberspace culture as employees develop negative attitudes towards the policies. In a different study, Chedid *et al.* (2019) additionally reported that some positive behaviours, such as collaboration and knowledge-sharing in higher education institutions, are influenced by positive attitudes such as trust. As such, this suggests that security policies may foster negative attitudes among employees, in turn leading to negative behaviours that may fail to support the attainment of the QNV 2030 in higher education and the achievement of a positive cyberspace culture.

4.7.1.3 Security Technical Competency and Best Practices

Security technical competency and best practices are also important in guiding learning and authorship in cyberspace.

4.7.1.3.1 Skills and Knowledge

The information security culture framework (Shabbaan, 2014) further stipulates a need for technical competency and, as such, recommends benchmarking, training, and implementation of information security controls within organisations. In the proposed cyberspace culture framework (Figure 4-7), two technical aspects are emphasised: 1) security and training and 2) awareness creation. The Researcher's motivation for focussing on these two aspects stems from the fact that learners who might be well versed with traditional learning approaches and textbook use in physical classrooms might experience a challenge when shifting to the virtual landscape. Similarly, educators familiar with traditional teaching and authorship tools might also experience similar adoption challenges. As such, training should be offered to ensure the actors can employ digital tools to effectively conduct learning, teaching, and authorship.

4.7.2 External Elements of the Cyberspace Culture Framework

External elements of the cyberspace culture framework (Figure 4-7) refer to the roles undertaken by the government, including the institution of appropriate national leadership and governance to develop strategies and manage the higher education sector; external technical assistance to facilitate technology adoption; development of appropriate policies and best practices; providing government regulation to encourage growth of the culture; and reliance on theoretical perspectives, particularly levels and dimensions of culture.

4.7.2.1 The QNV 2030 Directive

The QNV 2030 directive influences government leadership, regulation, and information security policy. This section examines each of these aspects and details how they influence attaining the higher education QNV 2030.

4.7.2.1.1 Leadership

According to Walsh (2019), effective leaders empower their followers to take more risks, and thus to achieve greater goals. Leadership describes a process of social influence whereby an individual who is influential or more prototypical than other group members in a given context empowers other members to either attain particular goals or to solve particular problems. Therefore, in this regard, attaining the higher education QNV 2030 is heavily influenced by the effectiveness of the steering committee or leadership that governs it. As such, this section assesses how leadership strategies influence the attainment of the higher education QNV 2030. In particular, this section expounds on how economic factors and external pressure shape the different leadership strategies and, subsequently, influence the achievement of the higher education vision.

4.7.2.1.1.1 Economic Factors

The QNV 2030 directive focusses on enhancing the quality of higher education by encouraging the development of independent institutions to offer training for the population. However, as revealed by Marshall (2018), higher education institutions remain challenged by increased costs of acquiring and maintaining technology, as well as in running operations as expected without sacrificing the institutions' soul and affecting the quality of delivered services. The implication is that the QNV 2030's directive to develop independent institutions is likely to influence the development of alternative sources of income, such as alliances with organisations, technology transfer, distance education, and provision of auxiliary services. Strategies such as adopting distance education and technology transfer influence both the achievement of the higher education QNV 2030 (setup of independent higher education institutions) and fostering cyberspace culture through introducing technology-mediated learning. The argument advanced is that some strategies implemented in higher education institutions are influenced by economic factors, such as

partnership with tech organisations to supply equipment at an affordable price level due to financial constraints. In the same vein, the institutions may decide to adopt distance learning rather than establish physical classrooms. While these strategies will positively influence the attainment of the QNV 2030 in higher education (increasing the number of financially viable institutions), they may also affect the cultivation of cyberspace culture due to interaction with digital technologies.

4.7.2.1.1.2 External Pressure

Likewise, the QNV 2030 directive regarding the policies implemented in higher education may lead to external pressure that adversely affects the higher education institutions' flexibility. In this context, external pressure refers to either leadership or managerial pressure by government regulation leading to reduced flexibility. The government is considered a major stakeholder in higher education by offering funding and facilitating partnerships with foreign investors (Cantwell, Coate and King, 2018). In other instances, political interests by the government may influence the policies and strategies that are implemented, such as the institution of leadership. Therefore, assessing how government regulation influences the revenue generation activities of higher education institutions, such as the research programmes they offer or their income-generating activities, this further emphasises how government regulation influences economic aspects.

4.7.2.1.2 Information Security Policy and Regulation

The government's information security policy directive, as well as its imposed regulations, further influence the attainment of the higher education QNV 2030 by influencing three core aspects: compliance and advanced security, environmental uncertainty, and political factors, as discussed in the section below.

4.7.2.1.2.1 Compliance and Advanced Security

According to Ghazvini, Shukur, and Hood (2018), information security policies represent a major requirement in higher education institutions such as universities due to the large amount of computer power they possess. Similarly, given their open-access policy, this allows higher education institutions to provide access to their networking infrastructure to both internal staff and students, as well as to visitors and worldwide authors. According to Tømte *et al.* (2019), the digitalisation of higher education is highly important in modern society, as it enhances both learning and teaching processes. Additionally, higher education institutions are also highly

knowledge-intensive and rely on information communication technologies (ICTs) to store, process, and disseminate information. Based on such aspects, higher education institutions must comply with diverse security policies as directed by the government. Mumper *et al.* (2011) cited a college president who argued that the institution had to comply with security policies because the government required them to do so, and that failure to comply was tantamount to violation of the law:

'We have a gazillion people working on compliance. The government requires us to do it

.... But if we don't have them, we would be in violation of the law' (Mumper et al., 2011

p.132).

The implication of compliance is that the higher education institutions may be required to implement new technologies that positively or negatively influence attaining the higher education QNV 2030, as well as the cyberspace culture developed as a result. This assertion arises from the fact that information security policies address diverse advanced technological aspects such as data and application security, security incident management, and security auditing.

4.7.2.1.2.2 Environmental Uncertainty

Uncertainty can be understood as a state of doubt or uncertainty regarding what will happen in the future, such as receiving funding in the context of higher education. Kongolo (2019) reported that, in higher education institutions, managers often worry over changes to government policy, as well as other arrangements, such as compliance with the requirements of government organs. In the same vein, managers also have to deal with the uncertainty of their students, who may feel anxious or limited in regards to acquiring quality education in the institutions or obtaining work positions upon graduation (Skakni *et al.*, 2019). Furthermore, given the high and wide uptake of the Internet and advanced technology in higher education institutions, concerns of information security policies and regulation also lead to greater uncertainty in the institutions. One such case would be the institution of new leadership in the education ministry following a change of government. As new leadership is instituted in the government, changes are anticipated in regards to the management of higher education institutions.

Mok (2016) observed that, as leaders understand the important role played by higher education in their country's social and economic development, the government provides significant support to the institutions. However, as government leadership transitions from one political power to another, this generates uncertainty regarding the management environment of higher education, as

the institutions are left unsure as to whether the support by the previous regime will be continued in the new government. This assertion was supported by Pacheco, Yoong, and Lips (2020), who reported diverse challenges experienced in the transition of higher education management in terms of both leadership and acquisition of digital technology. Anticipated changes in both leadership and management styles of the newly instituted leaders are also poised to generate uncertainty regarding whether the higher education QNV 2030 is attained. In turn, this influences the cultivation of a positive cyberspace culture.

Second, it is also important to consider the uncertainty that arises from the rapid pace of technological development, which requires higher education institutions to adopt new operational processes, such as adopting modern technology in learning (Hartman, Townsend and Jackson, 2019). Studies have observed that higher education institutions tend to be slow in adopting innovation due to the associated risks and time commitments required to explore the new ideas and tools. Due to the pressure exerted by competition from other higher education institutions, some universities may be under pressure to adopt modern technology in order to meet the students' demands in the institutions (Hartman, Townsend and Jackson, 2019). However, this is challenged by uncertainty regarding whether educators in the institutions are positioned to utilise the technology or whether it is likely to be effective in attaining the targeted objectives. In other instances, government regulation may pressure the institutions to adopt modern technology, an aspect that also leads to the same effects. Whether caused by a change of political leadership or pressure to adopt modern technology in higher education institutions, the implication of environmental uncertainty is that it negatively influences the attainment of the higher education QNV 2030. Uncertainty regarding technology's effectiveness or the tenure of senior management also adversely affects the attainment of the QNV 2030 in higher education, as well as the cultivation of a positive cyberspace culture.

4.7.2.1.2.3 Political Factors

Political influences from the government can also affect the attainment of the higher education QNV 2030, particularly by influencing the appointment of governing boards. Adamu (2019) reported that, until recently, the university community in Ethiopia was not involved in either nominating or appointing their leaders, as this function was undertaken by the Ministry of Education. A similar observation was also made in Sweden, where at least four of the 11 members of the university governing council were appointed by the Ministry of Education while the

remaining members were elected by academic and administrative staff (Pinheiro *et al.*, 2019). As leaders play an integral role in directing team members in attaining goals and the organisation's overall vision, politically motivated strategies can eventually influence the attainment of the higher education QNV 2030. In turn, this subsequently affects the cultivation of the cyberspace culture.

Beyond appointing leaders, political influences from the government can also arise from the requirement to implement different policies, such as supporting technological adoption, which can likewise influence the attainment of the higher education vision. The argument advanced in this case is that, since the government represents a major stakeholder in higher education, such as by providing funding through loans or sponsoring different partnerships, it also plays an important role in influencing the decisions made in higher education, as well as the implemented strategies. The shortcoming of this aspect concerns the fact that the government may influence the implementation of politically motivated strategies in the institutions, such as procuring technological equipment from a given vendor. These strategies will subsequently influence the attainment of the QNV 2030 in higher education and further affect the cyberspace culture fostered within the period.

4.7.2.2 Technological Advancement

The advancement of technology can also affect attaining the higher education QNV 2030 by directly influencing two aspects: 1) learning and teaching and 2) the management of the institutions.

4.7.2.2.1 Learning and Teaching

As modern technology is adopted across nearly all sectors of the economy, this also leads to changes in how learning and teaching are undertaken in higher education. Technology-mediated collaborative learning, for instance, enables learners at different levels to solve problems and, in the process, to enhance their higher-level thinking skills and soft skills such as oral communication and social interaction (Esteves, Matias and Pereira, 2018). Additionally, through distance learning, learners can enjoy flexible learning processes, as studies can be undertaken regardless of their locations (Mahlangu, 2018). Two studies in particular emphasised technology's role in improving collaboration among learners, as well as in facilitating flexible distance learning. Meanwhile, other authors have highlighted technology's role in improving engagement levels among learners. For instance, Bond and Bedenlier (2019) revealed that student engagement in terms of cognitive,

affective, and behavioural aspects increased as students were provided the opportunity to voice their concerns in learning environments. In this context, technology enabled students to contribute to their learning environments and to offer feedback on different aspects, thus enhancing their engagement levels.

In regards to teaching and pedagogy, Oliveira *et al.* (2019) reported that technology is also enhancing current pedagogical approaches by facilitating aspects such as computer simulations, virtual labs, robots, and mobile devices. Allsop *et al.* (2020) shared similar findings, highlighting the use of video-conferencing to deliver anatomy teaching to medical students. The authors further revealed that technology-enhanced learning was invaluable for training students dispersed across multiple sites, as they were challenged in acquiring the training due to travel constraints. However, with video conferencing and the use of interactive screens, higher participation was observed among students.

Smart campus technology is also innovating learning within higher education by introducing smart education approaches, such as e-learning, virtual classrooms, and personalised learning (Sari, Ciptadi and Hardyanto, 2017). Likewise, Radio Frequency Identification (RFID) technologies are being leveraged to develop more interconnected campuses connected to outside entities in realtime. For instance, smart parking provides information regarding available spaces in parking lots, including when they are unavailable. Additionally, Internet of Things (IoT) technology is also transforming libraries into smart learning spaces, which are then subsequently converted into spaces of peer collaboration, social learning, and discourse (Chan and Chan, 2018). Smart libraries are equipped with modern technologies such as smart bookshelves that immediately update system records to ease stock take; automatically detect books on shelves, thus eliminating queues; and provide real-time tracking and monitoring of books. Nachandiya et al. (2018) further observed that intelligent campuses (i-campus) provide an interactive environment whereby diversified technologies such as cloud computing, mobile-based learning, mobile apps, and RFID support a wide range of functions, including learning, administration, and networking. Directly, this fosters the creation of environments wherein learners and teachers can interact with one another to enhance learning. Furthermore, pedagogy is also significantly improved, as teachers can utilise a wide range of teaching tools and equipment, such as information boards and data centres (Nachandiya et al., 2018).

The implication of increased technology use is that the advanced technologies are fast-tracking the attainment of the higher education QNV 2030 by innovating both the teaching and learning processes. However, increased reliance on these technologies also influences the fostered culture as the learners interact with the technology. Therefore, it is important to not only evaluate how the advanced technologies are improving the attainment of the QNV 2030 in higher education, but also how this influences the cultivation of a cyberspace culture.

4.7.2.2.2 Management of Higher Education

Beyond this, advanced technology is also influencing the management of higher education institutions, such as by enhancing decision making through advanced data-driven approaches (Kurilovas, 2018). As the institutions generate significant amount of data, technologies such as artificial intelligence and data mining become highly useful for generating insights from the resultant large datasets. Several authors have accordingly demonstrated the impact of emerging technologies such as artificial intelligence and predictive analytics in generating important insights for managing higher education institutions. Jones (2019), for instance, reported that tracking, aggregating, and analysing profiles of students, along with their digital and analogue behaviour captured in their information systems, could promote learning mechanisms. Akçapınar, Altun, and Aşkar (2019) also demonstrated the use of learning analytics in developing early-warning systems for students at risk of dropping out of their classes. Such insights are useful for developing effective policies to ensure students complete their courses on time.

Continuing, information systems are also important for enhancing management operations by facilitating the creation, storage, and dissemination of diverse types of information (Singh, 2018). The direct implication of relying on advanced analytics to facilitate decision making in higher education is that this enhances attaining the QNV 2030 in higher education, as better strategies can be formulated from the generated insights. For instance, strategies to ensure students remain engaged during the learning process produces highly skilled graduates. Conversely, however, the increased reliance on analytical technology by higher education institutions' management also leads to a cultivation of a cyberspace culture. Examining technology's role in the management of higher education is highly important in this research, and the primary data-collection methods (interviews and focus groups) address this by collecting participants' views towards the same. As this research targets collecting data from senior management in higher education, this enables

understanding their perceptions regarding technology's role in the management of higher education.

4.7.2.3 Vision 2030

The QNV 2030 regarding higher education outlines long-term outcomes anticipated in the education sector, such as the modernisation of learning and increased emphasis on science and technology.

4.7.2.3.1 Implementation Framework

The 2030 vision highlights a roadmap to facilitate sustainable development in higher education as well as strategies such as investing in world-class scientific research and first-rate education in order to unlock the potential of its people. Therefore, the strategies outlined in the vision influence the fostering of cyberspace culture, as they can either enhance or hinder its attainment. As the QNV 2030 seeks to transform the country so as to facilitate sustainable development and improve living standards, it provides a guideline that influences the definition of country outcomes on a long-term basis and outlines a framework to facilitate the creation and implementation of national policies and plans (Qatar National Vision, 2030, 2008). Directly, this suggests that the framework is poised to influence the attainment of the higher education vision. For instance, guidelines in the QNV 2030 can be adopted in the higher education vision. Subsequently, a need to assess the value of the proposed cyberspace culture framework (Figure 4-7) is emphasised.

4.7.3 Theoretical Perspectives

In addition to the four core elements of the framework, theoretical perspectives on culture formation by Schein (1984) and Hofstede (1987) also play an important role in the development of a positive cyberspace culture.

4.7.3.1 Processes, Attitudes, and Values

First, Schein (1984) categorised culture along three levels of interaction: i) symbols and artefacts, ii) espoused values representing rules of conduct for the organisation, and iii) basic assumptions describing strongly embedded beliefs within the organisation that are considered a core element of culture. According to Schein (1984), culture is fostered as the three levels interact with one another; basic assumptions influence espoused values, which further influence the artefacts employed in organisations. Such perspectives are important for guiding leaders at different capacities to foster a positive culture in higher education institutions. Leaders guide their organisational members towards adopting positive behaviour and, eventually, fostering a positive

culture. The analysis of the different levels of culture by Schein (1984) highlights that culture manifests itself through visible processes, attitudes and behaviour, and values and beliefs. The different aspects are subsequently investigated in the research interviews in order to identify how they influence fostering cyberspace culture.

4.7.3.2 Cultural Dimensions

Hofstede's (1987) framework outlines five dimensions of culture: power distance, masculinity versus femininity, uncertainty avoidance, long-term orientation, and individualism versus collectivism. Hofstede's cultural model is important for identifying the type of culture prevalent among the population so as to guide leaders to leverage the population's cultural dimensions in fostering cyberspace culture. As Hofstede's cultural model outlines various dimensions of culture, understanding their influence on fostering cyberspace culture is emphasised in the research. For instance, where femininity is ranked lower than masculinity, strategies must be formulated to ensure equity in higher education institutions. Similarly, aspects such as power distance and the influence of individualism versus collectivism will significantly influence the nature of strategies adopted to foster cyberspace culture. Figure 4-7 below outlines the proposed framework:

4.7.4 Validation of the Cyberspace Culture Framework



Proposed cyberspace culture framework.

Source: The Researcher

Figure 4-7 above illustrates the cyberspace culture framework aligning the implementation of higher education vision to the cyberspace culture. The research framework (Figure 4-7) argues that attaining the higher education QNV 2030 is driven by three key elements: external aspects such as the QNV 2030 directives and advancements in technology, internal elements within higher education institutions, and theoretical perspectives governing the cultivation of culture. In turn, this is poised to foster cyberspace culture in higher education. Bi-directional arrows are employed to illustrate the interrelationship between the different variables, as each produces a commensurate influence on the other variable.

Internal elements include leadership and management strategies employed in the different higher education institutions, information security policies, and both security and technical best practices formulated and implemented in the individual institutions. The internal elements are also observed to be interrelated with the QNV 2030's implementation in higher education using bi-directional arrows, indicating that each of the elements affects and is affected by the implementation. For instance, the internal leadership strategy adopted in an organisation influences the implementation of the higher education vision. Likewise, the overall higher education QNV 2030 further influences the nature of the adopted leadership strategies. Similarly, information security training and policies are expected to be formulated and updated by the leadership in the given institutions in alignment with the framework's implementation. The proposed cyberspace culture framework thus outlines important dimensions of information security, such as training, which can be considered by leadership in an effort to further enhance their policies. Beyond this, security and technical best practices unique to the organisation also display a bi-directional association with the QNV 2030's attainment.

External elements, such as the QNV 2030 directive in terms of leadership and information security policy and regulation, influence and are influenced by the QNV 2030's implementation in higher education. Furthermore, advancement in technology also affects strategy implementation by directly influencing teaching and learning activities, as well as the management of higher education. As the trend in higher education shifts towards adopting technology to enhance the various processes of learning, teaching, and management, most universities tend to formulate strategies around the new technology. As a result, strategy implementation is influenced by the available technology employed. Finally, the QNV 2030 also affects the implementation of the 2030 vision for higher education by virtue of its strategic plan and pillars (Figure 2-2).

However, the interaction between cyberspace culture and theoretical perspectives—namely, levels and dimensions of culture—remains unidirectional, as the concepts only influence the nature of cyberspace culture. Bi-directional arrows are employed between cyberspace culture and the achievement of the higher education vision, as the two concepts interrelate with one another. In conclusion, cyberspace culture is argued to both influence and be influenced by the QNV 2030's attainment. However, the relationship between cyberspace culture and various theories remains solely unidirectional, as the culture does not influence the theories' position.

Cyberspace essentially describes the virtual and non-physical world present between computers and allows individuals to develop new selves and share data with one another. Subsequently, as individuals interact with each other and the resources in the virtual realm, this fosters a pattern of observable behaviour referred to as cyberspace culture. Cybersecurity, meanwhile, describes the protection of information and its critical elements, such as the systems and hardware that use, store, and transmit the information.

4.7.5 Contribution of the Cyberspace Culture Framework

The proposed framework contributes to attaining the higher education vision in several ways: First, it outlines the different internal and external elements involved in fostering positive cyberspace culture. These elements include leaders in different capacities across institutions of higher education, such as technical experts to offer training regarding the use of cyberspace technology to facilitate learning, teaching, and authorship; policy-makers to develop effective policies for minimising cyberspace insecurity and promoting best standards; the government as an oversight entity to facilitate the adoption of cyberspace culture; and finally, consultants and culture experts to enable institutions to adopt positive cyberspace culture.

Second, the framework argues that each of the highlighted elements significantly influence the development of cyberspace culture. As such, this implies that comprehensive analysis of each factor's influence on culture formation is required when implementing the framework. For instance, with the technical element, it will be important for institutions to map out all sorts of skills required by learners and educators, as well as the risk factors that could negatively influence the skills' development.

Third, by separating each framework component, this also facilitates easier evaluation of each element's contribution to the development of cyberspace culture and the commensurate attainment of the vision. For instance, evaluation procedures can examine leadership's influence on the progress made in attaining the vision and compare the results against the influence of policy, government regulation, and so on.

In addition to outlining the different contributions proposed by the cyberspace culture framework, it is also important to note that some elements have been drawn from other existent frameworks (refer to Table 4-3). For instance, the frameworks by Alnatheer and Nelson (2009), Shabbaan (2014), and Da Veiga and Elof (2010) all highlight the relevance of leadership and security culture in fostering positive culture. In the same vein, Shabbaan (2014) and Da Veiga and Elof (2010) also emphasised the importance of policy and technical competency. Nonetheless, some aspects that are unique in the proposed cyberspace culture framework include the identification of both internal

and external elements as influential elements in fostering cyberspace culture. Similarly, in the proposed framework, the impact of theoretical perspectives is also considered. These aspects form part of the unique contribution made by the Researcher.

4.8 Conclusion

The comprehensive review of various topical areas in the research revealed numerous significant findings. For instance, the review of various cultural theories (Schein and Hofstede) revealed important components required in culture formation (attitudes, beliefs, and artefacts), subsequently providing further insight into the requirements for developing the conceptual framework. However, in order to conceptualise the proposed cyberspace culture framework, the chapter also examined current information security frameworks. The critical evaluation of the different cyberspace culture frameworks indicated that, while cyberspace culture frameworks exist in numerous countries and organisations, none of them consider human participation. Proper training, education, and rewards or punishments established for employees can ensure that an organisation's cyberspace culture strategy is effectively implemented.

In response, the proposed framework (Figure 4-7) seeks to address current research gaps in delivering learning in cyberspace and empowering educators to use cyberspace technology to teach and develop content while ensuring their identities in cyberspace are not adversely affected. Additionally, the framework also strives to help researchers and leaders in higher education to visualise and comprehend how they can develop effective cyberspace culture in their own environments. Three core elements of the proposed framework (Figure 4-7) are outlined (theoretical underpinnings, external actors implementing the QNV 2030 directive, and internal actors involved in managing higher education institutions) in order to demonstrate the different kinds of elements required to implement the proposed framework. As a result, the roles of different elements are clearly set apart so as to minimise any possibility of overlap or confusion.

In the following chapter, the research methodology guiding the collection and analysis of primary data in this research is detailed. Furthermore, justification of the selected methodology is also provided in order to ensure the appropriate alternatives are implemented.

Chapter 5: Research Methodology

5.1 Introduction

Research methodology describes the science and philosophy behind a given research (Adams, Khan, and Raeside, 2014). This allows the researcher an in-depth understanding regarding the knowledge constraints and the different ways in which they can be created. Understanding how to construct the research methodology appropriately is also important for enabling the researcher to critically analyse the knowledge represented as fact. Therefore, a need arises to critically examine knowledge-creation processes so as to add value to society (Khan, 2008). Furthermore, as the researcher constructs the research methodology, diverse theoretical underpinnings are reviewed, issues pertaining to data collection and analysis are examined, and conclusions on the specific phenomenon under investigation are obtained.

Nevertheless, identifying the appropriate methodology for a given study can represent a daunting task, especially since this directly influences the credibility of the results. In the worst-case scenario, the chosen research methodology will fail to answer the research objectives identified at the onset of the research. However, to avoid such occurrences, the researcher can adopt the research onion framework to provide a structural guide for conducting the research work in a stepwise, logical, and analytical manner (Saunders *et al.*, 2009). An additional motivation for adopting this framework stems from the fact that it is widely accepted and deployed in the scientific community. The research onion is illustrated in Figure 5-1 below:



Figure 5-1: Research onion process. Source: Saunders *et al.* (2009)

Justification for employing the research onion framework arises from the fact that it provides a stepwise approach to undertaking research similar to peeling an onion, which allows further details to be observed (Saunders *et al.*, 2009). At the outermost layer, the research philosophy is outlined, and gradually, research aspects such as approaches, strategies, choices, time horizons, and techniques are revealed. Accordingly, this chapter is structured based on the different layers of the framework.

Section 5.2 details the rationale behind the research philosophy while Section 5.3 discusses the research approach. In Section 5.4, the research strategy is described further while Section 5.5 elaborates on the research design. In Section 5.6, the data collection methods—namely, semi-structured interviews—are discussed in greater detail. In Section 5.7, data analysis techniques are elaborated further. Due to adopting qualitative collection techniques, thematic analysis represents the main analytical approach utilised. Section 5.8 discusses the various considerations of research quality, particularly the validity and reliability concerns. Finally, a conclusion section is added to summarise the main findings from within the chapter.

5.2 Interpretivist Research Philosophy

In selecting a research philosophy, the researcher adopts a belief regarding the procedure through which data for a given phenomenon should be gathered, analysed, and utilised (Kothari and Garg, 2016). Collins (2010) viewed this as a paradigm and, as such, argued that it acts as the lens through which authors view the world. Adopting the appropriate research philosophy is thus important in this research, as this enables the Researcher to develop assumptions regarding the world being studied as well as the various ways in which authors ought to understand it. Furthermore, as diverse assumptions can be made about the world, this subsequently leads to the emergence of diverse philosophies to support scientific inquiry.

In this research, the interpretive philosophy is utilised due to its ontology, epistemology, and axiology. Kumar (2008) emphasises that these three features distinguish a philosophy, with ontology describing the author's view regarding the reality under study, epistemology referring to the author's view concerning what pertains to acceptable knowledge, and axiology outlining the author's view as to the role of values in research.

The interpretivist philosophy is considered the opposite of the positivist paradigm, as the researcher adopts a subjective rather than objective view in conducting the research (Engel and Schutt, 2014). Furthermore, with the interpretivist paradigm, reality is constructed socially with the authors, focussing on the understanding that reality is shaped by people, unlike in the positivist case, where emphasis is directed towards reality without actors' input. Collins (2010) added that interpretivism is aligned with idealism—the view that meaning resides within the world independent of consciousness.

Interpretivism is justified in the present research, as it aligns with the Researcher's aim of comprehending the cyberspace culture as it is experienced by different actors involved in implementing the QNV 2030 in higher education. Furthermore, values play an important role in determining how actors behave in the different cyberspaces, as positive reinforcement contributes fosters positive culture whereas negative reinforcement leads to nefarious acts. As such, a subjective perspective helps the Researcher understand the reality under study. In the following section, the research approach adopted here is discussed. Justification for adopting the inductive rather than deductive option is likewise detailed.

5.3 Research Approach

As scientific research involves applying systematic techniques and procedures to facilitate the creation of knowledge, two main forms of reasoning are employed: deductive and inductive (Spaulding *et al.*, 2013). The current section examines these two approaches and provides justification for the selected alternative.

5.3.1 Inductive Research

Over the years, the examination of deductive and inductive reasoning by numerous authors has led to the emergence of various definitions of the concepts. On the one hand, Goswami (2011) compared the concepts on the basis that they involve drawing new conclusions from given information. For this reason, the Researcher conceptualised inductive reasoning as requiring individuals to transcend the given information and make references that may not be valid deductively, such as in drawing an analogy.

Glaser (2014), on the other hand, differentiated the concepts based on their root definitions of deduction and induction, respectively. This approach derived from Decoo (1996), who argued that their root definitions serve as the least common denominator of the terms. According to Decoo (1996), deduction essentially describes a process that moves from consciously generated rules to their application and language use or, in other words, from general to specific; meanwhile, induction implies the inverse, describing a process that moves from the particular to the general, or from language examples to the rules that govern them.

Subsequently, Glaser (2014) differentiated inductive and deductive reasoning based on the order in which each theory is encountered; inductive reasoning encounters examples before inferring the rules that control them while deductive reasoning concerns the presentation of rules before examples are encountered. Trochim *et al.* (2016) further resonated with this definition, though they defined the terms informally as the top-down deductive and bottom-up inductive approaches.

Figure 5-2 below further illustrates the bottom-up inductive reasoning approach, demonstrating the development of theory from observations:



Figure 5-22: Bottom-up inductive research approach.

Source: Trochim et al. (2016)

In this research, the inductive approach is deemed appropriate for several underlying reasons: First, this research is exploratory in nature, as the focus resides on examining the different ways in which cyberspace culture shapes Qatar's 2030 higher educational vision. As such, this indicates that the Researcher should begin by interacting with specific observations, followed by identifying patterns and formulating tentative hypotheses before generalising theories that govern the relationships. Second, the goal of this research can only be attained after examining real-world observations, indicating that the deductive approach is not suitable, as it does not necessarily rely on real-world knowledge to reach conclusions. Third, the present research does not formulate any hypotheses to be tested through empirical observations, further justifying the inductive approach. The research strategy for data collection is discussed in the following section.

5.4 Research Strategy

The objective of a research strategy consists of outlining a general plan that will be employed to undertake the research work (Cresswell, 2007). In another study, Singh (2011) posits that the research strategy presents a formal plan to solve a problem by outlining research objectives and planned devices necessary to implement the strategy. In this research, two research strategies are adopted: semi-structured interviews and focus group discussions. In Sub-Section 5.4.1 below, the semi-structured interview strategy is discussed and justified regarding its use in the current research.

5.4.1 Semi-Structured Interviews

According to Wengraf (2011), semi-structured interviews differ from fully structured interviews in that they involve using open-ended questions and the subsequent development of follow-up questions that generate further evidence regarding a candidate's knowledge on a given asset. Smit

et al. (2008) further highlighted that, since a candidate's responses cannot be predicted in advance, the interviewer is required to improvise questions in alignment to the candidate's responses in order to gain an in-depth understanding of their knowledge. As such, this technique is more flexible than the conventional fully structured interview.

Given that this research seeks to investigate the strategic approach to cyberspace culture and how it aligns with the QNV 2030's strategic direction, semi-structured interviews were considered the most appropriate strategy, as they enable the Researcher to incorporate follow-up questions that facilitate extracting more knowledge from participants. In effect, this strategy helps the Researcher explore the problem area and gather as much data as possible regarding the selected topic. Second, semi-structured interviews were also chosen because this research is guided by the interpretivist philosophy, which emphasises collecting data from a small number of participants. Due to the reduced size of the research population, it was deemed necessary to employ a strategy that extracts as much data as possible from the participants. To facilitate this collection, semi-structured interviews represent the most appropriate option.

However, to conduct interviews, a researcher is still required to prepare standard questions that guide the interview process, similar to the fully structured interviews (Mitchell and Jolley, 2013). Wengraf (2011) added that semi-structured interviews will nonetheless require more preparation, as they incorporate follow-up questions. Similarly, greater creativity and discipline levels are required of the researcher in order to improvise the responses provided by a candidate. However, the author can be challenged in interpreting the responses due to the incorporation of follow-up questions (Edwards and Holland, 2013).

In this research, the Researcher was interested in interviewing diverse stakeholders directly involved in developing or implementing the QNV 2030 in higher education. As the stakeholders are drawn from different sectors, the semi-structured interview technique was appropriate for gaining an in-depth understanding of their perceptions regarding how cyberspace culture influences the vision's attainment. Creative use of follow-up questions further enabled the Researcher to gain deeper insights regarding the underlying factors that are likely to challenge attaining the vision.

5.4.2 Focus Group Discussions

Focus group discussions refer to unique qualitative research methods involving the discussion of a specific set of issues among a pre-determined group of people (Hennink, 2007). Such groups are comprised of 8–10 members who are knowledgeable about a certain topic or issue (Edmunds, 2001). However, adding more than 10 individuals per group lowers its efficiency, as interactions become difficult to control. Krueger and Casey (2015), though, argued that the number can range from as few as four to as many as 12 individuals. The decision of the group's size nonetheless lies with the author, albeit with the guideline that the group's size must be small enough to allow individuals to interact with one another while simultaneously being large enough to provide a wide range of perspectives.

Hennink (2007) posited that a focus group's core purpose is to provide a wide range of views concerning a given topic as well as to facilitate a deeper understanding of the highlighted issues from the individuals' perspectives. Subsequently, when organising focus group discussions, care must be taken to ensure that the appropriate group size is selected and that all individuals involved depict a certain level of homogeneity or similarity (Stewart *et al.*, 2007). The similarity aspect is employed as a basis of recruitment whereby only individuals who meet certain pre-determined criteria can be involved in the discussions. For instance, the recruitment criteria can be broadly defined on aspects such as age or gender. Alternatively, they can be defined narrowly, such as being restricted to individuals who participated in a given programme (Krueger and Casey, 2015). Therefore, a researcher needs to be aware that, under most circumstances, the focus group brings together individuals who do not know each other.

Justification for selecting focus group discussions stems from the fact that they provide qualitative data due to their exploratory nature (Edmunds, 2001). In this research, the Researcher established several focus groups comprising stakeholders drawn from different organisations directly or indirectly involved in implementing the QNV 2030 in higher education. Some of these include individuals in the Ministry of Transport and Communication (MOTC), stakeholders and management of higher education institutions, and government policy-makers. Additionally, utilising focus groups complements the results obtained from semi-structured interviews, where other important perspectives can be generated.

Despite the wide range of advantages associated with adopting focus groups, such as multiplicity of views, spontaneity of responses, and their ability to facilitate further clarification while also allowing the identification of new issues, a researcher is expected to exercise caution when conducting such discussions due to several limitations (Fern, 2001). Specifically, Hennink (2007) identified three highly problematic limitations: First, the Researcher must demonstrate adequate moderation skills so as to eliminate bias in selecting participants. Second, group dynamics may generate unprecedented issues and, as a result, adversely affect the discussions' effectiveness (Edmunds, 2001). For instance, some participants might be too dominating whereas others might shy from engaging in discussions. In other instances, hierarchies may develop, leading to agreement rather than engagement among participants. Third, issues might arise in the data analysis, especially where large textual information is generated (Stewart *et al.*, 2007).

As this section has elaborated on the use of focus groups and semi-structured interviews to collect primary data, the following section discusses the research design effectively adopted.

5.5 Research Design

According to Myers *et al.* (2010), adopting an appropriate research design is important in research, as this helps combine the study elements logically and coherently so as to address formulated research questions. In this regard, the adopted research design presents a blueprint that guides data collection, assessment, and evaluation (Gorard, 2013). The implication of selecting the wrong design is that this can lead to weak and unconvincing conclusions whereas appropriate research designs guarantee that the conclusions are valid and trustworthy (Anastas, 2012). As a result, the Researcher is tasked with identifying the philosophical assumptions, outlining the research method, identifying techniques for collecting and analysing data, and providing a summary of the research findings in order to align empirical findings with the research design. Figure 5-3 illustrates the research design employed in the current research:



Figure 5-3: Sequential research design adopted.

Source: The Researcher

As presented in Figure 5-4 above, the research design begins with the problem-identification phase followed by a comprehensive review of cyberspace culture literature with the goal of examining prevalent research gaps. The identified research gaps provide a basis for developing the theoretical framework aligning cyberspace culture to the QNV 2030 in higher education. Stage one concludes with the identification of a suitable research methodology. In stage two, the Researcher emphasises qualitative data collection and analysis using thematic analysis procedures. Finally, in stage three, the empirical data are combined with secondary literature findings and validation of the developed framework undertaken in the context of the cyberspace culture alignment with the QNV 2030. Additionally, the Researcher also focusses on answering the research questions by combining the primary and secondary data. The individual elements of the research design (research process and sample composition) are further discussed in the following sub-sections.

5.5.1 Research Process

This sub-section discusses the research processes adopted in this research. To begin, the desk research is expounded, followed by a discussion of the field research process in the second section.

5.5.1.1 Desk Research

The goal of the desk research was to identify a theoretical foundation facilitating the development of a framework to align cyberspace culture with the QNV 2030 in higher education, as illustrated in Chapter 4. However, a prerequisite for developing the framework involved conducting an indepth evaluation of literature on cyberspace culture (Chapter 2) and identifying current literature gaps (Chapter 3). To this end, the Researcher examined diverse literature sources ranging from books and journal articles to conference papers, case studies, and industry reports in order to gain a better understanding of cyberspace culture and how it influences attaining the QNV 2030 in higher education.

The four-quadrant framework (Althonayan, 2003) was employed in the literature evaluation chapter (Chapter 3) to facilitate identifying research gaps. The framework divided the research philosophy into visionary and implementational and the outcomes into descriptive and prescriptive (see Chapter 3, Tables 3-1 and 3-2). Chapter 3 concluded that few studies focussed on the implementation of cyberspace culture in different contexts. In particular, the Researcher identified a lack of cyberspace culture frameworks in the context of Qatar and the implementation of its QNV 2030. As a result, a need was identified to develop a framework that addressed such research gaps. The literature examined in Chapter 2 was further classified under the four framework quadrants: implementational-prescriptive (see Table 3-2). As a result, the Researcher was able to easily identify quadrants where significant work had been undertaken, thus eliminating the need for further work, as well as quadrants possessing a limited number of studies, thus indicating a need for further research. Beyond this, the classification further confirmed a need to develop a framework to facilitate implementing cyberspace culture in Qatar's context. Objectives of desk research included the following:

- 1. To identify core academic- and industry-based literature on cyberspace culture.
- 2. To establish and elaborate on current literature gaps.
- 3. To develop a theoretical foundation to facilitate the development of a framework to address prevalent research gaps.
- 4. To enhance subject knowledge on different thematic areas on cyberspace culture.

Desk research is important, as it facilitates the development of the proposed framework, and so addresses the different identified research gaps in addition to providing recommendations for both academia and the higher education sector in Qatar.

5.5.1.2 Field Research

As this research focusses on developing a framework aligning cyberspace culture to the QNV 2030 in higher education, the qualitative research approach was deemed appropriate. With qualitative research, the Researcher could interact with important stakeholders involved in both developing and implementing the QNV 2030. The utilised data sources are listed in the following:

- 1. Primary research data sources:
 - 1.1 Qualitative research
 - Semi-structured interviews with stakeholders from the Ministry of Transport and Communication (MOTC), government policy-makers, and senior practitioners in the higher education sector in Qatar.
 - Focus groups with senior practitioners in the Ministry of Transport and Communication (MOTC) and Qatar's government and higher education sector.
- 2. Secondary research data sources:
 - 2.1 Literature review

The objectives of field research (refer to Section 1.5) included the following:

- 1. To evaluate current cyberspace cultures' alignment with strategic vision.
- 2. To evaluate the various current frameworks regarding cyberspace culture and identify their strengths and weaknesses.
- 3. To analyse the current strategic approach to cyberspace culture and its alignment with the country's strategic directions.
- 4. To analyse the Qatar Vision, 2030 cyberspace strategy and how it is influenced by the attainment of the 2030 higher education vision.

5.5.2 Sample Composition

This sub-section discusses the adopted sampling procedures and the composition of the research sample utilised in the semi-structured interviews. As the research is qualitative in nature, it is associated with smaller sample sizes, unlike the quantitative research approach, which involves large random samples (Daymon and Holloway, 2011). Silverman (2010) further argued that, since

the validity of qualitative analysis depends more on the quality of the analysis rather than the sample size, qualitative studies are typically composed of small sampling sizes studied in depth.

The emphasis on smaller sample sizes indicates that the considered research population is homogeneous in nature (Shi, 2008). Creswell (2012) also justified the use of a homogeneous study population in a qualitative study by highlighting that this ensures the author can understand the underlying problem at a deeper level. Three sampling strategies are associated with qualitative research: convenience sampling, snowball sampling, and purposive sampling (Macnee and McCabe, 2008). The present research utilises the purposive sampling technique, which involves the intentional selection of participants based on predetermined criteria, such as specialist knowledge and capacity to take part in the research (Jupp, 2009). In this regard, this technique only considers participants who are involved in the development or implementation of Qatar's 2030 vision for higher education. Specialist knowledge and position held in different institutions represent some of the key factors considered in the selection criteria. The alternative snowballing and convenience sampling techniques, by contrast, were not adopted due to their misalignment with the research objectives. For instance, convenience sampling was rejected due to solely focussing on participants who meet pre-determined selection criteria and are readily available to participate. As such, this technique was critiqued for producing poor data quality that would lack intellectual credibility. The snowball sampling technique was also rejected for utilising individuals who are known and recommended by the current research participants.

Trochim *et al.* (2016) further classified sampling techniques into two types: probability and nonprobability. With probability sampling, the members of the population possess a known, equal chance of being included in the research sample, whereas with the non-probability case, their inclusion in the research depends on the researcher's judgment (Babbie, 2015). The present research utilises the non-probability sampling technique, as the Researcher is only interested in participants who display certain knowledge and who work in a predetermined environment.

Two qualitative techniques were employed to collect data in the present research: semi-structured interviews and focus group discussions. Initially, the Researcher planned to collect data from 15 interviews and two focus groups comprising five members each. However, with the Covid-19

pandemic, the research plan was adjusted due to social distancing directives by the World Health Organisation (WHO), meaning that physical one-on-one meetings could not take place. As a result, the Researcher increased the interviews to 18 in order to leverage online teleconferencing technology and reduced the focus group to only one comprising six members. The Researcher was also able to conduct the focus group before the WHO social distancing directive. However, for the interviews, only eight had been conducted on a one-on-one basis. As a result, 10 interviews had to be conducted via Microsoft Online Call. The semi-structured interviews were conducted with senior practitioners drawn from higher education institutions, MOTC, and regulatory organisations in Qatar. The Researcher considered respondents from different professional backgrounds in order to increase the robustness and validity of the findings. The second stage involved facilitating the focus group discussion, comprising six senior practitioners.

An additional consideration when determining sample sizes involved data saturation, an aspect in which no new data that emerge can be considered important for developing theory or achieving the research's agenda (Daymon and Holloway, 2011). The Researcher accordingly ensured that the sample size is neither so small that it does not achieve data saturation nor so large that the Researcher would be unable to capture the responses of individuals in the research sample. As such, a total of 24 participants were accordingly identified to provide the qualitative data (18 for the interviews, six focus group). In the following section, the considerations made regarding data collection are detailed further.

5.6 Data Collection Strategy

This section discusses the processes undertaken to prepare and undertake the interviews and focus groups.

5.6.1 Semi-Structured Interviews

Interviewing has been described as the most reactive data-collection method, as it involves asking people questions to which they provide answers (Bernard *et al.*, 2017). Three types of interviews exist: unstructured, semi-structured, and structured (Bernsen and Dybkjær, 2010). In this subsection, the different interviews are elaborated further and justification is provided for the adoption of semi-structured interviews in this research.

To begin, Myers and Shaw (2004) described unstructured interviews as lacking a formal agenda or fixed time limit. As a result, both parties are provided with freedom and opportunities to develop relevant issues. Leavy (2014) added that, since they lack a pre-set structure, the interviewer cannot prepare for such interviews beforehand, instead remaining a passive listener throughout the period. While such interviews sound like casual conversations, however, they require the interviewer to be highly disciplined in order to recall issues mentioned by the interviewee (Bernard *et al.*, 2017). The authors added that, despite providing a significant wealth of information, they remain challenged by the fact that the provided information can turn out to be insignificant for the research.

Structured interviews, meanwhile, closely resemble questionnaires that are delivered aloud on a face-to-face basis (Myers and Shaw, 2004). Here, the interviewer asks the respondent a given number of questions in the sequential order in which they appear in a printed form. Additionally, little conversation occurs between the interviewee and interviewer except when making clarifications (Bernsen and Dybkjær, 2010). Such interviews are appropriate when the researcher's interest is to make cross comparisons among groups of participants who answer identical questions. However, Leavy (2014) argued that such researchers are disadvantaged by not leveraging the dialogical potentials for knowledge production associated with human conversation.

The semi-structured interview combines features from both structured and unstructured interviews (Myers and Shaw, 2004). The interviewer (the Researcher) follows a guideline of set questions when conducting the interview and is also allowed some measure of freedom to ask follow-up questions during the interview. The interviewer (the Researcher) is also flexible and can modify the coverage of topics and details regarding how topics are covered (Bernard *et al.*, 2017). As the researcher is thus able to discover new knowledge aspects during the interview process, this was deemed appropriate for the current research. As such, semi-structured interviews were chosen, as they allow greater flexibility in collecting data.

The Researcher sought to interview respondents on a face-to-face basis, utilising telephone interviews when such meetings were impossible to establish. Data from the interviews were

recorded in audio tapes and later transcribed to extract individual responses (Smith *et al.*, 2008). However, as advised by Bernsen and Dybkjær (2010), the Researcher also took additional notes during the interview so as to reduce the time taken in analysing transcripts.

5.6.1.1 Design of the Interviews

The interview questions employed in this research were developed based on key research themes highlighted in the literature review:

- Challenges in cyberspace culture;
- Alignment of cyberspace culture to the higher education QNV 2030;
- Gaps in cyberspace culture and the higher education QNV 2030; and
- Alignment of proposed framework to the higher education QNV 2030.

The research interview comprised nine open-ended questions that were sub-divided into the following framework-aligned categories: 1) external elements, 2) internal elements, 3) theoretical perspectives, and 4) testing the bi-directional relationship between the attainment of the higher education QNV 2030 and cyberspace culture. The questions were adopted from the developed conceptual framework, which described the problem based on the four aspects above. The four categories were adopted from the developed conceptual framework in Figure 4-7.

Follow-up questions were also added so as to gain a deeper understanding regarding the various research themes. Each interview lasted from 30 minutes up to one hour and was divided into three sub-parts.

The first part involved establishing the participants' demographic profiles so as to ensure they possessed sufficient knowledge to provide the required data. However, since the purposive sampling strategy was employed to recruit the research participants, some of the details regarding their work positions had already been established.

The second part centred on identifying respondents' perspectives concerning cyberspace culture and its contribution to the QNV 2030 in higher education. The goals of this sub-section included 1) establishing the depth of understanding regarding how cyberspace culture influences attaining the QNV 2030 in higher education, 2) identifying organisations' critical success factors that affected alignment between the two, and 3) understanding the challenges faced in fostering cyberspace culture.

The third part focussed specifically on the strategic framework aligning cyberspace culture to the attainment of the QNV 2030 in higher education. To this end, questions investigated potential challenges and limitations likely to be encountered during the implementation as well as various solutions to the problems.

5.6.1.2 Selection of Interviewees

Initially, the Researcher targeted 15 participants for the semi-structured interviews. However, due to the Covid-19 pandemic, the Researcher increased the number to 18, as one focus group was impossible to undertake. The 18 participants were drawn from Qatar's Ministry of Transport and Communication (MOTC), higher education institutions, and regulatory organisations. The Researcher obtained the participants through existent networks in research academia as well as by directly contacting the MOTC to recruit additional participants. However, it is worth noting that only six interviews were conducted before the Covid-19 pandemic reached full effect. As a result, there was free interaction between the different senior practitioners and the Researcher without the restrictions of social distancing. The interviews were conducted in two phases whereby the first six participants were interviewed on a one-on-one basis before the Covid-19 pandemic took hold in March 2020. Thereafter, 12 interviews were conducted in 2020 through electronic means (utilising video-conferencing technology) due to the Covid-19 pandemic.

The participants were further grouped into clusters whereby the research dealt with participants in terms of groups as opposed to an individual basis. The clusters were selected based on the organisational structures of the targeted companies and included 1) a strategy team and 2) a technical team. However, the Researcher remained aware that management and leadership teams might be similar in some organisations. Different themes in the interviews were also outlined based on the cyberspace culture framework and included 1) external elements, 2) internal elements, 3) theoretical perspectives, and 4) the bi-directional relationship between the attainment of the higher education QNV 2030 and cyberspace culture.

5.6.2 Focus Group Discussions

This section discusses the organisation of the focus groups as well as the processes employed to select respondents.

5.6.2.1 Design of Focus Group Discussions

In order to ensure consistency, the Researcher utilised the same questions developed for the semistructured interviews in the focus group discussions. Refer to Sub-Section 5.6.1.1 to view the thematic areas that were considered. However, as argued by Breen (2006), several factors have to be considered in establishing focus group discussions due to their differences from semi-structured interviews. Table 5-1 below summarises the differences between the two:

Construct	One-on-one interview	Focus group discussions
Aim	Probe experience	Idea generation
Researcher's role	Interviewer	Moderator
Sample sizes	Target to reach saturation <12	Target to reach saturation 4–6 per
		group
Equipment used	Audio-recording devices, quiet	Audio-recording devices, quiet
	room	room, props (e.g. flash cards)

Table 5-1: Differences between focus groups and one-on-one interviews.

Source: Breen (2006)

First, the Researcher was required to play the role of moderator rather than interviewer. As such, it was necessary to undertake further research to prepare for the new role. Second, due to the respondents' differing work schedules, the Researcher had to contact the participants via email at least 3–4 weeks in advance and to share the focus group discussion questions in order to allow them adequate time to prepare for the discussions (Stewart *et al.*, 2007). Third, in order to manage the focus groups effectively, the Researcher divided the participants into two five-member groups and scheduled discussions on different dates. A private board-room was also outsourced to facilitate discussions in a round-table manner. Props such as leaflets and flash cards containing the research questions were prepared in advance in an effort to ensure the respondents did not stray from the breadth of the topic.

Conducting a pilot study was also important for gauging the respondents' understanding of the interview questions, revising the questions' structure by either adding or deleting some elements, and facilitating evaluation of the new moderator role's effectiveness. To that end, three participants were selected for a mock focus group discussion. The Researcher also focussed on assessing the process in which the focus groups were conducted. Oates and Alevizou (2018) summarised the focus group process in six definite steps, presented in Figure 5-5 below:



Figure 5-4: Focus group process. Source: Oates and Alevizou (2018)

Similarly, through the pilot study, the Researcher was able to assess the respondents' comfort levels as the discussions progressed from the introductory phase to the main discussion and the end of the discussions. Multiple audio-recording devices were utilised in order to eliminate risk of failure (Stewart *et al.*, 2007).

5.6.2.2 Selection of Respondents

The focus group discussions involved only six senior practitioners drawn from Qatar's MOTC, higher education institutions, and regulatory organisations. To facilitate the discussions, the participants were organised into one group comprising six members. It is worth noting, however, that the focus group discussion was conducted before the Covid-19 pandemic had been in full effect. This allowed free interaction between the different senior practitioners and the Researcher without the restrictions of social distancing. The 10th question in the interviews was discussed by participants in the focus group. The following section reviews the procedures utilised to analyse the collected data.

5.7 Qualitative Data Analysis

Data collected from the semi-structured interviews were saved in audio format and thereafter transcribed into text. It is important to emphasise that, although the use of technology proved beneficial in facilitating interviews during the Covid-19 pandemic, it was considered a limitation in qualitative data collection due to hindering the Researcher from observing other subtleties among the participants, such as their body language and tone. Therefore, to obtain research insights, the Researcher relied solely on the analysis of the specific words employed by participants and the various themes that were discussed. In an ideal scenario where face-to-face interviews would be undertaken, the researcher would analyse both the words used by participants and their body language as well as the tone in their speech. However, due to the limitations of technology, such aspects were not considered in the analysis.

In order to analyse the responses, thematic analysis, which essentially describes a process of encoding qualitative information, was employed (Boyatzis, 2009). At its most basic level, thematic analysis enables identifying patterns or themes from a qualitative dataset, and thereafter to derive meaning and value from them (Smith and Sparkes, 2016). Justification for adopting thematic analysis in this research stems from the fact that the Researcher organised the data from semi-structured interviews into several research themes (see Sub-Section 5.6.1.1). As a result, thematic analysis would facilitate encoding the information into useful insights.

Smith and Sparkes (2016) summarised the thematic analysis process in six phases, which can be further simplified into three steps. Figure 5-6 below summarises the six phases of thematic analysis:



Figure 5-6: Thematic analysis phases. Adopted from: Smith and Sparkes (2016)

In the first and second phases, a researcher can gain familiarity with the data by reading it over and over and making notes on interesting aspects that emerge. During the coding phase, important aspects of potential relevance to the research are highlighted, such as by using a computer programme. Smith and Sparkes (2016) argued that no definite stopping point exists for coding, and the researcher can add as many codes as deemed appropriate to capture the analytical aspects of the data. In the third, fourth, and fifth steps, the researcher organises the generated codes into themes and reviews them, thus developing a rich analysis of the data that captures the finalised themes. As a result, three main themes are highlighted by the Researcher: overarching themes that capture the general idea, themes that organise a concept, and sub-themes that capture a given facet of the concept. The final phase involves compiling the findings into a comprehensive report. In the following section, concerns for research quality (validity and reliability) are addressed.

5.8 Research Quality

As this research is largely qualitative in nature, it remains subject to scrutiny, since the Researcher's subjectivity and interpretations influence the obtained results. Subsequently, it is necessary to ensure the findings' reliability and validity. In this section, the Researcher accordingly discusses the standards adopted to ensure the research data remained reliable and valid.

5.8.1 Reliability

Baumgarten (2012) defined reliability as both the consistency associated with the measurement of a given concept by utilising an identical measurement procedure as well as the findings' replicability. Krippendorff (2004) further added that reliable data describe that which remains
145

constant throughout the variations of the measuring process. Two types of reliability are associated with data: internal reliability, which focusses on the consistency of indicators employed in the research and is specified by a correlation value between them; and external reliability, which outlines whether data can be generalised beyond the context of the research (Baumgarten, 2012).

Quantitative research studies that rely on the positivist philosophies easily evaluate data reliability using statistical procedures such as test-retest for stability estimates and Cronbach's Alpha for internal reliability (Mertens, 2005). Noble and Smith (2015), however, argued that reliability can also be achieved in qualitative research by making several considerations, such as demonstrating clear decision trails to ensure consistent interpretations. Each participant in a study should also be selected using similar parameters and consistent lines of questioning. As this research is qualitative in nature, reliability was ensured by adopting a dependable process whereby interviews and focus groups were conducted in a systematic manner. Each of the interviewees was asked the same questions, and all focus group participants were allowed sufficient time to respond to the asked question. Furthermore, non-probability sampling was utilised whereby only participants who met the pre-determined criteria (working in higher education and involved in implementing QNV 2030) were considered. This ensured that the research could be replicated at a later period by future researchers who would adopt a similar process to obtain similar results. In the same vein, reliability was also assured by virtue of the fact that the findings could be confirmed through previous literature studies. This aspect is further described in Chapter 7, wherein the findings from the primary data collection are compared against the previous literature review. As the primary findings corroborated the previously reviewed literature, this emphasises its trustworthiness and truthfulness.

5.8.2 Validity

Validity, meanwhile, concerns truths and cannot be tested through duplications, unlike reliability, which relies on the agreement of results in subsequent duplicates (Krippendorff, 2004). Thyer (2010) added that validity emphasises the findings' accuracy, and in qualitative research, it addresses data credibility. The researcher further highlighted three broad categories under which validity is classified: reactivity, which focusses on the distorting effects arising due to the researcher's presence in the field; research bias; and respondent bias.

In order to ensure validity in qualitative research, Noble and Smith (2015) recommended for the researcher to account for any personal bias that could have influenced the findings, acknowledge any existent bias in sampling, and engage with other researchers to reduce bias. In the present research, the Researcher ensured validity by accounting for the different forms of likely bias and soliciting the participants' views regarding the credibility of the findings and their interpretation. Second, the research was also considered valid and credible as established methods of data collection-namely, focus groups and interviews-were appropriate for addressing the formulated research questions. Third, further evaluation of the methodology also revealed that sampling was appropriate for addressing the research objectives, as the research participants were selected by a purposive sampling technique whereby only professionals who worked in higher education and were involved in implementing vision 2030 were selected. As such, this improved the findings' accuracy, as the participants were directly involved in the particular thematic areas of focus for the research. Additionally, utilising thematic analysis to extract insights from the collected data also further emphasised the validity and trustworthiness of the findings. As this section has indicated how the research quality is achieved here, the following section discusses ethical considerations emphasised in this research.

5.9 Research Ethics

In order to ensure research is conducted in an ethical and accepted fashion, several aspects are considered. To begin, since qualitative approaches are employed to collect data by interviewing different participants, it is important to ensure that the selected respondents take part in the research on a voluntary basis and without coercion (David and Sutton, 2009). For this reason, the researcher must prepare an introductory letter to seek consent from the various participants and to provide a detailed explanation regarding how the data obtained will be secured and utilised. This also confirms that the data gathered from the participants are purely for research basis and that the Researcher will not use it in other ways. Participants were required to sign the letter as confirmation of agreement to provide their data. However, they were also free to decline to take part.

Second, it is also imperative for a researcher to ensure privacy and confidentiality of the collected data (Groves *et al.*, 2009). To this end, pseudonyms such as #participant1 can be adopted to identify respondents. The data should also be stored in a confidential and secure manner and should

not be shared with other third parties without the respondents' consent. To this end, the researcher must strive to store all audio and text-based documentation from respondents in secure drives, safe from any damage or theft.

5.10 Summary

This chapter presented the research methodology guiding this research. Subsequently, theoretical and analytical underpinnings were elaborated in detail as well. An interpretivist philosophy was deemed appropriate for this research, as it aligns with the Researcher's goal of comprehending cyberspace culture as it is experienced by different actors involved in implementing the QNV 2030 in higher education. Inductive reasoning was also adopted, thus indicating that the Researcher would begin by interacting with specific observations, followed by identifying patterns and formulating tentative hypotheses before generalising theories that govern the relationships.

Data were collected using the qualitative approach by conducting focus group discussions and semi-structured interviews. However, in order to ensure validity and reliability of the collected data, the Researcher avoided randomisation of the samples and instead emphasised purposive non-probability sampling, wherein only participants who meet predefined criteria are selected for the research. In Chapter 6, the collected qualitative data are presented for further analysis.

Chapter 6: Primary Data Collection and Analysis

6.1 Introduction

This chapter presents the research findings obtained from conducting interviews and focus groups with the goal of contextualising the validity of the proposed cyberspace culture framework (Figure 4-7). To investigate the effectiveness of aligning the implementation of the 2030 higher education vision with the cultivation of cyberspace culture, this chapter analyses the research findings using a qualitative analysis technique as guided by the research objectives detailed in Section 1.5. Although qualitative analysis comprised the dominant technique employed to draw insights from the primary data, the Researcher also utilised content analysis, which includes a descriptive analysis of the findings. Descriptive statistics were employed to evaluate the particulars of the interviews and focus groups, including demographic details such as gender, work positions, and organisational sectors where participants worked. Thereafter, thematic analysis was undertaken to evaluate the findings thematically, with further interpretation undertaken in Chapter 7, where the research questions are addressed. Regarding the chapter's structure, Section 6.2 reiterates the data collection and analysis strategies employed in this research. In Sections 6.3 and 6.4, the particulars of the interviews and focus groups are presented using descriptive analysis. Section 6.5, which comprises the bulk of the chapter, presents the findings from the thematic analysis while Section 6.6 concludes the chapter.

6.2 Data Collection and Analysis Strategies

In this section, the strategies employed to collect and analyse the data are detailed. However, since the strategies were discussed comprehensively in the methodology chapter (Chapter 5), in this section, the Researcher only emphasises key points that were identified as opposed to duplicating the content. The objective of outlining semi-structured interviews and focus group discussions is to provide clarity regarding the procedures employed to collect and analyse the research data before detailing the findings that were obtained.

6.2.1 Data-Collection Strategy

As detailed previously in Section 5.6, the Researcher adopted two data-collection strategies: semistructured interviews and a focus group. For the semi-structured interviews, the Researcher prepared a guideline of nine open-ended questions in alignment with the thematic areas highlighted in the proposed cyberspace culture framework (Sub-Section 5.6.1.1). Interviewee selection was further accomplished using a purposive sampling technique (Sub-Section 5.6.1.2) whereby 18 participants were drawn from Qatar's Ministry of Transport and Communication (MOTC), higher education institutions, and regulatory organisations. The interviews were conducted using two primary techniques: face to face and via online calls using the Microsoft Team Online Calls application.

Furthermore, for the focus group (Sub-Section 5.6.2), the Researcher acted as a moderator of a group of six participants drawn from the same organisations where only one question was considered (Sub-Section 5.6.2.2). However, unlike the semi-structured interviews, guiding questions were not prepared. This is because the goal of the focus group was to evaluate the proposed cyberspace culture framework's effectiveness (Figure 4-7) in facilitating the attainment of the higher education 2030 vision and the achievement of a cyberspace culture.

6.2.2 Data Analysis Strategy

As revealed in Section 5.7, qualitative analysis represented the main procedure employed for the assessment of the data. In this section, the procedure is described further.

According to Elo *et al.* (2014), qualitative content analysis, as a research method, represents a systematic and objective approach to describing and quantifying phenomena. As such, the authors claimed that, in order to undertake content analysis successfully, raw data must be reduced to concepts that describe the research phenomena by creating categories, concepts, models, or conceptual systems. Hsieh and Shannon (2005) further posited that, rather than comprising a single method employed to draw important insights from text data, qualitative content analysis can be sub-divided into three approaches: conventional, summative, and directed. The authors further highlighted that the key difference between them stems from their coding schemes, code origins, and threats of trustworthiness. However, additional differences were highlighted by Pietiläinen *et al.* (2017): On the one hand, with the conventional analysis approach, coding categories are derived from the raw data during analysis; on the other hand, directed content analysis develops coding categories from prior theories or relevant research findings (Oh, 2018). The summative content analysis involves counting and comparing keywords or content, followed by interpretation of the underlying context.

This research adopts the directed content analysis approach whereby coding categories are derived from the proposed cyberspace culture framework. Therefore, the Researcher did not focus on generating codes and themes from the collected raw data. Instead, the various components of the developed cyberspace culture framework were employed to organise the raw data into themes and categories. The justification for considering the directed content analysis approach stems from the fact that the Researcher designed interviews and the focus group under the guidance of the proposed cyberspace culture framework. As a result, this helps identify coding categories and themes employed in the data analysis.

Erlingsson and Brysiewicz (2017) reiterated this claim, observing that, although the starting point for qualitative content analysis features transcribed interview text, the objective of the analysis consists of transforming the verbatim-transcribed interviews into a concise summary of abstracted key results. Figure 6-1 below outlines the overview process of the stepwise qualitative content analysis adopted in this research:



Figure 6-1: Synopsis of the adopted qualitative content analysis approach. Source: Erlingsson and Brysiewicz (2017)

As outlined in Figure 6-1, the analysis strategy in this research involves four activities: 1) condensing the transcribed text into shorter sentences; 2) developing codes, comprising labels that are one or two words long; 3) developing categories for group-related codes through either content or context; and 4) developing themes expressing the latent or underlying meaning communicated to the reader. Nonetheless, due to the extensive nature of the collected data, for this research, the Researcher employed NVivo 12—a qualitative data analysis software application that enhances processing and analysing the findings. It is worth noting, however, that this research primarily collected qualitative data by conducting interviews and moderating a focus group. Accordingly, the quantitative results presented in this chapter arose from analysing the qualitative data and were only incorporated to support the interview findings. For instance, several graphical charts were generated to display the distribution of responses. However, as the data were only collected at one point in time, this was considered cross-sectional in nature (Siegel 2012).

Furthermore, as reported by Zamawe (2015), computer-assisted qualitative data analysis software (CAQDAS) has emerged as a popular solution to lessen the burden of undertaking qualitative research projects due to their labour-intensive and time-consuming nature. While diverse CAQDAS applications are readily available, such as ATLAS, MAXqda, NVivo, and Q6, the Researcher adopted Nvivo 12 due to its wide-scale adoption in the scientific community and its robust functions, such as character-based coding, rich text capabilities, and multimedia functions (Edwards-Jones, 2014). This application enabled the Researcher to condense the transcribed text into smaller sentences and to identify codes or labels for the transcribed text. However, as argued by Zamawe (2015), qualitative analysis software only aids the analysis process; it does not analyse the data. As such, the researcher must maintain control of the analysis process. In this context, it was important that the Researcher determine the various categories and themes to facilitate data interpretation.

A total of 18 interviews and one focus group comprising six members were undertaken to provide the required data. The Researcher initially planned to collect the data over a period of only one month (4 weeks). However, with the Covid-19 pandemic, this was increased to two months (6–8 weeks) due to the difficulty of accessing the respondents. Interview responses provided data for the first three themes while the focus group responses provided input for the fourth theme. The various themes adopted in this research are outlined as follows:

- 1. Theme 1: Influence of external elements on the attainment of the QNV 2030 in higher education (Sub-Section 4.7.2).
- 2. Theme 2: Influence of internal elements on the attainment of the QNV 2030 in higher education (Sub-Section 4.7.1).
- 3. Theme 3: Influence of theoretical perspectives on the attainment of the QNV 2030 in higher education (Sub-Section 4.7.3).
- 4. Theme 4: Effectiveness of the proposed cyberspace culture framework in aligning cyberspace culture with the higher education QNV 2030 (Sub-Section 4.7.4).

6.3 Interview Particulars

To ensure validity and reliability of the findings—as discussed in Chapter 5, Sub-Sections 5.8.1 and 5.8.2—purposive sampling was utilised whereby only select participants were considered to take part in the interviews based on their knowledge of the research topic. The participants included senior staff members working in Qatar's MOTC, higher education institutions, and other government regulatory organisations. The Researcher categorised the interviewees into two core groups—strategy and technical—based on the fact that the proposed cyberspace culture framework (Figure 4-7) generally detailed these aspects. The grouping also arose from the nature of the participants' roles and the departments in which they worked in the different organisations, as some were in strategy while others resided in technically-oriented positions. Eight of the targeted interviews were conducted face to face (before the COVID-19 pandemic) where the Researcher interacted with the interviewees in person. However, due to the COVID-19 disruption and the resultant social distancing requirements outlined by the World Health Organisation (WHO), the remaining 10 interviews were conducted via Microsoft Team online calls. Table 6-1 illustrates some of the interview particulars. Refer to Appendix B for the full particular details.

No.	Respondent code	Duration	Words	Interviewee roles	
1	Respondent 1	29:27	3019	Dean	
2	Respondent 2	42:34	4492	VP	
3	Respondent 3	46:39	6632	Manager	
4	Respondent 4	49:30	3392	VP	
5	Respondent 5	25:48	3939	Director	
6	Respondent 6	31:00	4646	Associate Vice President	
7	Respondent 7	20:04	2736	Undersecretary	
8	Respondent 8	22:18	2792	Director	
9	Respondent 9	30:40	3908	Manager	
10	Respondent 10	19:00	1895	Manager	
11	Respondent 11	28:51	2659	President	
12	Respondent 12	17:30	2303	Section Head	
13	Respondent 13	27:59	3633	Teaching Assistant	
14	Respondent 14	44:21	6009	Director	
15	Respondent 15	43:47	6587	Director	
16	Respondent 16	25:05	3154	Section Head	
17	Respondent 17	17:48	2358	Director	
18	Respondent 18	10:27	1425	Associate Vice President	
Total 18 Interviews			Duration: 532.48 min		
			[8 hrs, 14 min, 58 sec]		

Table 6-1: Particulars of the interviews.

Total words: 65,579
10 online calls, eight face-to-face interviews

Source: The Researcher

6.3.1 Interview Demographic Data

The analysis of participants' geographical details revealed that all participants were drawn from Qatar and worked in Qatari-based organisations. This subsequently suggested that they were knowledgeable regarding current issues affecting local organisations.

6.3.1.1 Gender

First, the participants' gender was examined, as detailed in Figure 6-2:



Figure 6-2: Gender of participants. Source: The Researcher

The findings indicate that there were more male (77.8%) than female (22.2%) participants in the research. This suggests that senior positions in higher education and the regulatory sector in Qatar are male-dominated.

6.3.1.2 Nationality

Next, the participants' nationality was assessed, as detailed in Figure 6-3:





All participants worked in organisations based in Qatar and were conversant with the various aspects in the scope of the research. However, according to the analysis of the findings in Figure 6-3, there were more Qatari nationals (66.7%) compared to non-Qataris (33.3%). The existence of non-Qatari nationals among the interview participants further suggests that senior roles in higher education and regulatory organisations are occupied by professionals from different countries.

6.3.1.3 Job Categories

The participants' job categories were also evaluated, as detailed in Figure 6-4:



Figure 6-4: Job category of participants. Source: The Researcher

The findings indicate that the majority of the participants were from academic institutions (higher education; 61%) as compared to non-academic ones (regulatory; 39%). The findings also suggest that the participants were knowledgeable regarding the various issues that challenged the cultivation of cyberspace culture in higher education.

6.3.1.4 Organisational Sector

Further evaluation of the participants' data also revealed the different types of organisations in which the respondents worked. Figure 6-5 illustrates the distribution of work organisations:



Figure 6-5: Organisational sectors of the participants. Source: The Researcher

According to the results, 13 participants worked in higher education (72%) whereas five participants worked in regulatory organisations (28%). The implication is that most of the respondents could identify the issues faced in Qatari higher education. However, the additional members from the regulatory industry provided important insights into the role that regulation oversight plays in achieving the higher education QNV 2030. Furthermore, as the organisational sectors undertake different activities and face various opportunities and challenges, the Researcher

was able to collect diverse perspectives regarding the issues that would affect implementing the QNV 2030 in higher education.

6.3.1.5 Respondents' Professional Roles

Third, the Researcher was also interested in identifying the participants' professional experience and their actual responsibilities in the different organisations. Figure 6-6 illustrates these professional roles:



Figure 6-6: Professional roles of the participants.

Source: The Researcher

Based on the findings from Figure 6-6, all participants held senior roles in their organisations. The majority consisted of directors (27.8%), followed by managers (16.7%), vice presidents (11.1%), associate vice presidents (11.1%), and section heads (11.1%). Fewer participants held the highest senior roles, such as president (5.6%) and dean (5.6%), as well as minor roles, such as teaching assistants (5.6%) and undersecretaries (5.6%).

6.3.1.6 Respondents' Professional Experience

As disparity was identified regarding the participants' professional roles, which further suggested that they served in different operational capacities and were thus knowledgeable regarding different aspects that would affect the QNV 2030's implementation. The Researcher accordingly



investigated the key operations of each participant (either technical or strategy), as detailed in Figure 6-7:

Figure 6-7: Professional responsibilities of interview participants.

Source: The Researcher

The findings from Figure 6-7 indicate that the majority of participants (55.6%) handled strategy operations while 44.4% undertook technical responsibilities. A plausible explanation for the high number of participants in strategy operations stems from the high number of senior staff members reported in Figure 6-6, such as managers, directors, and vice presidents.

6.3.1.7 Respondents' Work Experience

Additionally, to ensure legitimacy and validity of the provided responses, the Researcher also investigated participants' work experience in terms of years worked in the same professional role. Figure 6-8 displays the findings:



Figure 6-8: Professional experience (years) of the participants. Source: The Researcher

From the findings in Figure 6-8, it was observed that all participants had worked in their respective roles for at least two years, and thus were knowledgeable regarding the various operations that took place. Further evaluation also indicated that the most experienced members had worked for eight years as directors, section heads, and managers. One of the managers had also worked in the same role for seven years and as a director for six years. Finally, the associate vice president, vice president, and chief were observed to work for at least five years.

6.4 Focus Group Particulars

After conducting the interviews with the various participants, the Researcher selected different professionals from both academic and non-academic institutions to participate in the focus group discussion. As the main objective of the focus group was to evaluate the proposed cyberspace culture framework and its role in attaining the higher education QNV 2030, the Researcher relied on the focus group to complement or support the interview findings. Accordingly, findings from

the interviews were presented first, followed by findings from the focus group. In this manner, both sets of findings were merged in a complementary manner such that the focus group supported the interview findings.

Next, the demographic details of the focus group participants are examined. Although the Researcher initially planned to facilitate two focus groups, the Covid-19 pandemic challenged this plan, and only one focus group was consequently facilitated. This comprised six participants discussing the role that the proposed cyberspace culture plays in attaining the higher education QNV 2030. Table 6-2 below lists some of the particulars of the focus group. Refer to Appendix C for the full details of the focus group.

No.	Respondent code	Interview setting	Professional roles	Professional experience
.1	Respondent 1	In person Face to Face	Section Head	8 years
.2	Respondent 2	In person Face to Face	Manager	6 years
.3	Respondent 3	In person Face to Face	Director	5 years
.4	Respondent 4	In person Face to Face	Manager	1 year
.5	Respondent 5	In person Face to Face	Director	5 years
.6	Respondent 6	In person Face to Face	Associate Vice President	2 years

Table 6-2: Particulars of the focus group.

Source: The Researcher

6.4.1 Respondents' Professional Responsibilities

First, the members' responsibilities were examined, as detailed in Figure 6-9:



Figure 6-9: Professional responsibilities of the focus group participants.

Source: The Researcher

The reported findings demonstrate a balance in the participants' professional roles: 50% were in strategy while the remaining 50% resided in the technical sector.

6.4.2 Organisational Sector

Next, the organisational sectors from which the participants were drawn were further evaluated, as detailed in Figure 6-10 below:





Source: The Researcher

The findings revealed that the majority of focus group participants (67%) came from the higher education sector rather than the regulatory sector (33%). This indicates that they were highly knowledgeable regarding academic issues that influence attaining the higher education 2030 vision.

6.4.3 Professional Roles

Since most members in the focus group were in higher education, their professional roles were further investigated, as detailed in Figure 6-11:



Figure 6-11: Professional roles of the focus group participants. Source: The Researcher

The findings revealed that all participants held senior roles in a bid to ensure greater validity and reliability of the data collected in the discussion. The majority of the participants comprised directors (33.3%) and managers (33.3%). This suggests that they were knowledgeable regarding high-level issues that challenge the implementation of different directives in higher education.

6.4.4 Professional Experience

The participants' work experience in terms of years served was evaluated as well, as detailed in Figure 6-12 below:



Figure 6-12: Professional experience of the focus group participants.

Source: The Researcher

Based on the findings, the section head had worked for the longest period (eight years), followed by one of the managers (six years) and the directors (five years). Only one of the managers had worked for a one-year period.

6.4.5 Gender

The participants' gender was further evaluated, as detailed in Figure 6-13:



Figure 6-13: Gender of the focus group participants.

Source: The Researcher

From the obtained results, it was observed that there the focus group comprised more male (83.3%) than female participants (16.7%). This finding aligned with the interview results, which had demonstrated a similar result, thus suggesting that senior roles in higher education and regulation industries were male-dominated.

6.4.6 Nationality

Next, a summary of the focus group participants' nationalities is presented in Figure 6-14:



Figure 6-14: Nationality of the focus group participants.

Source: The Researcher

The obtained results indicate that the majority of the participants in the focus groups consisted of Qatari nationals (66.7%). However, since two individuals were foreigners (33.3%), this further suggests that the higher education professionals came from different countries.

6.4.7 Job Category

Finally, the participants' job categories were examined, as detailed below in Figure 6-15:



Figure 6-15: Job category of the focus group participants. Source: The Researcher

Based on an analysis of the results, most of the participants worked in non-academic roles (66.7%) compared to academic ones (33.3%). As such, they held managerial and technical positions as opposed to serving as professors and lecturers.

From the analysis of the participants' demographic details in Sections 6.3 and 6.4, it was observed that the respondents were highly experienced in their fields, as the majority possessed more than five years' experience. Similarly, the participants held senior roles in higher education and regulatory sectors as directors, managers, and presidents, assuming both regulatory and technical responsibilities. Such findings highlight the validity of the participants' responses regarding the diverse issues affecting the QNV 2030's implementation in higher education.

With this understanding, the participants' responses to the different interview and focus group questions were analysed accordingly in the NVivo 12 application, as detailed in the following section. Furthermore, the coming section also details the procedure adopted for conducting the thematic analysis.

6.5 Thematic Analysis

As revealed earlier in Section 5.7, a thematic analysis approach was adopted to analyse the data collected in the research. The justification for using the approach is that the Researcher only collected qualitative data—namely from semi-structured interviews (Chapter 5, Sub-Section 5.6.1) and focus group discussions (Chapter 5, Sub-Section 5.6.2)—to address the research questions (Chapter 1, Section 1.5). As revealed in Sub-Section 5.6.1.1, the Researcher organised the qualitative analysis approach into three steps: i) evaluating the participants' demographic details, ii) assessing the participants' perspectives regarding how the QNV 2030's achievement contributes to the cultivation of cyberspace culture, and iii) evaluating the proposed cyberspace culture framework's strategic alignment with the QNV 2030's achievement in higher education. The first segment of the analysis was completed in Sections 6.3 (interview particulars) and 6.4 (focus group particulars), in which the participants' demographic details were examined.

The current section (Section 6.5) undertakes thematic analysis to draw useful insights from the interview responses in preparation for addressing the research questions in Chapter 7. It is worth noting that thematic analysis of interview responses facilitates assessing the participants' perspectives regarding how the QNV 2030's achievement contributes to the cultivation of cyberspace culture. Likewise, thematic analysis of the focus group responses helps assess the strategic alignment of the proposed cyberspace culture framework with the achievement of the QNV 2030 in higher education.

Thematic analysis, as described by Erlingsson and Brysiewicz (2017) in Figure 6-1, represents a step-wise approach utilised to draw high-level insights from raw data. As such, this involves labelling the data, creating codes, grouping them into related categories, and, finally, generating themes that express the latent meaning, used to address research questions. Pietiläinen *et al.* (2017) also emphasise the need to adopt a step-wise process to analyse data thematically, including reviewing interview transcripts in order to gain a general sense of the data, generate codes, group them into categories, and identify themes.

In the first step of the analysis, the Researcher began by reading through the collected responses in the transcript files to generally understand what participants were discussing. However, codes, categories, and themes were not developed in the transcript files, since a directed content analysis strategy was utilised whereby the proposed cyberspace culture framework guided the identification of themes and codes (Sub-Section 6.2.2). In the second phase, the files containing the interview transcripts were loaded into Nvivo 12 software as raw data. The Researcher then further preprocessed the data in preparation for the analysis. For instance, pseudonyms were assigned to each participant in order to conceal their identity (e.g. Respondent 1 - R1; Respondent 2 - R2). Likewise, the participants' professional categories (strategy and technical) were assigned to each pseudonym so as to easily identify them ('S' for strategy and 'T' for technical). The focus group transcript (named Focus Group) was also added, along with the interviews, in order to read through the content within the program. Upon loading the data into Nvivo, the Researcher further conducted a word-frequency query to gain a general idea of what the participants broadly discussed. This process confirmed the initial findings that the Researcher identified from reading through the transcript files. Figure 6-16 below illustrates the raw interview and focus group transcripts where participants were pseudonymised and a word-frequency query was undertaken:

DATA	Name ^	N 🕰 Unsaved Query				? ₄ ³
🕞 Files	Focus Group	Word Frequency Criteria				Run Query Save Query
File Classificati	📄 R 1 (S)					
Externals	R 10 (S)	Search in: Files and Externals Selected Items * Items in Selected Folders *				
	R 11 (T)	Finding matches: 💿 Exact match only	/ (e.g. "talk")			Display words: OAll
CODES	B 12 (T)	Include stemmed	d words (e.g. "talking")			 1000 most frequent
Rodes	■ R 13 (T)	With minimum length: 5				
	■ R 14 (T)					
e CASES	■ R 14 (1)	Summary Word Cloud				
NOTES		Word	Length	Count	Weighted Percentage >	
A NOTES		apaskar	3	024	0.72%	
📻 Memos	R 17(S)	education	9	524	1 50%	
Annotations	📕 R 18 (T)	think	5	492	1.45%	
\sub Memo Links	📄 R 2 (S)	higher	6	389	1.15%	
	📄 R 3 (T)	vision	6	366	1.08%	
D SEARCH	📄 R 4 (S)	culture	7	341	1.01%	
C Queries	R 5 (T)	people	6	336	0.99%	
Ouery Results	B 6 (S)	government	10	296	0.87%	
C Node Matrices		university	10	272	0.80%	
Node Matrices		qatar	5	269	0.79%	
Sets	R 8 (5)	believe	7	250	0.74%	
A MAPS	E R 9 (S)	technology	10	244	0.72%	
		example	7	222	0.66%	
🧑 Maps		important	9	214	0.63%	
		things	6	198	0.58%	
		security	8	174	0.51%	
		change	6	1/3	0.49%	
		students	8	166	0.49%	
		cyberspace	10	165	0.49%	
		policy	6	163	0.48%	
		achieving	9	157	0.46%	
		management	10	148	0.44%	
		level	5	147	0.43%	
		leadership	10	146	0.43%	
		actually	8	138	0.41%	
OPEN ITEMS	1	learning	8	137	0.40%	
		different	9	121	0.36%	
ABD (S)		cyber	5	118	0.35%	
		Institution	11	117	0.35%	
kk (S)		really	6	116	0.34%	
		naybe	5	115	0.34%	
C Unsaved Query		going	5	114	0.34%	
(A		question	0	113	0.33%	
0	0 item selected	🛱 Unsaved Query				

Figure 6-16: Synopsis of the interview responses.

Source: The Researcher

Table 6-2 further displays the word frequency of terms appearing with a word count of at least 300:

Word	Length	Count	Weighted Percentage
Speaker	7	924	2.73%
Education	9	538	1.59%
Think	5	492	1.45%
Higher	6	389	1.15%
Vision	6	366	1.08%
Culture	7	341	1.01%
People	6	336	0.99%

Table 6-2: Word frequency and weighted percentage of most common words.

Source: The Researcher

The results in Table 6-3 indicate that the most common terms arising in the interview responses, producing a count of at least 300, included speaker (2.73%), education (1.59%), think (1.45%), higher (1.15%), vision (1.08%), culture (1.01%), and people (0.99%). From an analytical perspective, it is argued that the terms' frequency emerged from references to the proposed cyberspace culture framework elements that respondents evaluated as well as the duplicated interview questions employed. Furthermore, the consistent reference to 'government' and 'university' also suggests that they played a significant role in addressing the research questions. However, as the data were not organised thematically, it can be further argued that the word 'query' did not generate any valuable output to address the research questions.

Therefore, the next step involved generating codes and themes from the raw data using a directed content analysis approach (Sub-Section 6.2.2). From Figure 4-7, which illustrates the proposed cyberspace culture framework, the Researcher identified four themes in relation to the framework's components. These aspects were considered themes because a) the interview questions were also organised based on the various issues and b) participants were questioned on

the different thematic aspects (refer to Appendix A for the interview questions). The themes in the research included the following:

- 1. External elements and attainment of the higher education QNV 2030 (Q1–Q5).
- 2. Internal elements and the higher education QNV 2030 (Q6–Q8).
- 3. Culture and the higher education QNV 2030 (Q9).
- 4. Proposed cyberspace culture framework (Focus group Q1).

Each theme was evaluated in the NVivo application in order to extract useful insights to address the research questions, as detailed in the following sections.

However, the Researcher began by generating a hierarchy chart to gain insight into the distribution of themes in terms of their weight in the research. Figure 6-17 displays the overall visualisation of theme distribution:



Figure 6-17: Sunburst illustrating distribution of research themes. Source: The Researcher

Figure 6-17 above illustrates that the responses more heavily emphasised the external rather than internal elements and culture. A treemap of the responses was also generated, as detailed in Figure 6-18, which further illustrates the actual elements to which the participants mostly referred:



Figure 6-18: Treemap illustrating distribution of research themes.

Source: The Researcher

Based on the size of the different tree nodes in Figure 6-18, the obtained findings also reiterate that the respondents placed greater emphasis on leadership, information security policy, and technological advancement compared to other aspects.

Furthermore, to identify the underlying factors leading to this distribution, the Researcher generated coding references and determined the number of coded items for each research theme,

as displayed in Table 6-3. Table 6-3 displays only some of the coding references per theme. However, in Appendix D, the full table details are displayed.

Theme	Sub-theme	Main code
		Leadership (Sub-Section 4.7.2.1.1)
External elements (Sub-Section 4.7.2)	Role of the QNV 2030 directive (Sub-Section 4.7.2.3)	Information security policy and regulation (Sub-Section 4.7.2.1.2)
	Role of technological advancement (Sub-Section 4.7.2.2)	Technological advancement (Sub- Section 4.7.2.2)
	Role of the QNV 2030 (Sub- Section 4.7.2.3)	The QNV 2030 framework (Sub- Section 4.7.2.3.1)
		Leadership and management strategies (Sub-Section 4.7.1.1)
Internal elements (Sub-Section 4.7.1)		Security policy (Sub-Section 4.7.1.2)
		Security technical and best practices
		Dimensions of culture
Role of culture		(Sub Section 4.7.3.2)
(Sub-Section 4.7.3)		Levels of culture (Sub Section 4.7.2.1)

Table 6-3: Number of coding references and items coded per theme.

Source: The Researcher

The evaluation of the results in Table 6-3 indicates that the external elements possessed the highest number of coded items, including economic factors (13), external pressure (16), compliance and

advanced security (4), environmental uncertainty (12), political factors (10), learning and teaching (9), management (4), and implementation framework (17). This was followed by internal elements, including accountability (5), risk management (1), security awareness (3), continuous learning (8), attitudes and behaviour (1), and security skills and knowledge (1). Culture produced the least number of coded factors (1). These results align with the previously generated sunburst (Figure 6-17) and treemap (Figure 6-18) results, which demonstrated similar insights.

In the NVivo application, the Researcher created and nested the nodes in a hierarchical format below the various main codes, sub-themes, and themes. Figure 6-17 below displays the hierarchical organisation of all research themes and codes in the NVivo application for the four themes:



Figure 6-17: Organisation of research themes and codes.

Source: The Researcher

From Figure 6-17, the categories of different themes can be drawn; for instance, under Theme 1, the three sub-themes of the research are identified—namely, the role of the Vision 2030 directive, the role of technological advancement, and the role of the QNV 2030. Under each sub-theme, individual factors are further outlined. The established hierarchy mirrors the proposed cyberspace culture framework developed in the research (Figure 4-7) and facilitates the analysis as findings identified under each code were easily compared against insights obtained from the review of previous literature studies (Chapters 2–4).

In the following sections, a detailed thematic analysis is undertaken in order to obtain useful insights for addressing the research questions.

6.5.1 Theme 1: External Elements and Attainment of the Higher Education 2030 Vision (Q1–Q5, Appendix A)

The first theme involved identifying how external elements influence attaining the QNV 2030 in higher education (Sub-Section 4.7.2). To this end, three main external elements were specified by the proposed cyberspace culture framework (Figure 4-7): a) the QNV 2030 directive (Sub-Section 4.7.2.1), b) technological advancement (Sub-Section 4.7.2.2), and c) the QNV 2030 (Sub-Section 4.7.2.3). The elements were also split further into individual factors, as detailed in the proposed cyberspace culture framework (Chapter 4, Section 4.7). For clarity purposes, Table 6-3 displays the various categories of the external elements (considered sub-themes) that were outlined:

Codes	Main Code	Sub-Themes	Theme
Economic factors (Sub-Section 4.7.2.1.1.1) External pressure (Sub-Section 4.7.2.1.1.2)	Leadership (Sub- Section 4.7.2.1.1)	Role of the QNV 2030 directive	External elements
Compliance and advanced security (Sub-Section 4.7.2.1.2.1) Environmental uncertainty (Sub- Section 4.7.2.1.2.2)	Information security policy and regulation (Sub-Section 4.7.2.1.2)	(Sub-Section 4.7.2.1)	

Table 6-3: External elements' codes, sub-themes, and themes.

Political factors (Sub-Section			(Sub-
4.7.2.1.2.3)			Section
Learning, teaching (Sub-Section		Role of	4.7.2)
4.7.2.2.1)	Technological	technological	
	advancement (Sub-	advancement	
Management (Sub-Section 4.7.2.2.2)	Section 4.7.2.2)	(Sub-Section	
		4.7.2.2)	
Implementation framework (Sub	The QNV 2030	Role of the QNV	
Section 4.7.2.2.1)	framework (Sub-	2030 (Sub-Section	
Secuol 4.7.2.3.1)	Section 4.7.2.3.1)	4.7.2.3)	

Source: The Researcher

The next process involved assigning each of the participants' responses to the various codes in order to organise the responses thematically. For instance, Interview Question 3 (Appendix A, Q3) assessed the participants' views concerning the role of advanced technologies for achieving the higher education 2030 vision in terms of management on the one hand and learning and teaching on the other. In Nvivo 12, the Researcher organised responses for this particular question from all participants under the respective codes—namely, management and both learning and teaching. In this manner, insights from all participants could be easily drawn from one location.

6.5.1.1 The QNV 2030 Directive – Leadership

In the first interview question, participants were asked to discuss how the government's (Vision 2030 directive) leadership in terms of strategy formulation influenced the attainment of the higher education 2030 vision. Literature suggests that the government represents a key stakeholder in cyberspace, alongside public authorities, commercial enterprises, and individuals (Canton, 2012). As a stakeholder, the government plays a role in the overall decision-making process in cyberspace, the development of rules governing the usage of cyberspace, and the management of security issues (Barzilai, 2004). Examining how the government's decisions and strategies influence the attainment of the higher education 2030 vision was therefore important in this research.

First, however, it was important to identify the contexts in which government leadership was discussed by the participants. Table 6-4 presents the word frequency of terms used in reference to leadership (word count of at least 100):

Word	Count	Weighted Percentage
Government	359	1.02%
University	332	0.94%
Technology	241	0.68%
Leadership	169	0.48%
Cyberspace	147	0.42%
Management	132	0.37%
Institution	120	0.34%
Universities	111	0.31%
Institutions	101	0.29%
Organisations	101	0.29%

Table 6-4: Word frequency and weighted percentage of leadership words.

Source: The Researcher

Analysis of the results in Table 6-4 indicates that leadership was discussed in the context of the government (1.02%), university (0.94%), technology (0.68%), cyberspace (0.42%), management (0.37%), institutions (0.34%), and organisations (0.29%). The findings emphasise that interview participants, at the least, identified the interaction of government leadership with different stakeholders such as universities and institutions in strategy formulation.

Further evaluation of the responses revealed that the government was considered an oversight authority that plays an important role in influencing the implementation of different strategies, regulations, and policies. In this light, four participants (22.2%) cited that the government is influential in developing implementation plans and policies that will be adopted by institutions. More specifically, Respondent 1 argued that the government is responsible for formulating policy, providing funding, and enhancing collaboration among stakeholders:
'The government is pushing all the regulatory, policy, and strategy and funding mechanisms and making the stakeholders do this kind of thing. The role of the government is very important'. (Respondent 1)

Respondent 9 added that the government, as a leader, provides implementation directions for the 2030 vision:

'It is extremely important; it's vital. I think the leadership should really give very explicit directions on how to execute the, 2030 vision and the implementation of Vision 2030. It is also important that it should be on top, not bottom-down'. (Respondent 9)

Analysis of the responses highlights the fact that the government's strategies are influential for achieving the higher education vision through policy development, funding, and guiding the implementation of the 2030 vision. The findings confirm previous literature in Sub-Section 2.4.1 suggesting that Qatar's government developed the QNV 2030 as a strategy to modernise the country by ensuring all sectors (economic, social, environmental, and human) are well developed. Policy and funding are emphasised in the QNV 2030's attainment.

Respondent 3 further viewed the government as playing an integral role as a source of direction for the different institutions, such as by providing guidelines regarding the adoption of e-learning:

'Since we have some ministries and the government role, both have guidelines for how they govern the systems, so I think there is a government role. So, they can set the direction, and then this will be led by the institutions'. (Respondent 3)

Respondent 10 also highlighted a similar role of the government in education development. In the respondent's words,

'Well, actually, I do believe that they have a major role in changing or implementing the Vision 2030, especially in education. I think educational decisions must be taken from, originally speaking, the government'. (Respondent 10)

The findings from these two participants further confirm previous literature in Sub-Section 2.4.2.2, which details the government's strategies for education development in the QNV 2030. The identified strategies include enhancing scientific research (Gasser and Drolshammer, 2015), boosting learning and teaching (Talebian *et al.*, 2014), and enhancing innovation and knowledge-searching processes (Zhou *et al.*, 2019). The findings also reiterate literature in Sub-Section 3.2.2.1, where quadrant II analysis emphasised cyberspace as a tool of learning and the government's strategies for guiding its use for that purpose.

Next, the role of the government's strategy was further investigated regarding its influence on economic factors and external pressure.

6.5.1.1.1 Economic Factors

In the first part of Interview Question 1 (Appendix A), participants were asked how the government's strategy concerning economic issues would likely affect the implementation of the higher education 2030 vision. Figure 6-18 displays the distribution of responses that were obtained:



Figure 6-18: Participants' views on the impact of economic factors. Source: The Researcher

Based on an evaluation of the responses, 14 participants (78%) agreed that funding was highly important while four respondents (22%) disagreed with the assertion. Among the participants who agreed, the role of funding in enhancing the implementation of the required education infrastructure was noted as the first aspect. Respondent 1 noted that the majority of higher education institutions in Qatar are fully funded by the government, which is mandated to support them financially:

'Of course, there is no education without funding, as it is critical. It requires massive Infrastructure If you take, for example, Qatar university, CCQ, CNAQ, university of Calgary, those are fully funded, and also all the QF universities'. (Respondent 1)

The findings confirm the literature in Sub-Section 3.2.2.1 stating that infrastructure invested in cyberspace as a tool of learning mediates collaborative work outside classroom environments through advanced tools and technologies (Fabos and Young, 1999; McLellan, 1998; Reynolds, 2002; Riel, 1995). The findings also resonate with work by researchers in Sub-Section 2.5.2.1, such as Johnson et al. (2018), who reported that online and video-based teaching playgrounds enable prospective teachers to prepare effectively and enhance teaching in online environments.

Likewise, Xu and Dai (2019) reported that adopting situational teaching positively affects the enhancement of online learning. As such, the findings indicate that cyberspace is increasingly being leveraged to improve learning and teaching practices.

Second, according to the findings, participants believe that a mutual relationship exists between the economy and education whereby funding education serves as an engine for building the economy. As such, government funding is observed to increase the existing workforce's capacity. Respondent 4 further argued that government funding plays a role in increasing the economic viability of higher education institutions, such as in creating more institutions:

'Leadership strategies to increase the economic viability of higher education institutions for example, creating more courses and expanding institutions—yeah, I think so'. (Respondent 4)

Third, Respondent 6 reported that sustainability in the educational sector is also linked to funding from the government whereby the provision of all required resources, such as land to build campuses, would benefit both current and future generations:

'We want the people who live in Qatar and their children to be educated here in different universities at different levels, and according to their skills'. (Respondent 6)

In summary, evaluating the responses indicates that the majority of participants (78%) agreed that the government's strategy towards funding education is vital for establishing cyberspace culture, as funding facilitates the acquisition of infrastructure, the setup of institutions, the improvement of technology and courses, and the sustainability of higher education. Similar findings have been reported in the literature reviewed in Sub-Section 2.4.2.2, where Qatar's Vision, 2030 focusses on funding education to improve its stature to a world-class level (Talebian *et al.*, 2014; Gasser and Drolshammer, 2015; Dunn and Kennedy, 2019). However, approximately 22% disagreed that funding is not a major requirement for advancing higher education institutions.

6.5.1.1.2 External Pressure

In the second part of Interview Question 1 (Appendix A), participants were also asked how the government's strategy directive, which culminates in external pressure, is likely to affect the implementation of the higher education 2030 vision. The government's external pressure was argued to arise from its role as a stakeholder in funding the development of the education sector and in formulating policies that guide higher education institutions (Sub-Section 4.7.2.1.1.2). Literature additionally confirms that the government plays an integral role in both funding the sector and fostering partnerships with foreign investors, thus highlighting a need to ensure higher education institutions are accountable (Cantwell, Coate and King, 2018).



Figure 6-19 below displays the distribution of the views obtained from the respondents:

Figure 6-19: Participants' views on the impact of external pressure. Source: The Researcher

From the evaluation of Figure 6-19, approximately nine of the participants (50%) argued that external pressure is highly important. In support of these positive findings, Respondent 6 agreed that current directives from the government exert pressure on the operations of institutions, primarily because the institutions report to the government. As a result, the institutions are required to consistently evaluate whether they are meeting the demands set by the government. Specifically, Respondent 6 asserted the following:

'Yeah, there is pressure. There is all of this, and we report to a higher authority all the time in the governments about our achievements, and they also check what we are doing'. (Respondent 6).

Respondent 9, meanwhile, explained that the Ministry of Higher Education is the source of pressure:

'The leadership pressure by the education ministry for higher education, in my opinion, you can do it from the other way around'. (Respondent 9).

However, an additional 50% of respondents further highlighted that external pressure does not exist in Qatar higher education. For instance, Respondent 7 revealed that external pressure from the government remains lacking:

'For now, we do not see the external pressure. The only type of pressure we see is internal, so maybe because, you know, we are not in higher administration, maybe the pressure is placed on the president of the university and his vice president'. (Respondent 7)

Supporting this view, Respondent 12 also denied the existence of external pressure:

'Are they actually bringing any pressure on them or not? I don't see it. To be honest, I have not seen this. It is not happening. But even if it does happen, I think that the university, the Qatar University administration, may or may not take it. But I have not seen anything so far'. (Respondent 12)

Among the nine participants (50%) who agreed, three of them (16%) argued that, although pressure is important, it does not exist in Qatar higher education institutions. For instance, Respondent 18 cited that, though pressure exists, it is positive and beneficial for the institutions, as it enables them to achieve more goals:

'I mean, if we consider strategic goals that are set by the country as pressure, I will say it is positive pressure'. (Respondent 18)

Respondent 16 agreed with this assertion and reported that a positive rapport exists between higher education institutions and the Ministry of Higher Education, and as a result, higher education institutions can more flexibly achieve their goals:

'They (the ministry) really allowed much more flexibility for Qatar university for the higher education component. So, I do not think we will have any pressure from the ministry to achieve anything'. (Respondent 16)

Therefore, regarding how external pressure influences the achievement of cyberspace culture, conflicting views were reported: only 34% of the participants agreed that there is external pressure to meet standards set by the Ministry whereas 50% denied its existence. However, 16% of the respondents cited that, although pressure is important, it does not exist in Qatar higher education institutions.

Respondents who denied the existence of pressure (11.1%) argued that universities are required to pressure themselves to achieve higher education objectives. The literature reviewed in Section 2.7 further highlights the need to develop transparent and unbiased rules to govern cyberspace (Goi, 2009); subsequently, this also emphasises the need for accountability among actors in cyberspace, thus leading to reports of external pressure.

6.5.1.2 The QNV 2030 Directive – Information Security Policy

In Interview Question 2 (Appendix A), the research investigated how, in terms of information security policy, the QNV 2030 directive influences the attainment of the higher education 2030 vision. The literature reviewed in Sub-Section 2.2.2.2 conceptualises cybersecurity as the protection of data, networks, and systems (Lindsay, 2015; Dorsey *et al.*, 2017). Additionally, studies reviewed in Section 2.3 underscore the challenges of insecurity in cyberspace's evolution (Galinec *et al.*, 2017; Deloitte, 2017). Therefore, the synthesis of findings in this section outlined the various ways in which information security policy from the government influences the achievement of the higher education 2030 vision.

First, findings revealed a consensus that information security policy is important not only in higher education, but also in other private sectors. Respondent 9, for instance, explicated the following:

'This is very important and will not only affect higher education, but also, it will affect everything ... health, private sector, government sector'. (Respondent 9)

In higher education in particular, Respondent 8 highlighted a need for standard policies that govern the entire sector to improve education for the whole country. Respondent 8 further described a need for different institutions to adopt similar information security policies. This highlights the notion of fragmented leadership in Qatar's higher education sector, which has subsequently led to individual differences in policies and procedures in the institutions. Gaps were subsequently identified in the institutions, as described by Respondent 8:

'If you have your own policy and procedures as one institution, what about the others? ... To be honest with you, we have a gap here ... we have to manage this gap between the government and the ministry of education, and also other kinds of institutions'. (Respondent 8)

Such findings from Respondent 8 regarding the fragmented nature of higher education and cybersecurity confirm the literature in Sub-Section 2.8.1, where gaps were identified in cyberspace security for both academic learning and private use (Brunn, 2014).

However, Respondent 1 noted that the government has made strides towards enhancing information security policies by establishing different committees and creating cybersecurity frameworks. The respondent argued that, over the years, information security policies have grown beyond their limited role in the IT department to affect other non-IT-related sectors, such as football and the World Cup. As a case in point, the Qatar 2022 Cybersecurity Framework was created in August, 2018 in preparation for the 2022 FIFA World Cup. The premise of this framework consisted of ensuring security for Qatar's inhabitants in cyberspace, as the tournament was expected to be digitally enabled and operated using innovations such as artificial intelligence,

the Internet of Things, and machine learning. Further analysis of the framework revealed that it specified how cybersecurity would be governed in terms of not only aspects such as endpoint, application, and network security, but also data protection, operations technology security monitoring, incident handling and recovery, and data privacy. The framework was argued to be developed in support of the larger 2030 QNV framework and, as such, indicated that the government was making progress in the institution of policies.

As guided by the proposed cyberspace culture framework (Figure 4-7), the Researcher further investigated how attainment of the higher education 2030 vision was influenced by three influential factors—namely, compliance and advanced security (Sub-Section 4.7.2.1.2.1), environmental uncertainty (Sub-Section 4.7.2.1.2.2), and political factors (Sub-Section 4.7.2.1.2.3).

6.5.1.2.1 Compliance and Advanced Security

In regard to how compliance and advanced security influence the attainment of the higher education 2030 vision, the perspectives obtained from the respondents are detailed in Figure 6-20:



Figure 6-20: Participants' views on compliance with advanced security. Source: The Researcher

According to the analysis, 14 respondents (78%) considered compliance and advanced security as highly important. However, 2 of the 14 participants (11%) cited that, although compliance is

important, it is not present in Qatar higher education. Likewise, four respondents (22%) argued that compliance is not important at all. However, the emergent theme from the responses was that the majority of participants vouched for compliance with advanced security. For instance, Respondent 2 argued that compliance enables higher education institutions to not only adhere to business practices in the international market during implementation of online systems such as networking and cloud computing, but also to adhere to the government's regulations. However, it was also argued that the regulations only affected how business was performed rather than the curriculum's delivery.

Respondent 2 asserted the following:

'But when we implement any online businesses, like networking or cloud, in this, we have to abide by the government regulations. This, we cannot do outside the space'. (Respondent 2)

Second, compliance with international guidelines and best practices was also emphasised to protect the international community and society. Therefore, institutions in Qatar must comply with established standards to transact with parties in foreign countries. The finding regarding compliance with international standards and protection of the international community additionally confirms the literature in Section 2.3, which emphasises that cyberspace represents a broad concept that transcends international boundaries and involves interaction with diverse community members.

Third, Respondent 12 argued that the government's oversight over institutions has also increased over time, and as such, institutions are required to demonstrate accountability by complying with set regulations:

'Now, we are seeing them (government) asking about our operations – for instance, what we are doing, and account for our actions. There are a lot of business requirements that they have been asking about'. (Respondent 12)

Additionally, findings indicated that, since the government provides support to higher education institutions in terms of funding, those institutions must comply with the directives issued by the government regarding information security. For instance, the privacy law introduced by the Qatar government requires institutions to manage the security of the data they collect. These findings were confirmed in the literature, whereby cybercrime laws were enacted in Qatar to preserve intellectual property (Foody *et al.*, 2017).

However, in the same vein, participants claimed that the government possesses a mandate to craft the policies carefully, as some institutions only comply with the regulations for the sake of, and not in consideration of, their value. The discussions by Respondents 2 and 12 emphasise insights from the literature reviewed in Section 2.7 regarding the openness of cyberspace culture and the need for guiding rules to direct actors in cyberspace.

6.5.1.2.2 Environmental Uncertainty and Political Factors

Although the two factors (environmental uncertainty and political factors) represent different constructs, the Researcher discussed them in one section due to the overlap identified from the analysis of the responses. Figure 6-21 below accordingly displays the respondents' views concerning environmental uncertainty and political factors:



Figure 6-21: Participants' views on environmental uncertainty and political factors. Source: The Researcher

Based on the evaluation of the findings in Figure 6-21, 10 of the respondents (56%) agreed that both environmental uncertainty and political factors are highly important for attaining the higher education 2030 vision. For instance, Respondent 2 cited that political factors such as the Qatar blockade where diplomatic and trade ties were cut between Qatar and UAE, Bahrain and Egypt, created uncertainty in the operational environment, as it led to disruption, thereby overlapping on the two factors:

'This blockade has changed the perspective of the country. As something that we did not anticipate, we did not plan for it. So, for instance, if this University QF decides to close its branches, it will have a huge impact on the higher education plan ... in general. So, the uncertainties, the unexpected, can change the whole dynamics'. (Respondent 2)

Respondent 14 further believes that the kind of leadership that has been instituted influences environmental uncertainty; changes in leadership, such as appointing a different minister of education, will subsequently affect the attainment of strategic objectives. As an explanation, it was argued that, by changing people (mostly senior leaders), visions and directions are also altered.

However, a counter-argument raised by eight respondents (44%) is that environmental uncertainty and political factors are not important for achieving the higher education 2030 vision. For instance, Respondent 14 highlighted that the change in leadership will not necessarily influence the achievement of strategic objectives and the set vision:

'I have at least witnessed the change of the upper management or leadership even in the country, which did not affect the vision. I believe it did not massively change the way things are going or under taking when it comes to the activities'. (Respondent 14)

Respondent 16 further added that, while disturbances in the operational environment, such as the Covid-19 pandemic, are not related in any way to the political leadership, they influence the attainment of the higher education vision:

'Lots of economic situations that we don't know what will happen ... will have a huge impact if it continues for a few months ... you will have problems with implementing your strategy, and so on'. (Respondent 16)

Nevertheless, evaluating the findings revealed a consensus in that environmental uncertainty is disadvantageous for the country and negatively influences the higher education sector's ability to achieve its vision. Findings on these two factors corroborate with the literature in Section 2.6, which emphasises the significant impact of politics on decision-making in cyberspace and its overall usage.

6.5.1.3 Technological Advancement

In Interview Question 3 (Appendix A), the Researcher investigated how advanced technology influenced the attainment of the higher education 2030 vision. Accordingly, two important aspects were examined: 1) the role of technology in learning and teaching and 2) the management of institutions.

6.5.1.3.1 Learning and Teaching

Figure 6-22 displays the participants' views regarding how technology for learning and teaching influences the attainment of the higher education 2030 vision:



Figure 6-22: Participants' views on technology for learning and teaching. Source: The Researcher

The evaluation of findings in Figure 6-22 revealed that 14 respondents (78%) identified technology as highly important while four participants (22%) argued that technology is not important for attaining the higher education vision. As the majority of the respondents agreed with the assertion, further analysis revealed the underlying factors that support this view; for instance, participants considered technology to be an enabler of learning, as it allows people to share knowledge with one another. This is confirmed by the literature reviewed in Sub-Section 2.5.2.1 wherein technology-mediated learning and teaching is more enhanced (Pugalee and Robinson, 1998). Respondent 1 specifically highlighted technology's influence on the development of blended learning, asserting the following:

'Taking that into the class room and outside the classroom absolutely, students learn within the classroom, and now, we are talking about blended learning. Allowing students to engage in the classroom, but also, they engage online'. (Respondent 1)

Additionally, findings suggest that technology facilitates establishing smart campuses that would utilise artificial intelligence to guide individuals within the school. For instance, new students could navigate to any room of their choosing simply by typing it into the smart campus system in their app, which would direct them there. Furthermore, Respondent 2 also identified technology as enabling efficient teaching operations, mentioning that exams could be conducted in a paperless manner and marked more quickly and efficiently:

'For example, the condition of processes, paperless. How many exams we can do without paper? We can put all our exams on the computer, and then marking will be quicker'. (Respondent 2)

Participants subsequently demonstrated agreement regarding the need to adopt modern technology in education as global trends move in a similar direction, and they also offered numerous opportunities. The findings are echoed in the literature in Sub-Section 2.5.2.1, where technology for learning and teaching was evaluated, as well as in Sub-Section 3.2.2.1, where cyberspace as a tool of learning was also reviewed. Furthermore, the findings resonate with research in Sub-

Section 2.4.2.2, where Bognar, Sablić, and Škugor (2018) reported that flipped learning also facilitates online discussions and consequently promotes the expression of thoughts and feelings in a friendly manner. Additionally, the findings mirror those by Alamri, Watson, and Watson (2020), who observed that blended learning environments facilitate personalised learning models in higher education, thereby supporting independent research.

However, challenges in adopting the technologies were also cited. For instance, Respondent 16 believes that there is resistance to technology and a lack of strategy to adopt it for learning and teaching purposes:

'We will have two types of resistance. We will have people who are not really capable of doing this, and they will find excuses why we should not do this rather than admitting we have an issue, letting us go along and improve ourselves, and moving on in this'. (Respondent 16)

Respondent 12 further emphasised this lack of strategy:

'The infrastructure can be, but it has to be driven by the business I have seen talks about it, but I have not seen anything in strategy. Now, back to the strategy itself ... about method of teaching, improving things'. (Respondent 12)

Based on the responses, challenges such as resistance to technology and the lack of strategy to improve learning and teaching hinder the overall success of technology.

6.5.1.3.2 Management of Institutions

Interview Question 4 (Appendix A) further examined technology's role in improving the management of higher education institutions. Figure 6-23 below displays the participants' views on this role:



Figure 6-23: Participants' views on technology for management. Source: The Researcher

According to the findings, 15 participants (83%) considered technology to be highly important for managing higher education institutions. Specifically, the reported findings suggest that technology is perceived to highly influence the management of higher education institutions by facilitating decision-making and managing resources. Such findings are confirmed by the literature in Sub-Section 2.5.2.2, where analytics and big data are employed to enhance decision-making (Liebowitz, 2016). Likewise, they mirror the findings by Luo (2019), who reported that using big data to manage student affairs is becoming a common trend in higher education and is advocated for improving normative management and leading to the adoption of scientific decision-making processes.

Respondent 1, meanwhile, cited that technology aids in making difficult decisions to streamline processes due to a decline in oil prices and the Covid-19 pandemic:

'We are bracing for major budget cuts now because of the corona virus and also the oil prices' decline. Institutions now need to make really tough decisions and streamline processes to be more efficient and effective; all that could be effectively enabled through technology'. (Respondent 1)

Second, the findings revealed that technology is also transforming administration in higher education institutions. As a result, current management bottlenecks are being effectively addressed. However, Respondent 12 argued that the business side of higher education institutions requires adopting such systems in administration:

'On the administrative level, they facilitate a lot of processes, make things a lot more secure, smoother ... everything should be better with technology'. (Respondent 12)

Third, Respondent 18 further believes that technology is transforming management by facilitating learning from previous experiences through the use of analytics and machine learning. In the literature, studies such as the one by Daniel (2014) have cited similar applications of analytics in evaluating student performance. Jha, Jha, and O'Brien (2019) have also reported that tracking student activity data using big data is also important for identifying students who are struggling to complete their course work, thereby leading to lower dropout rates:

'Nowadays, we are more focussed on what happened before. So, we are learning from the previous experience, how we were doing that, using the analytics, using machine learning in order to learn about the behaviours'. (Respondent 18)

The results indicate that technology enables higher education institutions to improve their processes by examining previous experiences via analytics and machine learning. However, three participants (17%) do not consider technology to be important for managing higher education institutions.

6.5.1.4 The QNV 2030 Implementation Framework

In Interview Question 5 (Appendix A), participants were asked how the QNV 2030 implementation framework affects achieving the higher education QNV 2030. In Sub-Section 2.4.1, literature on the QNV 2030 is detailed and its four pillars—namely, human, social, environmental, and economic—are outlined (Qatar National Vision, 2030, 2008). The particular impact of the QNV 2030's implementation framework on human development in terms of enhancement of higher education was examined in this section.

Table 6-6 below presents the word frequency of terms used in reference to the framework (word count of at least two):

Word	Count	Weighted Percentage
Universities	6	0.35%
Implementation	5	0.29%
Communication	3	0.17%
Capabilities	2	0.12%
Consideration	2	0.12%
Institutions	2	0.12%
Sustainability	2	0.12%

Table 6-6: Word frequency and weighted percentage of framework.

Source: The Researcher

The results indicate that the term 'framework' was employed in the context of implementation (0.29%), communication (0.17%), capabilities (0.12%), institutions (0.12%), and sustainability (0.12%).

Figure 6-24 below further displays participants' perceptions regarding the role of the implementation framework for attaining the higher education 2030 vision:



Figure 6-24: Participants' views on the role of the implementation framework. Source: The Researcher

The findings in Figure 6-24 illustrate that all 18 participants (100%) considered the implementation framework to be highly important for achieving the 2030 vision. In support of this finding, Respondent 1 believes that the QNV 2030 has helped shape the future for the development of a knowledge-based economy in the country whereby, instead of relying on hydrocarbons, technology would be employed to build the population's innovative capacities. In this respect, the implementation framework provides a guideline facilitating the uptake of modern technology. The findings are confirmed by the literature in Sub-Section 2.4.2.2 stating that Qatar's education system would modernise the economy by advancing scientific research and boosting learning. To this end, Respondent 1 asserted the following:

'Technology that can discover what else is beneath the sand in Qatar. Technology of, you know, tapping into the ingenuity and the capabilities that are in people. So, bring that out in terms of innovation that people can contribute to this'. (Respondent 1)

However, eight participants (44%) stated that, although the QNV 2030 framework is highly important, it does not yet exist; that is, the participants argued that the QNV 2030 framework is not effective, as it does not possess any definite milestones or key performance indicators (KPIs)

to assess what has been achieved overall. Respondent 16 argued that the QNV 2030 is similar to the American dream in that it appears promising only on paper, but not in terms of implementation. Recommendations were subsequently provided to improve the document further, such as by measuring progress made after four years. As Respondent 16 claimed,

'For sure, yeah, I think the Vision 2030 obviously is visionary, but each four years, the government will come up with a strategy for those. And at the end of each year, they measure the KPIs, and the end of the four years, they have a report about overall what has been achieved'. (Respondent 16)

The findings thus indicate that there is a need to sub-divide the QNV 2030 into milestones that can be easily measured.

6.5.2 Theme 2: Internal Elements and the Higher Education Vision, 2030 (Q6–Q8)

The second theme, assessed by Interview Questions 6, 7, and 8 (Appendix A), focussed on identifying the relationships between the internal elements within higher education institutions and the attainment of the higher education QNV 2030. Table 6-3 below displays the codes associated with the internal elements:

Codes	Main Code	Theme
Accountability (Sub-Section 4.7.1.1.1) Risk management (Sub-Section 4.7.1.1.2)	Leadership and management strategies (Sub-Section 4.7.1.1)	Internal
Security awareness (Sub-Section 4.7.1.2.1) Continuous learning (Sub-Section 4.7.1.2.2) Attitudes and behaviour (Sub-Section 4.7.1.2.3)	Security policy (Sub-Section 4.7.1.2)	elements (Sub-Section 4.7.1)

Table 6-2:	Internal	elements	codes	and	themes
1 4010 0 2.	Internet	erenter en	00000	will co	ci c

Security skills and knowledge	Security technical and best practices	
(Sub-Section 4.7.1.3.1)	(Sub-Section 4.7.1.3)	

Source: The Researcher

6.5.2.1 Internal Leadership and Management

Interview Question 6 (Appendix A) investigated how internal leadership and management strategies developed in higher education institutions influence the achievement of the higher education 2030 vision. Accordingly, two aspects were investigated: accountability and risk management. Findings from the literature in Sub-Section 4.7.1.1 previously revealed that leaders in different types of organisations must be accountable for both protecting the ethics of the business (Ghanem and Castelli, 2019) and implementing strategies that eliminate risks for the business (Fleming, 2004).

6.5.2.1.1 Accountability and Risk Management

First, a word frequency table for terms used in reference to accountability was developed, as detailed in Table 6-7 (word count at least four):

Word	Count	Weighted Percentage
Governance	9	3.54%
Educational	4	1.57%
Leadership	4	1.57%
People	4	1.57%
Think	4	1.57%

Table 6-7: Word frequency and weighted percentage of accountability.

Source: The Researcher

According to the results, 'accountability' was utilised in the context of the following words: governance (3.54%), educational (1.57%), leadership (1.57%), people (1.57%), and think (1.57%).

Table 6-8 further presents the word frequency of terms used in reference to 'risk' (word count at least two):

Table 6-8: Word frequency and weighted percentage of the word 'risk'.

Word	Count	Weighted Percentage
Decentralised	3	12.50%
Centralised	2	8.33%
Think	2	8.33%

Source: The Researcher

The results indicate that 'framework' was employed in the context of the words 'decentralised' (12.50%), 'centralised' (8.33%), and 'think' (8.33%).

Figure 6-25 further displays the distribution of the participants' views regarding the role of accountability and risk management for attaining the higher education 2030 vision:



Figure 6-25: Participants' views on the role of accountability and risk management. Source: The Researcher

Evaluating the findings in Figure 6-25 revealed that all 18 participants (100%) considered accountability and risk management to be highly important. However, five respondents (28%) argued that, although risk management and accountability are important, they do not exist in Qatar higher education institutions. Based on a detailed analysis of the responses, respondents believe that the leaders in the organisations possess a responsibility to be accountable for their employees' actions as well as to drive the implementation of different objectives. Respondent 15, for instance,

observed that top management support is essential for providing guidance and support to workers at lower levels:

'The leader is responsible for giving clear guidelines for the employees'. (Respondent 15)

A similar view was also reiterated by Respondent 11:

'If they are not trained, they should be trained. If they are not aware, there should be kind of campaigns and seminars and workshops to make them more aware. If they need resources, then they need to be supported by resources'. (Respondent 11)

The discussion between Respondents 15 and 11 concerned the fact that leaders play a role in guiding employees and must accordingly be well trained and provided with the resources necessary to execute the different objectives. The findings confirm the literature in Sub-Section 4.3.2, where Schein (2013) argued that leaders guide their employees' actions.

Second, in regard to risk management, Respondent 4 highlighted that the strategies from leadership are integral for eliminating risks by ensuring that things are done properly and identifying aspects that need to be corrected in order to meet expectations. Ensuring efficiency in higher education institutions would enhance the attainment of the higher education QNV 2030. Respondent 2 further argued that adopting decentralised systems is important for minimising the effects of risks. In the current scenario, the respondent cited that higher education institutions feature too much centralisation, asserting that

'We live in a very much centralised government, and institutions are also all centralised. For example, when you come to the Qatar University, the president has not only the ultimate power'. (Respondent 2)

A third finding, highlighted by Respondent 3, is that governance and risk management in higher education institutions represent a collaborative responsibility between different departments and should not be left to only one group: 'It is not a job of one, let us say, role or one department, or it is not their job. So, I think it is a joint role that everyone should think of'. (Respondent 3)

The discussion between Respondents 3 and 2 regarding the need for collaboration among leaders echoed the literature in Sub-Section 2.8.1 indicating that trust and co-operation are required between different entities to establish a beneficial working relationship (Silver, 2016).

Fourth, it was also reported that a clear implementation plan is crucial for managing any changes arising from a change in leadership and management. Respondent 6 vouched for mature implementation frameworks to facilitate the achievement of strategic goals. Such frameworks would outline the necessary KPIs that would highlight progress made.

As Respondent 6 stated,

'Therefore, you need to have someone at the top who has a very good vision and like this to establish, and the continuity is very important'. (Respondent 6)

Implementation metrics and models from literature were previously reviewed in Sub-Section 2.8.3, where their value for attaining strategic goals was reported.

Nevertheless, the fifth finding revealed that considerable emphasis is placed on skilled, effective, and competent leadership. Additionally, some participants also highlighted the importance of having young leaders aged 30 to 50 years to move the institutions forward.

6.5.2.2 Security Technical and Best Practices

Interview Question 7 (Appendix A) investigated how security technical and best practices influence the attainment of the higher education 2030 vision. The quadrant III analysis of the literature in Sub-Section 3.2.4 revealed that aspects such as learning and content creation in cyberspace are guided by best practices on the international platform. The synthesis of findings in this section yields important insights regarding how adopting security technical and best practices influence the achievement of the higher education 2030 vision. A word frequency table was

subsequently developed to identify terms used in reference to best practices (word count of at least seven). Table 6-9 details the results below:

Word	Count	Weighted Percentage
Practices	21	2.67%
Culture	9	1.15%
Right	9	1.15%
University	8	1.02%
Awareness	7	0.89%
Example	7	0.89%
Policies	7	0.89%
Practice	7	0.89%
World	7	0.89%

Table 6-9: Word frequency and weighted percentage of best practices.

Source: The Researcher

The results indicate that the word 'framework' was employed in the context of practices (2.67%), culture (1.15%), right (1.15%), university (1.02%), awareness (0.89%), example (0.89%), policies (0.89%), practice (0.89%), and world (0.89%).

The findings underscore the importance of best practices in Qatar's current social-cultural context when adopting them from other international countries. For example, Respondent 2 cited that a professor riding a bicycle in Qatar would be unacceptable despite being accepted in regions such as Japan. As such, for best practices to work well within the country's higher education sector, alignment and further modification are required.

As Respondent 2 explained,

'If any system fails here, it is not because it's bad, but because it does not fit and align with our social norms. So, the best thing to do is to take inspiration and see how we can modify it so that it best fits according to our acceptable conditions'. (Respondent 2) Respondent 4 shared a similar view, arguing that it is not sufficient to simply apply the best practices in an institution; rather, the proper types of practices must be adopted to achieve set goals. Additionally, the respondent also mentioned the need for guidelines and procedures to implement best practices:

'We need to have guidelines, handmade policies and procedures, because this is new, the culture is new, and in order to implement it the right way, we need to have some support'. (Respondent 4)

The discussion between Respondents 2 and 4 underscored the need for not only adopting the proper types of best practices, but also aligning them with cultural and social contexts to guarantee their success.

In contrast, Respondent 7 argued that best practices benchmarked from other countries represent their own method of achieving goals. As such, there is a need for Qatari institutions to develop their own best practices that consider their budgets, initial goals, and operational goals. However, this view was critiqued by Respondent 6, who pointed out the need to build practices from what others have done as opposed to beginning from scratch:

'You do not start from scratch; you look for best practices, you look for other experience in the same area in Qatar, or in the regions, and you build on that one with what you need in your institute'. (Respondent 6)

The aforementioned view held by Respondent 6 has been previously reported in the literature; specifically, reliance on third-party best practices in higher learning institutions was emphasised in Sub-Section 3.2.4. Moreover, previous findings suggest that the use of the practices is motivated by the inherent security and innovative capabilities (Gregg *et al.*, 2017). However, from the discussion, new findings on aligning best practices with cultural contexts have also been reported.

207

Further review revealed the various processes that institutions employ to create best practices, such as searching for current best practices and finding ways to customise them for a university in research centres. Respondent 9, for instance, highlighted the role played by research organisations and the ICT sector in developing the best practices employed by higher education institutions:

'Since I joined the university, we always searched for the best practice, and we check how this can be adapted and how can we customise it to be implemented here at the university'.

'I believe that country policy, including the cyberspace of Qatar, I believe it's not the university's prime job or business, but it's the ICT sector's'. (Respondent 9)

From the reported findings, it was observed that research institutions and the ICT sector play an integral role in the development of best practices adopted in higher education institutions.

6.5.2.3 Security Policy

Interview Question 8 (Appendix A) investigated the role that internal security policies implemented in higher education institutions play in the attainment of the higher education 2030 vision. To this end, three aspects were considered: attitudes and behaviour, continuous learning, and security awareness.

6.5.2.3.1 Attitudes and Behaviour

Figure 6-26 below displays the respondents' views regarding security skills and knowledge:



Figure 6-26: Participants' views on security skills and knowledge.

Source: The Researcher

Figure 6-26 indicated that 16 respondents (89%) considered security skills and knowledge to be highly important, and thus influential to employee behaviour and attitudes. For instance, Respondent 3 cited that, to effect behaviour change, regulations and policies should be created to serve as guidelines. Findings from the literature reviewed in Sub-Section 3.2.1.2 likewise revealed that cyberspace represents a unique cultural concept that influences the experiences and interactions of individuals who inhabit it, and guidelines are consequently required to protect these interactions.

As Respondent 3 asserted,

'I think the policies and procedures will make it clear to everyone what is going on and what they need to follow'. (Respondent 3)

A similar view was also reported by Respondent 13, who emphasised ensuring that security policies remain positive, clear, and easily understood. The argument was that having positive regulations will influence employees' attitudes:

'Of course, yes, it is very important if you have the right policy and the right procedure, and that means that you are on the right track'. (Respondent 13)

The discussions between Respondents 13 and 3 underscored the need for clear security policies to guide employees' behaviour within an institution. However, Respondent 7 argued that their creation is deterred by the bureaucracy involved in making changes to the existing policies, as they are not dynamic:

'Our problem is we have this tornado of committees on top of committees on top of committees. As a result, there is a lot of routine, and in order to change something, you need a lot of approvals I think is really hindering us from having the dynamic type of regulations and policies'. (Respondent 7)

Additionally, Respondent 18 argued that culture influences employees' attitudes regarding the adoption of different rules and regulations. Furthermore, the respondents cited that some cultures more readily adhere to set regulations than others, thus indicating a need to contextualise rules and policies relative to Qatar's culture.

As Respondent 18 explicated,

'I mean, you can never create a set of rules that are applicable in any organisation anywhere and so generic that it affects everyone'. (Respondent 18)

Respondent 18's claim that culture influences employees' attitudes regarding rules and regulations echoes previous literature in Sub-Section 2.2.4, where culture was observed to influence individuals, groups, and entire societies (Eliot, 2014; Spencer-Oatey, 2008). As such, the need remains for contextualisation of policies and rules adopted from other types of organisations to achieve success within a particular firm. However, two participants (11%) considered security skills and knowledge to be unimportant, and thus not influential on employees' attitudes and behaviour.

6.5.2.3.2 Continuous Learning and Security Awareness

Figure 6-27 displays the findings obtained regarding the second factor—namely, continuous learning and security awareness:



Figure 6-27: Participants' views on security awareness and continuous learning.

Source: The Researcher

From the reported findings, it was observed that all 18 participants (100%) considered continuous learning and security awareness to be highly important for achieving the higher education 2030 vision. In support of these findings, Respondent 3 cited that continuous learning is important for improving the efficiency of higher education institutions, as it is iterative in nature. Therefore, because institutions constantly review their processes, this eventually leads to better results. Additionally, the findings from the literature in Sub-Section 3.2.5 regarding the quadrant IV analysis previously revealed that digital technologies are dynamic in nature, resulting in a need for educators and teachers to develop dynamic identities that align with the changing technology (Martin-Albo and Gregorio-Godeo, 2017). Such findings emphasise continuous learning for the different actors in cyberspace.

As Respondent 3 asserted,

'Kind of an iterative, you revisit what you are doing, and then, if this needs a change, then you do the change and continue. So, I think this will lead to a better result'. (Respondent 3)

Respondent 18 added that continuous learning is important due to the constant dynamic changes that occur in the cybersecurity world, which require employees to remain abreast of new changes and developments. In the same vein, it was also mentioned that governments and institutions are required to create dynamic policies that are adaptable to changes in either the economy or the environment. Likewise, the findings from the literature in Sub-Section 2.2.2 previously highlighted that cyberthreats are constantly being advanced, meaning employees must continuously keep abreast of strategies to mitigate them.

As Respondent 18 stated,

'Setting rules or policies that are fixed is the fastest way to failure ... when you do procedures and processes, you did them with your own experience with what happened to

you and what is the best that you can imagine. But then, things happen in between. Some special cases happen, some economic changes happen, so you need to adapt to these things'. (Respondent 18)

The implication of the findings is that increased emphasis is placed on consistent adaptation and improvement of policies. This stems from the fact that the cybersecurity world experiences dynamic changes and, as a result, employees must remain up to date regarding new developments.

Regarding security awareness, participants linked this to continuous learning and the need to consistently keep one another abreast of the changes taking place in the market. In the higher education context, this suggests that employees are required to keep updating their knowledge on security aspects in order to facilitate delivery of the higher education QNV 2030.

6.5.3 Theme 3: Culture and the Higher Education QNV 2030 (Q9 Appendix A)

The third theme shifted its focus towards examining how culture influences attaining the higher education 2030 vision. In Interview Question 9 (Appendix A), respondents were accordingly asked to describe how culture influences the attainment of the vision. Table 6-10 displays the various codes associated with this theme:

Codes	Main Code	Theme
Culture dimensions (Sub-Section	Dimensions of culture	Pole of
4.7.3.2)	(Sub-Section 4.7.3.2)	culture
Values		(Sub-Section
Visible processes	Levels of culture (Sub Section $4.7.3.1$)	(500 Section
Attitudes and behaviour	Levels of culture (Sub-Section 4.7.5.1)	4.7.3)
(Sub-Section 4.7.3.1)		

Table 6-10: Culture codes and themes.

Source: The Researcher

6.5.3.1 Role of Culture in General

Although Table 6-10 displays two main culture categories (levels and dimensions), the participants discussed the two aspects together. First, a word frequency table of terms utilised in reference to culture was developed (word count of at least, 25), as presented in Table 6-11:

Word	Count	Weighted Percentage
Culture	57	2.96%
People	53	2.75%
Think	33	1.71%
Change	27	1.40%
Students	26	1.35%

Table 6-11: Word frequency and weighted percentage of culture.

Source: The Researcher

The obtained results indicate that 'framework' was used in the context of the following words: culture (2.96%), people (2.75%), think (1.71%), change (1.40%), and students (0.12%). These results suggest that culture is affiliated with individuals, change, and students.

Figure 6-28 displays the participants' views regarding the role of cultural influence for the attainment of the higher education 2030 vision:



Figure 6-28: Participants' views on the role of cultural influence.

Source: The Researcher

From the analysis of Figure 6-28, only three participants (17%) disregarded culture's influence on attaining the higher education 2030 vision. However, the majority of them—15 participants (83%)—considered culture to be highly important. For instance, as Respondent 6 claimed,

'Yeah, I think the culture of the people is a very important component'. (Respondent 6)

Respondent 3 held a similar view regarding culture and leading change:

'Yes, the culture has a big importance, actually, because at the end, they are leading the change'. (Respondent 3)

Respondent 5 added that culture plays an influential role in attaining success:

'Culture is one of the most important factors that could either drive the success or failure of the achievements or goals'. (Respondent 5)

The discussion between Respondents 6, 3, and 5 is confirmed in the literature in Sub-Section 2.2.5, where cyberspace culture facilitates knowledge construction (Haarmann, 2007) and advanced use of technology (Zeng *et al.*, 2016). However, approximately 17% of the respondents did not agree with the views, stating that culture does not exert any influence on the achievement of the higher education 2030 vision.

Respondent 3 further believed that the culture of individuals in an organisation must be aligned with changes to be introduced. For this reason, the respondent argued that, in order for cyberspace culture to be introduced successfully, an organisation must emphasise its value to members:

'So, you cannot implement something new and then fail to use it, or you just wasted your time. So, I think they need to be engaged from the beginning and understand what is going on. Then, they will support or start to support and believe in this change'. (Respondent 3)

Furthermore, Respondent 1 identified various challenges affecting the creation of a conducive cyberspace culture in the context of Qatar. For one, Qatar possesses a conservative social and cultural background that is strongly influenced by religion. As a result, educators would be challenged in gaining their learners' trust to use online tools for education purposes.

As Respondent 1 stated,

'You know, Qatar overall is a conservative cult culture based on religion, and also on culture and tribal society'. (Respondent 1)

Respondent 1 further added that technology is intrusive:

'And now, with the intrusion of technology in homes, for example, you can just turn on my camera now, and you get to see everything that is around me; you get to see my family, perhaps people without a hijab at home'. (Respondent 1)

The views by Respondent 1 align with the literature identifying internet culture as both chaotic in nature and bringing about cultural shocks and lifestyle changes (Gunter, 1995).

Second, Respondent 2 added that culture is formed over a long duration of time, and more time is thus required for new changes to be implemented:

'And this culture is built over time We at Qatar University have our own way of doing things. And our ways of doing things were developed from day one; now if we want to change this, it takes time'. (Respondent 2)

However, Respondent 10 argued that introducing more time also produces more risks, such as resistance to change. An example highlighted included faculty resistance towards the newly introduced blackboard system, with the faculty's age identified as the main deterrent to the system's adoption:

'So, you will always face resistance, especially in the educational sector, the university sector, because of the age, maybe, of the faculty'. (Respondent 10)

The discussions between Respondent 2 and Respondent 10 regarding the slow pace of culture formation and experienced risks are confirmed in the literature in Sub-Section 3.2.1.1, where the abstractness of cyberspace was identified and the difficulty in creating rules of interaction was observed.

Third, Respondent 8 observed that culture is important not only within an organisation, but also outside it and within the community. The suggestion was that, even if organisations succeed in fostering a positive cyberspace culture, input from the external environment, if not considered, will negatively influence the developed culture.

Respondent 8 added,

'Actually, yes, I do believe that culture is very important and that culture has two levels'.

A fourth identified challenge concerned the lack of basic computing knowledge among many students in Qatar universities. This would indicate that creating a cyberspace culture is not feasible, as students have yet to understand the basics of computing. The literature findings in Section 2.8 identified similar gaps in the creation of cyberspace culture and in guiding the use of technology. Furthermore, Respondent 10 argued that, without a basic understanding of computing, students would be more vulnerable to adopting negative cultures in cyberspace:

'There is a huge problem with using technology. Some students do not know even how to open their laptops; they do not know how to scan it, for example, and the number of such students is not small'. (Respondent 10)

'I had students who did not know how to send an email And this all is happening in 2020'. (Respondent 10)

Nevertheless, despite the different challenges highlighted, some of the participants offered recommendations for fostering a positive cyberspace culture. For instance, Respondent 14 emphasised the importance of leaders taking a central role to foster a positive cyberspace culture by adopting more effective change management practices. Similarly, they also cited the need for education on cyberspace aspects:

'The management, or the leadership, needs to put more effort into the change management to reach the minds of the people supporting them and let them allow them to support the management to achieve'. (Respondent 14)

In addition, Respondent 12 highlighted the need to educate people:

'You have to educate people and families. To familiarise them with a culture that, yes, we will provide you all this, but it is not going to be so easy for you'. (Respondent 12)

The discussions between Respondents 14 and 12 emphasised the role played by leadership in fostering culture among people. This role was likewise confirmed in the literature in Section 2.6 regarding the guiding principles of rule-making for cyberspace culture.

A third solution by Respondent 5 involved facilitating culture change by changing the mindset of the individuals and promoting campaigns of social behaviour change using change management techniques. This respondent also cited the need for collaboration between the education sector and the Ministry to equip students with better technology skills:

'Yes, usually, leaders who are focussing on tools and technologies and so on, and ignoring the culture, most probably would fail'. (Respondent 5)

Based on these findings, the key argument is that, in order for technology and tools to work as required, building the proper culture is crucial. Similar views have also been emphasised in the literature in Section 2.6 regarding the guiding principles of rule-making for cyberspace culture and in Sub-Section 3.2.1.2, where culture was perceived as a unique cultural concept.
6.5.4 Theme 4: Proposed Cyberspace Culture Framework (Focus Group Q1 Appendix A)

The fourth theme sought to validate the proposed framework in aligning cyberspace culture with the higher education 2030 vision. As such, participants engaged in a focus group discussion regarding the validity of three components of the proposed framework (internal, external, and culture). Table 6-12 lists the various themes and codes that were considered:

Tuble 0 12. Cyberspace culture muniework codes and sub memes
--

Codes	Theme
Internal elements (Sub-Section 4.7.1)	
External elements (Sub-Section 4.7.2)	Alignment of cyberspace culture framework with the H.E. vision
Culture (Sub-Section 4.7.3)	

Source: The Researcher

6.5.4.1 Internal Elements

With the internal elements, the participants discussed how they aligned cyberspace culture with the achievement of the higher education QNV 2030.

6.5.4.1.1 Leadership and Management Strategies

The participants agreed that this factor is influential for achieving the higher education QNV 2030, as leaders are responsible for steering their organisations towards achieving the objectives. Additionally, holding management accountable for actions undertaken by the institutions would pressure them to lead the firms in the appropriate directions. Such findings confirm the literature in Sub-Section 4.3.2, where Schein (2013) argued that leaders guide their employees' actions. Moreover, they also echo previous findings in Sub-Section 4.7.1.1, which previously revealed that leaders in different types of organisations must be accountable to protect the ethics of the business and eliminate risks.

As focus group Respondent 4 explained,

'So, if people on the management level will be held accountable, I guess the objectives that they will be achieving, I think we will be doing well; we have a clear path'. (Focus group Respondent 4)

Focus group Respondent 2 further added that the universities in which they worked had already taken steps to ensure accountability and minimise risks, such as by developing strategic plans and instituting a risk register to identify all possible risks and their respective corrective actions. However, the respondent also emphasised the need for frameworks to facilitate this alignment:

'But the point is what framework we are going to utilise in order to make sure our strategy is effective'. (Focus group Respondent 2)

The findings suggest a need to adopt frameworks to enhance accountability among higher education participants. In addition, focus group Respondent 5 acknowledged that accountability describes a mechanism that ensures work is performed correctly.

6.5.4.1.2 Security Policy

Regarding security policy, first, participants highlighted the importance of security policies for aligning cyberspace culture to the attainment of the higher education QNV 2030, such as by renewing security certificates like ISO to ensure compliance. Second, they recommended creating awareness among the employees of organisations, such as by developing email campaigns and offering external training regarding the importance of cybersecurity. Similar findings were previously reported in the literature in Sub-Section 3.2.5, where the quadrant IV analysis revealed that digital technologies are dynamic in nature and, as a result, there is a need for educators and teachers to develop dynamic identities that align with the changing technology (Martin-Albo and Gregorio-Godeo, 2017).

As an illustration, focus group Respondent 1 cited that some employees are prone to leaving their computers unlocked and sharing privileged university accounts with multiple individuals. Such

habits were identified as highly dangerous to the information integrity within the university, and awareness should thus be created among employees:

'We need to align our employees; we need to provide awareness. The point is, most of the employees, maybe, they are not highly aware about security'. (Focus group Respondent 1)

Furthermore, focus group Respondent 4 asserted that

'There is an issue and another initiative emerging from external environments or from the MOI, and it involves giving trainings to a lot of people on cyber security in the university. So, we're working on that'. (Focus group Respondent 4)

In addition to offering training and sharing awareness campaigns, focus group Respondent 3 also highlighted the need to provide security information to employees:

'I think, effective security awareness measures, but there is a part lacking, which is making sure that information is there'. (Focus group participant)

Fourth, alignment could also be enhanced by providing a sense of purpose to employees in order to create a mindset centred on a positive security culture.

The discussions between Respondents 1, 3, and 4 indicate that creating awareness and offering training are pivotal for enhancing cybersecurity. Moreover, the findings echo the previous literature in Sub-Section 3.2.1.2, which revealed that the uniqueness of cyberspace further necessitates the development of guidelines to protect interactions among its actors and inhabitants.

6.5.4.1.3 Security Technical and Best Practices

Regarding best practices, participants cited that such practices are useful for aligning cyberspace culture with the attainment of the higher education QNV 2030. These findings were confirmed by previous literature, whereby the quadrant III analysis in Sub-Section 3.2.4 demonstrated that aspects such as learning and content creation in cyberspace are guided by best practices on the international platform.

Focus group Respondent 2 highlighted two examples of best practices: 1) saving files in the cloud as a backup, as opposed to local disks, which are susceptible to damage or theft, and 2) maintaining backups as a means of enhancing the efficiency of universities and, subsequently, to positively affect the attainment of the vision.

Specifically, focus group Respondent 2 asserted that

'You start working on something, a laptop, but you don't use best practices of saving your files in the cloud and keeping them updated, etc. And one day, your laptop crashes; you have lost everything you worked on'. (Focus group Respondent 2)

Focus group Respondent 4 also highlighted the creation of security policies as a best practice to ensure accountability among employees:

'Put some policy in the place, then make awareness, and after the awareness, you can now start saying if they are complying or not complying with it, but you need to take care of them; then, you make them accountable'. (Focus group Respondent 4)

Furthermore, focus group Respondent 1 emphasised the role of privacy and awareness:

'If someone enters our website, we can enter your privacy. They will think about it ... we need to invest more in the policy infrastructure of the university, then after that, raise a lot of awareness'. (Focus group Respondent 1)

The discussion between Respondents 1, 4, and 2 demonstrate that participants place importance on creating awareness regarding different security issues as a best practice to guarantee employee accountability and improve security in higher education institutions. Likewise, in the literature, Alnatheer and Nelson (2009) also emphasise creating security awareness by offering training programmes and implementing security policies.

6.5.4.2 External Elements

Furthermore, the participants also discussed how the external elements aligned cyberspace culture with the achievement of the higher education QNV 2030.

6.5.4.2.1 The QNV 2030 Directives

Regarding the QNV 2030 directives from the government concerning strategy formulation for higher education institutions, participants agreed that the directives align cyberspace culture with the achievement of the higher education 2030 vision. First, they observed that the government's external pressure and economic strategies play a major role in education. One of the participants mentioned that a reason for the government's pressure on education arose from global changes in technology and the need to adopt an education-based economy. The findings are confirmed by the literature in Sub-Section 2.4.1 stating that Qatar's government developed the QNV 2030 as a strategy to modernise the country by ensuring growth in all sectors (economic, social, environmental, and human).

As focus group Respondent 1 explicated,

'Definitely, the economic factors and the external pressure will play a big role'. (Focus group Respondent 1).

Focus group Respondent 3, meanwhile, added the following:

'That's why they're trying to put the pressure on the education, because they are thinking that this generation should change the whole country and the whole economy. Now, the economy is changing'. (Focus group participant)

Focus group Respondent 5, however, claimed that pressure from the government, such as to ensure compliance with global data protection rules like the GDPR, stems from the fact that the government funds most operations in the institutions. This view is confirmed by the literature in Sub-Section 3.2.2.1, which indicates that infrastructural investments are necessary to enhance the role of cyberspace as a tool of learning.

Focus group Respondent 5 stated the following:

'As long as the government pays the universities to operate, then they will have some control'. (Focus group Respondent 5)

However, focus group Respondent 6 argued that the leadership in higher education is fragmented, thus making it difficult to align cyberspace culture with the vision's achievement, as different institutions are on different technology levels. The suggestion was that various challenges still need to be tackled before the government's leadership strategies can align cyberspace culture with the achievement of the higher education QNV 2030.

To this end, focus group Respondent 6 asserted the following:

'I don't think there is a leadership in higher education. We have a fragmented leadership; you have the issue, but the university has leadership, and you have other leadership ... and we're not all on the same page ... that's why we're suffering now in achieving and aligning our jobs with the outcomes that the Ministry of Education is trying to push us to achieve'. (Focus group Respondent 6)

Furthermore, participants reported that alignment could be fostered by the Ministry of Education protecting institutions' academic freedoms. Regarding political factors, the findings revealed that some of the participants believe that new leadership is likely to change operations in higher education institutions by disregarding what their predecessors have done. However, others argued that some leaders will continue the previous work, albeit with tweaks and improvements. The discussions regarding how political factors influence the attainment of the higher education vision mirror the findings from the literature in Section 2.7, where the openness of cyberspace culture was identified and the need to create rules to govern it were reported. Likewise, the discussion also echoes findings in Section 2.6, where guiding principles on rule-making in cyberspace were further outlined.

The general consensus among participants is that the government plays an important role in formulating leadership strategies that later facilitate aligning cyberspace culture with the achievement of the higher education QNV 2030. Leadership is accordingly viewed through economic incentives, external pressure, and policies developed.

6.5.4.2.2 Technological Advancement

The participants agreed that technological advancement facilitates aligning cyberspace culture with the achievement of the higher education QNV 2030, as this advancement represents an integral component for transforming how teaching and learning are accomplished in Qatar universities. In effect, participants argued that technology has the potential to entirely transform how education is approached in Qatar. The findings regarding the opportunities of technology-mediated learning are echoed in the literature in Sub-Section 2.5.2.1, where technology for learning and teaching was evaluated, as well as in Sub-Section 3.2.2.1, where cyberspace as a tool of learning was also reviewed.

Focus group Respondent 2 asserted the following:

'Absolutely, it's maybe the main driver for change in our education system ... this will allow us to transform the way we are giving lectures here in Qatar university'. (Focus group Respondent 2).

Focus group Respondent 1, however, argued that technology will not necessarily help improve teaching, as some courses cannot be taken online:

'Yeah, exactly. There are even courses that are non-transformable to online. You just can't. You just need to deliver them traditionally'. (Focus group Respondent 1).

The implication is that several participants within educational environments will not interact in cyberspace, as technology is not effective. Focus group Respondent 3 further identified bureaucracy as a factor that influences the implementation of technology-based learning and teaching, as some projects have yet to be started since their institution in September of 2018:

'After several years in Qatar university, I felt that the main problem is not the will to change, but that paperwork, the bureaucracy, that comes with applying these things. You come with the idea, "Let's do this. Let's do it, everybody's happy. Yeah, let's start", and then it gets delayed'. (Focus group Respondent 3).

Therefore, although technology advancement could align cyberspace culture with the achievement of the higher education QNV 2030, the discussions between Respondents 3 and 1 indicate that issues such as bureaucracy and the difficulty in transforming some courses into online platforms further affect this alignment.

6.5.4.2.3 The QNV 2030 Implementation Framework

Regarding the implementation framework aligning cyberspace culture with the achievement of the higher education QNV 2030, focus group Respondent 2 identified a key weakness in the lack of performance measures in the framework, making it difficult to align the two:

'But it's still. There are a lot of question marks, like when it comes to the identity and citizenship. For example, we couldn't align ourselves to it, because we don't know how to measure the loyalty and citizenship'. (Focus group Respondent 2).

Focus group Respondent 3 shared a similar view:

'It's like, since it's a vision for, 22 years, that's very difficult'. (Focus group Respondent 3).

The argument is that, although the QNV 2030 implementation framework provides a guideline for implementing the higher education QNV 2030, it nonetheless lacks performance measures, thus making it even more challenging to align cyberspace culture. Participants subsequently emphasised the value of adding performance indicators. The discussion between the participants regarding the lack of milestones for the QNV 2030 is also echoed in the literature in Sub-Section 2.4.1, wherein only the four pillars of development were outlined—human, social, environmental, and economic—without any indication of milestones to assess their completion.

6.5.4.3 Culture

The final discussion concerned the various ways in which culture aligns cyberspace culture with the achievement of the higher education QNV 2030. Participants largely discussed the various solutions that could be adopted to alleviate the current issues affecting the cultivation of a positive cyberspace culture. First, focus group Respondent 3 highlighted the role of developing implementation plans to overcome employees' resistance to change. Such implementation plans were conceptualised as frameworks in the literature in Section 2.6 to govern the usage of cyberspace.

First, focus group Respondent 3 asserted the following:

'And I guess, part of our implementation ... not only in Qatar, but everywhere ... even the human resisted change. So, this should be about the plan of implementation, how to overcome the changes' resistance'. (Focus group Respondent 3)

Second, focus group Respondent 4 identified the need to change employees' mindsets by highlighting the value of proposed changes:

'We need to share what they will gain from it ... for that organisation or to accept it, and to take it as agreed behaviour'. (Focus group Respondent 4)

Based on the findings, a need exists to share the value of proposed changes in order to encourage employees to accept changes to their behaviour. Such findings are mirrored in the literature in Sub-Section 3.2.4, where the need to create frameworks to guide the actions of actors in cyberspace was emphasised.

6.5.4.4 Validation of the Proposed Framework (Figure 4-7)

After evaluating the different components of the proposed cyberspace culture framework, participants were asked to validate how the different elements influenced cyberspace culture. The general consensus is that the framework is valid in explaining the alignment between cyberspace culture and the attainment of the higher QNV 2030. However, some participants argued that some of the relationships between the variables are not bi-directional, as displayed in Figure 4-7. One

such example involved the external elements, where focus group Respondent 3 argued that external elements are difficult to affect in a backward manner:

'If you look at leadership, it is, but if you look at the two points, physically, you cannot force updates on external pressure or force updates on economic factors'. (Focus group Respondent 3).

This finding implies a need to modify the proposed cyberspace culture framework (Figure 4-7) to incorporate only uni-directional effects, such as with external pressure and economic factors. However, for the majority of the components, a bi-directional relationship would be retained, as the factors both affect the attainment of the 2030 vision and are likewise affected by its achievement.

Furthermore, the framework's validation by the participants suggests that the theorised effects of each individual factor are confirmed in higher education institutions. Therefore, the Researcher identifies various uses of the proposed framework in streamlining the attainment of the higher education 2030 vision.

6.6 Conclusion

This chapter detailed the results and analysis of the primary data collected in this research. In the first section, the data analysis techniques adopted in the chapter were discussed. As thematic analysis was largely utilised and its basics were briefly described, along with a breakdown of how data would be evaluated in the chapter. The second section then presented participants' demographic details and professional experience; most of the interviewees held senior roles in higher education, such as managers and directors, and had also worked in those roles for more than five years. Such findings underscore the validity and robustness of the participants' responses, given that they are well experienced in higher education. The third section featured the thematic analysis of the different interview responses. Based on the analysis, the importance of the different elements of the proposed framework was identified, and the findings demonstrate that participants validated the proposed framework and highlighted several adjustments.

The conclusion from the data analysis is that the participants identified the three framework components—external, internal, and cultural—as factors that influence the attainment of the higher education QNV 2030. External elements, such as the government's QNV 2030 directives regarding information policy and leadership in terms of the higher education sector's external pressure and economic viability, were identified as critical for determining the overall progress attained. Likewise, according to the findings, technological advancement facilitates learning and teaching by transforming the traditional classroom. The QNV 2030 was also identified as influential for developing cybersecurity frameworks, such as the Qatar 2022 World Cup framework. Moreover, internal elements within institutions are associated with the attainment of the higher education QNV 2030, as leaders steer their institutions to achieve different goals.

However, apart from input from leaders, best practices and security policies were also emphasised as influential elements for achieving the QNV 2030 in higher education. For one thing, best practices will help institutions avoid developing their policies from scratch while security policies will ensure employees remain aware of loopholes leading to security vulnerabilities in the institutions. Finally, culture was identified as a critical aspect for achieving the QNV 2030, as Qatar is highly influenced by religion and culture. Therefore, insights were provided concerning how culture can be changed to facilitate attaining the higher education QNV 2030. In the following discussion (Chapter 7), the findings are explored in greater detail.

Chapter 7: Discussion

7.1 Introduction

The previous chapter detailed the findings from interviews with the research participants in a descriptive manner. To this end, the Researcher delineated important insights regarding how each research element (internal, external, and culture) influences the attainment of the 2030 higher education vision. As the findings were largely descriptive, this chapter evaluates them thematically so as to identify useful interpretations to help address the formulated research questions. In this regard, thematic analysis is employed to explore patterns and themes in the reported data, thus generating the necessary research insights (Smith and Sparkes, 2016). Furthermore, the findings are critically compared against current literature works in order to identify similarities to and differences from other related studies. The research questions addressed in this chapter (refer to Section 1.6) are as follows:

- 1. How does cyberspace culture strategically align with the country's strategic directions?
- 2. How are current cyberspace culture frameworks aligned with the country's strategic directions?
- 3. Why is cyberspace culture important, and what gaps exist in the literature regarding cyberspace culture?
- 4. What type of cyberspace strategy should the government deploy to align with the country's strategic directions?

The chapter is subsequently organised into three key sections: Section 7.2 presents the analysis of themes in order to extract important interpretations from the interview data. Following this, Section 7.3 discusses the key findings obtained from synthesising the interview data. Continuing, Section 7.4 explains the revisions to be implemented in the proposed cyberspace culture framework in order to align its components with the achievement of the 2030 higher education vision in Qatar. Finally, the conclusion is detailed in Section 7.5, delineating how the research questions were addressed and the research aim achieved.

7.2 Thematic Analysis of Research Findings

In performing the thematic analysis of the interview data in the Nvivo 12 application, the Researcher employed tools such as word frequency tables to understand the context of the ideas

generated by the respondents. However, as the proposed cyberspace culture framework (Figure 4-7) guided the development of interview questions, the research themes considered for the analysis were already known beforehand (Oh, 2018). Accordingly, the interview questions (Appendix A) were further grouped in alignment with the themes (Table 7-1). In turn, this implied that the thematic analysis would be directed as opposed to conventionally producing coding categories from the raw data during data analysis (Pietiläinen *et al.*, 2017).

In the proposed cyberspace culture framework (Figure 4-7), three main elements were identified to influence the attainment of the 2030 higher education vision: external elements (Chapter 4, Sub-Section 4.7.2), internal elements (Chapter 4, Sub-Section 4.7.1), and theoretical perspectives on culture (Chapter 4, Sub-Section 4.7.3). Within each category, individual factors were also identified; for instance, internal elements (Chapter 4, Sub-Section 4.7.1) were grouped into leadership and management strategies (Sub-Section 4.7.1.1), security policy (Sub-Section 4.7.1.2), and security technical and best practices (Sub-Section 4.7.1.3). The Researcher considered the explicit categories to be the dominant research themes and the individual factors in each category to be sub-themes. Moreover, code labels were added for the factors within sub-themes; for example, economic factors and external pressure were considered codes within the role of the 2030 vision directive sub-theme. A fourth theme was also added to capture the proposed cyberspace culture framework's validity or robustness in facilitating the achievement of the 2030 higher education vision in Qatar. For this particular theme, the Researcher also moderated a focus group discussion in order to complement the findings from semi-structured interviews (Sub-Section 6.2.1).

The first step subsequently undertaken by the Researcher consisted of developing theme and subtheme categories in the Nvivo application (Figure 6-11). Thereafter, tools such as word frequency tables were utilised to capture the main ideas identified by participants. A treemap and sunburst were also generated to illustrate the distribution of research themes. A thematic map of the themes and sub-themes considered in the research is presented in Table 7-1 below:

Theme 1: External elements and attainment of the 2030 higher education vision				
Interview	Main code	Sub-themes		
question		Sub memes		
Q1	Economic factors (Sub-Section	Sub-theme 1: Role of the Vision 2030		
Q2	4.7.2.1.1.1)	directive		
Q3	External pressure (Sub-Section			
Q4	4.7.2.1.1.2)	Leadership (Sub-Section 4.7.2.1.1)		
Q5	Compliance and advanced security			
	(Sub-Section 4.7.2.1.2.1)			
	Environmental uncertainty	Information security policy and regulation		
	(Sub-Section 4.7.2.1.2.2)	(Sub-Section 4.7.2.1.2)		
	Political factors			
	(Sub-Section 4.7.2.1.2.3)			
	Learning and teaching			
	(Sub-Section 4.7.2.2.1)	Sub-theme 2: Role of technological		
	Management (Sub-Section	advancement (Sub-Section 4.7.2.2)		
	4.7.2.2.2)			
	Implementation framework (Sub-	Sub-theme 3: Role of the Vision 2030		
	Section 4.7.2.3.1)	framework (Sub-Section 4.7.2.3.1)		
Theme 2: Internal elements and attainment of the 2030 higher education vision				
Q6	Accountability (Sub-Section			
Q7	4.7.1.1.1)	Sub-theme 1: Leadership and management		
Q8	Risk management (Sub-Section	strategies (Sub-Section 4.7.1.1)		
	4.7.1.1.2)			
	Security awareness (Sub-Section			
	4.7.1.2.1)	Sub-theme 2: Security policy (Sub-Section		
	Continuous learning (Sub-Section	4.7.1.2)		
	4.7.1.2.2)			
L				

Table 7-1: Thematic map of research findings.

Theme 3: C	Attitudes and behaviour (Sub- Section 4.7.1.2.3) Security skills and knowledge (Sub-Section 4.7.1.3.1) ulture and attainment of the 2030 high	Sub-theme 3: Security technical and best practices (Sub-Section 4.7.1.3) er education vision		
Q9	Culture dimensions (Sub-Section 4.7.3.2) Values Visible processes Attitudes and behaviour (Sub-Section 4.7.3.1)	Sub-theme 1: Role of culture (Sub-Section 4.7.3)		
Theme 4: Alignment of cyberspace culture framework with the 2030 higher education vision				
Focus group	External elements	Sub-theme 1: Alignment of external elements		
discussion	Internal elements	Sub-theme 2: Alignment of internal elements		
	Culture	Sub-theme 3: Alignment of culture		

Source: The Researcher

7.2.1 Theme 1: External Elements and the Higher Education Vision, 2030 Achievement

The goal of the first theme consisted of investigating how external elements influence the attainment of the 2030 higher education vision in Qatar. By discussing this theme, the Researcher was able to address the fourth research question regarding the cyberspace strategy that the government ought to deploy to align with the country's strategic directions. Likewise, the discussion generated insights that partially address the second research question regarding how the current cyberspace culture frameworks are aligned with the country's strategic directions.

Three main external influences were identified: 1) The Vision 2030 directive, which represents the government's control over higher education in terms of strategies for leadership, regulation, and information security policy; 2) technological advancement and its influence on learning, teaching,

and management; and 3) the Vision 2030 implementation framework, which also alludes to the QNV 2030. An overview of each theme is presented in the following sections.

7.2.1.1 Sub-Theme 1: Role of the Vision 2030 Directive

This sub-theme evaluates the role of the Vision 2030 directive, as defined by the government strategy regarding the achievement of the 2030 higher education vision in Qatar. In Chapter 2, literature confirmed the Qatar government's strategic control over the development of a knowledge-based economy in the QNV 2030, as detailed in Sub-Section 2.4.1 (Qatar National Vision, 2030, 2008). In the proposed cyberspace culture framework, government's strategic directives were found to influence higher education in two ways: leadership on the one hand and information security policy and regulation on the other.

7.2.1.1.1 Leadership

The synthesis of the findings revealed that participants perceive leadership in higher education to be embodied by the government and university management. The government, through the Ministry of Higher Education, was identified as an oversight authority that 1) provides guidelines, regulations, and directions for the higher education institutions; 2) influences the development and implementation of policies adopted by the higher education institutions; and 3) provides funding mechanisms to enhance the sector.

As Respondent 11 explicated,

'The government is pushing all the regulatory, policy and strategy, and funding mechanisms ... the role of the government is very important'. (Respondent 11)

This highlighting of the government's influential leadership and regulation roles resonates with previously studied cyberspace culture frameworks (Alnatheer and Nelson, 2009; Shabbaan, 2014; Al-Hogail, 2015). Likewise, in Chapter 2, the government's influential leadership role was identified in the Qatar government's funding of the acquisition of scientific research infrastructure (Gasser and Drolshammer, 2015) and the facilitation of mediated technology-based learning (Dunn and Kennedy, 2019), as well as in boosted collaboration between Qatar and its allies (Zhou *et al.*, 2019). Further analysis in Sub-Section 3.2.2.1 also highlighted enhancements for educators

and learners that improve the function of cyberspace as a tool of learning (Crossman, 1997; Harasim, 1993; Jonassen *et al.*, 1999; Khan, 1997; Land and Bayne, 2012; Zoto *et al.*, 2018).

This aspect further emphasises how the government's strategic leadership influences the fostering of cyberspace culture. Figure 7-1 below further demonstrates the overall leadership structure of higher education, as identified by the participants:



Figure 7-1: Leadership in Qatar higher education.

Source: The Researcher

Figure 7-1 illustrates that the government's strategic directives influence policy, funding, and the strategic directions adopted by higher education institutions. This influence is discussed further in relation to economic factors and external pressure in the following sections.

Category 1: Economic Factors

With this factor, the Researcher was interested in understanding how the government's strategic directives concerning economic issues will influence the implementation of the 2030 higher education vision in Qatar. To this end, four key findings were identified, as summarised in Figure 7-2:



Figure 7-2: Influence of the government's strategic economic directives on the H.E. Vision, 2030.

Source: The Researcher

The first reported influence was that the government's economic directives lead to the setup and development of higher education institutions, as most universities in Qatar are fully funded by the government. Noteworthy examples of such universities include the Qatar University, CCQ, CNAQ, and the University of Calgary (Respondent 1). The participants further revealed that the government's economic directives focus on increasing workforce capacity and the economic viability of higher education institutions. The cited findings-funding of higher education institutions by the Qatar government, enhancement of workforce capacity, and improvement of economic viability—reinforce the human development pillar of the QNV 2030, which seeks to develop the population by investing in scientific research and first-rate education (Qatar National Vision, 2030, 2008). Likewise, the findings reiterate diverse researchers (Fabos and Young, 1999; McLellan, 1998; Reynolds, 2002; Riel, 1995), who have also confirmed that technologies and learning infrastructure in cyberspace boost learning and collaboration outside the classroom. Such technologies are argued to be acquired through government financial assistance, as the government plays an integral role as a stakeholder in cyberspace (Baker, 2016; Barzilai, 2004). As the Qatar government focusses on building a knowledge-based economy, investment in higher education is considered paramount. The third finding in particular underscores this fact, since participants cited

that investment is meant to increase in the economic viability of higher education, such as by creating additional institutions.

Fourth, the findings also revealed that participants link government funding to the enhancement of sustainability in higher education, as they are empowered to acquire necessary resources, technologies, and infrastructure. Such findings align with Marshall (2018), who argued that, since higher education institutions are challenged by high operation costs, government support from the Vision 2030 directives helps to develop independent institutions to train the Qatar population. The literature in Sub-Section 3.2.4 further confirms that cyberspace represents a tool for learning and content creation, and thus necessitates acquiring appropriate tools and technological infrastructure (Lessig, 2002; Bell, 2009). Cantwell, Coate, and King (2018) also stated that the government comprises a major stakeholder in higher education, and thus provides the required funding. Therefore, the Qatar government's economic strategic directives can be understood to influence the achievement of the 2030 higher education vision by providing funding to set up higher education institutions, procure necessary infrastructure and technologies, enhance workforce capacities, and ensure sector sustainability on a long-term basis.

Category 2: External Pressure

With the second category, the Researcher focussed on understanding how government strategies, in exerting external pressure on higher education institutions, influence the attainment of the 2030 higher education vision. Two important findings were accordingly identified, as detailed in Figure 7-3 below:



Figure 7-3: Influence of the government's external pressure directives on the H.E. Vision, 2030. Source: The Researcher

According to the findings, some participants reported pressure from the government, exerted through the Ministry of Higher Education (Respondents 6 and 9). This pressure was further subdivided into two categories: First, the obligatory pressure was attributed to the Ministry's constant actions in evaluating whether set directives have been achieved by institutions. A plausible explanation for the obligatory pressure is that the government funds the institutions, thus requiring them to be accountable for their operations (Cantwell, Coate and King, 2018). Furthermore, in Section 2.6, the government was also identified as a policy-making body that influences overall decision-making and the usage of cyberspace (Barzilai, 2004; Baker, 2016). Other researchers in Chapter 4 (Alnatheer and Nelson, 2009; Shabbaan, 2014; Al-Hogail, 2015) have also reported that the government is integral to formulating and implementing policies and regulations for the higher education sector. As a result, external pressure is also expected to ensure that the set guidelines and regulations are adhered to.

Second, positive pressure was attributed to pushing institutions to achieve their goals. Such pressure was argued to arise from the positive rapport between higher education institutions and the Ministry of Higher Education.

As Respondent 18 asserted,

'Yeah, I mean, if we consider strategic goals that are set by the country as pressure, I will say it is positive pressure'. (Respondent 18)

In this context, it can be argued that the Ministry focusses on advancing higher education in the country, thus pushing the institutions to achieve more strategic objectives. One such case involves advancing scientific research to attain world-class levels, as stipulated by the QNV 2030 (Qatar National Vision, 2030, 2008). Furthermore, as revealed by the findings in Sub-Section 3.2.2 detailing the use of cyberspace as a tool of learning, numerous advantages are associated with the adoption of advanced cyberspace technologies, which lead to positive pressure. These advantages include solving problems (Duffy and Jonassen, 1992; Jonassen, 1993), boosting participation and engagement (Szücs *et al.*, 2013; Cope and Kalantzis, 2017), and improving collaboration (Fabos and Young, 1999; McLellan, 1998; Reynolds, 2002; Riel, 1995).

A second school of thought identified by some participants concerns the lack of external pressure from the government. The participants argued that the only pressure present within higher education institutions arises from senior management. However, critical analysis of the finding further suggests that the pressure is indirectly exerted by the government. Nonetheless, the key finding concerns the fact that the government's external pressure influences the attainment of the 2030 higher education vision by pushing institutions to achieve set objectives and ensuring that they comply with established regulations and guidelines.

7.2.1.1.2 Information Security Policy and Regulation

Within the same sub-theme regarding the government's Vision 2030 strategic directives, the Researcher further investigated how directives concerning information security policy and regulation influence the attainment of the 2030 higher education vision. The analysis of responses revealed that participants perceive the progress in developing information security policies in two different ways. Figure 7-4 details the participants' views on Qatar's information security policy:



Figure 7-4 : Participants' understanding of information security policy in Qatar. Source: The Researcher

One important finding is that information security policy is important not only in higher education, but also in other economic sectors. However, participants also mentioned current challenges concerning the lack of standardised policies across different sectors. This finding highlights a lack of coordination among different sectors despite the fact that the QNV 2030 seeks to promote growth for the entire economy (Qatar National Vision, 2030, 2008). The identification of extant challenges in developing information security policies alludes to the infancy of security culture in the country. This finding further resonates with Alnatheer and Nelson (2009), who cited that, in Saudi Arabian workplaces, information security culture remains in its infancy despite high economic development. Likewise, in Section 2.7 concerning the openness of cyberspace culture, researchers such as Fabre *et al.* (2016) have argued that resistance has been mounted against the development of standard laws to govern cyberspace, leading to the observed gap. This was further stressed in Section 2.8, where the lack of cyberspace culture frameworks to guide the attainment of strategic objectives was identified (Brunn, 2014).

A second noteworthy finding is that, although a lack of co-ordination represents a problem in Qatar, with the Qatar 2022 Cybersecurity Framework, the government has still made important strides in implementing information security policies in sectors such as football in order to help prepare for the 2022 FIFA World Cup. This finding also aligns with previous work by Foody *et*

al. (2017), which demonstrated that cybercrime laws were enacted in Qatar in 2014 to ensure protection against cybercrime and secure intellectual property. Fabre *et al.* (2016) further reported that solutions such as adopting cyberspace rules parallel to international public law and international environmental laws have been considered. Furthermore, the Qatar government has also widely adopted e-government solutions to deliver services to its citizens (Al-Hakim, 2007; Kettani and Moulin, 2015).

Following this, the Researcher evaluated how the government's strategic directives influence information security policy and regulation in terms of three core factors described in the next section.

Category 1: Compliance and Advanced Security

The first category sought to identify how the government's strategic directives regarding information security policy and regulation influence compliance and advanced security, as well as how this influences the attainment of the 2030 higher education vision (Sub-Section 4.7.2.1.2.1). Three important findings were accordingly identified, as summarised in Figure 7-5 below:



Figure 7-5 : Influence of compliance and advanced security on the H.E. Vision, 2030 attainment. Source: The Researcher

First, participants identified that compliance and advanced security directives serve the role of enabling higher education institutions to meet international market standards. In turn, this leads to

the second role of ensuring protection for the society and the community as Qatar interacts with foreign countries. These findings directly align with the QNV's human development pillar, which advocates for prioritising world-class scientific research and first-rate education (Qatar National Vision, 2030, 2008). Compliance with international best practices would ensure that Qatar achieves its vision of modernising its education sector. In the same vein, such compliance would improve international co-operation between Qatar and its allies, leading to better coexistence (Qatar National Vision, 2030, 2008). Previous findings by Cantwell, Coate, and King (2018) also revealed that the government facilitates partnerships with foreign investors in higher education, ensuring compliance with international market standards to enhance such relations. Furthermore, as the findings in Sub-Section 3.2.1.2 demonstrate, cyberspace as a unique cultural space adheres to the universality-without-totality concept, thereby extending to international boundaries (Le'vy, 2001). Subsequently, a need further arises to protect actors in cyberspace from negative activities such as cybercrime, espionage, and cyberterrorism (Galinec *et al.*, 2017).

The third role of the compliance and advanced security directives involves ensuring that institutions adhere to established government regulations. Participants cited the use of cloud computing as an example of a government directive that must be adhered to. The main argument advanced in support of compliance with regulations is that the government plays an oversight role in higher education by providing funding and developing policies. As a result, institutions must demonstrate accountability through compliance. This finding resonates with Mumper et al. (2011), who reported that institutions must comply with security policies and that failure results in violation of the law. Additionally, evaluating compliance with advanced security through the lens of cyberspace culture further reveals its importance in ensuring that Qatar's higher education institutions meet the demands of international law—a finding that aligns with Weber (2012). The argument is that, as cyberspace is not limited to a particular geographical area and exerts international influence, it is necessary to ensure that information security in higher education institutions meets international best practices in order to protect the society and community (Barzilai, 2003). This finding also resonates with the literature in Section 2.3, where security challenges in cyberspace, such as cyberterrorism, cybercrime, cyberwar, and espionage, were discussed (Galinec et al., 2017). The key argument is that compliance with international regulations would ensure that actors remain protected from adverse criminal activities.

Category 2: Environmental Uncertainty and Political Factors

The second category considered how the government's strategic directives regarding information security policy and regulation influence environmental uncertainty and political factors, as well as how this influences the attainment of the 2030 higher education vision (Sub-Section 4.7.2.1.2.2 and (Sub-Section 4.7.2.1.2.3). The findings are summarised in Figure 7-6 below:



Figure 7-6: Influence of environmental uncertainty and political factors on the H.E. Vision, 2030.

Source: The Researcher

The main finding reported from the synthesis of responses is that environmental uncertainty disrupts operations in higher education, thus hindering the attainment of the 2030 higher education vision. This disruption is further attributed to two main sources: political influences and non-political issues. To begin with, participants identified political influences such as changes in leadership within higher education institutions as a leading cause of environmental uncertainty and disruption of operations, as the change in senior leadership was argued to influence the attainment of the vision and strategic objectives. This view aligns with Skakni *et al.* (2019), who also claimed that changes in higher education leadership, such as in the Ministry, would lead to uncertainty, as doubt would arise concerning how the institutions would be managed. A similar finding was also reported by Adamu (2019), who posited that the Ministry of Education (political factors) previously played a role in nominating and appointing university leaders. Pinheiro *et al.* (2019) further reported that, in Sweden, the Ministry of Education appointed 4 out of 11 members of the university governing council. Such findings emphasise leadership change's negative influence on

the attainment of the long-term 2030 vision for higher education. However, several participants did not agree that changes in leadership would disrupt university operations.

Furthermore, the participants reported that disruption of operations would arise from external political influences, such as the blockade, which was argued to affect the country's higher education dynamics. For instance, it was suggested that the blockade would cause the Qatar Foundation University to close its branches (Respondent 1). Such external influences have previously been identified in the work of Mok (2016), who suggested that transitions in political power would generate environmental uncertainty as higher education institutions become unsure of receiving continued support in the new government. In this context, the change in power occurs in the central government rather than the Ministry of Higher Education or senior management within universities. The findings from the literature in Section 2.6 concerning rule-making in cyberspace likewise suggest that stakeholders involved in developing policies and rules in the digital space would influence the actors' actions and the usage of cyberspace, which is subsequently likely to lead to uncertainty (Baker, 2016).

In the context of cyberspace culture, researchers such as Barzilai (2004) have also previously reported that politics influences decision-making processes and overall governance. Therefore, the impact of external political events was identified as a causative factor of environmental uncertainty. Moreover, Bell (2001) reported that cyberspace is centred on people, their cognition of self, and the experiences they create in the space—a finding suggesting that any actions that affect their experiences will lead to environmental uncertainty. Finally, participants revealed that uncertainty could also arise from non-politically related factors, such as the current Covid-19 pandemic. Such external disturbances were argued to adversely influence strategy implementation.

7.2.1.2 Sub-Theme 2: Role of Technological Advancement

The second sub-theme investigated how technological advancement, as an external element, influences the attainment of the 2030 higher education vision.

7.2.1.2.1 Technological Advancement

The Researcher was keen to understand how the use of advanced technologies to facilitate learning and teaching, as well as the management of institutions, influences the achievement of the vision. These factors are discussed in the following section.

Category 1: Learning and Teaching

Regarding the use of technology to facilitate learning and teaching, Figure 7-7 summarises the primary findings that were obtained:



Figure 7-7: Influence of advanced learning and teaching tech on the H.E. Vision, 2030 attainment.

Source: The Researcher

Synthesis of the results revealed three important findings: First, technology comprises an enabler of collaboration and knowledge sharing, both explicitly in terms of documents and articles and implicitly in terms of ideas between learners. As such, technology was identified to improve interactions within learning environments. These findings align with the literature in Chapter 2, Sub-Section 2.5.2 as well as Chapter 3, Sub-Section 3.2.2.1. For instance, in the previous works of Talebian *et al.* (2014), ICT technologies were found to boost learning by eradicating the limits of time and place. Similarly, Dunn and Kennedy (2019) reported that technology-based learning enhances existing educational models by shifting from teacher-centred to learner-centred approaches that are more self-paced and self-directed. Esteves, Matias, and Pereira (2018) added that technology-mediated collaborative learning enables learners to enhance soft skills such as oral communication and social interaction. In addition, Mahlangu (2018) observed that technology-enabled learning is self-paced, and students can complete it in a flexible manner.

A second finding is that learning and teaching technologies facilitate the development of smart campuses where students can easily manoeuvre within their schools using artificial intelligence apps and without relying on any external assistance. This finding resonates with Ho *et al.* (2010), who revealed that modern learning approaches introduce the use of technological material and, in turn, lead to reduced reliance on traditional materials, such as hardcopy books. In this case, physical maps would represent the hardcopy version of the modern applications employed. Sari, Ciptadi, and Hardyanto (2017) further highlighted the role of smart campus technologies in innovating learning, as RFID technologies provide real-time connectivity to outside entities. Continuing, Chan and Chan (2018) reported that technologies such as IoT transform libraries into smart-learning spaces that improve peer collaboration and social learning.

A third finding is that technology enhances current teaching practices, such as by introducing paperless exams and having them marked within a short time period. This finding aligns with Oliveira *et al.* (2019), who previously reported that technology improves pedagogical processes through simulations, virtual labs, and robots, among other things. Allsop *et al.* (2020) also cited the use of video-conferencing technologies to deliver anatomical teaching to medical students. In the wider relation to cyberspace, diverse educational technologists (Crossman, 1997; Harasim, 1993; Jonassen *et al.*, 1999; Khan, 1997; Land and Bayne, 2012; Zoto *et al.*, 2018) have also argued that cyberspace could be considered a teaching site that maximises the available learning that could be achieved.

Nevertheless, despite the various cited advantages, the participants also reported that challenges such as resistance to technologies and the lack of an adoption strategy hinder the use of technology in learning and teaching.

As Respondent 12 stated,

'I have seen talks about it, but I have not seen anything in strategy. Now, back to the strategy itself ... about method of teaching improving things'. (Respondent 12)

Such findings further emphasise the need for Qatar higher education institutions to develop strategies to improve overall acceptance of the proposed technologies.

Evaluation of the findings in this section suggests that, in order to achieve QNV 2030, Qatar will increase its dependence on advanced technologies to facilitate attaining its objectives regarding the delivery of teaching and learning. These findings accordingly revealed that technology would enable knowledge-sharing and collaboration (Talebian *et al.*, 2014), facilitate the development of smart campuses (Ho *et al.*, 2010), and improve teaching practices (Oliveira *et al.* (2019). Despite this, however, there is also need to consider the likelihood of a retrograde shift in technology—an aspect that is implied by the current trend, whereby more post-digital experiences are being advocated, such as digital detoxing and abstinence from smartphones (Wilcockson, Osborne and Ellis, 2019). The argument for this retrograde shift in technology use concerns the fact that recent decades have witnessed widespread use and dependence on technology. As a result, this has subsequently led to technology misuse and unprecedented consequences such as addiction to smartphones and video games. The implied solution thus involves adopting retrograde activities such as reduced usage of smartphones as well as undertaking digital detox processes.

Second, as Hussein *et al.* (2020) postulate, students in the UAE were observed to develop negative attitudes regarding emergency online learning due to the Covid-19 pandemic. Noteworthy issues associated with e-learning included heavy workload, constant distractions, and problems with technology.

These two findings regarding the popularity of digital detox and negative attitudes towards emergency e-learning suggest a likely retrograde shift against e-education and the subsequent resurgence of traditional 'chalk-and-board' teaching approaches in the future. Therefore, there is also a need for the QNV 2030 to consider this aspect critically, including the various effects it is likely to exert on the attainment of the 2030 higher education vision.

Category 2: Management

Regarding the use of technology for managing higher education institutions, three important findings were reported, as presented in Figure 7-8 below:



Figure 7-8: Influence of management technologies on the H.E. Vision, 2030 attainment. Source: The Researcher

First, four participants (22.2%) revealed that technology facilitates tough decision-making and improves resource management, especially in the wake of declined oil prices and the Covid-19 pandemic. In this regard, technology is employed to streamline processes and facilitate difficult decisions due to the budgetary reductions experienced. A second, similar finding concerns the use of machine learning and analytics to facilitate learning from past experiences.

As Respondent 18 explicated,

'We are learning from the previous experience how we were doing that using analytics, using machine learning in order to learn about the behaviours'. (Respondent 18)

The two findings regarding technology-enabled decision-making and the use of analytics and machine learning to learn from previous experiences align with previous research detailed in Chapter 2, Sub-Section 2.5.2.2; for instance, Kurilovas (2018) demonstrated that artificial intelligence and data-mining techniques are becoming pivotal in higher education institutions, as they generate important insights from the large, existing datasets. Likewise, Liebowitz (2016) reported that analytics and big data are important for improving decision-making processes in different aspects of higher education. Daniel (2014) likewise highlighted the use of data analytics as a tool to formulate better policies that are data-driven.

As presented in Chapter 6, Sub-Section 6.3.1.3.2, participants revealed that technologies are also improving administration processes by making them 'smoother and secure' and by addressing current management bottlenecks. Such findings reiterate Singh's (2018) claim that information systems aid management by improving the creation, storage, and dissemination of different information types. Daniel (2014) also reported that administrators are able to make better data-driven decisions across different institutional departments by using technology. Furthermore, other researchers, such as Akçapınar, Altun, and Aşkar (2019), have reported that learning analytics systems allow university management to develop effective policies to ensure that students do not drop out of their classes by collecting insights concerning the risk factors that would lead to such an event. Natow *et al.* (2017) also cited the use of student support technologies to track performance and lower student attrition rates. Jones (2019), meanwhile, highlighted that digital technologies would allow one to track student behaviour and facilitate the analysis of their profiles to promote learning.

7.2.1.3 Sub-Theme 3: Role of Vision 2030

The final sub-theme of the external elements concerned the Vision 2030 implementation framework. Here, the Researcher sought to identify how the framework influences the attainment of the 2030 higher education vision.

7.2.1.3.1 Vision 2030 Implementation Framework

From the synthesis of the obtained responses, the key finding is detailed in Figure 7-9 below:



Figure 7-9: Influence of Vision 2030 implementation framework on the H.E. Vision, 2030 attainment.

Source: The Researcher

As detailed in Figure 7-9, the participants viewed the Vision 2030 implementation framework as a guide that helps shape the development of a knowledge-based economy by facilitating the uptake

of modern technologies. The participants further argued that, for many generations, Qatar's economy has relied on hydrocarbons in oil exports. However, with the uptake of technology, the economy will instead leverage its population's innovative capacities via their ingenuity and capabilities.

As Respondent 1 explicated,

'Technology that can discover what else is beneath the sand in Qatar ... tapping into the ingenuity and the capabilities that are in people'. (Respondent 1)

These findings directly align with the QNV 2030, which seeks to modernise the Qatar economy into one that is knowledge-based by enhancing human, social, environmental, and economic development (Qatar National Vision, 2030, 2008). In the context of higher education, the implementation framework provides a guideline for enhancing education to reach world-class standards in order to ultimately increase the Qatar population's capabilities and unlock their potential (Qatar National Vision, 2030, 2008). In the literature, the findings largely align with works reported in Chapter 2, Sub-Section 2.4.1, where the role of each pillar of the QNV 2030 was discussed.

Despite this view, other participants criticised the Vision 2030 implementation framework for its lack of performance indicators or definite milestones to signal what has been achieved over time. For instance, the use of KPIs was recommended over a four-year period to assess what has been achieved within that period. The argument is that following a long-term plan without any definite milestones would be highly challenging.

7.2.2 Theme 2: Internal Elements and the H.E. Vision, 2030 Achievement

Following the analysis of how external elements influence attaining the 2030 higher education vision, the second theme further investigated how internal elements influence the attainment of said vision. Three core sub-themes were accordingly analysed, as detailed in the coming sections.

7.2.2.1 Sub-Theme 1: Leadership and Management Strategies

With the first sub-theme, the concern was to identify how leadership and management strategies formulated within higher education institutions influence the attainment of the 2030 higher

education vision. To this end, the Researcher was particularly interested in understanding how the developed strategies promote accountability and risk management.

7.2.2.1.1 Accountability and Risk Management

From the synthesis of the responses, the key findings are summarised in Figure 7-10 below:



Figure 7-10: Influence of leadership and management strategies on the H.E. Vision, 2030. Source: The Researcher

Regarding the strategies' influence on accountability, participants agreed that leaders possess a responsibility to develop clear implementation plans that facilitate attaining strategic goals and responsibilities. Additionally, they cited that reliance on mature implementation frameworks would further improve such plans, as they outline necessary KPIs to evaluate progress. In turn, this would ensure successful achievement of strategic goals in higher education institutions. Such findings align with Ghanem and Castelli (2019), who previously reported that accountable leadership ensures the protection and development of ethical organisational perspectives.

A similar perspective can also be drawn from the development of the QNV 2030 by Qatar leadership to guide the implementation of strategic objectives related to the four human, social, economic, and environmental pillars (Qatar National Vision, 2030, 2008). Qatar leadership can be

argued to be accountable for achieving the overall 2030 vision through development of the vision plan, which serves as an implementation framework. Likewise, the development of the Qatar 2022 Cybersecurity Framework cited in an earlier section further underscores Qatar leadership's accountability in preparation for the FIFA World Cup. Regarding the implementation of technology-based learning, other researchers have also cited that implementation frameworks had been developed to guide the various processes in higher education, such as blended learning education (Garrison and Vaughan, 2013), blogging (Kerawalla *et al.*, 2009), and re-engineering of educational technology education (Kelley and Kellam, 2009).

A second finding is that leadership strategies would also facilitate accountability by ensuring that top leadership guides and supports the workers' actions. This suggests that it is not sufficient for leaders to simply craft implementation plans; rather, they must also support their employees and provide guidance in implementing those plans. Continuing, previous work by Leonard *et al.* (2016) further revealed that leadership's personal qualities, administrative behaviours, and actions influence culture formation and maintenance in organisations. In this context, it is argued that leadership accountability behaviour helps guide workers to implement strategic objectives and goals. Schein (2013) also cited that leadership represents the source of values and beliefs in organisations, and as such, leaders play an important role in dictating the values that their employees adhere to. The guiding role of Qatar as a leader in developing a knowledge-based economy was further emphasised in Chapter 2, Section 2.4, where the QNV 2030 was observed to account for development in all sectors—namely, human, social, economic, and environmental (Qatar National Vision, 2030, 2008). The participants, however, argued for the need to offer more training for leaders in higher education in order to improve their competency.

With risk management, participants revealed that appropriate leadership and management strategies help reduce risks by promoting efficiency through ensuring that activities are performed properly. These findings resonate with Hartman, Townsend, and Jackson (2019), who, for instance, reported high risks associated with adopting modern technology in higher education. One approach to eliminating these risks involves adopting existing frameworks, such as blended learning education (Garrison and Vaughan, 2013), blogging (Kerawalla *et al.*, 2009), and re-

engineering of educational technology education (Kelley and Kellam, 2009). Such frameworks provide a basis for understanding integration processes not specific to situations in Qatar.

However, the participants also cited that, in Qatar higher education, leadership and management are too centralised, making decentralisation necessary to promote collaboration among different departments.

7.2.2.2 Sub-Theme 2: Security Policy

The second sub-theme sought to identify how security policies influence the attainment of the 2030 higher education vision. To this end, participants were asked how security policy affects security awareness, continuous learning, and their attitudes and behaviour, as well as how this subsequently influences the achievement of the 2030 higher education vision. The findings are discussed in the following sections.

7.2.2.1 Security Awareness and Continuous Learning

Based on a review of the generated responses, the Researcher combined the views on continuous learning and security awareness due to the thematic similarity observed. The key findings are summarised in Figure 7-11 below:





Source: The Researcher

As detailed in Figure 7-11, two main findings were obtained: The first emphasises the importance of security awareness and continuous learning for improving the efficiency of processes and,

eventually, the results that were obtained. The suggestion is that, as security policies are constantly updated, this ensures process efficiency and leads to improved results in an iterative manner. This assertion is based on Robbins (1998), who reported that security policy guides organisations to develop strategies to help complete assigned tasks. This finding further aligns with Kanwar, Balasubramanian, and Carr (2019), who previously described continuous learning as a process of constant reskilling and upskilling to meet industry demands. In this context, updates to security policy would allow higher education institutions to constantly improve their awareness and skill levels, subsequently enhancing their efficiency in achieving the broader 2030 higher education vision.

The second finding is that security awareness and continuous learning would enable higher education institutions to keep abreast of the dynamic changes occurring in the cybersecurity world. Participants argued that dynamic rather than fixed security policies are required to ensure that institutions constantly adapt to changes in their environments. This finding directly aligns with Kanwar, Balasubramanian, and Carr's (2019) emphasis on the need for continuous learning, reskilling, and upskilling to meet industry demands. Additionally, the findings echo previous researchers (Landow, 1992; Apple and Jungck, 1996; Raschke, 2003; Aarseth, 1997) who have argued that identities in cyberspace are dynamic rather than static, as technologies continually morph. Therefore, continuous learning and improved self-awareness would allow higher education institutions to align with changing market demands as they focus on attaining the 2030 higher education vision.

7.2.2.2.2 Attitudes and Behaviour

In addition to positively influencing security awareness and continuous learning, security policies were also argued to influence the attitudes and behaviour of employees in higher education institutions. The main findings are detailed in Figure 7-12:


Figure 7-12: Perspectives on attitudes and behaviour influencing the H.E. Vision, 2030 attainment.

Source: The Researcher

By evaluating the responses, two main findings were identified: The first concerns the role of clear and easy-to-understand security policies that would guide employee actions. The suggestion is that implementing clear policies would positively influence behaviour and attitude change. Similar findings have also been reported by various researchers (Robbins, 1998; Da Veiga, 2008) arguing that information security behaviour within organisations must be aligned in order to promote positive security cultures.

The second finding is that the type of culture fostered within an organisation also influences employee attitudes towards adopting various guidelines and regulations stipulated in the security policies. As such, emphasis was placed on avoiding generic security policies that lack contextual meaning for employees. The findings regarding culture's role in influencing attitude and behaviour change resonated with Schein (1984, 2013), who claimed that culture within an organisation is formed over time and subsequently influences how individuals think, feel, and perceive in regard to solving problems. Furthermore, in Section 4.4, the Researcher posited a link between organisational culture and behaviour, with cyberspace culture influencing employees' behaviour in terms of how they interact with information assets and security. Likewise, the evaluation of different information security frameworks (Alnatheer and Nelson, 2009; Shabbaan, 2014; Da Veiga and Elof, 2010) indicates that culture within organisations fundamentally influences the security behaviours adopted by employees.

7.2.2.3 Sub-Theme 3: Security Technical Best Practices

In the third sub-theme, the Researcher sought to understand how security technical best practices adopted in higher education institutions influence the attainment of the 2030 higher education vision. Findings on best practices were accordingly discussed in regard to the enhancement of security skills and knowledge, as detailed in the following section.

7.2.2.3.1 Security Skills and Knowledge

From the synthesis of the responses, the main findings are summarised as illustrated in Figure 7-13:



Figure 7-13: Perspectives on security technical best practices influencing the H.E. Vision, 2030 attainment.

Source: The Researcher

All participants agreed that best practices are generally highly important for attaining the 2030 higher education vision. However, they argued that, as a prerequisite to implementing best practices in higher education institutions, it must be ensured that the proper types of practices are developed in the first place. Guidelines and procedures for implementing best practices were subsequently underscored to ensure that the proper practices were identified. The findings align with Gregg *et al.* (2017), who previously reported that most higher education institutions rely on major established tech vendors to facilitate implementing e-learning platforms due to their innovativeness and security. Likewise, Shabbaan (2014) advocated for adopting ISO/IEC 27002

to inform best practices in information security management. Furthermore, in Section 3.3, the Researcher argued that having best practices is important to guide novice learners in boosting their learning experiences in the digital space.

A second important finding is that, in addition to identifying the proper security technical best practices, it was further necessary to align them with Qatar's social-cultural context. The participants cited that, in order for the best practices to work effectively in higher education institutions, they needed to be modified further to fit Qatar's specific conditions.

As Respondent 2 asserted,

'If any system fails here, it is not because it's bad, but because it does not fit and align to our social norms'. (Respondent 2)

This finding led to a discussion regarding the various ways in which best practices could be formulated or created to meet the needs of Qatar higher education institutions. As a result, two important ideas were advanced: benchmarking from other existing frameworks and creating one's own best practices aligned with budgets, initial goals, and operational goals. The first idea on developing best practices by benchmarking from existing frameworks aligns with Shabbaan (2014), who highlighted the use of the ISO/IEC 27002 framework to derive best practices for information security management in organisations. Similarly, in tackling the problem of implementing digital learning in cyberspace, diverse frameworks were found to facilitate the processes, such as blended learning education (Garrison and Vaughan, 2013), blogging (Kerawalla *et al.*, 2009), and re-engineering of educational technology education (Kelley and Kellam, 2009). Some participants supported this idea, stating that one should avoid developing practices from scratch; instead, existing practices ought to be reviewed and modified to fit the Qatar context.

With the second idea on creating best practices within higher education institutions, the participants further suggested a need to establish research institutions to facilitate the process and aid in developing the proper types of practices. Such institutions were argued to help search for existing best practices and find ways to customise them to meet the needs of the individual higher

education institutions. This idea resonates with Lessig (2002), who previously argued that dominant vendors of e-learning systems in cyberspace act as gatekeepers that restrict creative freedom. Therefore, the creation of one's own best practices could be considered a strategy to express creative freedom in devising unique practices. Further findings in Section 3.3 likewise underscore the need for Qatar to craft its own unique frameworks to guide learners and educators in cyberspace.

7.2.3 Theme 3: Culture and the H.E. Vision, 2030 Achievement

With the third theme, the Researcher focussed on identifying the role that culture plays in attaining the 2030 higher education vision. In the proposed cyberspace culture framework (Figure 4-7), culture is represented as two theoretical perspectives: levels of culture (Schein, 1984, 1992) and dimensions of culture (Hofstede, 1984, 2010). However, when discussing culture's influence in higher education, participants considered its influence in general as opposed to from the two strict perspectives. The following section discusses the findings that were accordingly obtained.

7.2.3.1 Sub-Theme 1: Role of Culture

Based on the synthesis of the findings, three important insights were obtained, as detailed in Figure 7-14 below:



Figure 7-14: Perspectives on the influence of culture on the H.E. Vision, 2030 attainment. Source: The Researcher

As presented in Chapter 6, Sub-Section 6.3.3.1, participants noted that culture is highly important for achieving the higher education vision, as individuals in an organisational setting must be aligned with any change being introduced in order to implement it successfully. Additionally, participants reported that culture outside the organisation—that is, within the community and society—also influences the culture that is developed. Society's broader influence on cyberspace culture was similarly emphasised by Barzilai (2004), who described this as a global phenomenon that transcends geographical boundaries. In retrospect, the findings suggest that the 2030 higher education vision can only be achieved by aligning the culture of individuals and in higher education with the vision within both their organisations and society. This assertion is supported by previous work by Eliot (2014), who demonstrated that an individual's culture is influenced by the group to which they are attached—in this context, the organisation and society. As such, influencing higher education institutions to achieve the 2030 vision underscores the leaders' role in fostering a positive reinforcing culture both within the limits of their organisations and outside in society.

As presented in Chapter 6, Sub-Section 6.3.3.1, participants highlighted culture's reinforcing influence on achieving the 2030 vision. Additionally, they also cited numerous Qatar-specific challenges that hindered the fostering of a cyberspace culture to facilitate implementing the vision. To begin with, it was argued that the population's conservative social-cultural background would negatively influence trust behaviour regarding the use of technologies for educational purposes. For instance, participants claimed that the use of web cameras for learning would be argued to contravene participants' privacy. A second reported challenge was that, since culture is only formed after a long period of time, risks of resistance develop within that period, thus hindering overall progress in cultural development. A third challenge concerns the lack of basic computing skills and knowledge among university students-an aspect that further exacerbates efforts to foster positive cyberspace culture. An in-depth analysis of these issues emphasises the need for frameworks to facilitate the adoption of positive cyberspace culture. For instance, to address privacy concerns and freedom of expression, researchers such as Goi (2009) have suggested incorporating cultural diversity and freedom of expression to help develop the proper cyberspace culture frameworks. Figure 7-15 below summarises the Qatar-specific challenges of fostering cyberspace culture:



Figure 7-15: Challenges of fostering cyberspace culture in the Qatar context. Source: The Researcher

To tackle these challenges, participants cited several recommendations, such as emphasising the leaders' role in guiding the fostering of cyberspace culture. The participants highlighted that leaders in particular possess an important responsibility for educating and supporting individuals in adopting the proper cyberspace culture. This finding aligns with Schein (1984), who previously reported that leaders play an important role in guiding their members to adopt positive behaviour and foster positive culture. A second solution is to create awareness campaigns, while a third involves increasing collaboration between the Ministry and higher education institutions in order to equip more students with the required technical skills. Figure 7-16 below illustrates the recommended solutions for improving cyberspace culture in the Qatar context:



Figure 7-16: Solutions to foster cyberspace culture in the Qatar context. Source: The Researcher

From the synthesis of the findings, it can be argued that, in order to create a reinforcing cyberspace culture in Qatar, an appropriate framework is required due to the multiple factors identified. In the following section, the validity of the proposed cyberspace culture framework is examined from the participants' perspectives.

7.2.4 Theme 4: Proposed Cyberspace Culture Framework

The final theme investigated the effectiveness of the proposed cyberspace culture framework (Figure 4-7) for aligning cyberspace culture with the 2030 higher education vision. To this end, a focus group comprising six participants was moderated. As the Researcher had already determined the influence of each factor (theme) on the attainment of the higher education vision (Sub-Sections 7.2.1–7.2.3), this section focusses on identifying any additions to or criticisms raised against the proposed cyberspace culture framework. Therefore, this section discusses the additional participant-highlighted aspects that should be covered in the proposed cyberspace culture framework to further enhance it in facilitating the attainment of the 2030 higher education vision.

7.2.4.1 Sub-Theme 1: Alignment of External Elements

The first sub-theme focussed on identifying the alignment between external elements and the attainment of the 2030 higher education vision. The main findings are summarised in the following sections. First, the alignment of the Vision 2030 directives with the 2030 higher education vision is illustrated in Figure 7-17:



Figure 7-17: Alignment of Vision 2030 directives with the H.E. Vision, 2030. Source: The Researcher

The findings indicate that the strategic directives are aligned with the 2030 higher education vision through the provision of funding and exertion of external pressure by the Ministry. This exertion of pressure aligns with Cantwell, Coate, and King (2018), who cited that obligatory pressure is necessary because the government funds institutions, thus requiring them to be accountable for their operations. Likewise, the provision of funding has also been reported in the QNV 2030, as the Qatar government seeks to develop the population through investment in scientific research and first-rate education (Qatar National Vision, 2030, 2008). However, a noteworthy challenge is that higher education leadership remains fragmented in Qatar, thus making it difficult to align cyberspace culture with the 2030 vision.

The alignment of technological advancement with the 2030 higher education vision was further examined with the findings summarised in Figure 7-18 below:



Figure 7-18: Alignment of technological advancement with the H.E. Vision, 2030. Source: The Researcher

From the synthesis of the findings, it was observed that technological advancement is aligned with the 2030 higher education vision by enhancing learning and teaching. Such findings corroborate previous work by Talebian *et al.* (2014), who demonstrated that ICT technologies eliminate constraints of time and place in learning. Dunn and Kennedy (2019) likewise reported that technology-based learning enhances current educational models by shifting from teacher-centred to learner-centred approaches that are more self-paced and self-directed. Esteves, Matias, and Pereira (2018) added that collaborative learning mediated by technology allows learners to enhance soft skills such as oral communication and social interaction. Despite these benefits, however, two main challenges were reported in aligning technology with the 2030 vision: First, there is a need to consider courses that cannot be transformed online, and second, high bureaucracy levels were reported, as some projects have taken too long to launch.

The alignment of the Vision 2030 implementation framework with the 2030 higher education vision is displayed further in Figure 7-19:

Vision, 2030 implementation framework aligning to H.E. Vision, 2030

Improvements to enhance alignment

Need for performance measures to track progress

Figure 7-19: Alignment of the Vision 2030 implementation framework with the H.E. Vision, 2030.

Source: The Researcher

From the findings obtained, the participants argued that, in order to improve the alignment between the Vision 2030 implementation framework and the 2030 higher education vision, it is necessary to incorporate performance measures to facilitate progress tracking.

7.2.4.2 Sub-Theme 2: Alignment of Internal Elements

The second sub-theme further investigated the alignment of internal elements with the attainment of the 2030 higher education vision. The alignment of each individual factor is reviewed in the section below. To begin with, the alignment of leadership and management strategies is detailed in Figure 7-20 below:



Figure 7-20: Alignment of leadership and management strategies with the H.E. Vision, 2030. Source: The Researcher

The findings indicate that leaders in Qatar's higher education institutions have made substantial progress in aligning their leadership and management strategies with the 2030 vision by improving accountability and risk management. However, in order to further improve the alignment, the participants advocated for implementing frameworks to help identify possible risks as well as corrective actions to eliminate existing problems. The finding regarding the role of implementation frameworks has been highly emphasised in previous literature, such as in the development of the QNV 2030 by Qatar leadership to guide the implementation of strategic objectives for the four pillars—human, social, economic, and environmental (Qatar National Vision, 2030, 2008)—as well as in implementing different forms of technology-based learning, such as blended learning education (Garrison and Vaughan, 2013), blogging (Kerawalla *et al.*, 2009), and re-engineering of educational technology education (Kelley and Kellam, 2009).

Findings on security policy alignment with the 2030 higher education vision are illustrated further in Figure 7-20 below:



Figure 7-21: Alignment of security policy with the H.E. Vision, 2030. Source: The Researcher

The findings indicate that most higher education institutions have successfully aligned their security policies with the 2030 vision by complying with international standards, such as ISO. Such findings echo previous research works (Landow, 1992; Apple and Jungck, 1996; Raschke, 2003; Aarseth, 1997) arguing the need for continuous learning, as identities in cyberspace are dynamic and not static due to technologies continually morphing. However, several challenges were noted, and recommendations were accordingly provided to improve the alignment of security policy with the 2030 higher education vision. The recommendations include providing more training and security awareness campaigns, providing security information, and fostering a security culture among employees.

Finally, the alignment between security technical best practices and the 2030 higher education vision is detailed in Figure 7-22 below:



Figure 7-22: Alignment of security policy with the H.E. Vision, 2030. Source: The Researcher

The findings demonstrate that, in most higher education institutions, modern best practices such as backing up data have been adopted. This finding matches those of researchers such as Gregg *et al.* (2017), who previously reported that most higher education institutions rely on major established tech vendors to facilitate implementing e-learning platforms due to their innovativeness and security. However, to improve alignment, the participants further cited the need to create security policies to improve the adoption of best practices.

7.2.4.3 Sub-Theme 3: Alignment of Culture

The final sub-theme considered the alignment of culture with the attainment of the 2030 higher education vision. Figure 7-23 summarises the findings as follows:



Figure 7-23: Alignment of culture with the H.E. Vision, 2030. Source: The Researcher

The findings illustrate that the participants did not cite any successful incidences of the alignment of culture with the 2030 higher education vision. However, they emphasised the need to develop implementation plans to foster the proper cyberspace culture. Likewise, they also mentioned the need to educate employees regarding the value of adopting the proposed changes when a new culture is to be developed.

7.3 Validated Proposed Cyberspace Culture Framework

From the synthesis of the empirical research findings detailed in this chapter, the refined proposed cyberspace culture framework is presented in Appendix B. Furthermore, propose refinements from the analysis of empirical findings are provided alongside the individual elements in the originally proposed cyberspace culture framework (Figure 4-7).

Based on the analysis of the findings from the interviews and focus groups, several amendments to the proposed cyberspace culture framework were recommended (Figure 4-7).

7.3.1 Internal Elements

Concerning the internal elements, the participants confirmed that accountability and risk management represent important factors in leadership and management strategies. However, an additional factor was highlighted concerning the provision of an implementation framework to

guide the execution of various strategies. With the security policies, participants acknowledged that security awareness, continuous learning, and attitudes and behaviour influence the attainment of the higher education 2030 vision. However, to enhance their impact, there is a need to improve training and security awareness campaigns, provide security information to employees, and develop a reinforcing culture among employees. Finally, to further enhance security technical and best practices, participants cited that policies must be created that improve the adoption of the practices.

7.3.2 External Elements

Regarding the external elements, participants confirmed that the Vision 2030 directive, technological advancement, and the Vision 2030 implementation framework are important for achieving the Vision 2030 for higher education. To enhance the Vision 2030 directive, it was suggested for leaders to enhance collaboration with one another in different higher education organisations. Likewise, to improve technological advancement, participants mentioned the need to address courses that are not transformable online. Similarly, they also highlighted a need to address bureaucracy in the implementation of projects. Finally, to improve the Vision 2030 implementation framework, it was suggested to include performance measures in order to assess the overall progress made over time.

7.3.3 Culture

With culture, participants argued that implementation plans must be developed to help overcome resistance to the proposed challenges. Additionally, it was suggested that employees should be educated regarding the value of proposed changes to encourage overall adoption.

7.3.4 Refinement of the Proposed Cyberspace Culture Framework

Empirical findings contributing to the proposed framework are incorporated in this section to refine the framework further, thereby ensuring that it facilitates attaining the higher education 2030 vision. In the originally proposed cyberspace culture framework (Figure 4-7), the Researcher identified three constructs as influencing the implementation of the higher education 2030 vision:

- External elements Government's role, advanced technology, and the Vision 2030 implementation framework (Sub-Section 4.7.2).
- Internal elements Leadership and management strategies in higher education, security policies, and technical best practices (Sub-Section 4.7.1).

3. Cultural influences – Dimensions and levels of culture (Sub-Section 4.7.3).

Synthesis of the findings from the focus group discussions revealed that, in general, the participants agree that each of the constructs influences the attainment of the higher education vision. The evaluation of the findings in the chapter further reveals how each of the factors influences the attainment of that vision and the overall achievement of cyberspace culture. Such findings directly indicate the importance of cyberspace culture and partly answer the third research question. However, several gaps or aspects were identified where the framework could be refined further to facilitate attaining the higher education 2030 vision. Furthermore, refining the proposed cyberspace culture framework helped the Researcher to address the four research questions advanced in the research, as attaining the higher education vision is linked to achieving a positive cyberspace culture in Qatar. The refined proposed cyberspace culture framework is detailed in Figure 7-24 below:



Figure 7-24: Validated proposed cyberspace culture framework.

Source: The Researcher

The side-by-side comparison of the proposed cyberspace culture framework (Figure 4-7) and validated proposed cyberspace culture framework (Figure 7-24) reveals that all components in the original framework have been maintained. However, in the validation of the framework, the bidirectional relationship between the external elements and the attainment of the higher education vision was modified to one that is uni-directional. This is because respondents argued that the achievement of the higher education 2030 vision would not influence aspects such as technological advancement or information security policies, as they are already independent.

The first research question, for instance, sought to identify the strategic alignment of cyberspace culture with the country's strategic directions. As these strategic directions are linked to the QNV 2030 and its implementation framework, the Researcher addressed this question by identifying how external elements (Vision 2030 directives and Vision 2030 implementation framework) influence the attainment of the higher education 2030 vision. The main findings indicate that the government's external pressure does influence the attainment of the higher education 2030 vision through obligatory and positive pressure, which ensures that institutions adhere to the guidelines and regulations set by the government in addition to achieving their own objectives. Likewise, government funding would enable institutions to acquire technologies and improve their sustainability and overall viability. However, a further refinement was also suggested—namely, fostering more enhanced collaboration among leadership of higher education institutions, as this remains fragmented. Technology was also observed to influence the achievement of the higher education 2030 vision by enhancing learning, teaching, and management of institutions. To further enhance the role of technology, a few refinements were suggested, such as eliminating bureaucracy in the implementation of technology projects and developing strategies to address courses that are not transformable into online platforms. Furthermore, the findings revealed that the Vision 2030 implementation framework serves an important role in guiding higher education institutions. To enhance the framework further, performance measures were suggested to track overall progress. By adding these refinements, the Researcher argues that the country's strategic directions will be aligned with the cyberspace culture.

Beyond this, several refinements were also identified regarding the internal elements, which included the leadership and management strategies in higher education, security policies, and technical best practices. Regarding leadership and management of higher education institutions, the findings reveal that this factor has influenced the attainment of the higher education 2030 vision, as leaders have become more accountable and manage risks in their organisations. However, a persistent gap is noted concerning the lack of implementation frameworks to facilitate attaining various strategic objectives. Second, security policy was also found to positively influence the achievement of Vision 2030, as institutions have become more aware of cybersecurity risks and have also complied with global security standards, such as ISO. However, to further enhance the role of the Vision 2030 implementation framework, recommendations such as improving training and security awareness, availing security information, and reinforcing security culture among employees were made. With respect to best practices, refinements such as creating reinforcing policies to guide their adoption in higher education institutions were suggested. In relation to the final construct on culture, the following refinements were suggested: Developing implementation plans to overcome employees' resistance and educating employees regarding the value of the proposed changes.

Continuing, refining the external and internal elements of the proposed cyberspace culture framework and cultural factors helps address the fourth research question in the research, which sought to identify the type of cyberspace strategy that the government should deploy to align its strategic directions. Additionally, findings revealed an existent bi-directional relationship between the constructs and the attainment of the higher education 2030 vision. With the external elements, the findings indicate that, as the individual factors (government strategic directives, technological advancement, and Vision 2030 implementation framework) influence the achievement of the higher education 2030 vision could lead to improved strategies for implementing technologies for teaching, learning, and overall management of the institutions. Such aspects subsequently emphasise the current bi-directional relationship between the constructs and the achievement of the higher education 2030 vision.

7.4 Implementation Guidance

As presented in Chapter 1, Section 1.4, the primary goals of this research were to investigate the strategic approach to cyberspace culture and its alignment with Qatar's strategic direction, as well as to develop an effective framework for aligning cyberspace culture with the vision of Qatar's strategic direction, supported by implementation guidance focussed primarily on the higher education sector. In this section, a road map is presented detailing the various ways in which the proposed framework can be utilised.

7.4.1 Role of the Government (Vision 2030 Directive)

Based on the evaluation of the findings, the following contributions can be attributed to the government (Vision 2030 directive):

- Provide funding to support the operations of higher education institutions, such as to set up more institutions; expand their scientific research departments; and acquire necessary technology and infrastructure to support learning, teaching, and management of higher education institutions. This arises from the fact that all respondents agreed that funding is crucial to the success of higher education and the achievement of Vision 2030.
- Increase collaboration with higher education institutions in formulating policies and developing frameworks that guide the implementation of different aspects, such as advanced security features that meet international standards in the institutions. In the research, participants argued that leadership in higher education institutions is fragmented, thus derailing the standardisation of policies and implementation frameworks. Therefore, there is a need to increase collaboration between higher education institutions and the government in order to facilitate the development of policies and frameworks.
- Serve as an oversight authority that supervises the actions of all stakeholders in the higher education sector in order to boost fair treatment and prevent any nefarious activities, such as cybercrime. As the government represents a key stakeholder in cyberspace, its oversight authority can ensure that all higher education institutions receive fair treatment in terms of resource allocation and research opportunities.
- Formulate regulation, which fosters a conducive and enabling environment for higher education institutions to avoid unprecedented effects from environmental uncertainty, such as abrupt changes in the institutions' leadership. This is because respondents argued that

environmental uncertainty arising from political influences can derail the attainment of the higher education 2030 vision.

 Develop KPIs for the QNV 2030 to facilitate the evaluation of its progress over time and to further outline milestones that help track progress. Respondents in the research cited that, although the QNV 2030 is well-meaning in enhancing the economy, it still lacks milestones to track its progress. Such performance measures should be instituted to guide higher education institutions in assessing the progress of attaining their own vision.

7.4.2 Role of Higher Education Leadership

To achieve the higher education 2030 vision, the following roles are advocated for the leadership in higher education institutions, as leadership plays an important role as an internal element of the proposed cyberspace culture framework:

- Collaborate in developing frameworks to enable leaders to implement different strategic objectives. This assertion arose from findings in the research, which demonstrated that leaders are challenged, and thus require guidance in achieving strategic objectives.
- Provide necessary training and resources for leaders in higher education institutions to equip them with skills necessary to guide employees towards achieving strategic objectives. This is because respondents highlighted that leaders require support in terms of training to enable them to attain strategic objectives.
- Contextualise security best practices to Qatar's social-cultural environment in order to guarantee their success in local higher education institutions. The participants argued that adopting the proper security and technical best practices was not enough, as they would only succeed where they aligned with the culture of Qatari higher education institutions.
- Tackle the challenge of bureaucracy associated with updating existing security policies to align with requirements in the market. Respondents argued that the approval process involved in updating security policies is highly bureaucratic and can derail actual approvals. Therefore, it is expedient to fast-track the process in order to ensure that security policies adhere to market demands.

7.4.3 Role of the Culture in Organisations

The culture of higher education institutions plays an essential role in achieving the higher education 2030 vision. The following directives are accordingly advocated to ensure that culture is well cultivated to facilitate the achievement of that vision:

- Ensure that leaders in higher education institutions take responsibility for educating employees on cyberspace aspects in order to foster a positive cyberspace culture and enable them to adopt appropriate change-management practices.
- Promote campaigns to facilitate social behaviour change among employees in higher education institutions in an effort to enable them to develop the proper type of cyberspace culture. The reported findings indicate that most negative actions in cyberspace arise from the fact that employees remain unaware of their impact. Therefore, there is a need to undertake social campaigns to promote behaviour change among employees.
- Improve collaboration between the Ministry and the higher education sector in order to equip employees with better technology skills.

7.5 Conclusion

Based on the analysis of empirical findings in this chapter, several fundamental findings have been reported: First, each of the individual elements in the proposed cyberspace culture framework (internal, external, and culture) appear to influence the attainment of the 2030 higher education vision. With the external elements, the government's oversight authority as a regulator and financier of Qatar's higher education sector possesses significant implications for the industry's growth. Qatar's Higher Education Ministry also plays a role in promoting the adoption of cyberspace culture by providing funding to acquire necessary technologies and by facilitating partnerships between higher education framework also serves as an important guide to aid the development of cyberspace culture by specifying what actions should be taken in institutions. Technological advancement, meanwhile, both enhances the management of learning institutions.

Therefore, the external environment in which higher education operates considerably influences the attainment of the 2030 higher education vision as well as the overall creation of cyberspace

culture. However, to further improve the alignment of external elements with the 2030 vision for higher education, collaboration among leaders in higher education must be improved, as leadership currently remains fragmented. Additionally, performance measures should also be incorporated into the Vision 2030 implementation framework in order to facilitate the tracking of overall progress. Likewise, to enhance the adoption of technology, bureaucracy must be eliminated to fast-track such processes. Furthermore, courses that are not directly transformable into online platforms should be reviewed in order to identify optimal strategies to launch them into cyberspace.

Internal elements such as leadership and management strategies also influence the attainment of the 2030 higher education vision, as leaders steer their organisations to achieve targeted objectives. However, it was observed that implementation frameworks are required to help facilitate the process. Security policies are further observed to influence employees' security behaviours and attitudes as well as to raise awareness of security matters in higher education institutions. However, their implementation remains challenged by the lack of training and security knowledge. These factors should be addressed to improve overall attainment of the 2030 vision. Moreover, security best practices should be implemented alongside security policies to guide their adoption in the institutions. Finally, to improve the culture's influence on attaining the 2030 higher education vision, implementation frameworks must be adopted that help minimise the threat of resistance from employees. Likewise, educating employees on the value of new changes in cyberspace should also be emphasised to improve alignment with the 2030 vision.

In summary, the proposed cyberspace culture framework illustrated in Figure 7-24 has been refined further to facilitate attaining the 2030 higher education vision and to aid the formation of cyberspace culture.

Chapter 8: Conclusion and Recommendations

8.1 Introduction

In this chapter, the findings obtained in the research are synthesised in order to outline evidence regarding how the aim and objectives of the research were fulfilled. In particular, the Researcher describes how the research aims were fulfilled as well as the investigation process adopted to address research objectives and research questions. Additionally, the research's contributions to theory and practice and its potential impact are outlined. Beyond this, limitations that challenged this research are also described.

The chapter is sub-divided into seven sections: In Sections 8.2, 8.3, and 8.4, the Researcher describes how the results address the research aims, objectives, and questions. Limitations that challenged the research are further outlined in Section 8.5. Thereafter, in Sections 8.6 and 8.7, the research's contributions to practice and theory are discussed, and Section 8.8 highlights various recommendations from the research. Finally, Section 8.9 details the main conclusion of the research.

The following sections describe how the research results were identified in the research aims, objectives, and questions.

8.2 Results Determined by Research Aim (Chapter 1, Section 1.4)

In Chapter 1, Section 1.4, the research aim guiding this research undertaking was detailed as follows: To investigate the strategic approach to cyberspace culture and its alignment with Qatar's strategic direction and to develop an effective framework for aligning cyberspace culture with the vision of Qatar's strategic direction, supported by implementation guidance focussed primarily on the higher education sector. At an in-depth level, the research aim can be broken down into two parts:

Part 1: The first part of the research aim focusses on investigating the strategic approach to cyberspace culture and its alignment with Qatar's strategic direction. In addressing the research aim, and due to the lack of literature on the subject, this thesis began by conceptualising its own definition of cyberspace culture by merging the concepts of cyberspace and culture. To identify the strategic approach to developing cyberspace culture, previous literature and diverse

information security frameworks were examined. The reviewed literature provided important insights into the conceptualisation of the strategic approach to cybersecurity, cyberspace management, and information security culture. With this understanding, various strategic components of cyberspace culture were outlined, and a proposed cyberspace culture framework was developed. Findings from the respondents further revealed an alignment between the proposed strategic approach to cyberspace culture and the strategic directions in Qatar as guided by the pillars of the QNV 2030. Similarity was also observed between the findings from primary research and prior literature. However, several differences were also reported, considered the research contributions.

Part 2: The second part of the research aim emphasises the development of an effective framework for aligning cyberspace culture with the vision of Qatar's strategic direction, supported by implementation guidance focussed primarily on the higher education sector. With the first part of the research aim addressed, various components regarding the strategic approach to cyberspace culture were identified. Likewise, the examination of existing information security frameworks revealed various existent gaps that were yet to be addressed as well as an apparent lack of implementation guidance for the higher education sector. The Researcher accordingly synthesised the findings from the literature review (derivate one), information security frameworks (derivate two), and research gaps in the reviewed frameworks (derivate three) to conceptualise the proposed cyberspace culture framework (Figure 4-7). The developed framework included three elements: a) external elements (Vision 2030 directive, technological advancement, and Vision 2030); b) internal elements (leadership and management strategies, security policy, and security technical and best practices); and c) cultural influence. Findings from the respondents revealed that the framework is effective in aligning cyberspace culture with Qatar's strategic direction. Additionally, respondents highlighted important refinements to the proposed framework in order to enhance it and facilitate attaining the higher education 2030 vision. Implementation guidance for the developed framework was also provided whereby the actions of various actors were outlined in the higher education sector.

In achieving the research aim, this thesis highlights various strategic components of cyberspace culture by deriving a cyberspace culture framework in Figure 4-7 and further refining it in Figure

7-24. This framework is supported by findings from previous literature and seeks to bridge the research gaps in current information security frameworks. The findings from respondents further revealed that the framework is effective in aligning cyberspace culture with the vision of Qatar's strategic direction, thereby validating it. In the next section, the results are detailed per research objective.

8.3 Results Determined by Research Objectives (Chapter 1, Section 1.5)

This section presents a discussion of the results obtained by addressing the research objectives formulated in Chapter 1, Section 1.5.

Research Objective 1: The first research objective (Chapter 1, Section 1.5) sought to evaluate the alignment of existing cyberspace culture with the strategic vision in Qatar. However, to identify this alignment, it was important to define cyberspace culture and strategic vision. In Chapter 1, debates among academics and various practitioners regarding the emergence of cyberspace as a defining aspect of modern life were accordingly highlighted. Similarly, various components of cyberspace ranging from telecommunications devices, physical infrastructures, and computerised systems to associated software, networks linking computer systems together, and the data residing in those systems were identified. Cyberspace culture's role in guiding the actions of actors in the digital space was also delineated based on its definition as a set of beliefs that emerge through computer-mediated interaction.

In Chapter 2, the review of literature on cyberspace culture highlighted various advances that have emerged in the field as well as challenges affecting its evolution. Additionally, the review of the QNV 2030 led to the identification of various strategic visions of the Qatar government in transforming its economy into one that is knowledge-based. Therefore, to address this objective, focus was placed on identifying the alignment between cyberspace culture as a set of beliefs and the strategic visions specified in the QNV 2030. However, the findings from previous literature revealed several gaps in the alignment between existing cyberspace culture and the QNV 2030's strategic visions regarding aspects such as a lack of clear governance, guidance on the application of cyberspace, and the development of infrastructure.

In Chapter 3, similar findings were observed: Specifically, the review highlighted current literature research gaps in cyberspace culture and the attainment of strategic visions in higher education. Notable research gaps included the lack of guidance in using cyberspace for authorship, teaching, and learning as well as in the selection of best practices to facilitate attaining strategic visions. Therefore, the key findings emphasised that research gaps existed regarding the alignment of current cyberspace culture with strategic visions. A lack of guidance in using cyberspace, authoring content, and best practices represent some of the highlighted issues.

Research Objective 2: The second research objective (Chapter 1, Section 1.5) identified the strengths and weaknesses of the current cyberspace culture frameworks and their alignment with the strategic directions of the country (Qatar). This objective was met by examining previous frameworks detailed in Chapter 4. To this end, four frameworks were examined: 1) the ISF framework (Alnatheer and Nelson, 2009); 2) the ISCF framework (Da Veiga, 2008); 3) the Information Security Culture framework (Shabbaan, 2014); and 4) the Information Security Framework - STOPE (Al-Hogail, 2015). The evaluation of the different cyberspace culture frameworks subsequently revealed that each framework was effective in its own way at aligning with the strategic directions of different organisations. The ISF framework, for instance, identifies the role of leadership, government regulation, security culture, management, and organisation culture in fostering cyberspace culture. However, an emergent gap was observed in the frameworks in that they do not focus on the role of the human factor in fostering cyberspace culture; rather, they primarily focus on technical aspects. Furthermore, in Chapter 4, Table 4-3, the results suggest that none of the frameworks detail all aspects required to address existent cyberspace culture gaps highlighted in Chapter 2. Therefore, the frameworks fail to align with the strategic directions in the country, thereby requiring the development of the proposed cyberspace culture framework.

In Section 3.3, meanwhile, several literature gaps were identified regarding cyberspace culture and higher education. The first concerned the lack of frameworks guiding both parties in aligning cyberspace culture to the established goals. A second research gap concerned whether Qatar would rely on dominant entities to supply educational technology in order to facilitate attaining the QNV 2030 in higher education. Third, a research gap was also identified in guiding authorship among teachers in the digital landscape. Therefore, to address the second and third research gaps, the

280

Researcher identified a gap in creating frameworks and best practices that will foster positive cyberspace culture to boost the attainment of the QNV 2030. Beyond this, in quadrant IV, a fourth research gap was also identified concerning how cyberspace influences identity formation.

The results obtained for research objectives 1 and 2 were argued to address the various research gaps identified in the literature. For instance, in addressing the first gap, the findings revealed that, although frameworks existed for aligning cyberspace culture to strategic goals, they primarily focussed on technical aspects while ignoring the impact of the human factor. Therefore, this required developing a cyberspace culture framework. In turn, this helped address the second and third research gaps in the literature. By extension, the developed framework would also facilitate addressing cyberspace's influence on identity formation, thereby solving the fourth research gap.

Research Objective 3: The third research objective (Chapter 1, Section 1.5) analysed the current strategic approach to cyberspace culture and its alignment with the country's strategic directions. To address the objective, both literature findings and primary research were synthesised. From the literature review, it emerged that existent cyberspace culture frameworks are not effective in aligning with the country's strategic directions, as most authors focus on the technical aspects of cyberspace and direct less attention to human factors. As the review of the QNV 2030 revealed that the human development pillar is important for achieving Vision 2030, the failure of current frameworks to address human aspects in cyberspace culture means that the approach is ineffective in aligning with the country's strategic approach to cyberspace culture, citing aspects such as a lack of clear guidelines for using cyberspace and a lack of leadership and governance in cyberspace. The findings consequently emphasised the need to develop a more effective cyberspace culture framework to capture all necessary requirements that would ensure alignment with the country's strategic directives.

Research Objective 4: The fourth research objective (Chapter 1, Section 1.5) evaluated the Qatar Vision, 2030 cyberspace strategy and how it is influenced by attaining the higher education 2030 vision. In Chapter 2, Section 2.4, the review of the literature concerning the QNV 2030 revealed four pillars of the Vision 2030 strategy: human, economic, social, and environmental. Further

review indicated that cyberspace culture is linked to the attainment of the QNV 2030's strategic objectives concerning economic development, delivery of social services, and education development. Given that higher education serves as the engine of the economy in delivering skilled workers to drive businesses and social services in Qatar, the QNV 2030's cyberspace strategy is directly influenced by the achievement of the higher education 2030 vision. Additionally, findings from the participants further confirm that attaining the higher education vision in terms of using technology to deliver learning, to teach, and to manage higher education institutions improves higher education institutions' effectiveness in upskilling the population, and thus in driving a knowledge-based economy.

However, in Chapter 3, a literature evaluation revealed gaps in the current cyberspace strategy adopted in higher education institutions in Qatar. For instance, they lack frameworks to align cyberspace culture with strategic goals, to guide the actions of educators and learners in the digital space, and to enable novice learners to enhance their expertise in cyberspace. Addressing these gaps through the proposed cyberspace culture framework (Chapter 4, Figure 4-7) would thus ensure not only the attainment of the higher education vision, but also the achievement of the QNV 2030's strategic goals.

8.4 Results Determined by Research Questions (Chapter 1, Section 1.6)

The research questions re-addressed the research objectives and further focussed on the research aims. In this section, a discussion is undertaken regarding the results achieved from addressing the research questions.

Research Question 1 (Chapter 1, Section 1.6): *How does cyberspace culture strategically align with the country's strategic directions?*

This research question re-addresses the first research objective. In the context of Qatar, it emerged that cyberspace culture does not effectively align with the country's strategic directions, as several research gaps were identified. The evaluation of the QNV 2030, meanwhile, revealed that, in achieving its strategic objectives of transforming Qatar's economy from one that is hydro-carbon-based into one that is knowledge-based, the country primarily focusses on technical measures, such as a) investment in necessary infrastructure and technologies and b) collaboration with

different international stakeholders. Additionally, it was observed that Qatar has made important strides in enhancing aspects such as cybersecurity by passing important laws and developing frameworks, such as the Qatar 2022 Cybersecurity framework in preparation for the FIFA 2022 World Cup.

However, despite Qatar's progress in acquiring necessary technological infrastructure and enhancing cybersecurity, the review further demonstrated that important concerns regarding the human factor in cyberspace have not been adequately addressed. Specifically, the evaluation of Qatar's cyberspace strategy revealed gaps in guiding the actions of learners, educators, authors, and scientists in cyberspace, which must be addressed to effectively align cyberspace culture with the country's strategic directions.

Research Question 2 (Chapter 1, Section 1.6): *How are current cyberspace culture frameworks aligned with the country's strategic directions?*

To address this research question, current cyberspace culture frameworks (Chapter 4) were examined to establish their alignment with the country's strategic directions. To this end, four frameworks were examined: 1) the ISF framework (Alnatheer and Nelson, 2009), 2) the ISCF framework (Da Veiga, 2008), 3) the Information Security Culture framework (Shabbaan, 2014), and 4) the Information Security framework – STOPE (Al-Hogail, 2015). As Qatar's strategic directions primarily focus on transforming the economy into one that is knowledge-based, the evaluation of the frameworks revealed that none of the frameworks address all requirements needed to foster cyberspace culture. The targeted requirements include 1) leadership, 2) policy, 3) technical competency, 4) government regulation, 5) organisation culture, 6) management, 7) security culture, and 8) change management. Analysis revealed that the frameworks largely ignore the human aspect in cyberspace, which is emphasised in the country's strategic directions in aspects such as 1) delivering learning and teaching in cyberspace, 2) authorship in cyberspace, and 3) using cyberspace to guide novice learners to attain expertise in cyberspace.

Research Question 3 (Chapter 1, Section 1.6): *Why is cyberspace culture important, and what literature gaps exist regarding cyberspace culture?*

Based on the evaluation of literature, cyberspace culture was identified as crucial, primarily in guiding the actions of actors in cyberspace. In the context of higher education in Qatar, this largely includes learners, educators, and management who rely on computer-mediated technology to achieve their strategic objectives in higher education institutions. Aside from higher education, cyberspace culture is also influential in facilitating the service industry, such as in the delivery of e-government services to SMEs and the population. In other economic sectors, businesses leverage positive cyberspace culture to enhance decision-making and boost economic competitiveness. Therefore, different types of businesses (higher education, service industries) rely on positive cyberspace culture to achieve their strategic objectives. However, notable gaps in literature were also identified, such as the lack of frameworks to align cyberspace culture with strategic objectives and to ensure the protection of actors in cyberspace.

Research Question 4 (Chapter 1, Section 1.6): What type of cyberspace strategy should the government deploy to align with the country's strategic directions?

To address this research question, the findings from primary research and the literature were examined. The literature evaluation in Section 3.3 accordingly revealed a need for the government to develop implementation frameworks to guide the use of cyberspace as a learning tool for educators and learners. These frameworks would provide clarity on aspects such as content delivery, authorship in the digital landscape, and the adoption of best practices concerning the use of technology to achieve strategic objectives. Participants further added that it is important for the government to provide support for the acquisition of necessary technological infrastructure in higher education institutions through funding. The government's role in creating a conducive operational environment was also emphasised through aspects such as passing regulation to ensure fairness and enhancing collaboration with external parties from international regions.

8.5 Research Limitations

Despite addressing the research objectives, the research remained limited by several factors. In this section, the limitations encountered in the literature search and the research design are detailed. It is worth noting that in Chapter 1, Section 1.9, the Researcher outlined the limitations that were

anticipated in conducting the research. This section, however, discusses the limitations that were actually encountered in completing this research.

8.5.1 Limitations in Literature Search

In the literature-examination process, the research was limited by the lack of similar empirical studies on cyberspace culture that had been conducted in Qatar's higher education. Specifically, the Researcher noted a lack of studies that provide an official definition of cyberspace culture as well as previous research that examines the thematic issue in higher education. To compensate for this limitation, the researcher developed a definition of cyberspace culture (Sub-Section 2.2.5) that was utilised in the context of this thesis. Likewise, due to the limited number of available literature regarding cyberspace culture in Qatar, the researcher expanded the scope to consider studies that had also been undertaken in Gulf region countries, such as Saudi Arabia.

A second limitation is that, although the Researcher focussed on gathering literature from the past three decades (1990 to date), the constraint of the research timeline restricted this thesis to focussing primarily on recent literature in the current decade. This limitation arose from the fact that most studies detailing the use of cyberspace for learning and teaching concentrated on the current decade (2010 to 2020). Therefore, there was a likelihood of research bias, as literature from the past two decades was scant. However, the Researcher acknowledged the restriction and focussed on ensuring that a balanced literature review was undertaken.

8.5.2 Limitations in Research Design

The Researcher largely collected qualitative data by conducting semi-structured interviews and moderating a focus group. The first limitation encountered in this regard pertained to the COVID-19 pandemic, which made it difficult to conduct one-on-one sessions with the selected participants. To overcome this limitation, the Researcher utilised online tools such as Microsoft Online Calls to engage with the participants.

Second, the COVID-19 pandemic also made it difficult for the researcher to obtain a larger sample for the research purpose. This is because the onset of the pandemic led to the closure of most higher education institutions and a consequent difficulty in accessing their senior staff, who were the target audience for this research. However, the Researcher overcame this limitation by contacting

the research participants directly via email. As a result, 18 participants were interviewed, and a focus group comprising of six members was moderated.

Third, the reliance on solely qualitative data also introduced a limitation of possible subjectivity and the risk of Researcher bias. However, as the Researcher acknowledged this restriction, secondary literature was also reviewed in order to confirm the results from primary data. Finally, the proposed cyberspace culture framework was also limited due to having not been empirically tested in the real-life context. As a result, it relies solely on empirical results and previous literature to validate the research.

8.6 Research Contribution

This section details the theoretical and practical contributions identified from this research undertaking. Theoretical contributions generally describe the new insights and knowledge derived from the research whereas practical contributions comprise the actionable insights that can be adopted by stakeholders in Qatar's cyberspace industry.

8.6.1 Theoretical Contribution

The current research makes key contributions to theory in terms of understanding cyberspace culture and framing its association with the achievement of strategic objectives.

- Broader understanding of cyberspace culture: This research acknowledges the lack of diverse literature related to cyberspace culture. However, by examining a wide range of literature sources, this thesis provides a broader understanding of cyberspace culture and its influence on the achievement of strategic objectives.
- 2) Theoretical reframing of the relationship between cyberspace culture and strategic directions: This research provides a holistic understanding of the influence that cyberspace culture exerts on the achievement of strategic objectives set by either organisations or the government as a whole. This research further revealed that cyberspace culture is important not only in guiding the actions of actors in an abstract digital space, but also for ensuring that their sensitive information remains protected from threats in the digital landscape.
- 3) Identification of knowledge gaps: This thesis demonstrates that current cyberspace culture frameworks neglect the human aspect in cyberspace by failing to guide the attitudes and actions of individuals in the digital landscape. Additional literature gaps regarding the

development of implementation frameworks were also emphasised, thereby highlighting the need to develop more effective frameworks to guide the alignment of cyberspace culture and strategic objectives.

4) Development of a conceptual framework: This research developed the proposed cyberspace culture framework (Chapter 4, Figure 4-7), which indicates the various core elements required to facilitate the attainment of the higher education 2030 vision and to consequently foster a positive and reinforcing cyberspace culture.

8.6.2 Practical Contribution

This section details actionable insights that can be directly adopted by stakeholders in Qatar's higher education sector regarding the utilisation of cyberspace. These insights include a range of factors from the validation of the cyberspace culture framework to the government's role in guiding the development of cyberspace culture.

- Development of a cyberspace culture implementation framework: The research developed an implementation framework to guide the attainment of the higher education 2030 vision. This framework provides implementation guidance to facilitate the development of cyberspace culture and further outlines the responsibilities of different stakeholders in cyberspace, such as the government and management of higher education institutions.
- 2) Validation of the cyberspace culture framework in achieving the higher education 2030 vision: In this research, the proposed cyberspace culture framework was validated from the analysis of the interviewee responses and focus group discussions. The direct implication is that the research highlights two key stakeholders—the government (Vision 2030 directive) and higher education leadership—as the most important players for achieving the higher education 2030 vision. Implementation guidance is subsequently provided in Chapter 7, Sub-Section 7.4.1 for the government and Chapter 7, Sub-Section 7.4.2 for higher education leadership. For instance, identified guidance for the government includes funding, increasing collaboration with international parties, and acting as an oversight authority for the institutions. Likewise, guidance for higher education leadership best security practices and enhancing training.

- 3) **Conclusion that current cyberspace culture frameworks are ineffective in aligning strategic objectives:** This research demonstrates that currently utilised cyberspace culture frameworks are highly technical in nature, and thus fail to address concerns regarding the human factors in cyberspace. The proposed cyberspace culture framework, however, emphasises the need for culture in guiding the attitudes, actions, and behaviour of employees. In effect, the proposed framework addresses the existent gap regarding the lack of human factors in fostering cyberspace culture.
- 4) Empirical evidence of the government's role in the development of cyberspace culture: This research further indicates that, in order to achieve the QNV 2030, the government must play a leading role in funding higher education institutions, serve in a supervisory role to guide institutions, pass fair regulations to facilitate the uptake of best practices, and facilitate collaboration with other international stakeholders to enhance cyberspace culture in Qatar.
- 5) Empirical evidence of the role of organisational culture in reinforcing cyberspace culture: Next, this research revealed that, at the foundational level, cyberspace culture can only be fostered when individual attitudes and behaviours are modified towards the change. To that end, the role of organisational culture in guiding employee cyberspace actions is emphasised in this research.
- 6) Link between the attainment of the higher education 2030 vision and the QNV 2030: This research also suggests that attaining the higher education 2030 vision directly influences the successful achievement of the QNV 2030 because better education systems serve as the engine of the economy. Furthermore, the population becoming more learned and competent in leveraging cyberspace to achieve strategic objectives has also fuelled the QNV 2030, thereby leading to the economy's transformation into one that is knowledge-based. In this regard, the achievement of higher education goals facilitates attaining the QNV 2030 by producing a highly learned population that can manage diverse aspects of the economy by leveraging advanced technology.

8.7 Recommendations

The research has proposed the cyberspace culture framework as a solution to facilitate achieving the higher education 2030 vision and, in turn, to foster positive cyberspace culture and achieve the

QNV 2030. The current research further validates the proposed cyberspace culture framework through the collection of empirical data (a focus group discussion) and a review of previous literature. Through the validation process, this framework was refined by modifying the bidirectional relationships between the external elements and the achievement of the Vision 2030 into a uni-directional alternative. Therefore, the first recommendation concerns an increased focus on the validated cyberspace culture framework by stakeholders involved in fostering cyberspace culture (government and higher education institutions) in order to leverage the framework to achieve targeted strategic objectives.

Continuing, this research advocates for different stakeholders to adopt implementation guidance associated with the proposed cyberspace culture framework. Therefore, the second recommendation is for stakeholders to not only focus on the validated cyberspace culture framework, but also to adhere to the implementation guidance it proposes. The government, for instance, is recommended to take a leading role in guiding the development of cyberspace culture through funding higher education institutions, facilitating collaboration with international stakeholders, and passing regulations that promote fairness and the use of technology. Leadership in higher education institutions should also focus on being accountable and avoid implementing risk-generating strategies.

However, this research has also revealed that the adoption of cyberspace for educational advancement remains challenged by negative aspects such as cybercrime, cyberbullying, and cyber espionage. This suggests that, even as higher education institutions seek to adopt cyberspace to achieve their strategic objectives as well as Vision 2030, there remains a further need to invest in security best practices in order to ensure that the institutions' resources and actors in cyberspace remain protected from the increased threats. As such, the third recommendation emphasises the role of security preparedness for higher education institutions before migrating all their educational and managerial activities to cyberspace.

8.8 Recommendation for Future Research

Despite this research proposing and validating a cyberspace culture framework to align cyberspace culture with strategic objectives, future scholars should build on the current framework and test its components in real-world settings. This will subsequently ensure that the framework is evaluated
in terms of its effectiveness in aligning cyberspace culture and strategic objectives in higher education. Additionally, future scholars should subsequently test the proposed cyberspace culture framework in different higher education settings, such as universities, colleges, and tertiary institutions, in order to assess its effectiveness in facilitating the achievement of the higher education 2030 vision and to identify aspects that can be improved further.

Continuing, the review of the literature revealed a dearth of studies on cyberspace culture in the context of Qatar higher education. This suggests a need for future researchers to also focus more keenly on this topic in Qatar given the government's increased investment in higher education and adoption of a knowledge-based economy. Furthermore, the majority of studies that were examined indicate that considerable attention has been directed towards cyberspace and its application in higher education in the Western world while minimal studies have been undertaken in Gulf countries. This also highlights the need for researchers to further investigate cyberspace culture in Qatar's higher education sector.

A third avenue for future research is to investigate other additional factors that influence the achievement of the higher education 2030 vision in Qatar in order to enhance the proposed cyberspace culture framework. Despite the framework's effectiveness in aligning the strategic objectives of higher education with cyberspace culture, the Researcher acknowledges that other potential improvements can be made to the framework, specifically regarding the influence of contextual background factors that affect the use of cyberspace culture in higher education. Therefore, a further need exists to examine the socio-cultural factors that influence the adoption of cyberspace culture and its alignment with the strategic objectives developed.

8.9 Research Conclusion

The main conclusion from this research is that the Qatar higher education 2030 vision can be achieved by strategically conceptualising cyberspace culture and adopting the proposed cyberspace culture framework in order to align cyberspace culture with the targeted strategic directions. In this research, it emerged that existent cyberspace culture frameworks remain ineffective in aligning cyberspace culture with the strategic objectives in organisations due to their lack of consideration for the impact of human factors. Moreover, research concerning the alignment of cyberspace culture and the attainment of the higher education 2030 vision remains

scarce. This research thus bridges this gap by proposing the cyberspace culture framework (Figure 4-7), comprised of three strategic elements: a) external elements representing the government's Vision 2030 directives, b) internal elements representing higher education leadership, and c) the culture of organisations. This research further identifies two main stakeholders involved in the development of Qatar's cyberspace culture in higher education: a) The government, which plays multiple roles as a supervisory authority, a funding organisation, and a regulator; and b) higher educational leadership, which is involved in crafting strategies that ensure institutions achieve targeted strategic objectives. However, input from other external stakeholders such as international collaborators, scientists, and pioneering technology companies is also acknowledged. An ecosystem comprising stakeholders and resources is thus identified as highly important for fostering cyberspace culture and ensuring that cyberspace is utilised to facilitate attaining strategic objectives.

Additionally, this research posits that the achievement of the higher education 2030 vision fits into the attainment of the larger and more comprehensive QNV 2030, which seeks to transform Qatar from a hydrocarbon-based economy into one that is knowledge-based using its four pillars—namely, economic, environmental, human, and social development. This assertion arises from the fact that reliance on cyberspace culture to facilitate achieving the higher education vision is anticipated to lead to increased dependence on cyberspace for advancing other strategic objectives in other sectors of the economy, such as business. As a result, the majority of the population will leverage cyberspace to achieve their strategic objectives in sectors such as business and service industries.

In conclusion, this research argues that the proposed cyberspace culture framework is universal in its application, as it can be adopted across different sectors to ensure that cyberspace culture aligns with the achievement of strategic objectives. This research accordingly advocates for the framework's use in aligning cyberspace culture with strategic objectives, thereby leveraging cyberspace to advance economies

References

- Adams, J., Khan, H. and Raeside, R., 2014. *Research Methods for business and social science students*. New Delhi: SAGE Publications.
- Adamu, A., 2019. Selection and Appointment of Higher Education Leaders in Ethiopia: An Assessment of Implementation. Bahir Dar Journal of Education, 19(1).
- Adomi, E., 2011. Handbook of research on information communication technology policy. Hershey, Pa: IGI Global.
- Akçapınar, G., Altun, A. and Aşkar, P., 2019. Using learning analytics to develop earlywarning system for at-risk students. International Journal of Educational Technology in Higher Education, 16(1).
- Alamri, H., Watson, S. and Watson, W., 2020. Learning Technology Models that Support Personalization within Blended Learning Environments in Higher Education. TechTrends,.
- Al-Hakim, L., 2007. Global e-government. Hershey: Idea Group Publishing.
- Al-Hogail, A., 2015. Cultivating and Assessing an organisational information security culture; an empirical study. *International Journal of Security and Its Applications*, 9(7), pp.163-178.
- Al-Jaber, H., 2014. Qatar national cyber security strategy. [Online] Available at: http://www.motc.gov.qa/sites/default/files/national_cyber_security_strategy.pdf [Accessed 15/7/2019].
- Alnatheer, M. and Nelson, K., 2009. Proposed framework for understanding information security culture and practices in the Saudi context. *Proceedings of the 7th Australian Information Security Management Conference*, Perth, Western Australia
- Al Neaimi, A., Ranginya, T. and Lutaaya, P. 2015. A Framework for effectiveness of cyber security defenses, a case of the United Arab Emirates (Uae). *International Journal of Cyber-Security and Digital Forensics (Ijcsdf)*, **4**, pp. 290-301.
- Allsop, S., Hollifield, M., Huppler, L., Baumgardt, D., Ryan, D., van Eker, M., Spear, M. and Fuller, C., 2020. Using videoconferencing to deliver anatomy teaching to medical students on clinical placements. Translational Research in Anatomy, 19, p.100059.
- Altner, D. and Servi, L., 2016. A two-stage stochastic shift scheduling model for cybersecurity workforce optimization with on call options. McLean, Virginia: The MITRE Corporation.

Althonayan, A., 2003. *Integrating technology strategy with business strategy in the airline industry*. Buckinghamshire: Buckinghamshire Business School.

Alvesson M., 2002. Understanding organisational culture. London: Sage Publications Ltd

Alvesson, M., 2014. Understanding organizational culture. London: Sage Publications.

- Amant, K. S., Sapienza, F. and Sides, C. H. 2017. Culture, communication and cyberspace: rethinking technical communication for international online environments. London: Routledge.
- Andronache, A., 2019. Aligning Cybersecurity Management with Enterprise Risk Management In The Financial Industry. PhD. Brunel University.
- Apple, M. W. and Jungck, S., 1996. You don't have to be a teacher to teach this unit: teaching technology and control in the classroom. In: Hargreaves, A. and Fullan, M. G. Eds. *Understanding teacher development. Teacher development series*. London: Cassell, pp. 20–42.
- Aarseth, Espen J., 1997. *Cybertext: perspectives on ergodic literature*. The Johns Hopkins University Press.
- Audibert, L. and Murray, A., 2016. A principled approach to network neutrality. *SCRIPTed*, **13**(2), pp.118-143.
- Axelrod, R. 2014. A Repertory of Cyber Analogies. Cyber Analogies.

Babbie, E. R. 2015. The Practice of Social Research, Nelson Education.

- Baezner, M. and Robin, P., 2017. Hotspot analysis the use of cybertools in an internationalised civil war context: Cyber activities in the Syrian conflict. [Online]
 Available at: http://www.css.ethz.ch/content/dam/ethz/special-interest/gess/cis/center-for-securities-studies/pdfs/Cyber-Reports-2017-05.pdf [Accessed 7/6/2019].
- Baker, M. 2016. Striving For Effective Cyber Workforce Development. Software Engineering Institute, May. Https://Resources. Sei. Cmu. Edu/Library/Asset-View. Cfm.
- Baldauf, K. and Stair, R., 2011. *Succeeding with technology*. Boston, MA: Course Technology/Cengage Learning.
- Bambauer, D., E., 2013. "Privacy Versus Security", Arizona Legal Studies, Discussion Paper No. 13-06.
- Bangia, R., 2005. Internet and web design. New Delhi: Firewall Media.

- Barzilai-Nahon, K., Rafaeli, S. and Ahituv, N. Measuring Gaps In Cyberspace: Constructing A Comprehensive Digital Divide Index. Workshop On Measuring The Information Society, The Conference Of Internet Research, 2004.
- Baumgarten, M., 2012. Paradigm Wars Validity and Reliability in Qualitative Research. München: GRIN Verlag GmbH.
- Beck, U. 2014. The brave new world of work. John Wiley and Sons.
- Becker, H. J. 1994. How Exemplary Computer-Using Teachers Differ From Other Teachers: Implications For Realizing The Potential Of Computers In Schools. Journal Of Research On Computing In Education, 26, 291-321.
- Bednar, A. K., Cunningham, D., Duffy, T. M., and Perry, J. D. (1992). Theory into practice: How do we link? In Duffy, T. M. and Jonassen, D. H. (Eds.), Constructivism and the technology of instruction: a conversation, Hillsdale: Lawrence Erlbaum Associates, 17-34.
- Bell, D., 2001. An introduction to cyberspace cultures. London: Routledge.
- Bell, D., 2009. Cyberspace/cyberspace culture. International Encyclopaedia of Human Geography, pp.468-472.
- Bernard, H., Wutich, A. and Ryan, G., 2017. Analysing qualitative data. Thousand Oaks: SAGE.
- Benarous, L., Kadri, B. and Bouridane, A., 2016. A Survey on Cyber Security Evolution and Threats: Biometric Authentication Solutions. Signal Processing for Security Technologies, pp.371-411.
- Bernsen, N. and Dybkjær, L., 2010. Multimodal Usability. London: Springer London.
- Bendovschi, A. 2015. Cyber-Attacks–Trends, Patterns And Security Countermeasures. Procedia Economics And Finance, 28, 24-31.
- Benedikt, M. 2000. Cyberspace: First steps. In D. Bell and B. M. Kennedy (Eds.), The cyber cultures reader. New York, NY: Routledge
- Berdayes, V. and Murphy, J., 2000. *Computers, human interaction, and organisations*. Westport, Conn.: Praeger.
- Berman, P., 1999. The culture of cyberspace. *Proceedings of the Annual Meeting (American Society of International Law)*, **93**, pp.354-356.
- Bidgoli, H., 2004. The Internet encyclopaedia. Hoboken, N.J.: John Wiley and Sons.
- Bieda, D. and Halawi, L., 2015. Cyberspace: a venue for terrorism. *Issues in Information Systems*, **16**(3), pp.33-42.

- Billone, A. 2016. The Future Of The Nineteenth-Century Dream-Child: Fantasy, Dystopia, Cyberspace culture, Routledge.
- Blessing, L. and Chakrabarti, A., 2009. DRM, a Design Research Methodology.
- Bognar, B., Sablić, M. and Škugor, A., 2018. Flipped Learning and Online Discussion in Higher Education Teaching. Didactics of Smart Pedagogy, pp.371-392.
- Bond, M. and Bedenlier, S., 2019. Facilitating Student Engagement Through Educational Technology: Towards a Conceptual Framework. Journal of Interactive Media in Education, 2019(1).
- Bowers, C. A.: 1988, *The Cultural Dimensions of Educational Computing; Understanding the Non-Neutrality of Technology*, Teachers College Press, Columbia University, New York and London
- Bower, B. L. 2001. Distance Education: Facing The Faculty Challenge. Online Journal Of Distance Learning Administration, 4, 1-6.
- Boyatzis, R., 2009. Transforming qualitative information. Thousand Oaks: Sage.
- Boyle, J. 2017. Foucault In Cyberspace: Surveillance, Sovereignty, And Hardwired Censors. Law And Society Approaches To Cyberspace. Routledge.
- Breen, R., 2006. A Practical Guide to Focus-Group Research. Journal of Geography in Higher Education, 30(3), pp.463-475.
- Brown, J. D. 2014. The Cyberspace Evolution's Impact On the United States' Strategic Culture, Us Army War College.
- Brunn, S. D. 2014. Cyberspace Knowledge Gaps and Boundaries in Sustainability Science: Topics, Regions, Editorial Teams and Journals. Sustainability, 6, 6576-6603.
- Bryant, W., 2016. International conflict and cyberspace superiority. New York: Routledge.
- Buil, I., de Chernatony, L. and Martínez, E., 2012. Methodological issues in cross-cultural research: An overview and recommendations. Journal of Targeting, Measurement and Analysis for Marketing, 20(3–4), pp.223–234.
- Burda, M. and Harding, M. 2013. Panel Probit with Flexible Correlated Effects: Quantifying Technology Spillovers in The Presence of Latent Heterogeneity. Journal of Applied Econometrics, 28, 956-981.
- Bhushan, S., 2019. The future of higher education in India. New Delhi: Springer.
- Cantwell, B., Coate, H. and King, R., 2018. Handbook on the Politics of Higher Education. Cheltenham: Edward Elgar.

- Caragliu, A. and Del Bo, C., 2019. Smart innovative cities: the impact of Smart City policies on urban innovation. *Technological Forecasting and Social Change*, **142**, pp.373-383.
- Caton, J., 2012. Beyond domains, beyond commons: Context and theory of conflict in cyberspace. In: 4th International Conference on Cyber Conflict (CYCON 2012).
- Chan, H. and Chan, L., 2018. Smart Library and Smart Campus. Journal of Service Science and Management, 11(06), pp.543-564.
- Chang, L. Y. and Grabosky, P. 2017. The Governance Of Cyberspace. Regulatory Theory, 533.
- Chase, M., Macfadyen, L. P., Reeder, K. and Roche, J. 2002. Intercultural Challenges In Networked Learning: Hard Technologies Meet Soft Skills. University Of British Columbia.
- Chatman, J. A., and Eunyoung Cha, S., 2003. Leading by leveraging culture. California Management Review, 45, 19–34
- Chedid, M., Caldeira, A., Alvelos, H. and Teixeira, L., 2019. Knowledge-sharing and collaborative behaviour: An empirical study on a Portuguese higher education institution. Journal of Information Science, p.016555151986046.
- Cheung, H. and Chan, A., 2009. Education and competitive economy: how do cultural dimensions fit in? *Higher Education*, **59**(5), pp.525-541.
- Choo, K.-K. R., 2011. The Cyber Threat Landscape: Challenges And Future Research Directions. Computers and Security, 30, 719-731.
- Claessens, S., Glaessner, T. and Klingebiel, D., 2002. *E-finance in emerging markets*. Malden, MA: Blackwell Pub.
- Cohen, L., Manion, L., and Morison, K. 2007. Research Methods in Education. (6th ed.). London: Routledge.
- Collins, H., 2010. Creative research. London: AVA Publishing.
- Cope, B. and Kalantzis, M., 2017. E-learning ecologies. London: Routledge.
- Creswell, J. W., 2011. *Educational research: Planning, conducting, and evaluating quantitative and qualitative research* (7th ed.). Boston MA: Addison Wesley.
- Crossman, D. (1997). The evolution of the world wide web as an emerging instructional technology tool. In B. H. Khan (Ed.), Web-based instruction (pp. 19–23) Englewood Cliffs, NJ: Educational Technology Publications
- Cummings, E., Chau, S. and Turner, P., 2009. Assessing a patient-centred e-health approach to chronic disease self-management. *Patient-Centred E-Health*, pp.168-185.

- Curti, F. and Mihov, A., 2018. Fraud recovery and the quality of country governance. *Journal of Banking and Finance*, **87**, pp.446-461.
- Daniel, B., 2014. Big data and analytics in higher education: opportunities and challenges. *British Journal of Educational Technology*, **46**(5), pp.904-920.
- Dauber, D., Fink, G. and Yolles, M., 2012. A Configuration Model of Organisational Culture. SAGE Open, 2(1), p.215824401244148.
- David, M. and Sutton, C., 2009. Social research. London: Sage.
- Da Veiga, A., 2008. *Cultivating and assessing information security culture* (Doctoral dissertation, University of Pretoria).
- Da Veiga, A. and Eloff, J., 2010. A framework and assessment instrument for information security culture. Computers and Security, 29(2), pp.196-207.
- Dawson, J. and Thomson, R., 2018. The Future Cybersecurity Workforce: Going Beyond Technical Skills for Successful Cyber Performance. Frontiers in Psychology, 9.
- Daymon, C. and Holloway, I., 2011. Qualitative research methods in public relations and marketing communications. London: Routledge Taylor and Francis Group.
- De Jonge, J. H. M., 2006. Three Levels of Culture (Schein). 12manage B.V.-The Netherlands [online]. Available:

http://www.12manage.com/methods_schein_three_levels_culture.html

- De Witt, C. and Gloerfeld, C., 2017. Mobile learning and higher education. *The Digital Turn in Higher Education*, pp.61-79.
- Deloitte, 2017. Cyber reporting survey. [Online] Available at: https://www2.deloitte.com/content/dam/Deloitte/uk/Documents/audit/deloitte-ukgovernance-in-focus-cyber-risk-reporting.pdf [Accessed 22 Jun. 2019].
- Diki-Kidiri, M., 2007. Securing a place for a language in cyberspace. [Online] Available at: http://unesdoc.unesco.org/images/0014/001497/149786E.pdf [Accessed 7/6/2019].
- Ding, Z., Li, X., Jiang, C. and Zhou, M. 2018. Objectives And State-Of-The-Art Of Location-Based Social Network Recommender Systems. Acm Computing Surveys (Csur), 51, 18.
- Dohr, A., Modre-Opsrian, R., Drobics, M., Hayn, D. and Schreier, G., 2010. The Internet of Things for Ambient Assisted Living. 2010 Seventh International Conference on Information Technology: New Generations.
- Dorsey, D. W., Martin, J., Howard, D. J. and Coovert, M. D. 2017. Cybersecurity Issues In Selection. Handbook Of Employee Selection. Routledge.

- Dornyei, Z. 2007. Research Methods in Applied Linguistics: Quantitative Qualitative, and Mixed Methodologies. Oxford: Oxford University Press
- Douglass, F. (1995). Narrative of the Life of Fredrick Douglass. Tustin: Xist Publishing.
- Dunn, T. and Kennedy, M., 2019. Technology enhanced learning in higher education; motivations, engagement and academic achievement. *Computers and Education*, **137**, pp.104-113.
- Duffy, T. M. and Jonassen, D. H. (1992). Constructivism: New implications for instructional technology. In T. M. Duffy and D. H. Jonassen (Eds.), Constructivism and the technology of instruction: A conversation (pp. 1-16). Hillsdale, NJ: Lawrence Erlbaum Associates
- Dvornik, D. and Sabolić, D., 2007. Telecommunication liberalization and economic development in European countries in transition. *Technology in Society*, **29**(4), pp.378-387.
- Edwards, R. and Holland, J., 2013. What is qualitative interviewing?. London: Bloomsbury.
- Eliot, T., 2014. *Notes towards the definition of culture*. Boston, Mass: Houghton Mifflin Harcourt.
- Elo, S., Kääriäinen, M., Kanste, O., Pölkki, T., Utriainen, K. and Kyngäs, H., 2014. Qualitative Content Analysis. SAGE Open, 4(1), p.215824401452263.
- Erlingsson, C. and Brysiewicz, P., 2017. A hands-on guide to doing content analysis. African Journal of Emergency Medicine, 7(3), pp.93-99.
- Escobar, A., Hess, D., Licha, I., Sibley, W., Strathern, M. and Sutz, J. (1994). Welcome to Cyberia: Notes on the Anthropology of Cyberspace culture [and Comments and Reply]. Current Anthropology, 35(3), pp.211-231.
- Engel, R. and Schutt, R., 2014. Fundamentals of social work research. Thousand Oaks, California: Sage.
- Eriksson, P. and Kovalainen, A., 2008. Qualitative methods in business research. London: SAGE.
- Esteves, M., Matias, R. and Pereira, A., 2018. Collaborative Learning Environment in Higher Education: A Case Study. Lecture Notes in Computer Science, pp.17-24.
- Etikan, I., Musa, S.A. and Alkassim, R.S., 2016. Comparison of convenience sampling and purposive sampling. *American Journal of Theoretical and Applied Statistics*, *5*(1), pp.1-4.

- Fabos, B. and Young, M.D., 1999. Telecommunication in the Classroom: Rhetoric versus Reality. *Review of Educational Research*, 69(3), 217. Retrieved July 16, 2019 from <u>https://www.learntechlib.org/p/89199/</u>.
- Fabre, R., Messerschmidt-Mariet, Q. and Holvoet, M., 2016. New challenges for knowledge.Hoboken: Wiley.
- Fadun, O.S., 2013. Risk Management and Risk Management Failure: Lessons for Business Enterprises. International Journal of Academic Research in Business and Social Sciences, 3(2), pp.225–239.
- Fang, B., Jia, Y., Li, X., Li, A. and Wu, X., 2017. Big search in cyberspace. *IEEE Transactions on Knowledge and Data Engineering*, **29**(9), pp.1793-1805.
- Fang, T., 2003. A critique of Hofstede's fifth national culture dimension. International Journal of Cross Cultural Management, 3(3), 347–368. http://dx.doi.org/10.1177/1470595803003003006
- Fetterman, D. M. 1996. Research News And Comment: Videoconferencing On-Line: Enhancing Communication Over The Internet. Educational Researcher, 25, 23-27.
- Fiddner, D., 2015. Defining a Framework for Decision Making in Cyberspace. [ebook] Available at:

http://www.businessofgovernment.org/sites/default/files/Defining%20a%20Framewor k%20for%20Decision%20Making%20in%20Cyberspace.pdf [Accessed 15/7/2019].

- Fink, A. 1995. How to design surveys. Thousand Oaks, CA: Sage Publications.
- Fleming, D., 2004. Leadership wisdom from unlikely voices. El Cajon, CA: EmergentYS Books.
- Foody, M., Samara, M., El Asam, A., Morsi, H. and Khattab, A. 2017. A Review Of Cyberbullying Legislation In Qatar: Considerations For Policy Makers And Educators. International Journal Of Law And Psychiatry, 50, 45-51.
- Forest, J. and Kinser, K., 2003. *Higher education in the United States*. Santa Barbara: ABC-CLIO.
- Freire, P. (1970). Pedagogy of the oppressed. New York: Herder and Herder.
- Galinec, D., Možnik, D. and Guberina, B., 2017. Cybersecurity and cyber defence: national level strategic approach. Automatika, 58(3), pp.273-286.
- García, B. M. 2016. Cibercultura, Tic Y Redes Sociales: Nuevas Formas De Comunicación Para Las Familias. Cyberspace culture, Ict And Social Networks: New Forms Of Communication For Families. Pixel-Bit. Revista De Medios Y Educación, 195-206.

- Garrison, D. and Vaughan, N., 2013. *Blended learning in higher education*. San Francisco, Calif.: Jossey-Bass.
- Gartzke, E. and Lindsay, J. R. 2015. Weaving Tangled Webs: Offense, Defense, And Deception In Cyberspace. Security Studies, 24, 316-348.
- Gasser, U. and Drolshammer, J., 2015. The Brave New World of (Swiss) Law: Contours of a Framework and Call for a Strategy to Shape Law's Digital Revolution. SSRN Electronic Journal.
- Ghanem, K. and Castelli, P., 2019. Accountability and Moral Competence Promote Ethical Leadership. Journal of Values-Based Leadership, 12(1).
- Ghasemaghaei, M., 2019. Does data analytics use improve firm decision making quality? The role of knowledge sharing and data analytics competency. *Decision Support Systems*, 120, pp.14-24.
- Ghazvini, A., Shukur, Z. and Hood, Z., 2018. Review of Information Security Policy based on Content Coverage and Online Presentation in Higher Education. International Journal of Advanced Computer Science and Applications, 9(8), pp.410-423.
- Gibson, W. 1984. Neuromancer. New York: Ace. Freedom Of Expression And Regulation Of Information In Cyberspace, 125, 18-19.
- Glaser, K., 2014. Inductive or deductive?. Newcastle upon Tyne, England: Cambridge Scholars Publishing.
- Goi, C. L. 2009. Cyberspace culture: Impacts On Netisen. Asian Culture And History, 1, 140.
- Gómez-Diago, 2012. Cyberspace and Cyberspace culture. In Kosut, M. and Golson, J. Geoffrey (Eds).Encyclopedia of Gender in Media. SAGE
- Gómez-Diago, G., 2015. Communication In Crowdfunding Online Platforms. Creativity In The Digital Age. Springer.
- Goodman, S., Kirk, J. and Kirk, M., 2007. Cyberspace as a medium for terrorists. *Technological Forecasting and Social Change*, **74**(2), pp.193-210.
- Goodenow, R. (1996). The Cyberspace Challenge: Modernity, post-modernity and reflections on international networking policy. Comparative Education, 32(2), pp.197-216.
- Goswami, U., 2011. The Wiley-Blackwell handbook of childhood cognitive development. Malden, Mass.: Wiley-Blackwell.
- Gregg, A., Holsing, C. and Rocco, S., 2017. Quality online learning: e-learning strategies for higher education. *Leading and Managing e-Learning*, pp.13-27.
- Groves, R., Fowler, F., Couper, M., Lepkowski, J., Singer, E. and Tourangeau, R., 2009. Survey methodology. Hoboken, N.J.: Wiley.

- Gungor, V. C., Sahin, D., Kocak, T., Ergut, S., Buccella, C., Cecati, C. and Hancke, G. P. 2011. Smart Grid Technologies: Communication Technologies And Standards. Ieee Transactions On Industrial Informatics, 7, 529-539.
- Gunitsky, S., 2015. Corrupting the cyber-commons: social media as a tool of autocratic stability. *Perspectives on Politics*, **13**(1), pp.42-54.
- Gunter, H. 1995. Jurassic Management: Chaos And Management Development In Educational Institutions. Journal Of Educational Administration, 33, 5-20.

Haarmann, H., 2007. Foundations of culture. Frankfurt, M.: Lang.

- Haarstad, H. and Wathne, M., 2019. Are smart city projects catalysing urban energy sustainability? *Energy Policy*, **129**, pp.918-925.
- Hakken, D., 1999. *Cyborgs@cyberspace? An ethnographer looks to the future*. New York: Routledge.
- Hartman, R., Townsend, M. and Jackson, M., 2019. Educators' perceptions of technology integration into the classroom: a descriptive case study. Journal of Research in Innovative Teaching & Learning, 12(3), pp.236-249.
- Hanson, A. 2013. Technology And Cultural Tectonics: Shifting Values And Meanings, Springer.
- Hansen, L., and Nissenbaum, H., 2009. Digital Disaster, Cyber Security, And The Copenhagen School, International Studies Quarterly (2009) 53, 1155–1175
- Harasim, L. (1993). Global Networks: Computers and Communication. Cambridge: MIT Press.
- Haughey, C.J., 2020. Cybersecurity Framework: How To Create A Resilience Strategy. [online] Security Intelligence. Available at: https://securityintelligence.com/articles/how-to-create-a-cybersecurityframework/?category=risk-management [Accessed 26 Dec. 2020].

Helyer, R., 2015. The work-based learning student handbook. London: Red Globe Press.

- Henry, G.T., 1990. Practical sampling: Applied social research methods series. Vol. 21 [0—16. SAGE Publication, Inc
- Hina, S. and Dominic, P., 2018. Information security policies' compliance: a perspective for higher education institutions. Journal of Computer Information Systems, pp.1-11.
- Hine, C., 2000. Virtual ethnography. Los Angeles: Sage.
- Ho, L.-A., Kuo, T.-H. and Lin, B. 2010. Influence Of Online Learning Skills In Cyberspace. Internet Research, 20, 55-71.

- Hofstede, G. (1983). National Cultures Revisited. Behaviour Science Research, 18(4), pp.285-305.
- Hofstede, G., 2001. Culture's consequences (2nd ed.). Thousand Oaks, CA: Sage.
- Hofstede, G., 2002. Dimensions do not exist A reply to Brendan McSweeney. Human Relations, 55(11), 1355–1361. http://dx.doi. org/10.1177/0018726702055011921
- Hofstede, G., V. Garibaldi de Hilal, S. Malvezzi, B. Tanure, and H. Vinken, 2010a.
 Comparing regional cultures within a country: Lessons from Brazil. Journal of Cross-Cultural Psychology, 41(3), 336–352. http://dx.doi.org/10.1177/0022022109359696
- Hofstede, G., G. J. Hofstede, and M. Minkov, 2010b. Cultures and organisations Software of the mind. San Francisco: McGraw Hill.
- Hofstede, G., G. J. Hofstede, M. Minkov, and H. Vink., 2008. Value survey module manual 2008. <u>http://www.geerthofstede.com/vsm-08</u>
- Hogan, S. and Coote, L., 2014. Organisational culture, innovation, and performance: A test of Schein's model. Journal of Business Research, 67(8), pp.1609-1621.
- Holbeche, L., 2015. *The high performance organization: creating dynamic stability and sustainable success*. London: Routledge.
- Hope, A. 2016. Educational Heterotopia And Students' Use Of Facebook. Australasian Journal Of Educational Technology, 32.
- Howard, P., Duffy, A., Freelon, D., Hussain, M., Mari, W. and Mazaid, M., 2011. Opening closed regimes: what was the role of social media during the Arab spring? SSRN Electronic Journal.
- Howley, K. 2016. 'I Have A Drone': Internet Memes And The Politics Of Culture. Interactions: Studies In Communication and Culture, 7, 155-175.
- Hsieh, H. and Shannon, S., 2005. Three Approaches to Qualitative Content Analysis. Qualitative Health Research, 15(9), pp.1277-1288.
- Hung, D. and Chen, D-T., 2001. Situated Cognition, Vygotskian Thought and Learning from the Communities of Practice Perspective: Implications for the Design of Web-BasedE-Learning.Education Media International. Vol.38, No.1, pp3-12.
- Hunter, D. 2017. Cyberspace As Place And The Tragedy Of The Digital Anticommons. Law And Society Approaches To Cyberspace. Routledge.
- Hussein, E., Daouda, S., Alrabaiah, H. and Badawi, R., 2020. Exploring Undergraduate Students' Attitudes towards Emergency Online Learning during COVID-19: A Case from the UAE. *Children and Youth Services Review*, [online] 119, p.105699.

- Jacobson, M., 2010. Terrorist Financing and the Internet. *Studies in Conflict and Terrorism*, **33**(4), pp.353-363.
- Jahshan, P., 2007. Cybermapping and the writing of myth. New York: P. Lang.
- Janczewski, L. and Colarik, A., 2005. *Managerial guide for handling cyber-terrorism and information warfare*. Hershey: Idea Group Publ.
- Jarvis, D., 2014. Regulating higher education: Quality assurance and neo-liberal managerialism in higher education—A critical introduction. Policy and Society, 33(3), pp.155-166.
- Jha, M., Jha, S. and O'Brien, L., 2019. Re-engineering Higher Education Learning and Teaching Business Processes for Big Data Analytics. Business Information Systems, pp.233-244.
- Junqueira, E., 2015. The cyberspace culture theories and teacher preparation at the LIFE-UFC project in Brazil. *International Journal of Education and Development using ICT*, **11**(3).
- Johnson, H., Dunlap, J., Verma, G., McClintock, E., DeBay, D. and Bourdeaux, B., 2018.
 Video-Based Teaching Playgrounds: Designing Online Learning Opportunities to
 Foster Professional Noticing of Teaching Practices. TechTrends, 63(2), pp.160-169.
- Jonassen, D. H., Peck, K. L., and Wilson, B. G., 1999. Learning with technology: A constructivist per-spective. Upper Saddle River, NJ: Prentice Hall.
- Jonassen, D.H. (1993) 'Computers in the Classroom: Mindtools for Critical Thinking', Englewood Cliffs, New Jersey, Prentice Hall
- Jonassen, D. H., Howland, J. L., Moore, J. L. and Marra, R. M., 2003. Learning to Solve Problems with Technology: A Constructivist Perspective.New Jersey: Merrill Prentice Hall.
- Jones, S. (1995). CyberSociety. Thousand Oaks: Sage.
- Jones SG. 1998. CyberSociety 2.0: Revisiting Computer-Mediated Communication and Community. Thousand Oaks, CA:
- Jupp, V., 2009. The Sage dictionary of social research methods. London: Sage.
- Kaisler, S., 2016. Birthing the computer. Newcastle: Cambridge Scholars.
- Kanwar, A., Balasubramanian, K. and Carr, A., 2019. Changing the TVET paradigm: new models for lifelong learning. International Journal of Training Research, 17(sup1), pp.54-68.

- Kelley, T. R. and Kellam, N., 2009. A theoretical framework to guide the re-engineering of technology education. *Journal of Technology Education*, **20**(2), pp.37-49.
- Kerawalla, L., Minocha, S., Kirkup, G. and Conole, G., 2009. An empirically grounded framework to guide blogging in higher education. *Journal of Computer Assisted Learning*, 25(1), pp.31-42.
- Kettani, D. and Moulin, B., 2015. *E-government for good governance in developing countries*. London: Anthem Press.
- Khan, J., 2008. Research methodology. Delhi: APH Publishing.
- Khan, B. H. (Ed.). (1997). Web-based instruction. Englewood Cliffs, NJ: Educational Technology Publications.
- Khin, A. A., Chau, W. H., Chuan, C. S. and Tong, L. K. 2016. Factors Affecting For Online Marketing Strategy Of Business Development In Malaysia. British Journal Of Economics, Management and Trade, 11, 1-17.
- Kiliç, B. 2018. Communicative Potentials Of Internet Memes In Cyberspace Of Turkey: An Analysis Of Caps Culture.
- Kitching, F., Winbolt, M., Macphail, A. and Ibrahim, J. E. 2015. Web-Based Social Media For Professional Medical Education: Perspectives Of Senior Stakeholders In The Nursing Home Sector. Nurse Education Today, 35, 1192-1198.
- Kongolo, M., 2019. Management of Higher Education Institutions in Uncertain Times: The Perspectives of University College. Journal of Social Science Studies, 6(2), p.97.
- Kothari, C. and Garg, G., 2016. *Research methodology*. London: New Age International Limited, Publishers.
- Krippendorff, K., 2004. Content analysis. Thousand Oaks: Sage.
- Kumar, C., 2008. Research methodology. New Delhi: APH Pub. Corp.
- Kumar, H., Singh, M., Gupta, M. and Madaan, J., 2018. Moving towards smart cities: solutions that lead to the smart city transformation framework. *Technological Forecasting and Social Change*.
- Kurilovas, E., 2018. On data-driven decision-making for quality education. Computers in Human Behavior, p.105774.
- Kvale, S. and Brinkmann, S., 2009. Interviews. Learning the Craft of Qualitative Researcch Interviewing. (2nd Ed.). Thousand Okas, CA: Sage.
- Kwon, J. W., 2012. Does China have more than one culture? Asia Pacific Journal of Management, 29(1), 79–102. http://dx.doi. org/10.1007/s10490-010-9191-y

- Lacatus, M.L., 2013. Organizational Culture in Contemporary University. Procedia Social and Behavioral Sciences, 76, pp.421–425.
- Land, R. and Bayne, S., 2012. Education in cyberspace. London: Routledge.
- Landow, G. L. (1992). Hypertext, The convergence of contemporary critical theory and technology. Baltimore: Johns Hopkins University Press.
- Lankshear, C., and Knobel, M., 2008. Digital Literacies: concepts, policies and practices. New Literacies and Digital Epistemologies, 30. Peter Lang Publishing, New York, USA.
- Lave, J. and Wenger, E. (1991). Situated Learning.
- Leavy, P., 2014. The Oxford handbook of qualitative research. New York: Oxford University Press.
- Lehto, M. and Neittaanmäki, P., 2018. Cyber Security. Cham: Springer.
- Leonard, H., Lewis, R., Freedman, A. and Passmore, J., 2016. The Wiley-Blackwell handbook of the psychology of leadership, change, and organizational development. New Jersey: Wiley.
- Lessig, L. 2002. The Architecture of Innovation, Duke Law Journal, 51, 1783-1801.
- Lessig, L., 2004. Free culture. London: Penguin.
- Lévy, P., 2001. Cyberspace culture. Minneapolis: University of Minnesota Press.
- Li, F., Li, Z., Han, W., Wu, T., Chen, L. and Guo, Y. Cyberspace-Oriented Access Control: Model And Policies. Data Science In Cyberspace (Dsc), 2017 Ieee Second International Conference On, 2017. Ieee, 261-266.
- Liebowitz, J., 2016. Thoughts on recent trends and future research perspectives in big data and analytics in higher education. *Big Data and Learning Analytics in Higher Education*, pp.7-17.
- Lim, J., Chang, S., Maynard, S. and Ahmad, A., 2009. Exploring the Relationship between Organizational Culture and Information Security Culture. In: 7th Australian Information Security Management Conference.
- Lindsay, J. R. 2015. The Impact Of China On Cybersecurity: Fiction And Friction. International Security, 39, 7-47.
- Linhares de Souza, Y., Farinelli, F., Jamil, G., Lobo de Vasconcelos, M. and Dias, G., 2012. Virtual communities of practice as a support for knowledge sharing in social networks. Handbook of Research on Business Social Networking, pp.374-387.
- Liu, C. and Albitz, P., 2009. DNS and BIND. Sebastopol: O'Reilly Media, Inc.

- Lord, A., Zhou, Y. R., Jensen, R., Morea, A. and Ruiz, M. 2016. Evolution From Wavelength-Switched To Flex-Grid Optical Networks. Elastic Optical Networks. Springer.
- Louis, M.R. 1980. Organisations as culture-bearing milieux. In Organisational Symbolism. Edited by L.R. Pondy, et al. Greenwich, CT: JAI.
- Lovink, G. 2017. Social Media Abyss: Critical Internet Cultures And The Force Of Negation, John Wiley and Sons.
- Lucas, J. and Moeller, B., 2004. *The effective incident response team*. Boston: Addison-Wesley.
- Luo, J., 2019. Application of Big Data Technology in Student Management Decision. Advances in Intelligent Systems and Computing, pp.687-694.
- Luyegu, E., 2016. Ensuring presence in online learning environments. Handbook of Research on Strategic Management of Interaction, Presence, and Participation in Online Courses, pp.350-376.
- Mabry, L., 2008. Case study in social research. In P. Alasuutari, L. Brinkman, and J. Brannen (eds). SAGE Handbook of Social Research Methods (pp. 214-227). Thousand Oaks, CA: Sage
- Macfadyen, L., 2006. *The culture(s) of cyberspace*. Encyclopaedia of Human Computer Interaction, pp.143-149.
- Macfadyen, L. P., Roche, J. and Doff, S. 2004. Communicating Across Cultures In Cyberspace: A Bibliographical Review Of Intercultural Communication Online, Lit Verlag Münster.
- Macnee, C. and McCabe, S., 2008. Understanding nursing research. Philadelphia, Pa.: Lippincott Williams and Wilkins.
- Mahmoud, A. and Auter, P., 2009. The interactive nature of computer-mediated communication. American Communication Journal, 11(4).
- Mahlangu, V., 2018. The Good, the Bad, and the Ugly of Distance Learning in Higher Education. Trends in E-learning.
- Malcolmson, J., 2009. What is security culture? does it differ in content from general organisational culture? In the proceedings of the 43rd Annual International Carnahan Conference on Security Technology (pp. 361–366).
- Markham AN. 1998. Life Online: Research- ing Real Experience in Virtual Space. Walnut Creek, CA: AltaMira
- Marshall, S., 2018. Shaping the University of the Future.

- Martin-Albo, Á. and Gregorio-Godeo, E., 2017. *Identity and identification*. Cuenca: Ediciones de la Universidad de Castilla-La Mancha.
- Manuel., 2010. Globalisation, Networking, Urbanisation: Reflections on the Spatial Dynamics of the Information Age. Urban Studies, 47(13), 2737-2745
- Mayer, M., Martino, L., Mazurier, P., and Tzvetkova, G., 2014. How would you define Cyberspace? Draft Pisa,

https://www.academia.edu/7096442/How_would_you_define_Cyberspace

Mazurczyk, W., Drobniak, S. and Moore, S., 2016. Towards a Systematic View on Cybersecurity Ecology. Advanced Sciences and Technologies for Security Applications, pp.17-37.

McIlwraith, A., 2016. Information Security and Employee Behaviour.

- McLellan, J. 1998. Nanabosho: Weaving the dream into the classroom. Issues in the North 3, 107-110
- McLuhan, M. (1964). Understanding media. Toronto: Toronto Education Quarterly.
- McPherson, P. and Coleman, M., 2019. Research universities must bolster both security and openness (opinion) | Inside Higher Ed. [online] Insidehighered.com. Available at: https://www.insidehighered.com/views/2019/08/05/research-universities-mustbolster-both-security-and-openness-opinion [Accessed 10 Dec. 2019].
- McSweeney, B., 2002. The essentials of scholarship: A reply to Geert Hofstede. Human Relations, 55(11), 1363–1372. http://dx.doi. org/10.1177/00187267025511005
- Merendino, A., Dibb, S., Meadows, M., Quinn, L., Wilson, D., Simkin, L. and Canhoto, A., 2018. Big data, big decisions: the impact of big data on board level decision-making. *Journal of Business Research*, **93**, pp.67-78.
- Mertens, D., 2005. Research and Evaluation in Education and Psychology. Thousand Oaks, CA: Corwin Press.
- Meyers, E., Erickson, I. and Small, R., 2013. Digital literacy and informal learning environments: an introduction. Learning, Media and Technology, 38(4), pp.355-367.
- Mitchell, M. and Jolley, J., 2013. Research design explained. Belmont, Calif.: Wadsworth Cengage Learning.
- Mihalache, A., 2002. The cyber space-time continuum: meaning and metaphor. *The Information Society*, **18**(4), pp.293-301.
- Misty, V., 2015. Enhanced discovering computers 2017. Boston: Cengage Learning.

Mok, K., 2016. Promoting International Connectivity and Seeking Global Competitiveness: Issues and Challenges. Managing International Connectivity, Diversity of Learning and Changing Labour Markets, pp.1-21.

Mongkhonvanit, J., 2010. Industrial Cluster & Higher Education. Bloomington: Xlibris.

- Morin, O., 2016. How Traditions Live and Die. Oxford: Oxford University Press.
- Moteff, J. And Parfomak, P., 2004, October. Critical Infrastructure And Key Assets: Definition And Identification. Library Of Congress Washington Dc Congressional Research Service.
- Mucic, D., Hilty, D. M., Parish, M. B. and Yellowlees, P. M. 2016. Web-And Internet-Based Services: Education, Support, Self-Care, And Formal Treatment Approaches. E-Mental Health. Springer.
- Mumper, M., Gladieux, L., King, J. and Corrigan, M., 2011. The federal government and higher education. American higher education in the twenty-first century: Social, political, and economic challenges, pp.113-138.
- Murray, J. H. and Murray, J. H. 2017. Hamlet On The Holodeck: The Future Of Narrative In Cyberspace, Mit Press.
- Myers, N. 1990. The Gaia Atlas Of Future Worlds: Challenge And Opportunity In An Age Of Change, Robertson Mccarta.
- Myers, B. and Shaw, L., 2004. The essentials. Cheltenham: Nelson Thornes.
- Nachandiya, N., Gambo, Y., Joel, N. and Davwar, P., 2018. Smart Technologies for Smart Campus Information System. Asian Journal of Research in Computer Science, 2(2), pp.1-7.
- Nakamura, L., 2002. Cybertypes. New York: Routledge.
- Natow, R., Reddy, V. and Grant, M., 2017. How and why higher education institutions use technology in developmental education programmeming. [Online] Available at: https://files.eric.ed.gov/fulltext/ED578193.pdf [Accessed 12 Jun. 2019].
- Neaimi, A.A., and Ranginya, P.L., 2014. A critical analysis of the effectiveness of cyber security defenses in uae government agencies.
- Ng, E., 2010. Comparative blended learning practices and environments. Hershey: Information Science Reference.
- Noble, H. and Smith, J., 2015. Issues of validity and reliability in qualitative research. Evidence Based Nursing, 18(2), pp.34-35.

- Oates, C. and Alevizou, P., 2018. Conducting focus groups for business and management students. Los Angeles: Sage.
- Oh, I., Jang, W. and Kim, S. 2018. Selling Trust In Cyber Space: Social Networking Service (Sns) Providers And Social Capital Amongst Netisens In South Korea. Asia Pacific Business Review, 24, 196-211.
- Oh, S., 2018. Social Q&A. Social Information Access, pp.75-107.
- Okada, A., and Ferreira, G., 2012. Co-authorship in the age of cyberspace culture: open educational resources at the open university of the United Kingdom. *Revista Teias*, 13(30), pp. 227–353.
- Oliveira, A., Feyzi Behnagh, R., Ni, L., Mohsinah, A., Burgess, K. and Guo, L., 2019.
 Emerging technologies as pedagogical tools for teaching and learning science: A literature review. Human Behavior and Emerging Technologies, 1(2), pp.149-160.
- Ong, B. (1989). Research from Within: Blurring Boundaries and Developing New Methods. The Sociological Review, 37(3), pp.505-517.
- Orr, L. M., and W. J. Hauser., 2008. A re-inquiry of Hofstede's cultural dimensions: A call for the 21st century. Marketing Management Journal, 18(2), 1–19
- Ostroff, C., Kinicki, A. and Muhammad, R., 2012. Organisational Culture and Climate. Handbook of Psychology, Second Edition.
- Pacheco, E., Yoong, P. and Lips, M., 2020. Transition issues in higher education and digital technologies: the experiences of students with disabilities in New Zealand. Disability & Society, pp.1-23.
- Palfrey, J. and Gasser, U., 2008. Born Digital Understanding the First Generation of Digital Natives. Basic Books, New York
- Palloff, R. and Pratt, K., 1999. Building learning communities in cyberspace. San Francisco: John Wiley and Sons, Inc.
- Pietiläinen, V., Salmi, I., Rusko, R. and Jänkälä, R., 2017. Experienced Stress and the Value of Rest Stops in the Transportation Field. Handbook of Research on Human Factors in Contemporary Workforce Development, pp.249-267.
- Pinheiro, R., Geschwind, L., Foss Hansen, H. and Pulkkinen, K., 2019. Reforms, Organizational Change and Performance in Higher Education. Cham: Palgrave Macmillan.
- Popov, V., Brinkman, D., Biemans, H., Mulder, M., Kuznetsov, A. and Noroozi, O., 2012. Multicultural student group work in higher education. *International Journal of Intercultural Relations*, **36**(2), pp.302-317.

- Postman, N. 1985. Media And Technology As Educators. Education In School And Non-School Settings. The Eighty-Fourth Yearbook Of The National Society For The Study Of Education, Part I, 183-200.
- Pugalee, D. K. and Robinson, R. 1998. A Study Of The Impact Of Teacher Training In Using Internet Resources For Mathematics And Science Instruction. Journal Of Research On Computing In Education, 31, 78-88.
- Punch, K. F. 2006. Developing effective research proposals. London: SAGE
- Qatar National Vision 2030, 2008. Qatar national vision 2030. [Online] Available at: https://www.diwan.gov.qa/-/media/Diwan-Amiri/Files/Qatar-National-Vision-2030-EN.ashx?la=en [Accessed 11 Jun. 2019].
- Ramachandran, S., Rao, S., and Goles, T., 2008. Information security cultures of four professions: a comparative study. In Proceedings of the 41st Annual Hawaii International Conference on System Sciences (HICSS 2008) (pp. 454–454). Hawaii: Ieee.
- Raschke, C.A., 2003. The digital revolution and the coming of the postmodern university.
- Reamer, F., 2013. Social Work in a Digital Age: Ethical and Risk Management Challenges. Social Work, 58(2), pp.163-172.
- Repko, R. 2009. The Collaboration Imperative For Cyberspace Stakeholders. White Paper, Office Of The Air Force Deputy Chief Of Staff For Operations, Plans, And Requirements.
- Reynolds, C., 2002. Changing gender scripts and moral dilemmas for women and men in education, 1940–1970. In C. Reynolds (Ed.), Women and school leadership (pp.29-48). Albany, NY:SUNY Press
- Riel, V. C.B.M. (1995). Principles of Corporate Communication. Prentice-Hall, London.
- Robertson, J. and Arnold, L., 2018. Cyberwar: how nations attack without bullets or bombs. [Online] Available at: https://www.bloomberg.com/news/articles/2018-05-11/cyberwar-how-nations-attack-without-bullets-or-bombs-quicktake [Accessed 7/6/2019].
- Robbins, S., 1998. Organizational Behavior. Indianapolis: Prentice Hall.
- Roper, C., Fischer, L. and Grau, J., 2006. Security Education, Awareness and Training: SEAT from Theory to Practice. Burlington: Elsevier.

- Rowe, D., Lunt, B. and Ekstrom, J., 2011. The role of cyber-security in information technology education. Proceedings of the 2011 conference on Information technology education - SIGITE '11.
- Rost, J. and Glass, R., 2011. *The dark side of software engineering: the ethics and realities of subversi*. New Jersey: John Wiley and Sons.
- Rouse, R., & Dionisio, M., 2018. Looking Forward, Looking Back. Carnegie Mellon: ETC Press.
- Rowland. J., Rice, M. and and Shenoi, S., 2014. The Anatomy of a Cyber Power. International Journal of Critical Infrastructure Protection, Vol. 7, No. 2, pp. 3-11. doi: 10.1016/j.ijcip.2014.01.001. Retrieved from https://www.researchgate.net/profile/Sujeet_Shenoi/publication/259994578_The_Ana tomy_of_a_Cyber_Po wer/links/55b501a008ae092e965581a0.pdf
- Safa, N., Maple, C., Watson, T. and Von Solms, R., 2018. Motivation and opportunity based model to reduce information security insider threats in organisations. Journal of Information Security and Applications, 40, pp.247-257.
- Sari, M., Ciptadi, P. and Hardyanto, R., 2017. Study of Smart Campus Development Using Internet of Things Technology. IOP Conference Series: Materials Science and Engineering, 190, p.012032.
- Shaaban, H. k., 2008. Enhancing the governance of information security in developing countries: the case of Zanzibar (Doctoral dissertation, University of Bedfordshire).
- Saunders, M., Lewis, P. and Thornhill, A. 2009. Research Methods For Business Students, Pearson Education.
- Savrul, M., Incekara, A. and Sener, S., 2014. The potential of e-commerce for SMEs in a globalizing business environment. *Procedia - Social and Behavioural Sciences*, 150, pp.35-45.
- Schaller, M. and Crandall, C., 2004. *The psychological foundations of culture*. Mahwah, N.J.: Lawrence Erlbaum Associates.
- Scharfenort, N. 2012. Urban Development And Social Change In Qatar: The Qatar National Vision 2030 And The 2022 Fifa World Cup. Journal Of Arabian Studies, 2, 209-230.
- Schatz, D., Bashroush, R. and Wall, J., 2017. Towards a More Representative Definition of Cyber Security. The Journal of Digital Forensics, Security and Law.
- Schein, E. (1984). Culture as an environmental context for careers. Journal of Organisational Behaviour, 5(1), pp.71-81.
- Schein, E.H., 1992. Organisational culture and leadership Jossey-Bass. San Francisco, 42.

- Schlienger, T., and Teufel, S., 2003. Information security culture: From analysis to change. South African Computer Journal, 31, 46–52.
- Shi, L., 2008. Health services research methods. New York: Cengage.
- Shweder, R. A., Goodnow, J., Hatano, G., Levine, R. A., Markus, H. and Miller, P. 1998. The Cultural Psychology Of Development: One Mind, Many Mentalities.

Siegel, A.F., 2012. Time Series. Practical Business Statistics, pp.429–464.

- Signorini, P., R. Wiesemes, and R. Murphy., 2009. Developing alternative frameworks for exploring intercultural learning: A critique of Hofstede's cultural difference model. Teaching in Higher Education, 14(3), 253–264
- Silahtaroğlu, G. and Alayoglu, N., 2016. Using or not using business intelligence and big data for strategic management: an empirical study based on interviews with executives in various sectors. *Procedia Social and Behavioural Sciences*, **235**, pp.208-215.
- Silverman, D. (Ed.). 2016. Qualitative research. London: Sage.
- Silverman, D., 2010. Doing qualitative research. Thousand Oaks, Calif.: Sage Publications.
- Skakni, I., Calatrava Moreno, M., Seuba, M. and McAlpine, L., 2019. Hanging tough: post-PhD researchers dealing with career uncertainty. Higher Education Research & Development, 38(7), pp.1489-1503.
- Slowik, E., 2014. Hobbes and the phantasm of space. *Hobbes Studies*, **27**(1), pp.61-79.
- Smirnov, A., Levashova, T., & Kashevnik, A., 2019. Enterprise Ontology for Service Interoperability in Socio-Cyber-Physical Systems. Proceedings Of The I-ESA Conferences, 203-213. doi: 10.1007/978-3-030-13693-2_17
- Smith, J., Harré, R. and Langenhove, L., 2008. Rethinking methods in psychology. London: Sage.
- Smith, B. and Sparkes, A., 2016. Routledge handbook of qualitative research in sport and exercise. New York: Routledge.
- Snow, C. P. and Collini, S. 1959. The Two Cultures.
- Solis, B., 2009. Revealing the people defining social networks. [Online] Available at: http://www.briansolis.com/2009/10/revealing-the-people-defining-socialnetworks/?utm_source=twitterfeedandutm_medium=twitter [Accessed 11 Jun. 2019].
- Soreide, G., 2006. Narrative construction of teacher identity: positioning and negotiation. Teachers and Teaching, 12(5), pp.527-547.
- Spaulding, D., Lodico, M. and Voegtle, K., 2013. Methods in educational research. San Francisco, Calif.: Jossey-Bass.

- Spencer-Oatey, Helen, 1952-, ed., 2008. Culturally speaking: culture, communication and politeness theory, second edition. London, U.K. / New York, NY, U.S.A.: Continuum International Publishing Group, pp. 258-273. ISBN 978082649310
- Spivey, N.N. (1997). The constructivist metaphor. Reading, writing, and the making of meaning. San Diego, CA: Academic Press.
- Sriwindono, H. and Yahya, S., 2012. The effects of cultural dimension on MIS acceptance. 2012 IEEE International Conference on Management of Innovation and Technology (ICMIT).
- Stoll, C., 2005. *The cuckoo's egg: tracking a spy through the maze of computer espionage.* New York: Pocket Books.
- St. Amant, K., 2007. *Linguistic and cultural online communication issues in the global age*.Hershey, PA: Information Science Reference.
- Strauss, C. and Quinn, N., 2001. *A cognitive theory of cultural meaning*. Cambridge, UK: Cambridge University Press.
- Strate, L., Jacobson, R., Gibson, S. and Gumpert, G., 1996. *Communication and cyberspace: social interaction in an electronic environment*. Hampton Press, Incorporated.
- Street, B., 2003. What's "new" in New Literacy Studies? Critical approaches to literacy in theory and practice. Current Issues in Comparative Education, 5(2), 77-91.
- Sun, P., Ku, C. and Shih, D., 2015. An implementation framework for E-Government 2.0. *Telematics and Informatics*, **32**(3), pp.504-520.
- Szücs, A., Tait, A., Vidal, M. and Bernath, U., 2013. *Distance and e-learning in transition*. London: Wiley.
- Talebian, S., Mohammadi, H. and Rezvanfar, A., 2014. Information and communication technology (ICT) in higher education: advantages, disadvantages, conveniences and limitations of applying e-learning to agricultural students in Iran. *Procedia - Social* and Behavioural Sciences, 152, pp.300-305.
- Taye, M., Sang, G. and Muthanna, A., 2019. Organizational culture and its influence on the performance of higher education institutions: The case of a state university in Beijing. International Journal of Research Studies in Education, 8(2).
- Teixeira, M., de Lima Júnior, J., de Farias Júnior, I., de Aquino, C. and Teixeira, M., 2017.
 Mobility, cyberspace culture, app and digital citizenship: a case study of
 "universidade conectada". Advances in Intelligent Systems and Computing, pp.503-510.

Thyer, B., 2010. The Handbook of Social Work Research Methods. Thousand Oaks: Sage.

- Tømte, C., Fossland, T., Aamodt, P. and Degn, L., 2019. Digitalisation in higher education: mapping institutional approaches for teaching and learning. Quality in Higher Education, 25(1), pp.98-114.
- Tran, Y., Yonatany, M. and Mahnke, V., 2016. Crowdsourced translation for rapid internationalization in cyberspace: a learning perspective. *International Business Review*, 25(2), pp.484-494.
- Trochim, W., Donnelly, J. and Arora, K., 2016. Research methods. Boston, MA: Cengage Learning.
- Tsai, Y., 2011. Relationship between Organizational Culture, Leadership Behavior and Job Satisfaction. BMC Health Services Research, 11(1).
- Turkle, Sherry (1995), Life on the Screen, New York, NY: Simon and Schuster
- Valdez, V. and Omerbašić, D., 2015. Multimodal self-authoring across bi/multilingual educator and student learning spaces. *Bilingual Research Journal*, **38**(2), pp.228-247.
- Välimaa, J., 1998. Higher Education, 36(2), pp.119-138.
- Van Manen, M. and Adams, C., 2009. The phenomenology of space in writing online. *Educational Philosophy and Theory*, **41**(1), pp.10-21.
- Verizon, 2009. 2009 data breach investigations report. Retrieved from http://www.verizonenterprise.com/resources/security/reports/2009_databreach_rp.pdf
- Vestich, E., 1997. Book Review- Silicon Snake Oil: Second Thoughts on the Information Highway. Journal of Technology Education, 9(1.
- Viberg, O. and Grönlund, Å., 2013. Cross-cultural analysis of users' attitudes toward the use of mobile devices in second and foreign language learning in higher education: a case from Sweden and China. *Computers and Education*, **69**, pp.169-180.
- Von Solms, R., and Van Niekerk, J., 2013. From Information Security To Cyber Security. Computers and Security, 38, 97–102. doi:10.1016/J.Cose.2013.04.004
- Walsh, K., 2019. Lead With Value: How Leaders Unleash Their Vision, Empower Others, and Evolve Their Business. Connecticut: Revampologist.
- Wang, J. and Kissel, Z., 2015. *Introduction to network security*. Singapore: John Wiley and Sons Singapore Pte Ltd.
- Weber, R. H., 2012. Future Design Of Cyberspace Law: Laws Are Sand (Mark Twain, The Gorky Incident. J. Pol. and L., 5, 1.
- Wengraf, T., 2011. Qualitative research interviewing. London: Sage.
- Whitman, M. and Mattord, H., 2009. Principles of Information Security. Boston: Cengage.

- Williams, C., 2007. Research Methods. Journal of Business and Economic Research, 5(3), pp.65-72.
- Wilcockson, T.D.W., Osborne, A.M. and Ellis, D.A., 2019. Digital detox: The effect of smartphone abstinence on mood, anxiety, and craving. *Addictive Behaviors*, 99, p.106013
- Wilson, B. 1996. Cultural Assimilation Of The Internet: A Case Study.
- Xu, Y. and Dai, L., 2019. Research on the Influence of Situational Teaching Mode on Online Learning Experience. Lecture Notes in Computer Science, pp.514-527.
- Yar, M., 2006) Cybercrime and Society. London: Sage Publications.
- Yeh, R.-S., 1983. On Hofstede's treatment of Chinese and Japanese values, Asia Pacific Journal of Management, 6(1), 149–160.
- Zamawe, F., 2015. The Implication of Using NVivo Software in Qualitative Data Analysis: Evidence-Based Reflections. Malawi Medical Journal, 27(1), p.13.
- Zeng, J., Yang, L. and Ma, J., 2016. A System-Level Modeling and Design for Cyber-Physical-Social Systems. ACM Transactions on Embedded Computing Systems, 15(2), pp.1-26.
- Zetter, K., 2016. Inside the cunning, unprecedented hack of Ukraine's power grid. [Online] Available at: https://www.wired.com/2016/03/inside-cunning-unprecedented-hackukraines-power-grid/ [Accessed 7/6/2019].
- Zhou, Y., Dong, F., Kong, D. and Liu, Y., 2019. Unfolding the convergence process of scientific knowledge for the early identification of emerging technologies. *Technological Forecasting and Social Change*, 144, pp.205-220.
- Zoto, E., Kowalski, S., Frantz, C., Lopez-Rojas, E. and Katt, B., 2018. A pilot study in cyber security education using cyberAIMs: a simulation-based experiment. *IFIP Advances in Information and Communication Technology*, pp.40-54.
- Zoogah, D. and Beugré, C., 2013. Managing organizational behaviour in the African context. New York: Routledge.
- Žukauskas, P., Vveinhardt, J. and Andriukaitienė, R., 2018. Philosophy and Paradigm of Scientific Research. Management Culture and Corporate Social Responsibility.

Appendix A: Interview Questions

1. What influence do you think the vision 2030's directive of strategy formulation exerts on the attainment of the higher education vision 2030? Does the attainment of the higher education vision 2030 lead to a better and effective cyberspace culture?

Please rate one of the following in your response.

Factor	Description	Very Important	Important	Average	Less Important	Not Important
Economic	Leadership strategies to increase the economic viability of higher education institutions. E.g. creating more courses and expanding institutions.					
External leadership pressure to implement certain strategies.	Leadership pressure by the education ministry for higher education institutions to adhere to select strategies. For instance, implement online learning.					
Other, please specify						

The question investigates the participant's understanding on the influence government leadership on the fostering of cyberspace culture in higher education in terms of strategy formulation. The question identifies different types of strategies that the government uses to foster cyberspace culture.

2. Please describe the influence of vision 2030 directive (information security policy and regulation) on the attainment of the higher education vision 2030? the fostering of cyberspace culture in higher education? In what ways does this influence lead to a better and effective cyberspace culture?

Factor	Description	Very Important	Important	Average	Less Important	Not Important
Compliance and Advanced Security	Some security policies may require institutions to comply by adopting new security management strategies. For instance, compliance to ISO in an effort to enhance their security. The compliance is likely to influence the creation of new security culture.					
Environmental uncertainty	Uncertainty in the higher education environment can arise where government regulations are expected to change with new leadership.					
Political factors	Politics in the government can force higher education institutions to adopt new strategies, such as advocate, for new management boards, thereby affecting culture.					

Other, please
specify

The question investigates the participant's understanding regarding the influence of government policy of information security on the fostering of cyberspace culture in higher education. It identifies some existent policies that are already being implemented in the given space and other possible alternatives that may be implemented.

3. Advanced technology transforms higher education in different ways. Please describe the influence of advancement in technologies on the attainment of the higher education vision 2030?

Factor	Description	Very Important	Important	t Average Less In		Not Important
Learning and teaching	Describes new innovations on e- learning to share knowledge, improve collaboration, and enhance flexibility of learning. Similarly, technology transforms teaching leading to flipped classrooms and higher reliance on technology for teaching					
Management of learning institutions	Refers to the use of technology to gain new insights on management of higher education institutions; for instance, analytics in decision making. Cyber					

	management is further required to			
	ensure technology is used			
	appropriately.			
Other, please				
specify				

The question investigates the participant's understanding regarding the influence of advanced technology on the attainment of the higher education vision 2030. As such, it seeks to identify how the participants link the role of technology in the attainment of the higher education vision 2030.

4. Please describe whether the influence of the attainment of the higher education vision 2030 leads to a better and effective cyberspace culture?

Factor	Description	Very Important	Important	Average	Less Important	Not Important
	Describes new innovations on e-					
Learning and	learning to share knowledge, improve					
	collaboration, and enhance flexibility					
toophing	of learning. Similarly, technology					
teaching	transforms teaching leading to flipped					
	classrooms and higher reliance on					
	technology for teaching.					

	Refers to the use of technology to gain			
	new insights on management of			
Managament of	learning institutions, for instance,			
Management of learning institutions	analytics in decision making. Cyber			
	management is further required to			
	ensure technology is used			
	appropriately.			
Other, please				
specify				

The question investigates the participant's understanding regarding the influence of advancement in technology on the fostering of cyberspace culture in higher education. It identifies some strategies that are already being implemented in the given space

5. Qatar's vision 2030 emphasizes on human development in terms of education advancement and social growth. Please describe whether the attainment of the higher education vision 2030 lead to a better and effective cyberspace culture? Please rate one of the following.

Factor	Description	Very Important	Important	Average	Less Important	Not Important
The proposed	The framework proposes various					
implementation	objectives that are needed to be					
framework of vision	achieved in order to enhance					
induction of vision	education by 2030. These objectives					

2030 in education	and activities are expected to influence			
development.	the higher education vision 2030.			
Other, please specify				

The question investigates the participant's perceptions on whether attaining the higher education vision 2030 influences the cultivation of effective cyberspace culture. The assumption held is that since the vision 2030 framework leads to the attainment of the higher education vision, there is a necessity to assess whether achieving the higher education vision 2030 influences the cultivation of positive cyberspace culture.

6. Please describe the influence of the internal leadership strategy adopted in a higher education institution on the attainment of the higher education vision 2030? Does the attainment of the higher education vision 2030 lead to a better and effective cyberspace culture?

Ы	4		6 41	6 11	•	•		
Please	rate	one o	of the	tollo	wing	ın	vour	response.
1 ICube	1	0110 0		10110			J	response

Factor	Description	Very Important	Important	Average	Less Important	Not Important
Accountability of						
leadership towards	Describes the liability or answerability					
the fostering of	of leaders towards ensuring positive					
positive cyberspace	cyberspace culture					
culture						
Risk management	Describes the leadership's role in					
	identifying and eliminating risks that					

	challenge the cultivation of positive			
	cyberspace culture.			
Other, please specify				

The question seeks to investigate the participant's understanding on the impact of the leadership strategy in a higher learning institution on the fostering of cyberspace culture in higher education.

7. Please describe the influence of security, technical and best practices adopted in a higher education institution, on the attainment of the higher education vision 2030? Does the attainment of the higher education vision 2030 lead to a better and effective cyberspace culture?

Factor	Description	Very Important	Important	Average	Less Important	Not Important
Security skills and						
knowledge to	Describes security know-how that					
prevent threats and	may not be directly observable.					
risk.						
Other, please specify						

The question seeks to establish the participant's understanding on the impact of security best practices adopted in a higher learning institution on the fostering of cyberspace culture in higher education.

8. Please describe the influence of the internal security policies implemented in a higher education institution, on the attainment of the higher education vision 2030? Does the attainment of the higher education vision 2030 lead to a better and effective cyberspace culture?

Factor	Description	Very Important	Important	Average	Less Important	Not Important
Attitudes and behaviour	Describes the security attitudes and behaviour of employees in the institution.					
Continuous learning	Describes the willingness of users to continue enhancing their knowledge and skills on security aspects.					
Security Awareness	Describes the awareness of users regarding need for securing the data and resources within the institution.					
Other, please specify						

The question investigates the participant's understanding of the influence of internal security policy factors on the attainment of the higher education vision 2030. As a result, it links the attainment of the vision to the cultivation of cyberspace culture.

9. Please describe the influence of culture in the fostering of cyberspace culture and the attainment of vision 2030 in higher education Please rate one of the following in your response.

Factor	Description	Very Important	Important	Average	Less Important	Not Important
Visible processes	Includes language, visible traditions such as leadership practices and social rituals in the firm.					
Attitudes and behaviour	Describes the behaviour and attitudes of employees that are unique to the organisation					
Values developed	The ideals and beliefs that guide the employees					
Cultural dimensions	Describes high level culture in terms of how it influences the inherent behaviour of individuals.					
Other, please specify						

The question examines the impact of theoretical levels of culture on the fostering of cyberspace culture. The insight is highly important, as it indicates the impact of theory on the real-world.
FOCUS GROUP

- 10. Based on your observation, do you think that the proposed framework (internal and external elements as well as theoretical perspectives) can align cyberspace culture to the higher education vision? Does the attainment of the vision affect how cyberspace culture is fostered?
 - Which aspects of the framework do you think exert higher influence on the development of cyberspace culture?
 - Which reservations do you have regarding the framework and its role in aligning cyberspace culture to the higher education vision?
 - Any additional comments?

The last question investigates the effectiveness of the proposed cyberspace culture framework on the alignment to the higher education vision. The effect of attaining the higher education vision on cyberspace culture is additionally evaluated in order to assess the bidirectional relationship. Two focus groups will be organised in order to discuss the bi-directional relationship between the proposed framework and alignment of the higher education vision. Each focus group comprises of 5 participants drawn from different departments in higher education and the Qatar vision 2030. The researcher will conduct the focus group sessions differently in order to compare the findings.

Appendix B: Interview Particulars

Table 6-3: Particulars of the interviews.

No	Respondent code	Duratio n	Word s	Interview setting	Interview category	Organisation section	Interviewe e roles	Interviewe e experience	Gender	Nation ality	Job categorie s
1	Respondent 1	29:27	3019	In person	Strategy	Higher Education	Dean	3 years	Male	Qatari	Academi c
2	Respondent 2	42:34	4492	In person	Strategy	Higher Education	VP	5 years	Male	Qatari	Academi c
3	Respondent 3	46:39	6632	In person	Technical	Higher Education	Manager	8 years	Male	Non- Qatari	Non- academic
4	Respondent 4	49:30	3392	In person	Strategy	Higher Education	VP	5 years	Male	Qatari	Academi c
5	Respondent 5	25:48	3939	In person	Strategy	Higher Education	Director	8 years	Male	Qatari	Academi c
6	Respondent 6	31:00	4646	In person	Strategy	Higher Education	Associate Vice President	5 years	Male	Non- Qatari	Academi c
7	Respondent 7	20:04	2736	In person	Strategy	Regulatory Sector	Undersecr etary	3 years	Male	Qatari	Academi c
8	Respondent 8	22:18	2792	In person	Technical	Higher Education	Director	3 years	Female	Qatari	Academi c

9	Respondent	30:40	3908	Online	Technical	Regulatory Sector	Manager	7 years	Male	Non- Qatari	Non- Academi
	-			•••••							с
10	Respondent	19:00	1895	Online	Technical	Regulatory Sector	Manager	2 years	Female	Qatari	Non-
	10			call							academic
11	Respondent	28.51	2650	Online	Stratogy	Higher Education	President	5 years	Male	Qatari	Academi
11	11	28:51	2057	call	Strategy						с
12	Respondent	17.20	2202	Online	Tashnisal	Regulatory Sector	Section	4 years	Female	Qatari	Non-
12	12 12	17.50	2303	call	Technical		Head				academic
12	Respondent	27.50	2622	Online	Christia and	Higher Education	Teaching	3 years	Female	Qatari	Academi
15	13	27:39	59 5055	call	Strategy		Assistant				c
14	Respondent	44:21 6009	Online	Ctrata av	Higher Education	Director	4 years	Male	Qatari	Academi	
14	14		call	Strategy						c	
15	Respondent	42.47	(507	Online	Churchen and	Higher Education	Director	6 years	Male	Non-	Non-
15	15	45:47	0387	call	Strategy					Qatari	academic
16	Respondent	25.05	2154	Online	Testal	Regulatory Sector	Section	8 years	Male	Non-	Non-
10	16	25:05	3154	call	Tecnnical		Head			Qatari	academic
17	Respondent	17.48	2358	Online	Technical	Higher Education	Director	3 years	Male	Non-	Academi
17	17	17.40	2330	call	Teennear					Qatari	с
	Respondent			Online		Higher Education	Associate	3 years	Male	Qatari	Non-
18	18	10:27 1425		Technical		Vice				academic	
	10			Call			President				

	Duration: 532.48 min
Total 10 Interviews	[8 hrs, 14 min, 58 sec]
Total 18 Interviews	Total words: 65,579
	10 online calls, eight face-to-face interviews

Source: The Researcher

Appendix C: Focus Group Particulars

 Table 6-2: Particulars of the focus group

No.	Respondent code	Interview setting	Profession al category	Organisational sector	Professional roles	Professional experience	Gender	Nationality	Job categories
.1	Respondent 1	In person Face to Face	Technical	Regulatory Sector	Section Head	8 years	Male	Non-Qatari	Non- academic
.2	Respondent 2	In person Face to Face	Technical	Higher Education	Manager	6 years	Male	Non-Qatari	Non- academic
.3	Respondent 3	In person Face to Face	Technical	Higher Education	Director	5 years	Male	Qatari	Non- academic
.4	Respondent 4	In person Face to Face	Strategy	Higher Education	Manager	1 years	Male	Qatari	Academic
.5	Respondent 5	In person Face to Face	Strategy	Regulatory Sector	Director	5 years	Female	Qatari	Non- academic
.6	Respondent 6	In person Face to Face	Strategy	Higher Education	Associate Vice President	2 years	Male	Qatari	Academic

Source: The Researcher

Appendix D: Themes and Coding References Table 6-3 Number of coding references and items coded per theme

Theme	Sub-theme	Main code	Code	Number of	
Theme	Sub-theme	Wall Code		coded items	
			Economic factors (Sub-	13	
		Leadership (Sub-	Section 4.7.2.1.1.1)	15	
		Section 4.7.2.1.1)	External pressure (Sub-	16	
	Role of the		Section 4.7.2.1.1.2)	10	
	QNV 2030		Compliance and advanced		
	directive (Sub-	Information	security (Sub-Section	4	
	Section	Information	4.7.2.1.2.1)		
External	4.7.2.3)	regulation (Sub	Environmental uncertainty	12	
elements		Section $4.7, 2, 1, 2$	(Sub-Section 4.7.2.1.2.2)	12	
(Sub-		Section 4.7.2.1.2)	Political factors (Sub-	10	
Section			Section 4.7.2.1.2.3)	10	
(4,7,2)	Role of		Learning, teaching (Sub-	0	
4.7.2)	technological	Technological advancement (Sub- Section 4.7.2.2)	Section 4.7.2.2.1))	
	advancement		Managamant (Sub Saction		
	(Sub-Section			4	
	4.7.2.2)		4.7.2.2.2)		
	Role of the	The QNV 2030			
	QNV 2030	framework (Sub-	Implementation framework	17	
	(Sub-Section	Section 4.7.2.3.1)	(Sub-Section 4.7.2.3.1)	17	
	4.7.2.3)				
		Leadership and	Accountability (Sub-Section	5	
Internal		management	4.7.1.1.1)	5	
alamanta		strategies (Sub-	Risk management (Sub-	1	
(Sub		Section 4.7.1.1)	Section 4.7.1.1.2)		
(Sub-			Security awareness	2	
		Security policy	(Sub-Section 4.7.1.2.1)	5	
4./.1)		(Sub-Section	Continuous learning	0	
		4.7.1.2)	(Sub-Section 4.7.1.2.2)	0	
			Attitudes and behaviour	1	
I	I		1	1	

· ·			-
		(Sub-Section $4.7.1.2.\overline{3}$)	
	Security technical and best practices (Sub-Section 4.7.1.3)	Security skills and knowledge (Sub-Section 4.7.1.3.1)	1
Role of culture (Sub-	Dimensions of culture (Sub-Section 4.7.3.2)	Culture dimensions (Sub- Section 4.7.3.2)	1
Section 4.7.3)	Levels of culture (Sub-Section 4.7.3.1)	Values Visible processes Attitudes and behaviour (Section 4.7.3.1)	

Source: The Researcher